#### UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

#### FORM 10-K

ANNUAL REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the year ended December 31, 2012

□ TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

Commission file number 001-34835

commission me number 001-54055

Envestnet, Inc. (Exact name of registrant as specified in its charter)

Delaware (State or other jurisdiction of incorporation or organization)

35 East Wacker Drive, Suite 2400, Chicago, IL (Address of principal executive offices)

Title of Each Class:

Common stock, par value \$0.005 per share

20-1409613 (I.R.S Employer Identification No.)

> 60601 (Zip Code)

Registrant's telephone number, including area code:

(312) 827-2800

Securities registered pursuant to Section 12(b) of the Act:

Name of Each Exchange on Which Registered:

NYSE

Securities registered pursuant to Section 12(g) of the Act: None

Indicate by check mark whether the registrant is a well-known seasoned issuer, as defined in Rule 405 of the Securities Act. Yes 🛛 No 🗵

Indicate by check mark whether the registrant is not required to file reports pursuant to Section 13 or Section 15(d) of the Act. Yes 🛛 No 🗵

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes  $\Box$  No  $\boxtimes$ 

Indicate by check mark whether the registrant has submitted electronically and posted on its corporate Web site, if any, every Interactive Data File required to be submitted and posted pursuant to Rule 405 of Regulation S-T (232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit and post such files). Yes  $\Box$  No  $\boxtimes$ 

Indicate by check mark if disclosure of delinquent filers pursuant to item 405 of Regulation S-K is not contained herein, and will not be contained, to the best of the registrants knowledge, in definitive proxy or information statements incorporated by reference in Part III of this Form 10-K or any amendment to this Form 10-K.

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, or a smaller reporting company. See the definitions of "large accelerated filer," "accelerated filer" and "smaller reporting company" in Rule 12b-2 of the Exchange Act. (Check one):

Large accelerated filer	Accelerated filer	X
Non-accelerated filer	Smaller reporting company	

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Act). Yes 🛛 No 🗵

Aggregate market value of registrant's common stock held by non-affiliates of the registrant, based upon the closing price of a share of the registrant's common stock on June 30, 2012 as reported on The New York Stock Exchange on that date: \$205,972,224. For purposes of this calculation, shares of common stock held by (i) persons holding more than 5% of the outstanding shares of stock, and (ii) officers and directors of the registrant, as of June 30, 2012, are excluded in that such persons may be deemed to be affiliates. This determination is not necessarily conclusive of affiliate status.

As of June 12, 2013, 32,411,665 shares of the common stock with a par value of \$0.005 per share were outstanding.

### TABLE OF CONTENTS

		Page
	<u>PART I</u>	
Forward-L	ooking Statements	3
Item 1.	Business	4
Item 1A.	Risk Factors	17
Item 1B.	Unresolved Staff Comments	31
Item 2.	Properties	31
Item 3.	Legal Proceedings	32
Item 4.	Mine Safety Disclosures	32
	PART II	
Item 5.	Market for Registrant's Common Equity, Related Stockholder Matters and Issuer Purchases of Equity Securities	33
Item 6.	Selected Financial Data	35
Item 7.	Management's Discussion and Analysis of Financial Condition and Results of Operations	36
Item 7A.	Quantitative and Qualitative Disclosures About Market Risk	60
Item 8.	Financial Statements and Supplementary Data	61
Item 9.	Changes in and Disagreements with Accountants on Accounting and Financial Disclosure	96
Item 9A.	Controls and Procedures	96
Item 9B.	Other Information	101
	PART III	
Item 10.	Directors, Executive Officers and Corporate Governance	101
Item 11.	Executive Compensation	104
Item 12.	Security Ownership of Certain Beneficial Owners and Management and Related Stockholder Matters	115
Item 13.	Certain Relationships and Related Transactions, and Director Independence	116
Item 14.	Principal Accountant Fees and Services	118
	PART IV	
Item 15.	Exhibits and Financial Statement Schedules	120
<u>SIGNATI</u>	JRES	123

#### Forward-Looking Statements

This annual report on Form 10-K contains forward-looking statements regarding future events and our future results within the meaning of the Private Securities Litigation Reform Act of 1995. These forward-looking statements include, in particular, statements about our plans, strategies and prospects under the heading "Management's Discussion and Analysis of Financial Condition and Results of Operations." These statements are based on our current expectations and projections about future events and are identified by terminology such as "anticipate," "believe," "continue," "could," "estimate," "expect," "expected," "intend," "will," "may," or "should" or the negative of those terms or variations of such words, and similar expressions are intended to identify such forward-looking statements. In addition, any statements that refer to projections of our future financial performance, our anticipated growth and trends in our business and other characteristics of future events or circumstances are forward-looking statements. Forward-looking statements may include, among others, statements relating to:

- difficulty in sustaining rapid revenue growth, which may place significant demands on the Company's administrative, operational and financial resources,
- fluctuations in the Company's revenue,
- the concentration of nearly all of the Company's revenues from the delivery of investment solutions and services to clients in the financial advisory industry,
- the Company's reliance on a limited number of clients for a material portion of its revenue,
- the renegotiation of fee percentages or termination of the Company's services by its clients,
- the Company's ability to identify potential acquisition candidates, complete acquisitions and successfully integrate acquired companies,
- the impact of market and economic conditions on the Company's revenues,
- compliance failures,
- regulatory actions against the Company,
- the failure to protect the Company's intellectual property rights,
- the Company's inability to successfully execute the conversion of its clients' assets from their technology platform to the Company's technology platform in a timely and accurate manner,
- general economic conditions, and
- political and regulatory conditions, as well as management's response to these factors.

Although we believe that our plans, intentions and expectations are reasonable, we may not achieve our plans, intentions or expectations.

These forward-looking statements involve risks and uncertainties. Important factors that could cause actual results to differ materially from the forward-looking statements we make in this annual report are set forth in Part I under "Risk Factors"; accordingly, investors should not place undue reliance upon our forward-looking statements. We undertake no obligation to update any of the forward-looking statements after the date of this annual report to conform those statements to reflect the occurrence of unanticipated events, except as required by applicable law.

You should read this annual report on Form 10-K completely and with the understanding that our actual future results, levels of activity, performance and achievements may be different from what we expect and that these differences may be material. We qualify all of our forward-looking statements by these cautionary statements.

The following discussion and analysis should also be read along with our consolidated financial statements and the related notes included elsewhere in this annual report. Except for the historical information contained



herein, this discussion contains forward-looking statements that involve risks and uncertainties. Actual results could differ materially from those discussed below.

Unless the context requires otherwise, the words "Envestnet," "the Company," "we," "us" and "our" are references to Envestnet, Inc. and its subsidiaries as a whole.

Except where we have otherwise indicated or the context otherwise requires, dollar amounts presented in this Form 10-K are in thousands, except for Part III and per share amounts.

#### Item 1. Business

#### General

We are a leading provider of unified wealth management software and services to financial advisors and institutions. By integrating a wide range of investment solutions and services, our technology platform provides financial advisors with the flexibility to address their clients' needs. As of December 31, 2012, approximately 23,000 advisors used our technology platform, supporting approximately \$368 billion of assets in approximately 1.7 million investor accounts.

Envestnet empowers financial advisors to deliver fee-based advice to their clients. We work with both Independent Registered Investment Advisors ("RIAs"), as well as advisors associated with financial institutions such as broker dealers and banks. The services we offer and market to financial advisors address advisors' ability to grow their practice as well as to operate more efficiently – the Envestnet platform spans the various elements of the wealth management process, from the initial meeting an advisor has with a prospective client to the ongoing day-to-day operations of managing an advisory practice.

Our centrally-hosted technology platform, which we refer to as having "open architecture" because of its flexibility, provides financial advisors with access to a series of integrated services to help them better serve their clients. These services include risk assessment and selection of investment strategies and solutions, asset allocation models, research and due diligence, portfolio construction, proposal generation and paperwork preparation, model management and account rebalancing, account monitoring, customized fee billing, overlay services covering asset allocation, tax management and socially responsible investing, aggregated multi-custodian performance reporting and communication tools, as well as access to a wide range of leading third-party asset custodians.

We offer these solutions principally through the following product and services suites:

- Envestnet's wealth management software empowers advisors to better manage client outcomes and strengthen their practice. Our software unifies the applications and services advisors use to manage their practice and advise their clients, including financial planning; capital markets assumptions; asset allocation guidance; research and due diligence on investment managers and funds; portfolio management, trading and rebalancing; multi-custodial, aggregated performance reporting; and billing calculation and administration.
- Our Portfolio Management Consultants group ("Envestnet | PMC" 

   primarily engages in consulting services aimed at providing financial advisors with additional support in addressing their clients' needs, as well as the creation of proprietary investment solutions and products. Envestnet | PMC's investment solutions and products include managed account and multi-manager portfolios, mutual fund portfolios and Exchange Traded Fund ("ETF") portfolios. Envestnet | PMC also offers Prima Premium Research, comprising institutional-quality research and due diligence on investment managers, mutual funds, ETFs and liquid alternatives funds.
- Envestnet | Tamarac provides leading portfolio accounting, rebalancing, trading, performance reporting and client relationship management software, principally to high-end RIAs.

• Envestnet Reporting Solutions software aggregates and manages investment data, provides performance reporting and benchmarking, giving advisors an in-depth view of clients' various investments, empowering advisors to give holistic, personalized advice and consulting.

We believe that our business model results in a high degree of recurring and predictable financial results.

The majority of our revenues is derived from fees charged as a percentage of the assets that are managed or administered on our technology platform by financial advisors. We also generate revenues from recurring, contractual licensing fees for providing access to our technology platform and from professional services.

For the year ended December 31, 2012, we earned fees of \$127,213 from assets under management ("AUM") or assets under administration ("AUA") or collectively ("AUM/A") by advisors using our technology platform. Asset-based fees accounted for approximately 81%, 81% and 77% of our total revenues for the years ended December 31, 2012, 2011 and 2010, respectively. Licensing and professional services fees accounted for 19%, 19% and 23% of our total revenues for the years ended December 31, 2012, 2011 and 2010, respectively.

For over 85% of our asset-based fee arrangements, we bill customers at the beginning of each quarter based on the market value of customer assets on our technology platform as of the end of the prior quarter, providing for a high degree of revenue visibility in the current quarter. Furthermore, our licensing fees are highly predictable because they are generally set in multi-year contracts providing longer-term visibility regarding that portion of our total revenues.

As the tables below indicate, our business has experienced steady and significant growth over the last several years. We believe this growth is attributable not only to secular trends in the wealth management industry as described below but also to the uniqueness and comprehensiveness of our product set.

The following table sets forth the assets under management (AUM) or administration (AUA) as of the end of the quarters indicated:



### Quarterly Assets under Management or Administration (\$ In Billions)

The following table sets forth the number of accounts with assets under management (AUM) or administration (AUA) as of the end of the quarters indicated:



Quarterly Accounts under Management or Administration

(In Thousands)

The following table sets forth as of the end of the years indicated the number of financial advisors that had client accounts on our technology platform:





We were founded in 1999 and through organic growth and strategic transactions we have grown to become a leading independent provider of integrated wealth management software and services to financial advisors and institutions. Our headquarters are located in Chicago and we have offices in New York, New York; Denver, Colorado; Seattle, Washington; Sunnyvale, California; Boston, Massachusetts; Landis and Raleigh, North Carolina; and two locations in Trivandrum, India.



#### **Our Market Opportunity**

The wealth management industry has experienced significant growth in terms of assets invested by retail investors in the past several years. According to the Federal Reserve, U.S. household and non-profit organization financial assets totaled \$54.4 trillion as of December 31, 2012, up 7% from \$50.6 trillion at December 31, 2011, and exceeding the 2007 peak of \$52.2 trillion.

In addition to experiencing significant growth in financial assets, the wealth management industry is characterized by a number of important trends, including those described below, which we believe create a significant market opportunity for technology-enabled investment solutions and services like ours.

*Increase in independent financial advisors*. Based on industry news reports, we believe that over the past several years an increasing percentage of financial advisors have elected to leave large financial institutions and start their own financial advisory practices or move to smaller, more independent firms. According to an analysis done by Cerulli Associates, the number of RIAs and dually-registered advisors increased 7% annually from 34,000 in 2006 to 47,000 in 2011.

*Increased reliance on technology among independent financial advisors*. In order to compete effectively in the marketplace, independent financial advisors are increasingly relying on technology service providers to help them provide comparable services cost effectively and efficiently.

Increased use of financial advisors. We believe, based on an analysis done by Cerulli Associates, that the recent significant volatility and increasing complexity in securities markets have resulted in increased investor interest in receiving professional financial advisory services.

*Increased use of fee-based investment solutions.* Based on our industry experience, we believe that in order for financial advisors to effectively manage their clients' assets, they are seeking account types that offer the flexibility to choose among the widest range of investment solutions. Financial advisors typically charge their fees for these types of flexible accounts based on a percentage of assets rather than on a commission or other basis.

*More stringent standards applicable to financial advisors*. Increased scrutiny of financial advisors to ensure compliance with current laws, coupled with the possibility of new laws focused on a fiduciary standard, may require changes to the way financial advisors offer advice. In order to adapt to these changes, we believe that financial advisors may benefit from utilizing a technology platform, such as ours, that allows them to address their clients' wealth management needs.

#### **Our Business Model**

We believe that a number of attractive characteristics significantly contribute to the success of our business model, including:

- Positioned to capitalize on favorable industry trends. As a leading provider of unified wealth management software and services to
  financial advisors, we believe we are well positioned to take advantage of favorable secular trends in the wealth management industry,
  particularly the increased prevalence and use of independent financial advisors, the movement toward fee-based pricing structures and
  advisors' increased reliance on technology.
- *Recurring and resilient revenue base.* The substantial majority of our revenues is recurring and derived either from asset-based fees, which generally are billed at the beginning of each quarter or from fixed fees under multi-year license agreements. For the year ended December 31, 2012, we derived 81% of our total revenues from asset-based fees and 19% from licensing and professional services revenues.

- Strong customer retention. We believe that the breadth of access to investment solutions and the multitude of services that we provide allow
  financial advisors to address a wide range of their clients' needs and, as a result, financial advisors are less likely to move away from our
  technology platform. Because a technology platform is involved in nearly all of a financial advisor's activities needed to serve their clients,
  once a financial advisor has moved clients and their assets onto our technology platform, significant time, costs and/or resources would be
  required for the financial advisor to shift to another technology platform.
- Substantial operating leverage. Because we have designed our systems architecture to accommodate growth in the number of advisors and
  accounts we support and to provide the flexibility to add new investment solutions and services, our technology platform and infrastructure
  allow us to grow our business efficiently, without the need for significant additional expenditures as assets grow. This, combined with low
  marginal costs required to add additional accounts and new investment solutions and services, enables us to generate substantial operating
  leverage during the course of our relationship with a financial advisor as the advisor adds accounts to our platform and the assets of the
  advisor's clients grow through financial advisors' use of additional investment solutions and services we provide.

#### **Our Growth Strategy**

Envestnet serves the fastest growing segments of the wealth management industry: independent financial advisors; fee-based solutions; and outsourced investment and technology solutions. We intend to increase revenue and profitability by continuing to pursue the following strategies:

- Increase the advisor base. Through the outreach and marketing activities of our sales and client service teams, we continue the process of leveraging existing enterprise client relationships to add new financial advisors to our technology platform, and building new relationships to add additional advisors. Generally, when we establish an enterprise client relationship, we are provided access to the client's financial advisors and are given the opportunity to convert them to our technology platform. During the five-year period ended December 31, 2012, within existing enterprises, we increased the number of advisors with AUM or AUA on our platform at a compound annual growth rate of 9%. We added a comparable number of advisors through new enterprise relationships established during the past five years, resulting in the total number of advisors with AUM or AUA on our platform annual growth rate of 18%. Even with that steady growth, we continue to have the opportunity to increase the number of financial advisors we serve within our existing enterprise client relationships as advisors increasingly shift their businesses to fee-based practices.
- *Extend the account base within a given advisor relationship.* We work with existing clients to shift an increasing portion of their business to the Envestnet platform. During the five-year period ended December 31, 2012, the average number of AUM or AUA accounts per advisor on our technology platform grew from approximately 16 to 28, an increase of over 75%. As a result, total AUM or AUA accounts increased at a compound annual growth rate of 32% during the same period.
- *Expand the services utilized by each advisor.* In most cases, an advisor will leverage only a portion of Envestnet's services. Accordingly, through our sales and marketing efforts, we will continue to educate our financial advisor customers regarding our capabilities in order to expand the scope of our investment solutions and services they employ.
- **Obtain new enterprise clients.** Growing a fee-based offering has become a strategic priority for financial services firms. Envestnet is positioned in the marketplace to empower these firms to deliver fee-based solutions to their advisors. These enterprise clients provide us with access to a large number of financial advisors that may be interested in utilizing our technology platform, as well as to the assets that are managed by these financial advisors. We believe that the current market opportunity for enterprise conversions continues to be significant. New enterprise clients also provide further opportunities to execute on the other strategies discussed above.

- *Continue to invest in our technology platform.* We intend to continue to invest in our technology platform to provide financial advisors with access to investment solutions and services that address the widest range of financial advisors' front-, middle—and back-office needs. In the years ended December 31, 2012, 2011 and 2010, our technology development expenditures totaled \$8,659, \$6,424 and \$5,588, respectively.
- Continue to pursue strategic transactions and other relationships. We intend to continue to selectively pursue acquisitions, investments
  and other relationships that we believe can enhance the attractiveness of our technology platform or expand our client base. Given our scale of
  operations and record of past transactions, we believe we are well-positioned to engage in such transactions in the future.
- In December 2011, we completed the acquisition of FundQuest Incorporated ("FundQuest"), a subsidiary of BNP Paribas Investment Partners
  USA Holdings, Inc. FundQuest (which was renamed Envestnet Portfolio Solutions, Inc.), a provider of fee-based managed services and
  solutions with approximately \$15 billion in AUM or AUA, was integrated fully into Envestnet during 2012.
- In April 2012, we acquired Prima Capital Holding, Inc. ("Prima"). Prima, now part of Envestnet | PMC, provides Prima Premium Research, including investment manager due diligence, consulting, and custom research to the wealth management and retirement industries. Prima's clientele includes banks, independent RIAs, regional broker-dealers, family offices and trust companies.
- In May 2012, we acquired Tamarac, Inc. ("Tamarac"). Tamarac, now operating as Envestnet | Tamarac, provides leading portfolio accounting, rebalancing, trading, performance reporting and client relationship management software, principally to high-end RIAs.
- On April 11, 2013, we entered into a definitive agreement to acquire substantially all of the assets of the Wealth Management Solutions ("WMS") division of Prudential Investments for \$10,000 in cash upon closing, subject to certain post-closing adjustments, plus contingent consideration of up to a total of \$23,000 in cash, based upon meeting certain performance targets, to be paid over three years. WMS is a provider of technology solutions that enables financial services firms to develop and enhance their wealth management offerings. We anticipate the acquisition will be completed in the third quarter of 2013.

#### **Our Technology Platform**

Our proprietary Web-based platform provides financial advisors with access to investment solutions and services that address, in one unified, centrally-hosted platform, based on our knowledge of the industry, the widest range of front-, middle—and back-office needs in our industry. The "open architecture" design of our technology platform provides financial advisors with flexibility in terms of the investment solutions and services they access, and configurability in the manner in which the financial advisors utilize particular investment solutions and services. The multi-tenant platform architecture ensures that this level of flexibility and customization is achieved without requiring us to create unique applications for each client, thereby reducing the need for additional technology personnel and associated expenses. In addition, though our technology platform is designed to deliver a breadth of functions, financial advisors are able to select from the various investment solutions and services we offer, without being required to subscribe to or purchase more than what they believe is necessary.

Envestnet's Advisor Suite® enables advisors to better advise their clients, invest portfolios, manage their practice and those portfolios, and report on their clients' holdings.



*Advise.* Our technology platform provides financial advisors with a flexible proposal and presentation tool that is capable of creating highly customized documents, including signature-ready forms needed to open client accounts. Also, our platform includes a number of financial planning tools such as Monte Carlo simulations, portfolio diagnostics and retirement planning which enable financial advisors to create and implement financial plans that are tailored to each client's investment goals, risk tolerance and assets. Our technology platform provides financial advisors with a customizable risk tolerance questionnaire, which assists advisors in understanding the investment objectives and preferences of their clients and also helps the advisor comply with applicable regulatory requirements regarding the suitability of investments and fiduciary obligations. Based on answers to the questionnaire, the advisor can analyze whether the current portfolio is appropriate to reach the client's goals and suggest an investment policy. We also empower advisors to improve their research and advice through the institutional quality research we provide on investment managers, funds, and alternative investments, and asset allocation guidance through Envestnet | PMC and Prima Premium Research.

*Invest.* Once the investment solutions have been selected, our technology platform, through relationships we have established with a variety of investment managers, allows the financial advisor to access and choose from a wide range of investment programs, including separately managed accounts, unified managed accounts, third-party strategist programs, mutual fund and ETF programs, and others, depending on the financial advisor's assessment of the client's needs. Because our technology platform supports nearly every investment program type that is currently available, financial advisors are able to keep more of a client's assets on one technology platform, thereby simplifying the operation of their practice, saving time and lowering costs. Envestnet | PMC provides consulting services aimed at providing financial advisors with additional support in addressing their clients' needs, including asset allocation modeling, asset manager and mutual fund due diligence, manager and fund selection and ongoing monitoring, investment portfolio construction and overlay services, principally relating to ongoing portfolio management and asset allocation rebalancing. Envestnet | PMC also creates proprietary investment solutions and products, including separate account strategies, multi-manager portfolios, mutual funds, mutual fund wrap and ETF asset allocation strategies.

*Manage.* Once a financial advisor has created a client account and selected investment solutions and programs, our technology platform provides access to ongoing account management services. Additionally, Envestnet | PMC portfolio managers review all Envestnet | PMC models and proprietary portfolios to determine when to rebalance across asset classes. We also offer overlay services that can help enhance an advisor's ability to carry out his or her fiduciary responsibility. Financial advisors can receive risk and style drift alerts, enabling them to adjust their clients' portfolios to ensure that the portfolios remain in compliance with their clients' stated investment objectives and risk tolerance levels. These services include ongoing review of investment portfolios for compliance with asset allocation criteria, with rebalancing recommendations made as necessary, assistance with investment portfolio tax management and review of investment accounts to ensure that investment decisions are consistent with the client's investment objectives. We also offer a socially responsible overlay which the financial advisor may use to maintain compliance with clients' investment restrictions. These may include securities issued by specific companies or from issuers in certain industries that clients want to exclude from their investment accounts.

Envestnet | Tamarac Advisor Xi integrates portfolio accounting, model management, monitoring, rebalancing, trading, performance reporting, and enterprise-level client relationship management ("CRM") software. Advisor Xi also empowers advisors to receive drift alerts and rebalancing notifications to help ensure that asset allocations conform to a client's investment policy. Envestnet | Tamarac's rebalancing software also can be configured to rebalance assets in the most tax-efficient manner for the client.

*Report.* Our technology platform helps advisors to better report on their clients' consolidated holdings through our Envestnet | Vantage and Envestnet | Tamarac offerings, which provide trade-ready, fully-reconciled aggregated performance reports to advisors and their end clients. Our technology platform provides financial advisors with access to over 30 third-party custodians, real-time data and Web-based service tools. Our platform generates highly configurable aggregated reports showing holdings, investment performance, capital gains and losses and other information for financial advisors to provide to their clients that can be downloaded, viewed online or printed. In addition, through our India operations, our technology platform provides financial advisors with access to client account data reconciled daily with records maintained by multiple custodians. Accordingly, when securities markets open each day, financial advisors have the most up-to-date account data available. Our technology platform also supports a wide range of fee and billing structures. These include breakpoint pricing, where lower fee rates are applied as asset levels meet or exceed pre-established thresholds, fees based on aggregated client funds across several accounts held by family members, fees tailored to different investment programs and investment solution types and other customized fee and billing arrangements. Our technology platform includes configurable Fiduciary Oversight Notes ("FONs") that help advisors understand implications of the decisions they make via educational text boxes, and then memorialize those decisions for client service and reporting. The FONs may be searched and reviewed by home offices or the advisors themselves.

#### **Our Customers**

Independent financial advisors that are working alone or as part of financial advisory firms. Our principal value proposition aimed at independent financial advisors working alone or as part of financial advisory firms is that our technology platform allows them to compete effectively with financial advisors employed by large financial institutions. We provide independent financial advisors with access to as many or more of the investment solutions and services that are typically available to financial advisors working at the largest firms. An example of one of our independent financial advisory firm clients is Commonwealth Financial Network.

*Enterprise clients*. We provide enterprise clients with a customized, private-labeled technology platform that enables them to support their affiliated financial advisors with a broad range of investment solutions and services. Our contracts with enterprise clients establish the applicable terms and conditions, including pricing terms, service level agreements and basic platform configurations. For the years ended December 31, 2012, 2011 and 2010, revenues associated with our relationship with our single largest enterprise client, FMR LLC, an affiliate of FMR Corp., or Fidelity, accounted for 22%, 31% and 31%, respectively, of our total revenues. No other client accounted for more than 10% of our total revenues. Examples of our other enterprise clients include Northwestern Mutual, National Financial Partners, National Planning Holdings and Russell Investments.

#### Sales and Marketing

Our sales and marketing staff of more than 100 employees is organized by sales channel and product offering. Our enterprise sales team focuses on entering into agreements with enterprise clients, which enable us to sell our platform capabilities to those firms and the advisors associated with them. Our advisory sales team sells to the individual financial advisors of broker dealers and enters into agreements with RIA firms. Envestnet | Tamarac sells rebalancing, performance reporting and CRM solutions principally to large RIA firms. Envestnet | PMC offers portfolio and investment management consulting services, including Prima Premium Research and due diligence capabilities.

The principal aim of our marketing efforts is to create greater visibility of our company and our brands, and to provide thought leadership to the wealth management industry. Our marketing efforts are focused on our core markets: financial advisors and enterprise clients. We use advertising and public relations to communicate our message to these target markets. Examples of these marketing efforts include:

- quotes in wealth management industry publications regarding our views on financial advisor trends and challenges;
- advertising and other marketing materials promoting our investment solutions and services;
- frequent participation in industry conferences and tradeshows, including events sponsored by our custodian partners, by sponsorship, making presentations and speaking on panels;
- hosting conferences on wealth management solutions;
- providing insights on industry trends through internal research and sponsoring and writing industry white papers; and
- creating marketing tools for financial advisors to better communicate with their current and prospective clients.

To implement our marketing efforts, we generally employ paid print and online advertisements in a variety of industry publications, as well as promotions that include e-blast campaigns and sponsored webinars. We also partner with independent broker dealers ("IBDs") on direct mail campaigns targeting such firms' financial advisors to describe the investment solutions and services that we offer, produce brochures and presentations for financial advisors to use with their clients and we create Internet pages or sites to promote our investment solutions and services.

#### Competition

We generally compete on the basis of several factors, including the breadth and quality of investment solutions and services to which we provide access through our technology platform, the number of custodians that are connected through our technology platform, the price of our investment solutions and services, the ease of use of our technology platform and the nature and scope of investment solutions and services that each client believes are necessary to address their needs. Our competitors offer a variety of products and services that compete with one or more of the investment solutions and services provided through our technology platform, although, based on our industry experience, we believe that none offers a more comprehensive set of products and services than we do. Our principal competitors include:

 Turnkey Asset Management Platform Providers. Providers of turnkey asset management platforms, including SEI Investments Company, Genworth Financial Inc. and Lockwood Advisors (a subsidiary of BNY Mellon Corporation), typically provide financial advisors with one or more types of products and services but generally offer fewer choices in terms of custodians, asset managers, technology features and functionality.

- **Providers of Specific Service Applications**. A number of our competitors, including Advent Software, Inc., provide financial advisors with a product or service designed to address one specific issue or need, such as financial planning or performance reporting. While our technology platform also provides access to these investment solutions or services, financial advisors may elect to utilize a single application rather than a fully integrated platform.
- Custodians. A number of leading asset custodians, such as Pershing (a subsidiary of BNY Mellon Corporation) and The Charles Schwab Corporation, have expanded beyond their custodial businesses to also offer advisor trading tools that compete with our financial advisordirected solutions.

#### Technology

Our technology platform features a three-tier architecture integrating a Web-based user interface, an application tier that houses the business logic for all of the platform's functionality and a SQL Server database. The application tier resides behind load balancers which distribute the workload demands across our servers. We believe our technology design allows for significant scalability.

We devote significant resources to ensuring sufficient platform capacity and system uptime. In 2012, our actual uptime was 99.8%. We have achieved Service Organization Control Report ("SOC1") compliance with our platform and we maintain multiple redundancies, back up our databases and safeguard technologies and proprietary information consistent with industry best practices. We also maintain a comprehensive business continuity plan and company-wide risk assessment program that is consistent with industry best practices and that complies with applicable regulatory requirements.

We have historically made significant investments in platform development in order to enhance and expand our technology platform and expect to continue to make significant investments in the future. In the years ended December 31, 2012, 2011 and 2010, we incurred technology development expenditures totaling approximately \$8,659, \$6,424 and \$5,588, respectively. Of these expenditures, we capitalized approximately \$2,350, \$1,482 and \$1,340, respectively, as internally developed software. The increase in the amount of technology development expenditures and the increase in the amount we capitalized in 2012 compared to 2011 is a result of technology development expenditures related to Envestnet | Tamarac products. We expect to continue focusing our technology development efforts principally on adding features to increase our market competitiveness, enhancements to improve operating efficiency and reduce risk, and client-driven requests for new capabilities.

#### **Intellectual Property and Proprietary Rights**

We rely on a combination of trademark, copyright and trade secret protection laws to protect our proprietary technology and our intellectual property. We seek to control access to and distribution of our proprietary information. We enter into confidentiality agreements with our employees, consultants, customers and vendors that generally provide that any confidential or proprietary information developed by us or on our behalf be kept confidential. In the normal course of business, we provide our intellectual property to third parties through licensing or restricted use agreements. We have proprietary know-how in algorithms, implementation and business on-boarding functions, along with a wide variety of applications software. We have registered the mark "ENVESTNET" with the U.S. Patent and Trademark Office in addition to several marks commonly used by Envestnet. We also pursue the registration of certain of our other trademarks and service marks in the United States. In addition, we have registered our domain name, www.envestnet.com and several other websites, such as www.envestnetpmc.com, www.investpmc.com, www.fiduciaryopportunity.com and www.envestnetadvisor.com. We have established a system of security measures to protect our computer systems from security breaches and computer viruses. We have employed various technology and process-based methods, such as clustered and multi-level firewalls, intrusion detection mechanisms, vulnerability assessments, content filtering, antivirus software and access control mechanisms. We also use encryption techniques for data transmissions. We control and limit access to confidential and proprietary information on a "need to know" basis.

#### Regulation

#### **Overview**

The financial services industry is among the most extensively regulated industries in the United States. We operate investment advisory, broker-dealer and mutual fund advisory businesses, each of which is subject to a specific regulatory scheme, including regulation at the federal and state level, as well as regulation by self-regulatory organizations and non-U.S. regulatory authorities. In addition, we are subject to numerous laws and regulations of general application.

Our wholly-owned subsidiaries, Envestnet Asset Management, Inc., Portfolio Management Consultants, Inc. and Envestnet Portfolio Solutions, Inc. operate investment advisory businesses. These subsidiaries are registered with the U.S. Securities and Exchange Commission ("SEC") as "investment advisers" under the Investment Advisers Act of 1940, as amended (the "Advisers Act"), and are regulated thereunder. As described further below, many of our investment advisory programs are conducted pursuant to the non-exclusive safe harbor from the definition of an "investment company" provided for under Rule 3a-4 of the Investment Company Act of 1940, as amended (the "Investment Company Act"). If Rule 3a-4 were to cease to be available, or if the SEC were to modify the rule or its interpretation of how the rule is applied, it could have a substantial effect on our business. Envestnet Asset Management, Inc. serves as the investment adviser to two mutual funds and a series of collective investment trusts managed towards retirement target dates. Mutual funds are registered as "investment companies" under the Investment Company Act. The Advisers Act and the Investment Company Act, together with related regulations and interpretations of the SEC, impose numerous obligations and restrictions on investment advisers and mutual funds, including recordkeeping requirements, limitations on advertising, disclosure and reporting obligations, prohibitions on fraudulent activities, and detailed operating requirements, including restrictions on transactions between an adviser and its clients, and between a mutual fund and its advisers and affiliates. The fiduciary obligations of investment advisers to their clients require advisers to, among other things, consider the suitability of the investment products and advice they provide, seek "best execution" for their clients' securities transactions, conduct due diligence on third-party products offered to clients, consider the appropriateness of the adviser's fees, and provide extensive and ongoing disclosure to clients. The application of these requirements to wrap fee programs is particularly complex and the SEC has in the past scrutinized firms' compliance with these requirements. The SEC is authorized to institute proceedings and impose fines and sanctions for violations of the Advisers Act and the Investment Company Act and has the power to restrict or prohibit an investment adviser from carrying on its business in the event that it fails to comply with applicable laws and regulations. Although we believe we are in compliance in all material respects with the requirements of the Advisers Act and the Investment Company Act and the rules and interpretations promulgated thereunder, our failure to comply with such laws, rules and interpretations could have a material adverse effect on us. Collective investment trusts are unregistered funds operated by a trust company or a bank regulated by the Office of the Comptroller of Currency (the "OCC"). In our role as advisor to these funds, we operate under substantially similar obligations to those discussed above for mutual funds.

Portfolio Brokerage Services, Inc., ("PBS"), our broker-dealer subsidiary, is registered as a broker-dealer with the SEC under the Securities Exchange Act of 1934, ("Exchange Act"), in all 50 states and the District of Columbia. In addition, PBS is a member of the Financial Industry Regulatory Authority ("FINRA"), the securities industry self-regulatory organization that supervises and regulates the conduct and activities of broker-dealers. Broker-dealers are subject to regulations that cover all aspects of their business, including sales practices, market making and trading among broker-dealers, use and safekeeping of customers' funds and securities, capital structure, record-keeping and the conduct of directors, officers, employees, representatives and associated persons. FINRA and the SEC conduct periodic examinations of the operations of its members, including PBS. Violation of applicable regulations can result in the suspension or revocation of a broker-dealer's registration, the imposition of censures or fines and the suspension or expulsion of the brokerdealer from FINRA. PBS is subject to minimum net capital requirements under the Exchange Act, SEC and FINRA rules and conducts its business pursuant to the exemption from the SEC's customer protection rule provided by Rule 15c3-3(k)(2)(i) under the Exchange Act. As of December 31, 2012, PBS was required to maintain a minimum of \$100 in net capital and its actual net capital was \$968.

Our regulated subsidiaries are subject to various federal and state laws and regulations that grant supervisory agencies, including the SEC, broad administrative powers. In the event of a failure to comply with these laws and regulations, the possible sanctions that may be imposed include the suspension of individual employees, limitations on the permissibility of our regulated subsidiaries and our other subsidiaries to engage in business for specified periods of time, censures, fines, and the revocation of registration as a broker-dealer or investment adviser, as applicable. Additionally, the securities laws applicable to us and our subsidiaries provide for certain private rights of action that could give rise to civil litigation. Any litigation could have significant financial and non-financial consequences including monetary judgments and the requirement to take action or limit activities that could ultimately affect our business.

Additional legislation and regulations, including those relating to the activities of investment advisers and broker-dealers, changes in rules imposed by the SEC or other regulatory authorities and self-regulatory organizations, or changes in the interpretation or enforcement of existing laws and rules may adversely affect our business and profitability. Our businesses may be materially affected not only by regulations applicable to it as an investment adviser or broker-dealer, but also by regulations that apply to companies generally.

#### Investment Advisory Program Conducted Under Rule 3a-4

Under the Investment Company Act, an issuer that is engaged in the business of investing, reinvesting or trading in securities may be deemed an "investment company," in which case the issuer may be subject to registration requirements and regulation as an investment company under the Investment Company Act. In order to provide assurance that certain discretionary investment advisory programs would not be considered investment company for programs that meet the requirements of the rule. We conduct the following programs pursuant to the Rule 3a-4 safe harbor:

- Separately managed accounts;
- Unified managed account portfolios;
- Mutual fund portfolios and exchange-traded fund portfolios; and
- Advisor as portfolio manager.

We believe that, to the extent we exercise discretion over accounts in any of these programs, these programs qualify for the safe-harbor because all of the programs have the following characteristics, which are generally required in order for a program to be eligible for the Rule 3a-4 safe harbor:

- Each client account is managed on the basis of the client's financial situation, investment objectives and reasonable client-imposed investment restrictions;
- At the opening of the account, the client's financial advisor obtains information from the client and provides us with the client's financial situation, investment objectives and reasonable restrictions;
- On no less than an annual basis, the client's financial advisor contacts the client to determine whether there have been any changes in the client's financial situation or investment objectives, and whether the client wishes to impose any reasonable restrictions on the management of the account or reasonably modify existing restrictions. This information is communicated to us and reflected in our management of client accounts;
- On a quarterly basis, we or another designated person (in most cases this will be the client's financial advisor) notify the client to contact us or another designated person if there have been any changes to the client's financial position or investment objectives or if the client wishes to impose any reasonable restrictions on the management of the account;
- We, the client's financial advisor and the manager of the client's account, all of whom are knowledgeable about the account and its management, are reasonably available to the client for consultation;

- All of the programs allow each client to impose reasonable restrictions on the management of his or her account;
- On at least a quarterly basis, the client is provided with a statement containing a description of all activity in the client's account during the preceding period, including all transactions made on behalf of the account, all contributions and withdrawals made by the client, all fees and expenses charged to the account, and the value of the account at the beginning and end of the period; and
- For all of the programs, each client retains, with respect to all securities and funds in the client's account, the right to withdraw securities or cash, vote securities, or delegate the authority to vote securities to another person, receive written confirmation or other notification of each securities transaction by the client's independent custodian, and proceed directly as a security holder against the issuer of any security in the client's account without the obligation to include us or any other client of the program in any such action as a condition precedent to initiating such proceeding.

#### Employees

As of December 31, 2012, we had 769 employees, including 115 in sales and marketing, 235 in engineering and systems, 342 in operations, 23 in investment management and research, and 54 in executive and corporate functions. Of these 769 employees, 326 were located in India. None of our employees is represented by a labor union. We have never experienced a work stoppage and believe our relationship with our employees is good.

#### **Executive Officers of the Registrant**

The following table summarizes information about each one of our executive officers.

Name Age		Position(s)
Judson Bergman	56	Chairman, Chief Executive Officer, Director
William Crager	49	President
Peter D'Arrigo	45	Chief Financial Officer
Scott Grinis	51	Chief Technology Officer
Shelly O'Brien	47	Chief Legal Officer, General Counsel and Corporate Secretary
Brandon Thomas	49	Chief Investment Officer

Judson Bergman, See information disclosed in "Item 10. Directors, Executive Officers and Corporate Governance."

*William Crager*, Age 49. Mr. Crager has served as our President since 2002. Prior to joining us, Mr. Crager served as Managing Director of Marketing and Client Services at Rittenhouse Financial Services, Inc., an investment management firm affiliated with Nuveen Investments, Inc. ("Nuveen"), a diversified investment manager. Mr. Crager received an MA from Boston University and a BA from Fairfield University, with a dual major in economics and English.

Peter D'Arrigo, Age 45. Mr. D'Arrigo has served as our Chief Financial Officer since 2008. Prior to joining us, Mr. D'Arrigo worked at Nuveen where he served as Treasurer since 1999, as well as holding a variety of other titles after joining them in 1990. Mr. D'Arrigo received an MBA from the Northwestern University Kellogg Graduate School of Management and an undergraduate degree in applied mathematics from Yale University.

Scott Grinis, Age 51. Mr. Grinis has served as our Chief Technology Officer since 2004. Prior to joining us, Mr. Grinis co-founded Oberon Financial Technology, Inc., our subsidiary, prior to its acquisition by us. Mr. Grinis received a BS and an MS degree in electrical engineering from Stanford University.

Shelly O'Brien, Age 47. Ms. O'Brien has served as our Chief Legal Officer, General Counsel and Corporate Secretary since 2002. Prior to joining us, Ms. O'Brien was General Counsel and Director of Legal and

Compliance for ING (U.S.) Securities, Futures & Options Inc., a broker-dealer, and futures commission merchant. Ms. O'Brien received a degree in political science from Northwestern University, a JD from Hamline University School of Law, and an LLM in taxation from John Marshall Law School.

*Brandon Thomas*, Age 49. Mr. Thomas is a co-founder of our company and has served as Chief Investment Officer and Managing Director of Portfolio Management Consultants, our internal investment management and portfolio consulting group, since 1999. Prior to joining us, Mr. Thomas was Director of Equity Funds for Nuveen. Mr. Thomas received an MBA from the University of Chicago, a JD from DePaul University and is a graduate of Brown University.

#### Item 1A. Risk Factors

Investment in our securities involves risk. An investor or potential investor should consider the risks summarized in this section when making investment decisions regarding our securities. These risks and uncertainties include, but are not limited to, the risk factors set forth below. The risks and uncertainties described in this section are not the only ones we face. Additional risks and uncertainties not presently known to us or that we currently believe are immaterial may also affect our business. If any of these known or unknown risks or uncertainties actually occurs, our business, financial condition and results of operations could be materially adversely affected.

#### **Risks Related to Our Business**

# We have experienced rapid revenue growth, which may be difficult to sustain and which may place significant demands on our administrative, operational and financial resources and any inability to maintain or manage our growth could have a material adverse effect on our results of operations, financial condition or business.

Our revenues during the three years ended December 31, 2012 have grown at a compound annual growth rate of 27%. We expect our growth to continue, which could place additional demands on our resources and increase our expenses. Our future growth will depend on, among other things, our ability to successfully grow our total assets under management and administration and add additional clients. If we are unable to implement our growth strategy, develop new investment solutions and services and gain new clients, our results of operations, financial condition or business may be materially adversely affected.

Sustaining growth will also require us to commit additional management, operational and financial resources and to maintain appropriate operational and financial systems. In addition, continued growth increases the challenges involved in:

- recruiting, training and retaining sufficiently skilled technical, marketing, sales and management personnel;
- preserving our culture, values and entrepreneurial environment;
- successfully expanding the range of investment solutions and services offered to our clients;
- developing and improving our internal administrative infrastructure, particularly our financial, operational, compliance, record-keeping, communications and other internal systems; and
- maintaining high levels of satisfaction with our investment solutions and services among clients.

There can be no assurance that we will be able to maintain or accelerate our growth, and any failure to do so could adversely affect our results of operations, financial condition or business.

#### Our revenue can fluctuate from period to period, which could cause our share price to fluctuate.

Our revenue may fluctuate from period-to-period in the future due to a variety of factors, many of which are beyond our control. Factors relating to our business that may contribute to these fluctuations include the following events, as well as other factors described elsewhere in this document:

- a decline or slowdown of the growth in the value of financial market assets, which may reduce the value of assets under management and administration and therefore our revenues and cash flows;
- negative public perception and reputation of the financial services industry, which would reduce demand for our investment solutions and services;
- unanticipated changes to economic terms in contracts with clients, including renegotiations;
- · downward pressure on fees we charge our clients, which would therefore reduce our revenue;
- changes in laws or regulations that could impact our ability to offer investment solutions and services;
- failure to obtain new clients;
- cancellation or non-renewal of existing contracts with clients;
- failure to protect our proprietary technology and intellectual property rights;
- unanticipated delays in connection with the conversion of client assets onto our technology platform;
- · reduction in the suite of investment solutions and services provided to existing clients; or
- changes in our pricing policies or the pricing policies of our competitors to which we have to adapt.

As a result of these and other factors, the results of operations for any quarterly or annual period may differ materially from the results of operations for any prior or future quarterly or annual period and should not be relied upon as indications of our future performance.

# We operate in a highly competitive industry, with many firms competing for business from financial advisors on the basis of a number of factors, including the quality and breadth of investment solutions and services, ability to innovate, reputation and the prices of services and this competition could hurt our financial performance.

We compete with many different types of companies that vary in size and scope, including Pershing LLC (a subsidiary of BNY Mellon Corporation), The Charles Schwab Corporation, SEI Investments Company, Genworth Financial Inc., Advent Software, Inc., and Lockwood Advisors (a subsidiary of BNY Mellon Corporation) and which are discussed in greater detail under "Business—Competition" included in this Form 10-K. In addition, some of our clients have developed or may develop the in-house capability to provide the technology and/or investment advisory services they have retained us to perform. These clients may also offer internally developed services to their financial advisors, obviating the need to hire us, and they may offer these services to thirdparty financial advisors or financial institutions, thereby competing directly with us for that business.

Many of our competitors have significantly greater resources than we do. These resources may allow our competitors to respond more quickly to changes in demand for investment solutions and services, to devote greater resources to developing and promoting their services and to make more attractive offers to potential clients and strategic partners, which could hurt our financial performance.

We may lose clients as a result of the sale or merger of a client, a change in a client's senior management, competition from other financial advisors and financial institutions and for other reasons. We also face increased competition due to the current trend of industry consolidation. If large financial institutions that are not our clients are able to attract assets from our clients, our ability to grow revenues and earnings may be adversely affected.

Our Envestnet | PMC group competes with other providers of investment solutions and products. These competitors may offer broader solutions and/or products and their solutions and/or products may have better investment returns during one or more periods. If the investment returns on our investment products are not perceived to be competitive, we could experience outflows of assets from these products and face difficulty attracting new assets to these products.

Our failure to successfully compete in any of the above-mentioned areas could have a material adverse effect on our results of operations, financial condition or business. Competition could also affect the revenue mix of services we provide, resulting in decreased revenues in lines of business with higher profit margins.

# We derive nearly all of our revenues from the delivery of investment solutions and services to clients in the financial advisory industry and our revenue could suffer if that industry experiences a downturn.

A decline or lack of growth in demand for financial advisory services would adversely affect our clients and, in turn, our results of operations, financial condition and business. For example, the availability of free or low-cost investment information and resources, including research and information relating to publicly traded companies and mutual funds available on the Internet or on company websites, could lead to lower demand by investors for the services provided by financial advisors. In addition, demand for our investment solutions and services among financial advisors could decline for many reasons. Consolidation or limited growth in the financial advisory industry could reduce the number of our clients and potential clients. Events that adversely affect our clients' businesses, rates of growth or the numbers of customers they serve, including decreased demand for our clients' products and services, adverse conditions in our clients' markets or adverse economic conditions generally, could decrease demand for our investment solutions and services and thereby decrease our revenues. Any of the foregoing could have a material adverse effect on our results of operations, financial condition or business.

# A limited number of clients account for a material portion of our revenue. Termination of our contracts with any of these clients could have a material adverse effect on our results of operations, financial condition or business.

For the years ended December 31, 2012, 2011 and 2010, revenues associated with our relationship with our single largest client, FMR LLC, an affiliate of FMR Corp., or Fidelity, accounted for 22%, 31% and 31% respectively, of our total revenues and our ten largest clients accounted for 47%, 64% and 62%, respectively, of our total revenues. Our license agreements with large financial institutions are generally multi-year contracts that may be terminated upon the expiration of the contract term or prior to such time for cause, which may include breach of contract, bankruptcy, insolvency and other reasons. The license fee payments pursuant to our license agreement with Fidelity were renegotiated and reduced as of December 31, 2011 and were extended for an additional five years. The agreement, as amended, includes receiving ongoing platform services fees through the Fidelity relationship based upon asset-based fees. No assurance can be given as to whether any projected increase in asset-based fees will offset the reduction in license fees. A majority of our agreements with financial advisors generally provides for termination at any time. If our contractual relationship with Fidelity were to terminate, or if a significant number of our most important clients were to terminate their contracts with us and we were unable to obtain a significant number of new clients, our results of operations, financial condition or business could be materially adversely affected.

# Our clients that pay us an asset-based fee may seek to negotiate a lower fee percentage or may cease using our services, which could limit the growth of, or decrease, our revenues.

A significant portion of our revenues are derived from asset-based fees. Our clients may, for a number of reasons, seek to negotiate a lower asset-based fee percentage. For example, an increase in the use of index-linked investment products by the clients of our financial advisor clients may result in lower fees being paid to our clients, and our clients may in turn seek to negotiate lower asset-based fee percentages for our services. In addition, as competition among our clients increases, they may be required to lower the fees they charge to their

clients, which could cause them to seek to decrease our fees accordingly. Any of these factors could result in fluctuation or a decline in our asset-based fees, which would have a material adverse effect on our results of operations, financial condition or business.

# Changes in market and economic conditions could lower the value of assets on which we earn revenues and could decrease the demand for our investment solutions and services.

Asset-based fees make up a significant portion of our revenues. Asset-based fees represented 81%, 81% and 77% of our total revenues for the years ended December 31, 2012, 2011 and 2010, respectively. We expect that asset-based fees will continue to represent a significant percentage of our total revenues in the future. Significant fluctuations in securities prices may materially affect the value of the assets managed by our clients and may also influence financial advisor and investor decisions regarding whether to invest in, or maintain an investment in, a mutual fund or other investment solution. If such market fluctuation led to less investment in the securities markets, our revenues and earnings derived from asset-based fees could be materially affected.

We provide our investment solutions and services to the financial services industry. The financial markets, and in turn the financial services industry, are affected by many factors, such as U.S. and foreign economic conditions and general trends in business and finance that are beyond our control. In the event that the U.S. or international financial markets suffer a severe or prolonged downturn, investors may choose to withdraw assets from financial advisors, which we refer to as "redemptions", and transfer them to investments that are perceived to be more secure, such as bank deposits and Treasury securities. For example, in late 2007 and through the first quarter of 2009, the financial markets experienced a broad and prolonged downturn, our redemption rates were higher than our historical average, and our results of operations, financial condition and business were materially adversely affected. Any prolonged downturn in financial markets or increased levels of asset withdrawals could have a material adverse effect on our results of operations, financial condition or business.

# Investors' decisions regarding their investment assets are affected by many factors and investors may redeem or withdraw their investment assets generally at any time. Significant changes in investing patterns or large-scale withdrawal of investment funds could have a material adverse effect on our results of operations, financial condition or business.

The clients of our financial advisors are generally free to change financial advisors, forgo the advice and other services provided by financial advisors or withdraw the funds they have invested with financial advisors. These clients of financial advisors may elect to change their investment strategies, by moving their assets away from equity securities to fixed income or other investment options, or by withdrawing all or a portion of their assets from their accounts to avoid all securities markets-related risks. These actions by investors are outside of our control and could materially adversely affect the market value of the investment assets that our clients manage, which could materially adversely affect the asset-based fees we receive from our clients.

#### We are subject to liability for losses that result from a breach of our fiduciary duties.

Our investment advisory services involve fiduciary obligations that require us to act in the best interests of our clients, and we may be sued and face liabilities for actual or claimed breaches of our fiduciary duties. Because we provide investment advisory services, both directly and indirectly, with respect to substantial assets we could face substantial liability to our clients if it is determined that we have breached our fiduciary duties. In certain circumstances, which generally depend on the types of investment solutions and services we are providing, we may enter into client agreements jointly with advisors and retain third-party investment money managers on behalf of clients. As a result, we may be included as a defendant in lawsuits against financial advisors and third-party investment money managers that involve claims of breaches of the duties of such persons, and we may face liabilities for the improper actions and/or omissions of such advisors and third-party investment money managers. In addition, we may face claims based on the results of our investment advisory recommendations, even in the absence of a breach of our fiduciary duty. Such claims and liabilities could therefore have a material adverse effect on our results of operations, financial condition or business.

#### We are subject to liability for losses that result from potential, perceived or actual conflicts of interest.

Potential, perceived and actual conflicts of interest are inherent in our existing and future business activities and could give rise to client dissatisfaction, litigation or regulatory enforcement actions. In particular, we pay varying fees to third-party asset managers and custodians and our financial advisor customers, or their clients, could accuse us of directing them toward those asset managers or custodians that charge us the lowest fees. In addition, we offer proprietary mutual funds and portfolios of mutual funds through our internal investment management and portfolio consulting group, and financial advisors or their clients could conclude that we favor our proprietary investment products because of their belief that we earn higher fees when our proprietary investment products are used. Adequately addressing conflicts of interest is complex and difficult and if we fail, or appear to fail, to adequately address potential, perceived or actual conflicts of interest, the resulting negative public perception and reputational harm could materially adversely affect our client relations or ability to enter into contracts with new clients and, consequently, our results of operations, financial condition and business.

#### If our reputation is harmed, our results of operations, financial condition or business could be materially adversely affected.

Our reputation, which depends on earning and maintaining the trust and confidence of our clients, is critical to our business. Our reputation is vulnerable to many threats that can be difficult or impossible to control, and costly or impossible to remediate. Regulatory inquiries or investigations, lawsuits initiated by our clients or stockholders, employee misconduct, perceptions of conflicts of interest and rumors, among other developments, could substantially damage our reputation, even if they are baseless or satisfactorily addressed. In addition, any perception that the quality of our investment solutions and services may not be the same or better than that of other providers, can also damage our reputation. Any damage to our reputation could harm our ability to attract and retain clients, which would materially adversely affect our results of operations, financial condition and business.

# If our investment solutions and services fail to perform properly due to undetected errors or similar problems, our results of operations, financial condition and business could be materially adversely affected.

Investment solutions and services we develop or license may contain undetected errors or defects despite testing. Such errors can exist at any point in the life cycle of our investment solutions or services, but are frequently found after introduction of new investment solutions and services or enhancements to existing investment solutions or services. We continually introduce new investment solutions and services and new versions of our investment solutions and services. Despite internal testing and testing by current and potential clients, our current and future investment solutions and services may contain serious defects or malfunctions. If we detect any errors before release, we might be required to delay the release of the investment solutions, services or enhancements until after they are deployed, and we may need to provide enhancements to correct such errors. Errors may occur that could have a material adverse effect on our results of operations, financial condition or business and could result in harm to our reputation, lost sales, delays in commercial release, third-party claims, contractual disputes, contract terminations or renegotiations, or unexpected expenses and diversion of management and other resources to remedy errors. In addition, negative public perception and reputational damage caused by such claims would adversely affect our client relationships and our ability to enter into new contracts. Any of these problems could have a material adverse effect on our results of operations, financial condition and business.

#### We could face liability or incur costs to remediate operational errors or to address possible customer dissatisfaction.

Operational risk generally refers to the risk of loss resulting from our operations, including, but not limited to, improper or unauthorized execution and processing of transactions, deficiencies in our operating systems, business disruptions and inadequacies or breaches in our internal control processes. We operate in diverse markets and are reliant on the ability of our employees and systems to process large volumes of transactions often within short time frames. In the event of a breakdown or improper operation of systems, human error or improper action by employees, we could suffer financial loss, regulatory sanctions or damage to our reputation.

In addition, there may be circumstances when our customers are dissatisfied with our investment solutions and services, even in the absence of an operational error. In such circumstances, we may elect to make payments or otherwise incur increased costs or lower revenues in order to maintain a strong customer relationship. In any of the forgoing circumstances, our results of operations, financial condition or business could be materially adversely affected.

#### We may become subject to liability based on the use of our investment solutions and services by our clients.

Our investment solutions and services support the investment processes of our clients, which, in the aggregate, manage billions of dollars of assets. Our client agreements have provisions designed to limit our exposure to potential liability claims brought by our clients or third parties based on the use of our investment solutions and services. However, these provisions have certain exceptions and could be invalidated by unfavorable judicial decisions or by federal, state, foreign or local laws. Use of our products as part of the investment process creates the risk that clients, or the parties whose assets are managed by our clients, may pursue claims against us for very significant dollar amounts. Any such claim, even if the outcome were to be ultimately favorable to us, would involve a significant commitment of our management, personnel, financial and other resources and could have a negative impact on our reputation. Such claims and lawsuits could therefore have a material adverse effect on our results of operations, financial condition or business.

Furthermore, our clients may use our investment solutions and services together with software, data or products from other companies. As a result, when problems occur, it might be difficult to identify the source of the problem. Even when our investment solutions and services do not cause these problems, the existence of these errors might cause us to incur significant costs and divert the attention of our management and technical personnel, any of which could materially adversely affect our results of operations, financial condition or business.

## Our business relies heavily on computer equipment, electronic delivery systems and the Internet. Any failures or disruptions in such technologies could result in reduced revenues, increased costs and the loss of customers.

Our business relies heavily on our computer equipment (including our servers), electronic delivery systems and the Internet, but these technologies are vulnerable to disruptions, failures or slowdowns caused by fire, earthquake, power loss, telecommunications failure, terrorist attacks, wars, Internet failures, computer viruses and other events beyond our control. Furthermore, we rely on agreements with our suppliers, such as our current data hosting and service provider, to provide us with access to certain computer equipment, electric delivery systems and the Internet. We are unable to predict whether a future contractual dispute may arise with one of our suppliers that could cause a disruption in service, or whether our agreements with our suppliers can be obtained or renewed on acceptable terms, or at all. An unanticipated disruption, failure or slowdown affecting our key technologies or facilities may have significant ramifications, such as data-loss, data corruption, damaged software codes or inaccurate processing of transactions. We maintain off-site back-up facilities for our electronic information and computer equipment, but these facilities could be subject to the same interruptions that may

affect our primary facilities. Any significant disruptions, failures, slowdowns, data-loss or data corruption could have a material adverse effect on our results of operations, financial condition or business and result in the loss of customers.

#### We could face liability related to disclosure or theft of the personal information we store on our technology platform.

Clients may maintain personal investment and financial information on our technology platform and we could be subject to liability if we were to inappropriately disclose any user's personal information, inadvertently or otherwise, or if third parties were able to penetrate our network security or otherwise gain access to any user's name, address, portfolio holdings or other financial information. Any such event could subject us to claims for misuses of personal information, such as unauthorized marketing or unauthorized access to personal portfolio information and could therefore have a material adverse effect on our results of operations, financial condition or business.

#### We could incur significant costs protecting the personal information we store on our technology platform.

Users of our investment solutions and services are located in the United States and around the world. As a result, we collect and store the personal information of individuals who live in many different countries. Privacy regulators in some of those countries have publicly stated that foreign entities (including entities based in the United States) may render themselves subject to those countries' privacy laws and the jurisdiction of such regulators by collecting or storing the personal data of those countries' residents, even if such entities have no physical or legal presence there. Consequently, we may be obligated to comply with the privacy and data security laws of such foreign countries. Our exposure to foreign countries' privacy and data security laws impacts our ability to collect and use personal information, increases our legal compliance costs and may expose us to liability.

We have incurred, and will continue to incur, expenses to comply with privacy and security standards and protocols imposed by law, regulation, industry standards or contractual obligations. Increased domestic or international regulation of data utilization and distribution practices could require us to modify our operations and incur significant additional expense, which could have a material adverse effect on our results of operations, financial condition or business.

#### We could face liability for certain information we provide, including information based on data we obtain from other parties.

We may be subject to claims for securities law violations, negligence, breach of fiduciary duties or other claims relating to the information we provide. For example, individuals may take legal action against us if they rely on information we have provided and it contains an error. In addition, we could be subject to claims based upon the content that is accessible from our website through links to other websites. Moreover, we could face liability based on inaccurate information provided to us by others. Defending any such claims could be expensive and time-consuming, and any such claim could materially adversely affect our results of operations, financial condition or business.

# We depend on our senior management team and other key personnel and the loss of their services could have a material adverse effect on our results of operations, financial condition or business.

We depend on the efforts, relationships and reputations of our senior management team and other key personnel, including Judson Bergman, our Chief Executive Officer, William Crager, our President, and Scott Grinis, our Chief Technology Officer, in order to successfully manage our business. We believe that success in our business will continue to be based upon the strength of our intellectual capital. The loss of the services of any member of our senior management team or of other key personnel could have a material adverse effect on our results of operations, financial condition or business.

# Our operations are subject to extensive government regulation, and compliance failures or regulatory action against us could adversely affect our results of operations, financial condition or business.

The financial services industry is among the most extensively regulated industries in the United States. We operate investment advisory, broker-dealer, mutual fund and collective investment trust lines of business, each of which is subject to a specific and extensive regulatory scheme. In addition, we are subject to numerous laws and regulations of general application. It is very difficult to predict the future impact of the legislative and regulatory requirements affecting our business and our clients' businesses.

Certain of our subsidiaries are registered as "investment advisers" with the SEC under the Advisers Act and are regulated thereunder. In addition, many of our investment advisory services are conducted pursuant to the non-exclusive safe harbor from the definition of an "investment company" provided under Rule 3a-4 under the Investment Company Act. If Rule 3a-4 were to cease to be available, or if the SEC were to modify the rule or its interpretation of how the rule is applied, our business could be adversely affected. Certain of our registered investment adviser subsidiaries provide advice to mutual fund clients and collective investment trusts. Mutual funds are registered as "investment companies" under the Investment Company Act. The Advisers Act and the Investment Company Act, together with related regulations and interpretations of the SEC, impose numerous obligations and restrictions on investment advisers and mutual funds, including requirements relating to the safekeeping of client funds and securities, limitations on advertising, disclosure and reporting obligations, prohibitions on fraudulent activities, restrictions on transactions between an adviser and its clients, and between a mutual fund and its advisers and affiliates, and other detailed operating requirements, as well as general fiduciary obligations. Collective investment trusts are unregistered funds operated by a trust company or a bank regulated by the OCC. In our role as advisor to these funds, we operate under substantially similar obligations to those discussed above for mutual funds.

In addition, PBS, our broker-dealer subsidiary, is registered as a broker-dealer with the SEC and with all 50 states and the District of Columbia, and is a member of FINRA, a securities industry self-regulatory organization that supervises and regulates the conduct and activities of its members. Broker-dealers are subject to regulations that cover all aspects of their business, including sales practices, market making and trading among broker-dealers, use and safekeeping of customer funds and securities, capital structure, recordkeeping and the conduct of directors, officers, employees, representatives and associated persons. FINRA conducts periodic examinations of the operations of its members, including PBS. As a broker-dealer, PBS is also subject to certain minimum net capital requirements under SEC and FINRA rules. Compliance with the net capital rules may limit our ability to withdraw capital from PBS.

All of the foregoing laws and regulations are complex and we are required to expend significant resources in order to maintain our compliance with such laws and regulations. Any failure on our part to comply with these and other applicable laws and regulations could result in regulatory fines, suspensions of personnel or other sanctions, including revocation of our registration or that of our subsidiaries as an investment adviser or broker-dealer, as the case may be, which could, among other things, require changes to our business practices and scope of operations or harm our reputation, which, in turn could have a material adverse effect on our results of operations, financial condition or business.

### Changes to the laws or regulations applicable to us or to our financial advisor clients could adversely affect our results of operations, financial condition or business.

We may be adversely affected as a result of new or revised legislation or regulations imposed by the SEC or other U.S. or foreign governmental regulatory authorities or self-regulatory organizations that supervise the financial markets around the world. In addition, we may be adversely affected by changes in the interpretation or enforcement of existing laws and rules by these governmental authorities and self-regulatory organizations. It is impossible to determine the extent of the impact of any new laws, regulations or initiatives that may be proposed, or whether any current proposals will become law, and it is difficult to predict how any changes or potential changes could affect our business. Changes to laws or regulations could increase our potential liability in

connection with the investment solutions and services that we provide. The introduction of any new laws or regulations could make our ability to comply with applicable laws and regulations more difficult and expensive. Any of the foregoing could have a material adverse effect on our results of operations, financial condition or business.

# A deemed "change of control" of our company could require us to obtain the consent of our clients and a failure to do so properly could adversely affect our results of operations, financial condition or business.

Under the Advisers Act, the investment advisory agreements entered into by our investment adviser subsidiaries may not be assigned without the client's consent. Under the Investment Company Act, advisory agreements with registered funds terminate automatically upon assignment and, any assignment of an advisory agreement must be approved by the board of directors and the shareholders of the registered fund. Under the Advisers Act and the Investment Company Act, such an assignment may be deemed to occur upon a change of control of the Company. A change of control includes both gaining or losing a "controlling person." Whether someone is a controlling person for these purposes depends significantly on the specific facts and circumstances. There can be no assurance that if we undergo a change of control, we would be successful in obtaining all necessary consents or that the method by which we obtain such consents could not be challenged at a later time. If we are unable to obtain all necessary consents or if such a challenge were to be successful it could have a material adverse effect on our results of operations, financial condition or business.

## We rely on exemptions from certain laws and if for any reason these exemptions were to become unavailable to us, we could become subject to regulatory action or third-party claims and our business could be materially and adversely affected.

We regularly rely on exemptions from various requirements of the Exchange Act, the Investment Company Act and the Employee Retirement Income Security Act of 1974 in conducting our activities. These exemptions are sometimes highly complex and may in certain circumstances depend on compliance by third parties whom we do not control. If for any reason these exemptions were to become unavailable to us, we could become subject to regulatory action or third-party claims and our business could be materially and adversely affected.

# If government regulation of the Internet or other areas of our business changes, or if consumer attitudes toward use of the Internet change, we may need to change the manner in which we conduct our business or incur greater operating expenses.

The adoption, modification or interpretation of laws or regulations relating to the Internet or other areas of our business could adversely affect the manner in which we conduct our business. Such laws and regulations may cover sales practices, taxes, user privacy, data protection, pricing, content, copyrights, distribution, electronic contracts, consumer protection, broadband residential Internet access and the characteristics and quality of services. Moreover, it is not clear how existing laws governing these matters apply to the Internet. If we are required to comply with new regulations or legislation or new interpretations of existing regulations or legislation, we may be required to incur additional expenses or alter our business model, either of which could have a material adverse effect on our results of operations, financial condition or business.

# We are substantially dependent on our intellectual property rights, and a failure to protect these rights could adversely affect our results of operations, financial condition or business.

We have made substantial investments in software and other intellectual property on which our business is highly dependent. We rely on trade secret, trademark and copyright laws, confidentiality and nondisclosure agreements and other contractual and technical security measures to protect our proprietary technology. Any loss of our intellectual property rights, or any significant claim of infringement or indemnity for violation of the intellectual property rights of others, could have a material adverse effect on our results of operations, financial condition or business.

None of our technologies, investment solutions or services is covered by any copyright registration, issued patent or patent application. We are the owner of eight registered trademarks in the United States, including "ENVESTNET," and we claim common law rights in other trademarks that are not registered. We cannot guarantee that:

- our intellectual property rights will provide competitive advantages to us;
- our ability to assert our intellectual property rights against potential competitors or to settle current or future disputes will not be limited by our
  agreements with third parties;
- our intellectual property rights will be enforced in jurisdictions where competition may be intense or where legal protection may be weak;
- any of the trademarks, copyrights, trade secrets or other intellectual property rights that we presently employ in our business will not lapse or be invalidated, circumvented, challenged or abandoned;
- our trademark applications will lead to registered trademarks; or
- competitors will not design around our intellectual property rights or develop similar technologies, investment solutions or products; or that we
  will not lose the ability to assert our intellectual property rights against others.

We are also a party to a number of third-party intellectual property license agreements. Some of these license agreements require us to make one-time payments or ongoing subscription payments. We cannot guarantee that the third-party intellectual property we license will not be licensed to our competitors or others in our industry. In the future, we may need to obtain additional licenses or renew existing license agreements. We are unable to predict whether these license agreements can be obtained or renewed on acceptable terms, or at all. In addition, we have granted our customers certain rights to use our intellectual property in the ordinary course of our business. Some of our customer agreements restrict our ability to license or develop certain customized technology or services within certain markets or to certain competitors of our customers. For example, our agreement with Fidelity restricts our ability to develop an enterprise-level integration or combination of products and services substantially similar to the technology platform we have developed for Fidelity. Some of our customers. In addition, some of our customer agreements require us to deposit the source code to the customized technology and investment solutions with a source code escrow agent, which source code may be released in the event we enter into bankruptcy or are unable to provide support and maintenance of the technology or investment solutions we have licensed to our customers. These provisions in our agreements may limit our ability to grow our business in the future.

## Third parties may sue us for intellectual property infringement or misappropriation which, if successful, could require us to pay significant damages or make changes to the investment solutions or services that we offer.

We cannot be certain that our internally developed or acquired technologies, investment solutions or services do not and will not infringe the intellectual property rights of others. In addition, we license content, software and other intellectual property rights from third parties and may be subject to claims of infringement if such parties do not possess the necessary intellectual property rights to the products they license to us. The risk of infringement claims against us will increase if more of our competitors are able to obtain patents for investment solutions or services or business processes. In addition, we face additional risk of infringement or misappropriation claims if we hire an employee who possesses third party proprietary information who decides to use such information in connection with our investment solutions, services or business processes without such third-party's authorization. We have in the past been and may in the future be subject to legal proceedings and claims that we have infringed or misappropriated the intellectual property rights of a third party. These claims sometimes involve patent holding companies who have no relevant product revenues and against whom our own proprietary technology may therefore provide little or no deterrence. In addition, third parties may in the future

assert intellectual property infringement claims against our customers, which, in certain circumstances, we have agreed to indemnify. Any intellectual property related infringement or misappropriation claims, whether or not meritorious, could result in costly litigation and could divert management resources and attention. Moreover, should we be found liable for infringement or misappropriation, we may be required to enter into licensing agreements, if available on acceptable terms or at all, pay substantial damages or make changes to the investment solutions and services that we offer. Any of the foregoing could prevent us from competing effectively, result in substantial costs to us, divert management's attention and our resources away from our operations and otherwise harm our reputation.

## If our intellectual property and proprietary technology are not adequately protected to prevent use or appropriation by our competitors, our business and competitive position would suffer.

Our future success and competitive position depend in part on our ability to protect our intellectual property rights. The steps we have taken to protect our intellectual property rights may be inadequate to prevent the misappropriation of our proprietary technology. There can be no assurance that others will not develop or patent similar or superior technologies, investment solutions or services. Unauthorized copying or other misappropriation of our proprietary technologies could enable third parties to benefit from our intellectual property rights without paying us for doing so, which could harm our business. Policing unauthorized use of proprietary technology is difficult and expensive and our monitoring and policing activities may not be sufficient to identify any misappropriation and protect our proprietary technology. In addition, third parties may knowingly or unknowingly infringe our trademarks and other intellectual property rights, and litigation may be necessary to protect and enforce our intellectual property rights. If litigation is necessary to protect our our intellectual property rights, any such litigation could be very costly and could divert management attention and resources. If we are unable to protect our intellectual property rights or if third parties independently develop or gain access to our or similar technologies, investment solutions or services, our results of operations, financial condition and business could be materially adversely affected.

#### The use of "open source code" in investment solutions may expose us to additional risks and harm our intellectual property rights.

To a limited extent, we rely on open source code to develop our investment solutions and support our internal systems and infrastructure. While we monitor our use of open source code to attempt to avoid subjecting our investment solutions to conditions we do not intend, such use could inadvertently occur. Additionally, if a third-party software provider has incorporated certain types of open source code into software we license from such third party for our investment solutions, we could, under certain circumstances, be required to disclose the source code for our investment solutions. This could harm our intellectual property position and have a material adverse effect on our results of operations, financial condition and business.

# Confidentiality agreements with employees, consultants and others may not adequately prevent disclosure of trade secrets and other proprietary information.

We have devoted substantial resources to the development of our proprietary technologies, investment solutions and services. In order to protect our proprietary rights, we enter into confidentiality agreements with our employees, consultants and independent contractors. These agreements may not effectively prevent unauthorized disclosure of confidential information or unauthorized parties from copying aspects of our technologies, investment solutions or products or obtaining and using information that we regard as proprietary. Moreover, these agreements may not provide an adequate remedy in the event of such unauthorized disclosures of confidential information and we cannot assure you that our rights under such agreements will be enforceable. In addition, others may independently discover trade secrets and proprietary information, and in such cases we could not assert any trade secret rights against such parties. Costly and time-consuming litigation could be necessary to enforce and determine the scope of our proprietary rights, and failure to obtain or maintain trade secret protection could reduce any competitive advantage we have developed and cause us to lose customers or otherwise harm our business.

# Our failure to successfully integrate acquisitions could strain our resources. In addition, there are significant risks associated with growth through acquisitions, which may materially adversely affect our results of operations, financial condition or business.

We expect to grow our business by, among other things, making acquisitions. In December 2011, we completed the acquisition of FundQuest and in the second quarter of 2012, we completed the acquisitions of Prima, Tamarac, and on April 11, 2013, we entered into a definitive agreement to acquire substantially all of the assets of the WMS, which we anticipate will be completed in the third quarter of 2013. Acquisitions involve a number of risks. They can be time-consuming and may divert management's attention from day-to-day operations. Financing an acquisition could result in dilution from issuing equity securities or a weaker balance sheet from using cash or incurring debt. Acquisitions might also result in losing key employees. In addition, we may fail to successfully integrate acquisitions successfully. We may also fail to generate enough revenues or profits from an acquisition to earn a return on the associated purchase price.

To the extent we grow our business through acquisitions, any such future acquisitions could present a number of other risks, including:

- incorrect assumptions regarding the future results of acquired operations or assets or expected cost reductions or other synergies expected to be realized as a result of acquiring operations or assets;
- · failure to integrate the operations or management of any acquired operations or assets successfully and on a timely and cost effective basis;
- insufficient knowledge of the operations and markets of acquired businesses;
- loss of key personnel;
- failure to obtain necessary customer consents or retain key customers;
- diversion of management's attention from existing operations or other priorities;
- increased costs or liabilities as a result of undetected or undisclosed legal, regulatory or financial issues related to acquired operations or assets; and
- inability to secure, on terms we find acceptable, sufficient financing that may be required for any such acquisition or investment.

In addition, if we are unsuccessful in completing acquisitions of other businesses, operations or assets or if such opportunities for expansion do not arise, our results of operations, financial condition or business could be materially adversely affected.

# Our failure to successfully execute the conversion of our clients' assets from their technology platform to our platform in a timely and accurate manner could have a material adverse effect on our results of operations, financial condition or business.

When we begin working with a new client, or acquire new client assets through an acquisition or other transaction, we are often required to convert the new assets from the clients' technology platform to our technology platform. These conversions present significant technological and operational challenges, can be time-consuming and may divert management's attention from other operational activities. If we fail to successfully complete our conversions in a timely and accurate manner, we may be required to expend more time and resources than anticipated, which could erode the profitability of the client relationship. In addition, any such failure may harm our reputation and may make it less likely that prospective clients will commit to working with us. Any of these risks could materially adversely affect our results of operations, financial condition or business.

#### Our business will suffer if we do not keep up with rapid technological change, evolving industry standards or changing requirements of clients.

We expect technological developments to continue at a rapid pace in our industry. Our success will depend, in part, on our ability to:

- continue to develop our technology expertise;
- recruit and retain skilled technology professionals;
- enhance our current investment solutions and services;
- develop new investment solutions and services that meet changing client needs;
- advertise and market our investment solutions and services;
- protect our proprietary technology and intellectual property rights; or
- influence and respond to emerging industry standards and other technological changes.

We must accomplish these tasks in a timely and cost-effective manner and our failure to do so could materially adversely affect our results of operations, financial condition or business.

# We must continue to introduce new investment solutions and services and investment solution and service enhancements to address our clients' changing needs, market changes and technological developments and failure to do so could have a material adverse effect on our results of operations, financial condition or business.

The market for our investment solutions and services is characterized by shifting client demands, evolving market practices and, for some of our investment solutions and services, rapid technological change. Changing client demands, new market practices or new technologies can render existing investment solutions and services obsolete and unmarketable. As a result, our future success will continue to depend upon our ability to develop new investment solutions and services and investment solution and service enhancements that address the future needs of our target markets and respond to technological and market changes. We incurred technology development expenditures of \$8,659, \$6,424 and \$5,588 in the years ended December, 31, 2012, 2011 and 2010, respectively. We expect that our technology development expenditures will continue at this level or they may increase in the future. We may not be able to accurately estimate the impact of new investment solutions and services on our business or how their benefits will be perceived by our clients. Further, we may not be successful in developing, introducing, marketing and licensing our new investment solutions or services or investment solutions or services and enhancements may not adequately meet the requirements of the marketplace or achieve market acceptance. In addition, clients may delay purchases in anticipation of new investment solutions or services or enhancements. Any of these factors could materially adversely affect our results of operations, financial condition or business.

#### In preparing our financial statements for 2012, we identified a material weakness in our internal control over financial reporting, and our failure to remedy this or other material weaknesses that we may identify in the future could result in material misstatements in our financial statements.

Our management is responsible for establishing and maintaining adequate internal control over our financial reporting, as defined in Rule 13a-15(f) under the Exchange Act. Our management identified a material weakness in our internal control over financial reporting as of December 31, 2012. A material weakness is defined as a deficiency, or combination of deficiencies, in internal control over financial reporting, such that there is a reasonable possibility that a material misstatement of our annual or interim financial statements will not be prevented or detected on a timely basis.

The material weaknesses identified by management as of December 31, 2012 consisted of inadequate and ineffective controls over income tax accounting. As a result of this material weakness, our management concluded as of December 31, 2012 that our internal control over financial reporting was not effective based on criteria set forth by the Committee of Sponsoring Organization of the Treadway Commission in Internal Control—An Integrated Framework (September 1992).

If the remedial measures we have begun implementing that are designed to address this material weakness are insufficient to address this material weakness, or if additional material weaknesses or significant deficiencies in our internal control are discovered or occur in the future, our consolidated financial statements may contain material misstatements and we could be required to restate our financial results.

#### **Risks Related to our Common Stock**

#### Our share price may be volatile, and the value of an investment in our common stock may decline.

An active, liquid and orderly market for our common stock may not be sustained, which could depress the trading price of our common stock. The price of our common stock has been, and is likely to continue to be, volatile, which means that it could decline substantially within a short period of time. The market price of shares of our common stock could be subject to wide fluctuations in response to many risk factors listed in this section, many of which are beyond our control, including:

- actual or anticipated fluctuations in our financial condition and operating results;
- changes in the economic performance or market valuations of other companies engaged in providing wealth management software and services;
- loss of a significant amount of existing business;
- actual or anticipated changes in our growth rate relative to our competitors;
- actual or anticipated fluctuations in our competitors' operating results or changes in their growth rates;
- issuance of new or updated research reports by securities analysts;
- our announcement of actual results for a fiscal period that are higher or lower than projected results or our announcement of revenue or earnings guidance that is higher or lower than expected;
- regulatory developments in our target markets affecting us, our customers or our competitors;
- fluctuations in the valuation of companies perceived by investors to be comparable to us;
- share price and volume fluctuations attributable to inconsistent trading volume levels of our shares;
- sales or expected sales of additional common stock;
- terrorist attacks or natural disasters or other such events impacting countries where we or our customers have operations; and
- general economic and market conditions.

Furthermore, the stock markets have experienced extreme price and volume fluctuations that have affected and continue to affect the market prices of equity securities of many companies. These fluctuations often have been unrelated or disproportionate to the operating performance of those companies. These broad market and industry fluctuations, as well as general economic, political and market conditions such as recessions, interest rate changes or international currency fluctuations, may cause the market price of shares of our common stock to decline. In the past, companies that have experienced volatility in the market price of their stock have been subject to securities class action litigation. We may be the target of this type of litigation in the future. Securities litigation against us could result in substantial costs and divert our management's attention from other business concerns, which could seriously harm our business.

#### Our insiders who are significant stockholders may have interests that conflict with those of other stockholders.

Our directors and executive officers, together with members of their immediate families, as a group, beneficially own, in the aggregate, approximately 24% of our outstanding capital stock as of December 31, 2012. As a result, when acting together, this group has the ability to exercise significant influence over most matters requiring our stockholders' approval, including the election and removal of directors and significant corporate transactions. The interests of our insider stockholders may not be aligned with the interests of our other stockholders and conflicts of interest may arise. In addition, the concentration of our shares may have the effect of delaying, deterring or preventing significant corporate transactions which may otherwise adversely affect the market price of our shares.

#### The future sale of shares of our common stock may negatively impact our stock price.

If our stockholders sell substantial amounts of our common stock, the market price of our common stock could fall. A reduction in ownership by a large stockholder could cause the market price of our common stock to fall. In December 2012, we filed a registration statement on Form S-3 for the potential resale of shares representing up to approximately 24% of our outstanding common stock. While that Form S-3 is no longer effective, such registration could indicate a present desire to sell those shares. In addition, the average daily trading volume in our stock is relatively low. The lack of trading activity in our stock may lead to greater fluctuations in our stock price. Low trading volume may also make it difficult for stockholders to execute transactions in a timely fashion.

## Certain provisions in our charter documents and agreements and Delaware law may inhibit potential acquisition bids for our company and prevent changes in our management.

Our certificate of incorporation and bylaws contains provisions that could depress the trading price of our common stock by acting to discourage, delay or prevent a change of control of our company or changes in management that our stockholders might deem advantageous. As a result of these provisions in our certificate of incorporation, the price investors may be willing to pay for shares of our common stock may be limited.

In addition, we are subject to Section 203 of the Delaware General Corporation Law, which imposes certain restrictions on mergers and other business combinations between us and any holder of 15% or more of our common stock.

## We do not currently intend to pay dividends on our common stock for the foreseeable future and, consequently, your ability to achieve a return on your investment will depend on appreciation in the price of our common stock.

We do not anticipate paying any cash dividends to holders of our common stock in the foreseeable future. Consequently, investors must rely on sales of their common stock after price appreciation, which may never occur, as the only way to realize any future gains on their investment. Investors seeking cash dividends should not purchase our common stock.

#### Item 1B. Unresolved Staff Comments

None.

#### Item 2. Properties

Our headquarters are located in Chicago, Illinois, and consist of approximately 38,000 square feet of leased space. We also lease office space in Denver, Colorado; New York, New York; Sunnyvale, California; Boston, Massachusetts; Seattle, Washington; Raleigh and Landis, North Carolina and two locations in Trivandrum, India. We believe that our office facilities are adequate for our immediate needs and that additional or substitute space is available if needed to accommodate the foreseeable growth of our operations.

### Item 3. Legal Proceedings

We are involved in litigation arising in the ordinary course of our business. We do not believe that the outcome of any of these proceedings, individually or in the aggregate, would, if determined adversely to us, have a material adverse effect on our results of operations, financial condition or business.

### Item 4. Mine Safety Disclosures

This section is not applicable.

#### PART II

#### Item 5. Market for Registrant's Common Equity, Related Stockholder Matters and Issuer Purchases of Equity Securities

#### (a) Market Information

Our common stock is listed on the New York Stock Exchange.

The following table sets forth, for the periods indicated, the high and low closing sale prices of our common stock, for each of the periods presented below as reported in the consolidated reporting system for the New York Stock Exchange Composite Transactions.

	2011
Quarter ended March 31, 2011	\$16.78 to \$11.91
Quarter ended June 30, 2011	\$15.02 to \$12.99
Quarter ended September 30, 2011	\$14.94 to \$9.67
Quarter ended December 31, 2011	\$12.24 to \$9.60
	2012
Quarter ended March 31, 2012	2012 \$13.01 to \$11.34
Quarter ended March 31, 2012 Quarter ended June 30, 2012	
	\$13.01 to \$11.34

#### (b) Holders

The approximate number of common stockholders was 228 as of June 12, 2013.

#### (c) Dividends

We have not paid dividends for the most recent two years.

#### **Common Stock**

As of December 31, 2012, we had 500,000,000 common shares authorized at a par value of \$0.005, of which 32,355,675 shares were outstanding.

#### **Preferred Stock**

As of December 31, 2012, we had 50,000 preferred shares authorized at a par value of \$0.001, of which no shares were outstanding.

#### (d) Securities Authorized for Issuance Under Equity Compensation Plan

For a description of securities authorized under our equity compensation plans, see note 13 to the notes to consolidated financial statements in Part II, Item 8.

#### (e) Stock Performance Graph

The following graph compares the cumulative return to stockholders on our common stock relative to the cumulative total returns of the Russell @2000Index and The S&P North American Technology-Services Index<sup>TM</sup> from the effective date of our initial public offering on July 28, 2010 through December 31, 2012. In calculating total annual stockholder return, reinvestment of dividends, if any, is assumed. The indices are included for comparative purposes only. This graph is not "soliciting material," is not deemed filed with the SEC and is not to be incorporated by reference in any of our filings under the Securities Act of 1933, as amended, or the Exchange Act, as amended, whether made before or after the date hereof and irrespective of any general incorporation language in any such filing.

### 29 MONTH STOCK PERFORMANCE GRAPH



	7/28/2010	12/31/2010	12/31/2011	12/31/2012
Envestnet, Inc. ("ENV")	\$ 100.00	\$ 189.55	\$ 132.89	\$ 155.00
Russell® 2000 Index	100.00	119.02	114.05	132.70
S&P North American Technology-Services Index™	100.00	112.38	126.21	153.33

The stock price performance included in this graph is not necessarily indicative of future stock price performance.

### (f) Recent Sales of Unregistered Securities

None

### (g) Issuer Purchases of Equity Securities

None

### Item 6. Selected Financial Data

### **Consolidated Statements of Operations**

	Year ended December 31,										
	2012			2011 2010				2009		2008	
			(In	thousands, ex	ccept sha	re and per sha	re infor	mation)			
Revenues:											
Assets under management or administration	\$	127,213	\$	99,236	\$	75,951	\$	56,857	\$	71,738	
Licensing and professional services		30,053		23,942		22,101		21,067		20,104	
Total revenues		157,266		123,178		98,052		77,924		91,842	
Operating expenses:											
Cost of revenues		56,119		42,831		31,444		24,624		34,604	
Compensation and benefits		54,973		40,305		37,027		28,763		28,452	
General and administration		30,617		21,856		21,607		15,726		15,500	
Depreciation and amortization		12,400		6,376		5,703		4,499		3,538	
Restructuring charges		115		434		961		-		-	
Total operating expenses		154,224		111,802		96,742		73,612		82,094	
Income from operations		3,042		11,376		1,310		4,312		9,748	
Total other income (expense)		26		(796)		(403)		(3,368)		115	
Income before income tax provision		3,068		10,580		907		944		9,863	
Income tax provision		2,603		2,975		1,533		1,816		4,608	
Net income (loss)		465		7,605		(626)		(872)		5,255	
Less preferred stock dividends		-		-		(422)		(720)		(203)	
Less net income allocated to participating convertible preferred stock		_		_		-		<u>-</u>		(2,406)	
Income (loss) attributable to common											
shareholders	\$	465	\$	7,605	\$	(1,048)	\$	(1,592)	\$	2,646	
Net income (loss) per share attributable to common stockholders											
Basic	\$	0.01	\$	0.24	\$	(0.05)	\$	(0.12)	\$	0.20	
Diluted	\$	0.01	\$	0.23	\$	(0.05)	\$	(0.12)	\$	0.19	
Weighted average common shares outstanding:											
Basic	32	,162,672	3	1,643,390	20	20,805,911		12,910,998		13,354,845	
Diluted		,341,615		2,863,834		),805,911		12,910,998		13,354,845	
		, , 0				.,,		,		,,0 .0	

### **Consolidated Balance Sheet Data**

	December 31,									
	2012		2011			2010		2009		2008
					(In tho	ousands)				
Cash and cash equivalents	\$	29,983	\$	64,909	\$	67,668	\$	31,525	\$	28,445
Working capital		14,178		64,944		62,979		27,262		21,405
Goodwill and intangible assets		92,794		33,559		3,213		3,261		4,331
Total assets		162,399		137,702		141,029		74,064		71,257
Stockholders' equity		125,996		115,639		101,480		57,252		57,589

#### Item 7. Management's Discussion and Analysis of Financial Condition and Results of Operations

Except where we have otherwise indicated or the context otherwise requires, dollar amounts presented in this Form 10-K are in thousands, except for Part III and per share amounts.

#### Overview

We are a leading provider of unified wealth management software and services to financial advisors and institutions. By integrating a wide range of investment solutions and services, our Web-based platform provides financial advisors with the flexibility to address their clients' needs.

Envestnet empowers financial advisors to deliver fee-based advice to their clients. We work with both independent advisors (RIAs), as well as advisors associated with financial institutions (broker dealers, banks). The services we offer and market to financial advisors address advisors' ability to grow their practice as well as operate more efficiently – the Envestnet platform spans from the initial meeting an advisor has with a prospective client to the ongoing day-to-day operations of managing an advisory practice.

Our centrally-hosted technology platform, which we refer to as having "open architecture" because of its flexibility, provides financial advisors with access to a series of integrated services to help them better serve their clients. These services include risk assessment and selection of investment strategies and solutions, asset allocation models, research and due diligence, portfolio construction, proposal generation and paperwork preparation, model management and account rebalancing, account monitoring, customized fee billing, overlay services covering asset allocation, tax management and socially responsible investing, aggregated multi-custodian performance reporting and communication tools, as well as access to a wide range of leading third-party asset custodians.

We offer these solutions principally through the following product and services suites:

- Envestnet's wealth management software empowers advisors to better manage client outcomes and strengthen their practice. Our software
  unifies the applications and services advisors use to manage their practice and advise their clients, including financial planning; capital
  markets assumptions; asset allocation guidance; research and due diligence on investment managers and funds; portfolio management, trading
  and rebalancing; multi-custodial, aggregated performance reporting; and billing calculation and administration.
- Our Portfolio Management Consultants group ("Envestnet | PMC" 
   Pmc" primarily engages in consulting services aimed at providing financial advisors with additional support in addressing their clients' needs, as well as the creation of proprietary investment solutions and products. Envestnet | PMC's investment solutions and products include managed account and multi-manager portfolios, mutual fund portfolios and ETF portfolios. Envestnet | PMC also offers Prima Premium Research, comprising institutional-quality research and due diligence on investment managers, mutual funds, ETFs and liquid alternatives funds.
- Envestnet | Tamarac provides leading portfolio accounting, rebalancing, trading, performance reporting and client relationship management software, principally to high-end RIAs.
- Envestnet Reporting Solutions software aggregates and manages investment data, provides performance reporting and benchmarking, giving advisors an in-depth view of clients' various investments, empowering advisors to give holistic, personalized advice.

We believe that our business model results in a high degree of recurring and predictable financial results.

#### Revenues

#### **Overview**

We earn revenues primarily under two pricing models. First, a majority of our revenues is derived from fees charged as a percentage of the assets that are managed or administered on our technology platform by
financial advisors. These revenues are recorded under revenues from assets under management ("AUM") or administration ("AUA") or collectively ("AUM/A"). Our asset-based fees vary based on the types of investment solutions and services that financial advisors utilize. Asset-based fees accounted for approximately 81%, 81% and 77% of our total revenues for the years ended December 31, 2012, 2011 and 2010, respectively. In future periods, the percentage of our total revenues attributable to asset-based fees is expected to vary based on fluctuations in securities markets, whether we enter into significant license agreements, the mix of AUM or AUA, and other factors. As of December 31, 2012, approximately \$98 billion of investment assets subject to asset-based fees were managed or administered utilizing our technology platform by approximately 16,100 financial advisors through approximately 450,000 investor accounts.

We also generate revenues from recurring, contractual licensing fees for providing access to our technology platform. These revenues are recorded under revenues from licensing and professional services. Licensing fees are generally fixed in nature for the contract term and are based on the level of investment solutions and services provided, rather than on the amount of client assets on our technology platform. Licensing fees accounted for 15%, 16% and 20% of our total revenues for the years ended December 31, 2012, 2011 and 2010, respectively. Fees received in connection with professional services accounted for the remainder of our total revenues. As of December 31, 2012, approximately \$270 billion of investment assets for which we receive licensing fees for utilizing our technology platform were serviced by approximately 6,900 financial advisors through approximately 1,228,000 investor accounts.

The following table provides information regarding the amount of assets utilizing our platform, financial advisors and investor accounts in the periods indicated.

	As of December 31,					
		2012		2011		2010
		(in millions	except ac	counts and a	dvisors da	ta)
Platform Assets						
Assets Under Management (AUM)	\$	30,970	\$	22,936	\$	14,486
Assets Under Administration (AUA)		67,368		47,148		49,202
Subtotal AUM/A		98,338		70,084		63,688
Licensing		269,729		69,514		75,668
Total Platform Assets	\$	368,067	\$	139,598	\$	139,356
Platform Accounts	_					
AUM		156,327		124,636		65,663
AUA		293,151		216,038		241,162
Subtotal AUM/A		449,478		340,674		306,825
Licensing		1,228,016		588,038		603,950
Total Platform Accounts		1,677,494		928,712		910,775
Advisors						
AUM/A		16,085		13,887		13,833
Licensing		6,941		5,709		7,746
Total Advisors		23,026		19,596		21,579

### Revenues from assets under management or administration

We generally charge our customers fees based on a higher percentage of the market value of AUM than the fees we charge on the market value of AUA, because we provide fiduciary oversight and/or act as the investment advisor in connection with assets we categorize as AUM. The level of fees varies based on the nature of the investment solutions and services we provide, as well as the specific investment manager, fund and/or custodian chosen by the financial advisor. A portion of our revenues from assets under management or administration include costs paid by us to third parties for sub-advisory, clearing, custody and brokerage services. These expenses are recorded under cost of revenues. We do not have fiduciary responsibility in connection with AUA and, therefore, charge lower fees on these assets. Our fees for AUA vary based on the nature of the investment solutions and services we provide.

For over 85% of our revenues from assets under management or administration, we bill customers at the beginning of each quarter based on the market value of customer assets on our technology platform as of the end of the prior quarter. For example, revenues from assets under management or administration recognized during the fourth quarter of 2012 were primarily based on the market value of assets as of September 30, 2012. Our revenues from assets under management or administration are generally recognized ratably throughout the quarter based on the number of days in the quarter.

Our revenues from assets under management or administration are affected by the amount of new assets that are added to existing and new client accounts, which we refer to as gross sales, and the amount of assets that are withdrawn from client accounts, which we refer to as redemptions. We refer to the difference between asset in-flows and outflows as net flows. Positive net flows indicate that the market value of assets added to client accounts exceeds the market value of assets that have been withdrawn from client accounts.

Our revenues from assets under management or administration are also affected by changes in the market values of securities held in client accounts due to fluctuations in the securities markets. Certain types of securities have historically experienced greater market price fluctuations, such as equity securities, than other securities, such as fixed income securities, though in any given period the nature of securities that experience the greatest fluctuations may vary. For example, from October 2007 to March 2009, the equity markets, as measured by the value of the S&P 500 index, declined in value by approximately 57%, which significantly contributed to the 21% decrease in our revenues from assets under management or administration from 2008 to 2009.

The following table provides information regarding the degree to which gross sales, redemptions, net flows and changes in the market values of assets contributed to changes in AUM or AUA in the periods indicated.

			Asset Rollforw	vard - 2012			
	As of	Gross	Redemp-	Net	Market	As of	
	12/31/11	Sales	tions	Flows	Impact	12/31/12	
	(in millions except account data)						
Assets under Management (AUM)	\$ 22,936	\$ 12,487	\$ (6,850)	\$ 5,637	\$2,397	\$ 30,970	
Assets under Administration (AUA)	47,148	28,381	(12,520)	15,861	4,359	67,368	
Subtotal AUM/A	\$ 70,084	\$ 40,868	\$ (19,370)	\$ 21,498	\$ 6,756	\$ 98,338	
Number of Fee-Based Accounts	340,674	191,551	(82,747)	108,804		449,478	

During 2012, we added \$10.4 billion of conversions included in the above AUM/A gross sales figures, and an additional \$13.2 billion of conversions in licensing.

		Asset Rollforward - 2011						
	As of	Gross	Redemp-	Net	Market	Fund	As of	
	12/31/10	Sales	tions	Flows	Impact	Quest	12/31/11	
	(in millions except account data)							
Assets under Management (AUM)	\$ 14,486	\$ 7,737	\$ (4,795)	\$ 2,942	\$ (317)	\$ 5,825	\$ 22,936	
Assets under Administration (AUA)	49,202	24,873	(18,537)	6,336	(1,089)	(7,301)	47,148	
Subtotal AUM/A	\$ 63,688	\$ 32,610	\$ (23,332)	\$ 9,278	\$ (1,406)	\$ (1,476)	\$ 70,084	
Number of Fee-Based Accounts	306,825	135,963	(92,060)	43,903		(10,054)	340,674	

On December 13, 2011, we completed our acquisition of FundQuest. At that time, \$5.8 billion of FundQuest assets previously reported as AUA were reclassified to AUM. Also during the fourth quarter, one of FundQuest's clients with \$1.5 billion in assets transitioned to licensing for a flat fee and is no longer reflected in AUA as of December 31, 2011. During 2011, we added \$8.3 billion of conversions, which are included in the above AUM/A gross sales figures and an additional \$1.0 billion in conversions under licensing agreements.

The mix of assets under management and assets under administration was as follows for the periods indicated:

		December 31,			
	2012	2011	2010		
Assets under management (AUM)	31%	33%	23%		
Assets under administration (AUA)	<u>69</u> %	67%	<u> </u>		
	100%	100%	100%		

We expect the percentage of AUM and AUA will fluctuate in future periods. The nature and type of services requested by our customers are the key drivers in determining whether customer assets are classified as AUM or AUA. Therefore, we do not have direct control over the mix of AUM and AUA.

#### Revenues from licensing and professional services fees

Our revenues received under license agreements are recognized over the contractual term. To a lesser degree we also receive revenues from professional services fees by providing customers with certain technology platform software development services. In the years ended December 31, 2012, 2011 and 2010, our revenues from professional services fees were \$6,145, \$3,818 and \$2,907, respectively. These revenues are generally recognized under a proportional performance model utilizing an output based approach. Our contracts have fixed prices, and generally specify or quantify interim deliverables.

We may enter into license agreements in future periods if requested by our customers and commercially attractive to us.

### Expenses

The following is a description of our principal expense items.

#### Cost of revenues

Cost of revenues primarily includes expenses related to our receipt of sub-advisory and clearing, custody and brokerage services from third parties. The largest component of cost of revenues, sub-advisory fees paid to third party investment managers, relates only to AUM since a sub-advisor is not utilized in connection with AUA. Clearing, custody and brokerage services are provided by third-party providers. These expenses are typically calculated based upon a contractual percentage of the market value of assets held in customer accounts measured as of the end of each fiscal quarter and are recognized ratably throughout the quarter based on the number of days in the quarter. Also included in cost of revenues are vendor specific expenses related to the direct support of revenues associated with the Envestnet | Tamarac products.

### Compensation and benefits

Compensation and benefits expenses primarily relate to employee compensation, including salaries, commissions, non-cash stock-based compensation, incentive compensation, benefits and employer-related taxes.

# General and administration

General and administration expenses include occupancy costs and expenses relating to communications services, research and data services, website and system development, marketing, professional and legal services and travel and entertainment.

#### Depreciation and amortization

Depreciation and amortization expenses include depreciation and amortization related to:

- fixed assets, including computer equipment and software, leasehold improvements, office furniture and fixtures and other office equipment;
- internally developed software; and
- intangible assets, primarily related to customer lists, proprietary technology and trade names, the value of which are capitalized in connection with our acquisitions.

Furniture and equipment is depreciated using the straight-line method based on the estimated useful lives of the depreciable assets. Leasehold improvements are amortized using the straight-line method over their estimated economic useful lives or the remaining lease term, whichever is shorter. Improvements are capitalized, while repairs and maintenance costs are recorded as expenses in the period they are incurred. Assets are tested for recoverability whenever events or circumstances indicate that the carrying value of the assets may not be recoverable.

Internally developed software is amortized on a straight-line basis over its estimated useful life. We evaluate the useful lives of these assets on an annual basis and test for impairment whenever events or changes in circumstances occur that could impact the recoverability of these assets.

Intangible assets are depreciated using an accelerated basis over their estimated economic useful lives and are reviewed for possible impairment whenever events or changes in circumstances occur that could impact the recoverability of these assets.

### **Recent Developments**

### **2013 Developments**

### Wealth Management Solutions

On April 11, 2013, we entered into a definitive agreement to acquire substantially all of the assets of the Wealth Management Solutions ("WMS") division of Prudential Investments for \$10,000 in cash upon closing, subject to certain post-closing adjustments, plus contingent consideration of up to a total of \$23,000 in cash, based upon meeting certain performance targets, to be paid over three years. WMS is a provider of technology solutions that enables financial services firms to develop and enhance their wealth management offerings. We anticipate the acquisition will be completed in the third quarter of 2013.

# **2012 Developments**

# Prima Capital Holding, Inc. Agreement

On April 5, 2012, we completed the acquisition of Prima. In accordance with the stock purchase agreement, we acquired all of the outstanding shares of Prima for consideration of \$13,925. Prima, now part of Envestnet | PMC, provides investment management due diligence, research applications, asset allocation modeling and multi-manager portfolios to the wealth management and retirement industries. Prima's clientele includes banks, independent RIAs, regional broker-dealers, family offices and trust companies.

# Tamarac, Inc. Agreement

On May 1, 2012, we completed the acquisition of Tamarac. In accordance with the merger agreement, a newly formed subsidiary of Envestnet merged with and into Tamarac, and Tamarac became a wholly-owned subsidiary of Envestnet. Under the terms of the merger agreement, net consideration was \$48,427 for all of the outstanding stock of Tamarac. Tamarac provides leading portfolio accounting, rebalancing, trading, performance reporting and client relationship management software, principally to high-end RIAs.

In accordance with the terms of the merger agreement between Envestnet and Tamarac, Tamarac senior management were required to apply at least 50% (up to 100%) of the aggregate proceeds of the Tamarac change of control payment totaling \$2,759 to purchase registered shares of Envestnet common stock (232,150 shares) in an amount equal to 95% multiplied by the Envestnet closing market price on the day before the merger closed (See notes 3 and 12 to the notes to consolidated financial statements).

In addition, we adopted the Envestnet, Inc. Management Incentive Plan for Envestnet | Tamarac Management Employees (the "2012 Plan"). The 2012 Plan provides for the grant of up to 559,551 shares of unvested common stock. The unvested common stock vests based upon Tamarac meeting certain performance conditions and then a subsequent two-year service condition. We also granted to certain Tamarac employees 232,150 stock options to acquire Envestnet common stock at an exercise price of \$12.51. These stock options vest on the second anniversary of the grant date (See notes 3 and 13 to the notes to consolidated financial statements).

## **2011 Developments**

## FundQuest Agreement

On December 13, 2011, we acquired all of the outstanding shares of FundQuest for total consideration of \$27,796. FundQuest, operating as Envestnet Portfolio Solutions, Inc., provides managed account programs, overlay portfolio management, mutual funds, institutional asset management and investment consulting to registered investment advisors, independent advisors, broker-dealers, banks and trust organizations. Upon closing of the transaction, the existing platform services agreement between us and FundQuest was terminated (see note 3 to the notes to consolidated financial statements) and approximately \$5.8 billion of FundQuest's assets were reclassified to assets under management from assets under administration. In addition, one of FundQuest's clients with \$1.5 billion in assets transitioned to licensing from assets under administration.

### Fidelity Agreement

For the years ended December 31, 2012, 2011 and 2010, revenues associated with our relationship with our single largest client, Fidelity, accounted for 22%, 31% and 31%, respectively, of our total revenues. As of December 31, 2011, we renegotiated a five-year license agreement with Fidelity which resulted in a reduction in 2012 license fee revenues. In addition, as a part of the renegotiated agreement, we will continue to receive ongoing platform services fees through the Fidelity relationship based upon asset-based fees.

### **Critical Accounting Policies**

Our consolidated financial statements are prepared in conformity with accounting principles generally accepted in the United States, or ("U.S. GAAP"). The accounting policies described below require management to apply significant judgment in connection with the preparation of our consolidated financial statements. In particular, judgment is applied to determine the appropriate assumptions to be used in calculating estimates that affect certain reported amounts in our consolidated financial statements. These estimates and assumptions are based on historical experience and on various other factors that we believe to be reasonable under the circumstances. If different estimates or assumptions were used, our results of operations, financial condition and cash flows could have been materially different than those reflected in our consolidated financial statements. For additional information regarding our critical accounting policies, see note 2 to the notes to the consolidated financial statements.

### **Revenue** recognition

We recognize revenues when all four of the following criteria have been met:

- Persuasive evidence of an arrangement exists;
- The product has been delivered or the service has been performed;

- The fee is fixed or determinable; and
- Collectability is reasonably assured.

# Types of revenues

We generate revenues from assets under management or administration and from licensing and professional service fees. Revenues from assets under management or administration are generated from fees based on a contractual percentage of assets under management or administration valued at each quarterend. These fees are generally collected at the beginning of a quarter in advance based upon the previous quarter-end values. In less than 15% of our contracts, fees are collected at the end of the quarter based upon the average daily balance. The contractual fee percentages vary based upon the level and type of services we provide to our customers. Pursuant to the contracts with our customers, we calculate our fees based on the asset values in the customer's account, without making any judgment or estimates. None of our fees is earned pursuant to performance-based or other incentive-based arrangements.

We generate revenues from licensing fees pursuant to recurring contractual fixed-fee agreements, principally with a portion of our enterprise clients. Our licensing fees vary based on the type of services we provide. We generate revenues from professional service fees by providing customers with customized technology platform software development services. These revenues are received pursuant to contracts that detail the nature of the services to be provided by us, the estimated number of hours such work will require and the total contract fee amount.

### **Recognition of revenues**

Application of the applicable accounting principles of U.S. GAAP requires us to make judgments and estimates in connection with the measurement and recognition of revenues. Revenues are recognized in the period in which the related services are provided. In certain cases, management is required to determine whether revenues should be recognized in an amount equal to the gross fees we receive or as a net amount reflecting the payment of expenses to thirdparties, such as sub-advisors and custodians, that provide services to us in connection with certain of our financial advisors' client accounts. When fees are collected for sub-advisory, clearing, custody or brokerage services in circumstances where we do not have a direct contract with the third-party provider, the fees are recorded as revenue on a net basis. Fees we received in advance of the performance of services are recorded as deferred revenues on our consolidated balance sheet and are recognized as revenues when earned, generally over three months.

Revenues from licensing are recognized over the contractual term. Contracts with nonstandard terms and conditions may require contract interpretation to determine the appropriate revenue recognition policy to apply. For non-software multiple-element arrangements, we allocate the revenue in the arrangement using our best estimate of selling price, when vendor-specific objective evidence ("VSOE") or third-party evidence ("TPE") of selling price is not available. We generally recognize revenue for consideration allocated to implementation and consulting services in a multiple-element arrangement as services are performed because these services have standalone value separate from the license fees.

Revenues from professional services are recognized under a proportional performance model utilizing an output-based approach. Our contracts have fixed prices, and generally specify or quantify interim deliverables.

Our revenue recognition is also affected by our judgment in determining appropriate allowances for uncollectible receivables. We consider customerspecific information related to delinquent accounts and past lost experience, as well as current economic conditions in establishing the amount of the allowance.

#### **Customer inducements**

In certain instances, we have entered into agreements which include inducement payments to the customer. In accordance with U.S. GAAP, inducement payments made to customers are capitalized and amortized against



revenue on a straight-line basis over the term of the agreement. Customer inducement amortization totaled zero, \$4,568 and \$3,238 for 2012, 2011 and 2010, respectively. Customer inducement assets are reviewed for impairment whenever events or circumstances occur that may impact the fair value of these assets. There were no customer inducement assets as of December 31, 2012 and 2011.

# Purchase accounting

During the fourth quarter of 2011, we completed the acquisition of FundQuest for consideration totaling \$27,796 and in the second quarter of 2012, we completed the acquisitions of Prima and Tamarac for consideration totaling \$13,925 and \$48,427, respectively. For more information on the acquisitions see note 3 to the notes to consolidated financial statements.

Assigning fair market values to the assets acquired and liabilities assumed at the date of an acquisition requires knowledge of current market values, and the values of assets in use, and often requires the application of judgment regarding estimates and assumptions. While the ultimate responsibility resides with management, for material acquisitions, we retain the services of certified valuation specialists to assist with assigning estimated values to certain acquired assets and assumed liabilities, including intangible assets.

Acquired intangible assets, excluding goodwill, are valued using a discounted cash flow methodology based on future cash flows specific to the type of intangible asset purchased. This methodology incorporates various estimates and assumptions, the most significant being projected revenue growth rates, earnings margins, and forecasted cash flows based on the discount rate and terminal growth rate. Management projects revenue growth rates, earnings margins and cash flows based on the historical operating results of the acquired entity adjusted for synergies anticipated to be achieved through integration, expected future performance, operational strategies, and the general macroeconomic environment. We review finite-lived intangible assets for triggering events such as significant changes in operations, customers or future revenue that might indicate the need to impair the assets acquired or change the useful lives of the assets acquired. There was no impairment or change in useful lives recognized on other intangible assets in 2012, 2011 or 2010.

Assumed liabilities are valued based on estimates of anticipated expenditures to be incurred to satisfy the assumed obligations, including contractual liabilities assumed, which require the exercise of professional judgment.

Assumed contracts may have favorable or unfavorable terms that must be valued as of the acquisition date. Such valuation is subject to management judgment regarding the evaluation and interpretation of contract terms in relation to other economic circumstances, such as the market rates for office space leases.

If we assume a performance obligation to customers as of the acquisition date, a deferred revenue obligation is recognized. Judgment is required to evaluate whether a future performance obligation exists and to assign a value to the performance obligation.

Estimation of working capital settlement amounts, if not resolved prior to the first reporting period after an acquisition, but before the end of the purchase measurement period, requires exercise of management judgment. We measure these amounts at the acquisition date fair value, if their fair value can be determined during the measurement period. If these estimated working capital settlement amounts are not resolved prior to the first reporting period after acquisition, we recognize the asset or liability if it can be reasonably estimated. Subsequent adjustments to these provisional working capital settlement amounts are evaluated by management to determine the proper accounting treatment under relevant U.S. GAAP authoritative guidance.

Assumed acquired tax liabilities for uncertain tax positions are dependent on assessing the past practices of the acquisition target based on our review of actual tax filings and information obtained through due diligence procedures. Evaluation of the validity of tax positions taken by the acquisition target are subject to management judgment.

Transaction costs associated with business combinations are expensed as they are incurred.

# Internally developed software

Costs relating to internally developed software that are incurred in the preliminary stages of development are expensed as incurred. Management determines when projects have met the criteria of the application development stage. This typically occurs when the conceptual formulation and evaluation of software functionality are finalized.

Once work on a software application has passed the preliminary stages, internal and external costs, if direct and incremental, are capitalized until the software application is substantially complete and ready for its intended use. These costs include expenditures related to software design, technical specifications, coding, installation of hardware and parallel testing. We cease capitalizing these costs upon completion of all substantial testing of the software application.

We also capitalize costs related to specific upgrades and enhancements of our internally developed software when we conclude that it is probable that the expenditures will result in additional functionality. Our maintenance and training costs are expensed as incurred.

As of December 31, 2012 and 2011, we had net capitalized internally developed software of \$4,324 and \$3,524, respectively. We capitalized \$2,350, \$1,482 and \$1,340 in internally developed software during the years ended December 31, 2012, 2011 and 2010, respectively.

Internally developed software is amortized on a straight-line basis over its estimated useful life. We evaluate the useful lives of these assets on an annual basis and test for impairment whenever events or changes in circumstances occur that could impact the recoverability of these assets. There were no impairments to internally developed software during the years ended December 31, 2012, 2011 and 2010.

# Non-cash stock-based compensation expense

Since our 2004 Stock Incentive Plan, 2010 Long-Term Incentive Plan and the 2012 Plan were adopted, stock-based awards have been an important component of our compensation structure. We expect that this will continue to be the case in the future. Our Board of Directors is responsible for determining the timing and magnitude of all equity grants. Prior to our initial public offering on July 28, 2010, our Board of Directors was responsible for determining the fair value of our common stock on the date of each stock option grant. The Board of Directors had delegated certain of its responsibilities to the Compensation Committee of the Board of Directors and certain members of management. As required under our 2004 Stock Incentive Plan and our 2010 Long-Term Incentive Plan, all of our stock options are granted with exercise prices at or above the fair value of our common stock on the grant date.

The following table provides information regarding stock options granted from January 1, 2009 through our initial public offering on July 28, 2010:

						Intrinsic Value As of	
Date	Shares	Stoc	k Price	Exerc	ise Price	Decemb	oer 31, 2012
Options:							
2/16/2009							
	1,000	\$	7.85	\$	7.85	\$	6.10
4/8/2009	8,230		7.85		7.85		6.10
5/15/2009	232,732		7.15		7.15		6.80
7/6/2009	10,000		7.15		7.15		6.80
11/16/2009	12,000		11.50		11.50		2.45
2/22/2010	71,000		13.45		13.45		0.50

Prior to our initial public offering in July 2010, there was no market for our common stock and therefore no readily available price to reference when determining the fair value of our common stock in connection with the granting of stock options. The value of our common stock was dependent upon our company valuation and, as described below, we had periodically obtained independent valuations and performed internal valuations of our

common stock. In each case, such valuations had been performed contemporaneously and we had determined the fair market value of our company in conformity with commonly accepted corporate valuation techniques and methodologies.

Prior to our initial public offering, we generally had obtained contemporaneous independent valuations at least annually and at the time of broad-based option grants, such as on May 15, 2009. For our internal valuations, we applied the same approach and methodology used by the independent valuation firm. For any option grants made between quarterly valuations of our common stock, our Board of Directors assessed all available information in determining whether the stock price in effect at the time of the grant should otherwise be adjusted. As a private company, we had performed our quarterly valuations such that they were effective approximately 45 days following the end of each calendar quarter to approximate the date upon which, if we were a reporting company, we would be required to disclose to the public through filings with the Securities and Exchange Commission ("SEC") our financial performance and associated operating metrics, which include assets under management and administration. Until such date, any information about a given quarter's financial performance, ending asset values, and other information that could be deemed material to investors, would not be known to the public even if we were a reporting company and therefore is not included in the valuation of our common stock during interim periods.

In the specific cases of option grants made after the dates of our quarterly valuations during the period under review, our Board of Directors concluded that no adjustment should have been made to the most recent valuation of our common stock based on its assessment that, had we been a reporting company, no new material information would have been available to the public since the date of the prior valuation of our common stock.

Prior to our initial public offering, our company valuation, whether prepared by an independent valuation firm or performed internally, considered an income approach, also known as a discounted cash flow analysis, incorporating our historical and expected financial performance, the relevant market and industry and economic trends. Our valuation also considered a market approach, including recent capital transactions involving either our company or comparable companies, and comparable public company valuations. The resulting calculation assigned a value for 100% of our company's equity on a marketable equivalent, non-controlling interest basis. We considered, but did not include, an asset approach, as we did not believe the book value of our assets provides meaningful input into our expected revenue and earnings, or the value of our company.

We believe the value of our common stock had the potential to change each fiscal quarter in the normal course of our business, since the majority of our total revenues earned in a given quarter is calculated based on the value of AUM and AUA as of the end of the previous fiscal quarter. These revenues, and our historical resulting projections for earnings and cash flow, were inherently subject to fluctuations from quarter to quarter.

Accordingly, prior to our initial public offering, we calculated the value of our common stock at least once each fiscal quarter. The historical quarterly valuations did at times fluctuate significantly as the market value of our assets under management or administration drives our near term financial results and longer term projections. The value of our common stock could also change if a material financing transaction or other significant event occurred within a given fiscal quarter. In such circumstances we performed an additional valuation of our common stock at the time of the transaction or event, using the same valuation methodology that was utilized in connection with our quarterly valuations.

After we determined a value for our company, we allocated the value to each class of our shares, including our common stock. Our value allocation methodology applied the principles set forth in the AICPA Practice Aid—Valuation of Privately-Held-Company Equity Securities Issued as Compensation, or the Practice Aid defines appropriate methods to allocate enterprise value to common shares when multiple share classes exist. Based on various factors, including the stage of a company's life and the timing and likelihood of various liquidity events, one method of allocation may be more appropriate than the others. We considered, but did not use, the probability-weighted expected return method due to the number of assumptions for each scenario that are difficult to estimate, and the fact that our most likely liquidation event was an initial public offering.

Additionally, we did not apply the liquidation method because, as the Practice Aid indicates, it would be inappropriate for a later-stage company such as ours to use that method to allocate value to the various share classes. Furthermore, the more imminent a liquidity event becomes, the more aligned the liquidation model and option pricing model become in attributing value to each share class. Accordingly, we used the option pricing method, as defined in the Practice Aid, which treats each class of equity as having a "call option" on the enterprise value. The option pricing method considers the economic preferences and other rights attributable to each share class, resulting in a price for each of our share classes, including our common stock. Our valuations of our common stock also reflected a discount for lack of marketability, adjusted over time to reflect the expected likelihood and timing of a liquidity event subsequent to each valuation date. No other discounts were applied in determining the value of our common stock.

During 2009 and through the date our initial public offering on July 28, 2010, we performed the following contemporaneous valuations of our common stock:

	Fair Value of
Date	Common Stock
2/15/2009	\$ 7.85
5/15/2009	7.15
8/15/2009	9.90
11/15/2009	11.50
2/15/2010	13.45

As described above, the assets under management or administration on our technology platform at the end of a given quarter have a significant impact on our short—and long-term financial projections and resulting valuation. For example, the valuation conducted on May 15, 2009 incorporated financial projections based on assets under management or administration as of March 31, 2009. The value of those assets was 6% below the value of the assets as of December 31, 2008. This contributed to the decline in the estimated fair value of our common stock between periods. Conversely, assets under management or administration increased 16% between March 31, 2009 and June 30, 2009, contributing to an increase in the estimated fair value of our common stock between May 15, 2009 and August 15, 2009. In addition, assets under management or administration increased 15% between June 30, 2009 and September 30, 2009, which contributed to the increase in the fair value of our common stock between August 15, 2009 and November 15, 2009. A 4% increase in assets under management or administration between September 30, 2009 and December 31, 2009, as well as the platform services agreement signed with FundQuest in February of 2010, contributed to the increase in the fair value of our common stock between November 15, 2009 and February 15, 2010. The decrease between the fair value of our common stock of comparable companies and the difficult conditions in the market for initial public offerings at and immediately prior to our determination of the initial public offering price. Other factors, such as updated financial projections not related to changes in our assets under management or administration, as well as fluctuations in the value of comparable publicly-traded companies, also contributed to the differences in the estimated fair value of our common stock between periods.

Since our initial public offering on July 28, 2010, we have not performed internal valuations or obtained independent valuations in order to determine the Company's stock price to reference when determining the fair value of our common stock in connection with the granting of stock options or restricted stock.

Non-cash stock-based compensation expense for stock option and restricted stock grants is estimated at the grant date based on each grant's fair value, calculated using the Black-Scholes option pricing model for stock options, and intrinsic value for restricted stock. Compensation and benefits expenses are recognized over the vesting period for each grant. The fair value of our stock options and the resulting expenses are based on various assumptions, including the expected volatility of our stock price, the expected term of the stock options, estimated forfeiture rates and the risk-free interest rate. The use of different assumptions would result in different fair values and compensation and benefits expenses for our option grants.

The 2012 Plan provides for the grant of up to 559,551 shares of unvested common stock ("Target Incentive Awards"). The Target Incentive Awards vest based upon Tamarac meeting certain performance conditions and then a subsequent two-year service condition. We measured the cost of these awards based on the estimated fair value of the award as of the market closing price on the day before the acquisition closed. We are recognizing the estimated expense on a graded-vesting method over a requisite service period of three to five years, which is the estimated vesting period. We have estimated expected forfeitures at the grant date and will recognize compensation expense only for those awards expected to vest. The initial forfeiture assumption is reassessed by management in subsequent periods and may change based upon new facts and circumstances. Changes in the forfeiture assumptions may impact the total amount of expense ultimately recognized over the vesting period.

### Income taxes

We are subject to income taxes in the United States and India. Significant judgment is required in evaluating our tax positions and determining our provision for income taxes.

We use the asset and liability method to account for income taxes. Deferred tax assets and liabilities are recognized for the future tax consequences attributable to differences between the financial statement carrying amounts of existing assets and liabilities and their respective tax bases, and for net operating loss carryforwards. Deferred tax assets and liabilities are measured using enacted tax rates expected to apply to taxable income in the years in which those temporary differences are expected to be recovered or settled. The effect on deferred tax assets and liabilities of a change in tax rates is recognized in the Company's income tax provision in the period that includes the enactment date. We record a valuation allowance to reduce deferred tax assets to an amount that we determine is more-likely-than-not to be realized in the future.

In our ordinary course of business, we may enter into transactions for which the ultimate tax determination is uncertain. In such cases, we establish reserves for tax-related uncertainties based on our estimates of whether, and the extent to which, additional taxes will be due. The reserves are established when we believe that certain positions are likely to be challenged and may not be fully sustained on review by tax authorities. We adjust these reserves in light of changing facts and circumstances, such as the closing of a tax audit or refinement of an estimate. Although we believe our reserves are reasonable, no assurance can be given that the final outcome of these matters will not be different from that which is reflected in our historical income tax provisions and accruals. To the extent that the final tax outcome of these matters is different than the amounts recorded, such differences will be reflected in our provision for income taxes includes the impact of reserve provisions and changes to reserves that are considered appropriate.

The amount of income tax we pay is subject to audits by federal, state and foreign tax authorities, which may result in proposed assessments. Our estimate of the potential outcome for any uncertain tax issue is highly judgmental. We believe that we have adequately provided for the foreseeable outcome related to these matters. However, our future results may include favorable or unfavorable adjustments to our estimated tax liabilities in the period the assessments are made or resolved, audits are closed or when statutes of limitations on potential assessments expire. Additionally, the jurisdictions in which our earnings or deductions are realized may differ from our current estimates. As a result, our effective tax rate may fluctuate significantly on a quarterly basis.

Significant judgment is also required in determining any valuation allowance recorded against deferred tax assets. In assessing the need for a valuation allowance, we consider all available evidence, including past operating results, estimates of future taxable income and the feasibility of tax planning strategies. In the event that we change our determination as to the amount of deferred tax assets that can be realized, we will adjust our valuation allowance with a corresponding impact to the provision for income taxes in the period in which such determination is made.

Our effective tax rates differ from the statutory rates primarily due to adjustments in valuation allowances, unrecognized tax benefits, state income taxes and changes in rates. Our provision for income taxes varies based on, among other things, changes in the valuation of our deferred tax assets and liabilities, the tax effects of non-cash stock-based compensation or changes in applicable tax laws, regulations and accounting principles or interpretations thereof.

As of December 31, 2012, we had net operating loss carryforwards for federal and state income tax purposes of \$42,912 and \$30,271, respectively, available to reduce future income subject to income taxes. The federal and state net operating loss carryforwards expire through 2033. In addition, we have alternative minimum tax credit carryforwards of approximately \$445, which are available to reduce future federal regular income taxes, if any, over an indefinite period.

We are subject to examination of our income tax returns by the U.S. Internal Revenue Service and other tax authorities. We assess the likelihood of adverse outcomes resulting from these examinations to determine the adequacy of our provision for income taxes. There can be no assurance that the outcomes from these examinations will not have a material adverse effect on our results of operations, financial condition and cash flows.

Our India subsidiary is currently under examination by the India Taxing Authority for the fiscal years ended March 31, 2009 and March 31, 2010. Based on the outcome of examinations of our subsidiary or the result of the expiration of statutes of limitations, it is reasonably possible that the related unrecognized tax benefits could change from those recorded in the consolidated balance sheets. It is possible that one or more of these audits may be finalized within the next twelve months.

# **Results of Operations**

Year ended December 31, 2012 compared to year ended December 31, 2011

	Year	Year Ended December 31,			Increase (Decreas		
	2012	2	2011		nount	%	
		(In the	ousands)				
Revenues:							
Assets under management or administration	\$ 127	213 \$	99,236	\$	27,977	28%	
Licensing and professional services	30	.053	23,942		6,111	26%	
Total revenues	157,	266	123,178		34,088	28%	
Operating expenses:							
Cost of revenues	56,	119	42,831		13,288	31%	
Compensation and benefits	54	973	40,305		14,668	36%	
General and administration	30	617	21,856		8,761	40%	
Depreciation and amortization	12	400	6,376		6,024	94%	
Restructuring charges		115	434		(319)	-74%	
Total operating expenses	154	224	111,802		42,422	38%	
Income from operations	3	,042	11,376		(8,334)	-73%	
Other income (expense):							
Interest income		29	77		(48)	-62%	
Interest expense		(3)	(786)		783	-100%	
Other income		-	1,100		(1,100)	-100%	
Other expense		-	(1,183)		1,183	-100%	
Unrealized gain (loss) on investments		<u> </u>	(4)		4	-100%	
Total other income (expense)		26	(796)		822	-103%	
Income before income tax provision	3	068	10,580		(7,512)	-71%	
Income tax provision	2	603	2,975		(372)	-13%	
Net income	\$	465 \$	7,605	\$	(7,140)	-94%	

### Revenues

Total revenues increased 28% from \$123,213 in 2011 to \$157,266 in 2012. The increase was primarily due to an increase in revenues from assets under management or administration of \$27,977. Revenues from assets under management or administration comprised 81% of total revenues in both 2012 and 2011.

### Assets under management or administration

Revenues earned from assets under management or administration increased 28% from \$99,236 in 2011 to \$127,213 in 2012. This increase was primarily due to an increase in asset values applicable to our quarterly billing cycles in 2012, relative to those used in 2011. Our 2012 revenues were positively affected by new account growth and positive net flows of AUM and AUA during 2011 and through September 2012, as well as an increase in revenues related to FundQuest.

The number of financial advisors with AUM or AUA on our technology platform increased from 13,887 as of December 31, 2011 to 16,085 as of December 31, 2012 and the number of AUM or AUA client accounts increased from approximately 341,000 as of December 31, 2011 to approximately 450,000 as of December 31, 2012.

### Licensing and professional services

Licensing and professional services revenues increased 26% from \$23,942 in 2011 to \$30,053 in 2012. This increase was primarily due to an increase in licensing revenue of \$3,787 and an increase in professional services revenue of \$2,327. The increase in licensing revenue was primarily a result of the acquisitions of Prima and Tamarac, partially offset by the renegotiated license agreement with Fidelity.

## Cost of revenues

Cost of revenues increased 31% from \$42,831 in 2011 to \$56,119 in 2012, primarily due to the corresponding increase in revenues from AUM or AUA. As a percentage of total revenues, cost of revenues increased from 35% in 2011 to 36% in 2012.

# Compensation and benefits

Compensation and benefits increased 36% from \$40,305 in 2011 to \$54,973 in 2012, primarily due to an increase in salaries, benefits and commissions of \$12,416 related to an increase in headcount as a result of the FundQuest, Prima and Tamarac acquisitions. In addition, incentive compensation and non-cash compensation expense increased \$1,137 and \$1,280, respectively, over the prior year period. Headcount increased from an average of 486 in 2011 to an average of 655 in 2012, primarily to support the growth of our operations, as well as increased headcount from acquisitions. As a percentage of total revenues, compensation and benefits increased from 33% in 2011 to 35% in 2012.

#### General and administration

General and administration expenses increased 40% from \$21,856 in 2011 to \$30,617 in 2012, primarily due to increases in transaction-related costs of \$2,051, occupancy costs of \$1,814, communication, research and data services expense of \$1,283, travel and entertainment costs of \$998, marketing expenses of \$655 and website hosting and development expenses of \$600. As a percentage of total revenues, general and administration expenses increased from 18% in 2011 to 19% in 2012.

### Depreciation and amortization

Depreciation and amortization expense increased 94% from \$6,376 in 2011 to \$12,400 in 2012, primarily due to an increase in intangible asset amortization of \$5,230 and fixed asset depreciation of \$823. The increase in intangible asset amortization was due to an increase in intangible assets as a result of the FundQuest, Prima and



Tamarac acquisitions (see note 3 to the notes to consolidated financial statements). The increase in depreciation and amortization expense was primarily due to increases in capitalized computer equipment and software to support the growth of our operations. As a percentage of total revenues, depreciation and amortization increased from 5% in 2011 to 8% in 2012.

#### **Restructuring charges**

In 2012, we incurred restructuring charges of \$115 primarily for severance charges related to the termination of certain Prima and Tamarac employees related to these acquisitions. In 2011, we incurred restructuring charges of \$434 primarily for severance charges related to the termination of certain FundQuest and Envestnet employees related to the FundQuest acquisition.

## Interest expense

Interest expense decreased from \$786 in 2011 to \$3 in 2012, primarily due to eleven months of imputed interest on payments due to FundQuest in 2011 compared to no imputed interest in 2012. As discussed in note 3 to the notes to consolidated financial statements, as a result of the FundQuest acquisition and the related termination of the Platform Services Agreement with FundQuest, we ceased imputing interest expense as of the date of acquisition.

#### Other income

Other income decreased from \$1,100 in 2011 to zero in 2012. In 2011, the Company received proceeds from an insurance recovery (see note 15 to the notes to consolidated financial statements).

#### Other expense

Other expense decreased from \$1,183 in 2011 to zero in 2012. In 2011, the Company incurred non-cash contract settlement charges related to the termination of the Platform Services Agreement between the Company and FundQuest (see note 3 to the notes to consolidated financial statements).

#### Income tax provision

Year Ende	d December 3	۱,
2012		2011
(in t	nousands)	
\$ 2,603	\$	2,975
84.8%		28.1%

Our 2012 effective tax rate differs from the statutory rate primarily as a result of unrecognized tax benefits recorded in India, the effect of state tax rate changes, permanent differences, and changes in the valuation of federal and state net operating losses and adjustments to state deferred tax assets. The unrecognized tax benefits recorded relate to India tax exposure resulting from an examination performed by the India Taxing Authority. The change in state tax rates was primarily related to recognizing the benefit of state tax deductions on our federal tax return as well as changes in state tax laws regarding the sourcing of state taxable income. The adjustments to state deferred tax assets result from an analysis performed on the tax basis of fixed assets. It was determined that our net deferred tax assets did not properly reflect the future state tax benefits that will be recorded, and therefore we adjusted our balances accordingly.

Our 2011 effective tax rate differs from the statutory rate primarily as a result of changes in our estimates of our state income tax obligations for prior years, changes in state tax rates and the effect of permanent differences. Our 2011 effective tax rate also differs from the statutory rate primarily as a result of the reversal of certain deferred income tax liabilities totaling \$1,186 related to the termination of the Platform Services Agreement between Envestnet and FundQuest (see note 3 to the notes to consolidated financial statements).

Year ended December 31, 2011 compared to year ended December 31, 2010

		Year Ended December 31,			Increase (Decrease)		
	2011 2010		2010	Amount		%	
			(In th	iousands)			
Revenues:							
Assets under management or administration	\$	99,236	\$	75,951	\$	23,285	31%
Licensing and professional services		23,942		22,101		1,841	8%
Total revenues		123,178		98,052		25,126	26%
Operating expenses:							
Cost of revenues		42,831		31,444		11,387	36%
Compensation and benefits		40,305		37,027		3,278	9%
General and administration		21,856		21,607		249	1%
Depreciation and amortization		6,376		5,703		673	12%
Restructuring charges		434		961		(527)	-55%
Total operating expenses		111,802		96,742		15,060	16%
Income from operations		11,376		1,310		10,066	*
Other income (expense):							
Interest income		77		149		(72)	-48%
Interest expense		(786)		(564)		(222)	39%
Other income		1,100		-		1,100	100%
Other expense		(1,183)		-		(1,183)	100%
Unrealized gain (loss) on investments		(4)		12		(16)	-133%
Total other (expense)		(796)		(403)		(393)	49%
Income before income tax provision		10,580		907		9,673	*
Income tax provision		2,975		1,533		1,442	94%
Net income (loss)	\$	7,605	\$	(626)	\$	8,231	*

\*Not meaningful.

# Revenues

Total revenues increased 26% from \$98,052 in 2010 to \$123,178 in 2011. The increase was primarily due to an increase in revenues from assets under management or administration of \$23,285. Revenues from assets under management or administration comprised 81% and 77% of total revenues in 2011 and 2010, respectively.

# Assets under management or administration

Revenues earned from assets under management or administration increased 31% from \$75,951 in 2010 to \$99,236 in 2011. This increase was primarily due to an increase in asset values applicable to our quarterly billing cycles in 2011, relative to those used in 2010. Our 2011 revenues were positively affected by new account growth and positive net flows of AUM and AUA during the fourth quarter of 2010 through September 30, 2011. This increase was partially offset by a decrease in the market value of AUM and AUA from the fourth quarter of 2010 to September 30, 2011.

New account growth and positive net flows of AUM and AUA resulted from continued efforts to increase the number of financial advisors and accounts on our technology platform and the implementation of the FundQuest assets on our technology platform. The number of financial advisors with AUM or AUA on our technology platform increased from 13,833 as of December 31, 2010 to 13,887 as of December 31, 2011 and the number of AUM or AUA client accounts increased from approximately 307,000 as of December 31, 2010 to approximately 341,000 as of December 31, 2011.

### Licensing and professional services

Licensing and professional services revenues increased 8% from \$22,101 in 2010 to \$23,942 in 2011, primarily due to an increase in licensing revenue of \$935 and an increase in professional services revenue of \$911.

### Cost of revenues

Cost of revenues increased 36% from \$31,444 in 2010 to \$42,831 in 2011, primarily due to an increase in revenues from assets under management or administration. As a percentage of total revenues, cost of revenues increased from 32% in 2010 to 35% in 2011.

## Compensation and benefits

Compensation and benefits increased 9% from \$37,027 in 2010 to \$40,305 in 2011, primarily due to an increase in salaries and commissions of \$2,048 related to an increase in headcount, an increase in non-cash stock-based compensation expense of \$1,331 primarily due to the grant of stock options on the date of our initial public offering and an increase in benefits and payroll taxes of \$960, offset by a decrease in incentive compensation expense of \$976. Headcount increased from an average of 437 in 2010 to an average of 486 in 2011, primarily to support the growth of our operations, as well as increased headcount from acquisitions. As a percentage of total revenues, compensation and benefits decreased from 38% in 2010 to 33% in 2011.

# General and administration

General and administration expenses increased 1% from \$21,607 in 2010 to \$21,856 in 2011. Significant changes from 2010 to 2011 include a decrease of \$2,668 in bad debt expense related to the uncollectible portion of accounts and notes receivable from Fetter Logic (see note 16 to the notes to consolidated financial statements) and a decrease of \$1,897 in legal fees related to the Fetter Logic litigation, offset by increases in communication, research and data services expense of \$925, occupancy costs of \$580, insurance and bank charges of \$404, travel and entertainment costs of \$401 and professional and other legal fees of \$804. As a percentage of total revenues, general and administration expenses decreased from 22% in 2010 to 18% in 2011. Excluding bad debt expense of \$2,668 and legal fees of \$1,933 related to the Fetter Logic litigation, general and administration expenses as a percentage of total revenues would have been 17% in 2010.

### Depreciation and amortization

Depreciation and amortization expense increased 12% from \$5,703 in 2010 to \$6,376 in 2011, primarily due to an increase in fixed asset depreciation and amortization of \$855. The increase in depreciation and amortization expense was primarily due to increases in capitalized computer equipment and software to support the growth of our operations. As a percentage of total revenues, depreciation and amortization decreased from 6% in 2010 to 5% in 2011.

#### Restructuring charges

Effective March 31, 2010, we closed our Los Angeles office in order to more appropriately align and manage our resources and incurred restructuring charges of approximately \$961 in 2010 and \$53 in 2011. These expenses related to vacating rental office space, relocation expenses and severance charges. In 2011, we incurred restructuring charges of approximately \$381 primarily for severance charges related to the termination of certain FundQuest and Envestnet employees related to the FundQuest acquisition (see note 9 to the notes to consolidated financial statements).

#### Interest expense

Interest expense increased from \$564 in 2010 to \$786 in 2011, primarily due to imputed interest on the payments due to FundQuest. Due to the FundQuest acquisition and the related termination of the Platform Services Agreement with FundQuest, we have ceased imputing interest expense as of the date of acquisition (see note 3 to the notes to consolidated financial statements).

# Other income

Other income increased from zero in 2010 to \$1,100 in 2011, due to the proceeds from an insurance recovery in 2011 (see note 15 to the notes to consolidated financial statements).

## Other expense

Other expense increased from zero in 2010 to \$1,183 in 2011, due to the contract settlement charges related to the termination of the Platform Services Agreement between Envestnet and FundQuest (see note 3 to the notes to consolidated financial statements).

#### Income tax provision

	Year Ended December 31,			
	2011		010	
	 (in thousands)			
Income tax provision	\$ 2,975	\$	1,533	
Effective tax rate	28.1%		*	

#### \*Not meaningful.

Our 2011 effective tax rate differs from the statutory rate primarily as a result of changes in our estimates of our state income tax obligations for prior years, changes in state tax rates and the effect of permanent items. The changes in state tax rates were primarily related to changes in state tax laws regarding the sourcing of state taxable income. Our 2011 effective tax rate also differs from the statutory rate primarily as a result of the reversal of certain deferred income tax liabilities totaling \$1,187 related to the termination of the Platform Services Agreement between Envestnet and FundQuest (see note 3 to the notes to consolidated financial statements).

Our 2010 effective tax rate differs from the statutory rate primarily as a result of changes in our estimates of our state income tax obligations for prior years and changes in state tax rates. The changes in state tax rates were primarily related to changes in state tax laws regarding the sourcing of state taxable income. Our 2010 effective tax rate also differs from the statutory rate primarily as a result of an increase in our tax valuation allowance we recorded in 2010. In 2010, our management determined that newly generated deferred tax assets related to capital losses from investments and bad debts were not expected to be utilized and correspondingly, we increased our tax valuation allowance.

# Non-U.S. GAAP Financial Measures

	 Year Ended December 31,				
	2012		2011	2	2010
	 (in thousands)				
Adjusted revenues	\$ 158,514	\$	123,178	\$	98,052
Adjusted EBITDA	23,988		27,436		18,115
Adjusted net income	10,570		13,754		7,629
Adjusted net income per share	0.32		0.42		0.24

"Adjusted revenues" excludes the effect of purchase accounting on the fair value of acquired deferred revenue. Under U.S. GAAP, we record at fair value the acquired deferred revenue for contracts in effect at the time the entities were acquired. Consequently, revenue related to acquired entities for periods subsequent to the acquisition does not reflect the full amount of revenue that would have been recorded by these entities had they remained stand-alone entities.

"Adjusted EBITDA" represents net income (loss) before deferred revenue fair value adjustment, interest income, interest expense, income tax provision, depreciation and amortization, non-cash compensation expense,

restructuring charges and transaction costs, severance, litigation-related expense, gain (loss) on investments, other income, impairment of customer inducement asset, contract settlement charges, bad debt expense and customer inducement costs.

"Adjusted net income" represents net income (loss) before deferred revenue fair value adjustment, non-cash compensation expense, severance, amortization of acquired intangibles, litigation-related expense, non-recurring tax items, bad debt expense, customer inducement costs, contract settlement charges, contract settlement – reversal of deferred taxes, impairment of customer inducement asset, other income and imputed interest expense. Reconciling items, excluding bad debt expense, contract settlement charges, contract settlement – reversal of deferred taxes and non-deductible transaction costs, are tax effected using the income tax rates in effect on the applicable date.

"Adjusted net income per share" represents adjusted net income attributable to common stockholders divided by the diluted number of weightedaverage shares outstanding.

Our Board of Directors and our management use adjusted revenues, adjusted EBITDA, adjusted net income and adjusted net income per share:

- As measures of operating performance;
- For planning purposes, including the preparation of annual budgets;
- To allocate resources to enhance the financial performance of our business;
- To evaluate the effectiveness of our business strategies; and
- In communications with our Board of Directors concerning our financial performance.

Our Compensation Committee, Board of Directors and our management may also consider adjusted EBITDA, among other factors, when determining management's incentive compensation.

We also present adjusted revenues, adjusted EBITDA, adjusted net income and adjusted net income per share as supplemental performance measures because we believe that they provide our Board of Directors, management and investors with additional information to assess our performance. Adjusted revenues provide comparisons from period to period by excluding the effect of purchase accounting on the fair value of acquired deferred revenue. Adjusted EBITDA provide comparisons from period to period by excluding potential differences caused by variations in the age and book depreciation of fixed assets affecting relative depreciation expense and amortization of internally developed software, amortization of acquired intangible assets, amortization of customer inducement costs, litigation-related expense, severance, gain on investments, and changes in interest expense and interest income that are influenced by capital structure decisions and capital market conditions. Our management also believes it is useful to exclude non-cash stock-based compensation expense from adjusted EBITDA and adjusted net income because non-cash equity grants made at a certain price and point in time do not necessarily reflect how our business is performing at any particular time.

We believe adjusted revenues, adjusted EBITDA, adjusted net income and adjusted net income per share are useful to investors in evaluating our operating performance because securities analysts use adjusted revenues, adjusted EBITDA, adjusted net income and adjusted net income per share as supplemental measures to evaluate the overall performance of companies, and we anticipate that our investor and analyst presentations will include adjusted revenues, adjusted EBITDA, adjusted net income and adjusted net income per share.

Adjusted revenues, adjusted EBITDA, adjusted net income and adjusted net income per share are not measurements of our financial performance under U.S. GAAP and should not be considered as an alternative to revenues, net income, operating income or any other performance measures derived in accordance with U.S. GAAP, or as an alternative to cash flows from operating activities as a measure of our profitability or liquidity.

We understand that, although adjusted revenues, adjusted EBITDA, adjusted net income and adjusted net income per share are frequently used by securities analysts and others in their evaluation of companies, these measures have limitations as an analytical tool, and you should not consider them in isolation, or as a substitute for an analysis of our results as reported under U.S. GAAP. In particular you should consider:

- Adjusted revenues, adjusted EBITDA, adjusted net income and adjusted net income per share do not reflect our cash expenditures, or future requirements for capital expenditures or contractual commitments;
- Adjusted revenues, adjusted EBITDA, adjusted net income and adjusted net income per share do not reflect changes in, or cash requirements for, our working capital needs;
- Adjusted revenues, adjusted EBITDA, adjusted net income and adjusted net income per share do not reflect non-cash components of employee compensation;
- Although depreciation and amortization are non-cash charges, the assets being depreciated and amortized often will have to be replaced in the future, and adjusted EBITDA does not reflect any cash requirements for such replacements;
- Due to either net losses before income tax expenses or the use of federal and state net operating loss carryforwards in 2012, 2011 and 2010, we had cash income tax payments of \$796, \$813 and \$171 in the years ended December 31, 2012, 2011 and 2010, respectively. Income tax payments will be higher if we continue to generate taxable income and our existing net operating loss carryforwards for federal and state income taxes have been fully utilized or have expired; and
- Other companies in our industry may calculate adjusted revenues, adjusted EBITDA, adjusted net income and adjusted net income per share differently than we do, limiting their usefulness as a comparative measure.

Management compensates for the inherent limitations associated with using adjusted revenues, adjusted EBITDA, adjusted operating income, adjusted net income and adjusted net income per share through disclosure of such limitations, presentation of our financial statements in accordance with U.S. GAAP and reconciliation of adjusted revenues to revenues, the most directly comparable U.S. GAAP measure and adjusted EBITDA, adjusted net income and adjusted net income per share to net income per share, the most directly comparable U.S. GAAP measure. Further, our management also reviews U.S. GAAP measures and evaluates individual measures that are not included in some or all of our non-U.S. GAAP financial measures, such as our level of capital expenditures and interest income, among other measures.

The following table sets forth a reconciliation of total revenues to adjusted revenues based on our historical results:

	Ye	Year Ended December 31,					
	2012	2011	2010				
		(in thousands)					
Total revenues	\$ 157,266	\$ 123,178	\$ 98,052				
Deferred revenue fair value adjustment	1,248						
Adjusted revenues	\$ 158,514	\$ 123,178	\$ 98,052				

The following table sets forth the reconciliation of net income (loss) to adjusted EBITDA based on our historical results:

	Yea	Year Ended December 31,			
	2012	2011	2010		
		(in thousands)			
Net income (loss)	\$ 465	\$ 7,605	\$ (626)		
Add (deduct):					
Deferred revenue fair value adjustment	1,248	-	-		
Interest income	(29)	(77)	(149)		
Interest expense	3	786	564		
Income tax provision	2,603	2,975	1,533		
Depreciation and amortization	12,400	6,376	5,703		
Non-cash compensation expense	4,037	3,062	1,731		
Restructuring charges and transaction costs	2,718	1,054	961		
Severance	278	698	570		
Litigation-related expense	265	128	1,933		
(Gain) loss on investments	-	4	(12)		
Other income	-	(1,100)	-		
Impairment of customer inducement asset	-	174	-		
Contract settlement charges	-	1,183	-		
Bad debt expense	-	-	2,668		
Customer inducement costs	-	4,568	3,239		
Adjusted EBITDA	\$ 23,988	\$ 27,436	\$18,115		

The following table sets forth the reconciliation of net income (loss) to adjusted net income and adjusted net income per share based on our historical results:

	Year Ended December 31,				
	2012 *	2011 *	2010 *		
		(in thousands)			
Net income (loss)	\$ 465	\$ 7,605	\$ (626)		
Add:					
Deferred revenue fair value adjustment	746	-	-		
Non-cash compensation expense	2,414	1,831	1,077		
Restructuring charges and transaction costs	1,810	630	598		
Severance	166	417	355		
Amortization of acquired intangible assets	3,687	559	694		
Litigation-related expense	158	77	1,202		
Non-recurring tax items	1,124	-	-		
Bad debt expense	-	-	2,668		
Customer inducement costs	-	2,732	2,015		
Contract settlement charges	-	1,183	-		
Contract settlement - reversal of deferred taxes	-	(1,187)	-		
Impairment of customer inducement asset	-	104	-		
Other income	-	(658)	-		
Imputed interest expense	-	461	340		
Adjusted net income	10,570	13,754	8,323		
Less: Preferred stock dividends	-	-	(422)		
Less: Net income allocated to participating preferred stock	-	-	(2,069)		
Adjusted net income attributable to common stockholders	\$ 10,570	\$ 13,754	\$ 5,832		
Basic number of weighted-average shares outstanding	32,162,672	31,643,390	20,805,911		
Effect of dilutive shares:					
Options to purchase common stock	954,056	974,192	992,753		
Common warrants	177,257	211,495	154,364		
Restricted stock	47,630	34,757	-		
Diluted number of weighted-average shares outstanding	33,341,615	32,863,834	21,953,028		
Adjusted net income per share	\$ 0.32	\$ 0.42	\$ 0.27		

\* Adjustments, excluding non-recurring tax items, bad debt expense, contract settlement charges, contract settlement – reversal of deferred taxes and nondeductible transaction costs, are tax effected using income tax rates as follows: for 2012 and 2011—40.2%; for 2010—37.8%.

# Liquidity and Capital Resources

As of December 31, 2012, we had total cash and cash equivalents of \$29,983, compared to \$64,909 as of December 31, 2011. We plan to use existing cash as of December 31, 2012 and cash generated in the ongoing operations of our business to fund our current operations and capital expenditures in 2013.



# Cash Flows

The following table presents information regarding our cash flows and cash and cash equivalents for the periods indicated:

	Year Ended December 31,					
	 2012	2	2011		2010	
		(In the	ousands)			
Net cash provided by operating activities	\$ 28,541	\$	24,721	\$	1,467	
Net cash used in investing activities	(69,707)		(30,133)		(5,501)	
Net cash provided by financing activities	6,240		2,653		40,177	
Net increase (decrease) in cash and cash equivalents	(34,926)		(2,759)		36,143	
Cash and cash equivalents, end of period	29,983		64,909		67,668	

### **Operating** Activities

Net cash provided by operating activities in 2012 increased by \$3,820 compared to 2011, primarily due to a decrease in net earnings of \$7,140 in 2012 compared to the prior year period, a decrease in amortization of customer inducements of \$4,568, offset by an increase in \$6,024 in depreciation and amortization expense, and an overall net increase in operating assets and liabilities of \$12,939.

Net cash provided by operating activities in 2011 increased by \$23,254 compared to 2010, primarily due to an increase in net earnings of \$8,231 for the year ended December 31, 2011 compared to the prior year period and a decrease in customer inducement liability payments of \$10,300, primarily a result of a \$1,000 payment to FundQuest in 2011 compared to a payment of \$11,300 to FundQuest in 2010.

### **Investing** Activities

Net cash used in investing activities in 2012 increased by \$39,574 compared to 2011. In 2012, the Company acquired Prima and Tamarac for net cash totaling \$62,352 combined and in 2011, the Company acquired FundQuest for net cash totaling \$23,719 (see notes 3 to the notes to consolidated financial statements). Additionally, cash disbursements in 2012 and 2011 totaled \$7,188 and \$6,280, respectively, for purchases of property and equipment and capitalization of internally developed software.

Net cash used in investing activities in 2011 increased by \$24,632 compared to 2010. In 2011, the Company acquired FundQuest for net cash totaling \$23,719. Additionally, cash disbursements in 2011 and 2010 totaled \$6,280 and \$5,509, respectively, for purchases of property and equipment and capitalization of internally developed software.

### **Financing** Activities

Net cash provided by financing activities in 2012 increased by \$3,587 compared 2011, primarily a result of the proceeds from the issuance of restricted stock in the Tamarac acquisition of \$2,759 (see notes 3 and 12 to the notes to consolidated financial statements) offset by a decrease in proceeds from the exercise of stock options of \$678.

Net cash provided by financing activities in 2011 decreased by \$37,524 million compared to 2010, primarily due to the receipt of net proceeds of \$42,066 from our initial public offering after deducting underwriting discounts and offering costs in 2010, net proceeds of \$1,525 for the exercise of warrants in 2010 and net proceeds of \$1,925 from the exercise of stock options in 2010, partially offset by \$3,993 used in purchases of our common stock in 2010 and payments of preferred stock dividends of \$1,346 in 2010. In 2011, we received net proceeds of \$2,747 from the exercise of stock options.

# Commitments

The following table sets forth information regarding our contractual obligations as of December 31, 2012:

				Payments	Due by Peri	od			
	Total		ss than vear	1-3	3 vears	3_4	5 vears		re than vears
	 10141	1	ycai		housands)		5 years	5	years
Operating leases (1)	\$ 46,571	\$	4,488	\$	10,364	\$	10,646	\$	21,073
Purchase obligations	710		637		73		-		-
Total	\$ 47,281	\$	5,125	\$	10,437	\$	10,646	\$	21,073

(1) We lease facilities under non-cancelable operating leases expiring at various dates through 2023.

The table above does not reflect the following:

- Amounts estimated for uncertain tax positions since the timing and likelihood of such payments cannot be reasonably estimated.
- Voluntary employer matching contributions to our defined contribution benefit plans since the amount cannot be reasonably estimated. For the years ended December 31, 2012, 2011 and 2010, we made voluntary employer matching contributions of \$660, \$474 and \$427, respectively.

### **Off-Balance Sheet Arrangements**

Other than operating leases as indicated above, we do not have any other off-balance sheet arrangements.

#### **Recent Accounting Pronouncements**

In June 2011, the FASB issued authoritative guidance that amends ASC Topic 220, Comprehensive Income, to require that all non-owner changes in stockholders' equity be presented either in a single continuous statement of comprehensive income or in two separate but consecutive statements, and it eliminates the option to present components of other comprehensive income as a part of the statement of changes in stockholders' equity. In addition, this guidance requires an entity to present on the face of the financial statements reclassification adjustments for items that are reclassified from other comprehensive income to net income in the statement(s) where the components of net income and the components of other comprehensive income are presented. These amendments are to be applied retrospectively and are effective for fiscal years, and interim periods within those years, beginning after December 15, 2011; however, early adoption is permitted. The adoption of this guidance on January 1, 2012 did not have a material impact on our consolidated financial statements.

In September 2011, the FASB issued authoritative guidance regarding the testing of goodwill for impairment. This guidance allows companies to perform a "qualitative" assessment to determine whether or not the current two-step quantitative testing method, in which a company compares the fair value of reporting units to their carrying amounts including goodwill, must be followed. If a qualitative assessment indicates that it is more-likely-than-not that the fair value of a reporting unit is greater than its carrying amount, then the quantitative impairment test is not required. A company may choose to use the qualitative assessment on none, some, or all of its reporting units or to bypass the qualitative assessment and proceed directly to the two-step quantitative testing method. This guidance is effective for annual and interim goodwill impairment tests performed for fiscal years beginning after December 15, 2011; however, early adoption is permitted. The adoption of this guidance on January 1, 2012 did not have a material impact on our consolidated financial statements.

# Item 7A. Quantitative and Qualitative Disclosures About Market Risk

# Market risk

Our exposure to market risk is directly related to revenues from asset management or administration services earned based upon a contractual percentage of AUM or AUA. In the years ended December 31, 2012, 2011 and 2010, 81%, 81% and 77% of our revenues, respectively, were derived from revenues based on the market value of AUM or AUA. We expect this percentage to vary over time. A decrease in the aggregate value of AUM or AUA may cause our revenue and income to decline.

# Foreign currency risk

The expenses of our India subsidiary, which primarily consist of expenditures related to compensation and benefits, are paid using the Indian Rupee. We are directly exposed to changes in foreign currency exchange rates through the translation of these monthly expenditures into U.S. dollars. As of December 31, 2012, we estimate that a hypothetical 10% increase in the value of the Indian Rupee to the U.S. dollar would result in a decrease of \$470 to pre-tax earnings and a hypothetical 10% decrease in the value of the Indian Rupee to the U.S. dollar would result in a \$385 increase to pre-tax earnings.

### Interest rate risk

We have no floating interest rate debt, therefore we are not directly exposed to interest rate risk.

# Item 8. Financial Statements and Supplementary Data

## **Report of Independent Registered Public Accounting Firm**

To the Board of Directors and Stockholders Envestnet, Inc.:

We have audited the accompanying consolidated balance sheets of Envestnet, Inc. and subsidiaries (the "Company") as of December 31, 2012 and 2011, and the related consolidated statements of operations, stockholders' equity, and cash flows for each of the years in the three-year period ended December 31, 2012. These consolidated financial statements are the responsibility of the Company's management. Our responsibility is to express an opinion on these consolidated financial statements based on our audits.

We conducted our audits in accordance with the standards of the Public Company Accounting Oversight Board (United States). Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements. An audit also includes assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion.

In our opinion, the consolidated financial statements referred to above present fairly, in all material respects, the financial position of the Company as of December 31, 2012 and 2011, and the results of their operations and their cash flows for each of the years in the three-year period ended December 31, 2012, in conformity with U.S. generally accepted accounting principles.

We have also audited, in accordance with the standards of the Public Company Accounting Oversight Board (United States), the Company's internal control over financial reporting as of December 31, 2012, based on the criteria established in *Internal Control—Integrated Framework* issued by the Committee of Sponsoring Organizations of the Treadway Commission and our report dated June 14, 2013 expressed an adverse opinion on the effectiveness of the Company's internal control over financial reporting.

/s/ KPMG LLP

Chicago, Illinois June 14, 2013

# **Consolidated Balance Sheets**

# (In thousands, except share information)

	Decemb	oer 31,
	2012	2011
Assets		
Current assets:		
Cash and cash equivalents	\$ 29,983	\$ 64,909
Fees receivable	9,188	9,644
Deferred tax assets, net	2,089	2,605
Prepaid expenses and other current assets	2,501	4,929
Total current assets	43,761	82,087
Property and equipment, net	11,791	11,091
Internally developed software, net	4,324	3,524
Intangible assets, net	27,150	12,225
Goodwill	65,644	21,334
Deferred tax assets, net	6,194	4,279
Other non-current assets	3,535	3,162
Total assets	\$162,399	\$ 137,702
Liabilities and Stockholders' Equity		
Current liabilities:		
Accrued expenses	\$ 20,201	\$ 14,919
Accounts payable	2,614	1,974
Note payable	-	171
Deferred revenue	5,768	79
Total current liabilities	28,583	17,143
Deferred rent liability	2,195	1,414
Lease incentive liability	3,886	2,933
Other non-current liabilities	1,739	573
Total liabilities	36,403	22,063
Commitments and contingencies		
Stockholders' equity		
Preferred stock	-	-
Common stock, par value \$0.005, 500,000,000 shares authorized as of December 31, 2012 and 2011; 44,071,564 and 43,515,899 shares issued as of December 31, 2012 and 2011, respectively;		
32,355,675 and 31,810,726 shares outstanding as of December 31, 2012 and 2011, respectively	220	218
Additional paid-in capital	173,611	163,584
Accumulated deficit	(37,277)	(37,742)
Treasury stock at cost, 11,715,889 and 11,705,173 shares as of December 31, 2012 and 2011, respectively	(10,558)	(10,421)
Total stockholders' equity	125,996	115,639
Total liabilities and stockholders' equity	\$162,399	\$ 137,702

See accompanying notes to Consolidated Financial Statements.

# **Consolidated Statements of Operations**

# (In thousands, except share and per share information)

		Year ended December 31,				
	2012	2011	2010			
Revenues:						
Assets under management or administration	\$ 127,213	\$ 99,236	\$ 75,951			
Licensing and professional services	30,053	23,942	22,101			
Total revenues	157,266	123,178	98,052			
Operating expenses:						
Cost of revenues	56,119	42,831	31,444			
Compensation and benefits	54,973	40,305	37,027			
General and administration	30,617	21,856	21,607			
Depreciation and amortization	12,400	6,376	5,703			
Restructuring charges	115	434	961			
Total operating expenses	154,224	111,802	96,742			
Income from operations	3,042	11,376	1,310			
Other income (expense):						
Interest income	29	77	149			
Interest expense	(3)	(786)	(564)			
Other income	-	1,100	-			
Other expense	-	(1,183)	-			
Gain (loss) on investments		(4)	12			
Total other income (expense)	26	(796)	(403)			
Income before income tax provision	3,068	10,580	907			
Income tax provision	2,603	2,975	1,533			
Net income (loss)	465	7,605	(626)			
Less preferred stock dividends	-	-	(422)			
Net income (loss) attributable to common stockholders	\$ 465	\$ 7,605	\$ (1,048)			
Net income (loss) per share attributable to common stockholders:						
Basic	\$ 0.01	\$ 0.24	\$ (0.05)			
Diluted	\$ 0.01	\$ 0.23	\$ (0.05)			
Weighted average common shares outstanding:						
Basic	32,162,672	31,643,390	20,805,911			
Diluted	33,341,615	32,863,834	20,805,911			

See accompanying notes to Consolidated Financial Statements.

Consolidated Statements of Stockholders' Equity

(In thousands, except share information)

	Preferre	d Stock	Common	Stock	Treasury Stock							Total
	Shares	Amount	Shares	Amount	Common Shares	Preferred Shares	Amount	Additional Paid-in Capital		cumulated Deficit		ockholders , Eauity
Balance, December 31, 2009	76,643	S -	13,524,276	\$ 68	(613,600)		\$ (6,334)	\$ 106,893	\$	(43,375)	\$	57,252
Exercise of Series B warrants	1.497		<i>, , , , , , , , , , , , , , , , , , , </i>					1.497				1.497
Conversion of preferred stock to common stock effective upon	1,497	-	-	-	-	-	-	1,497		-		1,497
initial public offering	(45,890)	_	7,842,034	39	_		_	(39)		_		_
Merger transactions between EnvestNet	(13,050)		7,012,051	57				(37)				
Group, Inc. and Envestnet, Inc.:												
Issuance of Envestnet, Inc. common stock to												
EnvestNet Group, Inc. shareholders	-	-	10,680,000	54	-	-	-	(54)		-		-
Conversion of Envestnet, Inc. preferred and common			.,,									
stock held by EnvestNet Group,												
Inc., into treasury stock (common												
equivalents)	(32,250)	-	5,160,000	26	(10,680,000)	-	-	(26)		-		-
Net operating loss tax benefit recognized from												
EnvestNet Group, Inc. merger	-	-	-	-	-	-	-	839		-		839
Initial public offering of common stock, net of offering costs			5,411,325	27	-	-	-	42,039		-		42,066
Issuance of warrant to FundQuest, Inc.	-	-	-	-	-	-	-	2,946		-		2,946
Exercise of common warrants	-	-	154,548	-	-	-	-	28		-		28
Exercise of stock options	-	-	296,188	1	-	-	-	1,924		-		1,925
Stock-based compensation	-	-	-	-	-	-	-	1,731		-		1,731
Purchase of treasury stock (at cost)	-	-	-	-	(381,480)	(122)	(3,993)	-		-		(3,993)
Conversion of preferred stock in treasury to common stock	-	-	-	-	(24,469)	122	-	-		-		-
Preferred stock dividends	-	-	-	-	-	-	-	-		(1,346)		(1,346)
Net loss		-								(626)		(626)
Balance, December 31, 2010	-	\$ -	43,068,371	\$ 215	(11,699,549)	-	\$ (10,327)	\$ 157,778	\$	(45,347)	\$	102,319
Examples of stark antions			447,528	3	_			2,744				2,747
Exercise of stock options Stock-based compensation	-	-	447,528	5	-	-	-	3,062		-		3,062
Purchase of treasury stock (at cost)	-	-	-	-	(5,624)	-	(94)	5,002		-		(94)
Net income	-			-	(3,024)	-	(94)	-		7.605		7.605
									0		•	
Balance, December 31, 2011	-	\$ -	43,515,899	\$ 218	(11,705,173)	-	\$ (10,421)	\$ 163,584	\$	(37,742)	\$	115,639
Exercise of stock options	-	-	298,947	1	-	-	-	2,068		-		2,069
Issuance of common stock:			_, .,,					_,				_,
Vesting of restricted stock	-	-	24,568	-	-	-	-	-		-		-
Issuance of restricted stock	-	-	232,150	1	-	-	-	2,758		-		2,759
Stock-based compensation	-	-	-	-	-	-	-	4,342		-		4,342
Tax benefit attributable to exercise of stock options	-	-	-	-	-	-	-	1,549		-		1,549
Reversal of net operating loss tax benefit recognized from												
EnvestNet Group, Inc. merger	-	-	-	-	-	-	-	(690)		-		(690)
Purchase of treasury stock (at cost)	-	-	-	-	(10,716)	-	(137)	-		-		(137)
Net income	-	-	-	-	-	-	-	-		465		465
Balance, December 31, 2012	-	ş -	44,071,564	\$ 220	(11,715,889)	-	\$(10,558)	\$ 173,611	\$	(37,277)	\$	125,996

See accompanying notes to Consolidated Financial Statements.

# Consolidated Statements of Cash Flows

(In thousands)

	Y	Year ended December 31,			
	2012	2011	2010		
OPERATING ACTIVITIES:					
Net income (loss)	\$ 465	\$ 7,605	\$ (626		
Adjustments to reconcile net income (loss) to net cash provided by operating activities:	12 400	( )7(	5 702		
Depreciation and amortization Amortization of customer inducements	12,400	6,376 4,568	5,703 3,238		
Amontzation of customer inducements Deferred rent and lease incentive	1.389	4,308	5,238		
Deterred rend and lease incentive Deferred income taxes	1,389	2,162	1,215		
Stock-based compensation	4.342	3,062	1,213		
Excess tax benefits from stock-based compensation	(1,549)	5,002	1,751		
Non-cash interest expense	(1,549)	786	564		
Los (gain) on investments	-	4	(12)		
Loss (gain) on investments Impairment of customer inducement asset	-	174	(12)		
Contract settlement charges		1,183	-		
Provision for doubtful accounts	-	-	2,668		
Changes in operating assets and liabilities, net of acquisitions:			2,000		
Fees receivable	1,017	1,940	(3,718)		
Prepaid expenses and other current assets	4,645	(1,988)	(5,710)		
Other non-current assets	(188)	(1,006)	(52)		
Customer inducements. net	(100)	(1,000)	(11,300)		
Accrued expenses	3,100	802	2,437		
Accounts payable	640	267	(185)		
Deferred revenue	1,028	(507)	208		
Other non-current liabilities	1,166	(39)	137		
Net cash provided by operating activities	28,541	24,721	1.467		
	20,541	27,721	1,407		
INVESTING ACTIVITIES:	(4.020)	(4.700)	(11(0)		
Purchase of property and equipment	(4,838)	(4,798)	(4,169)		
Capitalization of internally developed software Repayment of notes payable assumed in acquisition	(2,350) (174)	(1,482) (162)	(1,340)		
Proceeds from repayment of notes receivable	(1/4)	(162)	985		
Floceeds from repayment on motes receivable	-	-	(90)		
Increase in noise receivance	- 7	28	30		
Acquisition of businesses, net of cash acquired	(62,352)	(23,719)	(917)		
Net cash used in investing activities	(69,707)	(30,133)	(5,501)		
FINANCING ACTIVITIES:					
Proceeds from exercise of stock options	2,069	2,747	1,925		
Issuance of restricted stock	2,759	-	-		
Excess tax benefits from stock-based compensation expense	1,549	-	-		
Proceeds from exercise of warrants	-	-	1,525		
Net proceeds from issuance of common stock	-	-	42,066		
Purchase of treasury stock	(137)	(94)	(3,993)		
Preferred stock dividends	<u> </u>	<u> </u>	(1,346)		
Net cash provided by financing activities	6,240	2,653	40,177		
INCREASE (DECREASE) IN CASH AND CASH EQUIVALENTS	(34,926)	(2,759)	36,143		
CASH AND CASH EQUIVALENTS, BEGINNING OF PERIOD	64,909	67,668	31,525		
CASH AND CASH EQUIVALENTS, END OF PERIOD	\$ 29,983				
CASH AND CASH EQUIVALENTS, END OF PERIOD	\$ 29,983	\$ 64,909	\$ 67,668		
Supplemental disclosure of cash flow information - cash paid during the period for income taxes, net of refunds	\$ 796	\$ 813	\$ 171		
Supplemental disclosure of non-cash investing and financing activities:	1.054	401			
Leasehold improvements funded by lease incentive	1,054	491	119		
Non-cash consideration issued in a business acquisition	-	4,897	-		
Issuance of warrant for customer inducement	-	-	2,946		
Customer inducement payable	-	-	19,261		
Note payable assumed in a business acquisition	-	-	300		
Contingent consideration issued in a business acquisition	-	-	150		

See accompanying notes to Consolidated Financial Statements.

# Envestnet, Inc. Notes to Consolidated Financial Statements (In thousands, except share and per share amounts)

### 1. Organization and Description of Business

Envestnet, Inc. ("Envestnet") and its subsidiaries (collectively, the "Company") provide open-architecture wealth management services and technology to independent financial advisors and financial institutions. These services and related technology are provided via the Envestnet Advisor Suite \*, Envestnet | PMC\*, Envestnet | Vantage<sup>TM</sup>, and Envestnet | Tamarac<sup>TM</sup>.

Advisor Suite is a platform of integrated, internet-based technology applications and related services that provide portfolio diagnostics, proposal generation, investment model management, rebalancing and trading, portfolio performance reporting and monitoring solutions, billing, and back-office and middle-office operations and administration.

The Company's investment consulting group, Envestnet | PMC, provides investment manager due diligence and research, a full spectrum of investment offerings supported by both proprietary and third-party research, and overlay portfolio management services.

Envestnet | Tamarac provides leading portfolio accounting, rebalancing, trading, performance reporting and client relationship management software, principally to high-end Registered Investment Advisors ("RIAs").

Through these platform and service offerings, the Company provides open-architecture support for a wide range of investment products (separately managed accounts, multi-manager accounts, mutual funds, exchange-traded funds, stock baskets, alternative investments, and other fee-based investment solutions) from Envestnet | PMC and other leading investment providers via multiple custodians, and also account administration and reporting services.

Envestnet operates four RIAs and a registered broker-dealer. The RIAs are registered with the Securities and Exchange Commission ("SEC"). The broker-dealer is registered with the SEC, all 50 states and the District of Columbia and is a member of the Financial Industry Regulatory Authority ("FINRA").

### 2. Summary of Significant Accounting Policies

The Company follows accounting standards established by the Financial Accounting Standards Board ("FASB") to ensure consistent reporting of financial condition, results of operations and cash flows. References to Generally Accepted Accounting Principles ("GAAP") in these footnotes are to the FASB *Accounting Standards Codification*<sup>TM</sup>, sometimes referred to as the codification or ASC.

Principles of Consolidation—The consolidated financial statements include the accounts of Envestnet and its wholly-owned subsidiaries: Oberon Financial Technology, Inc. ("Oberon"); NetAssetManagement, Inc. ("NAM"); Envestnet Asset Management, Inc.; Envestnet Portfolio Solutions, Inc. ("EPS") (formerly "FundQuest Incorporated"); Prima Capital Holding, Inc.; Tamarac, Inc. ("Tamarac"); Sigma Asset Management, LLC; PMC International, Inc. and its wholly-owned subsidiaries, Portfolio Management Consultants, Inc. and Portfolio Brokerage Services, Inc. ("PBS"). All significant intercompany transactions and balances have been eliminated in consolidation. Accounts denominated in a non-U.S. currency have been re-measured using the U.S. dollar as the functional currency.

*Management Estimates*—Management of the Company has made certain estimates and assumptions relating to the reporting of assets, liabilities, revenues and expenses and the disclosure of contingent assets and liabilities to prepare these consolidated financial statements in conformity with GAAP. Significant areas requiring the use of management estimates relate to estimating uncollectible receivables, revenue recognition, costs capitalized for internally developed software, valuations and assumptions used for impairment testing of goodwill, intangible and other long-lived assets, fair value of stock and stock options issued, fair value of customer inducement assets and liabilities, realization of deferred tax assets and assumptions used to allocate purchase prices in business combinations. Actual results could differ materially from these estimates under different assumptions or conditions.

# Notes to Consolidated Financial Statements (Continued) (In thousands, except share and per share amounts)

*Revenue Recognition*—The Company recognizes revenue from services related to asset management and administration, licensing and professional services fees. The Company recognizes revenue when all of the following conditions are satisfied: (i) there is persuasive evidence of an arrangement, (ii) the service or product has been provided to the customer and no uncertainties exist surrounding product acceptances (iii) the amount of fees to be paid by the customer is fixed or determinable; and (iv) the collection of fees is reasonably assured.

- Asset management and administration fees The Company derives revenues from fees charged as a percentage of the assets that are
  managed or administered on its technology platform by financial advisors, financial institutions, and their clients (collectively "customers")
  and for services the Company provides to its customers. Such services include investment manager due diligence and research, portfolio
  diagnostics, proposal generation, investment model management, rebalancing and trading, portfolio performance reporting and monitoring
  solutions, billing, and back office and middle-office operations and administration. Investment decisions for assets under management or
  administration are made by our customers. The asset management and administration fees the Company earns are generally based upon a
  contractual percentage of assets managed or administered on our platform based on preceding quarter-end values. The contractual fee
  percentages vary based on the level and type of services the Company provides to its customers. Fees related to assets under management or
  administration increase or decrease based on values of existing customer accounts. The values are affected by inflows or outflows of customer
  funds and market fluctuations.
- Licensing and professional services fees—

*Licensing*—The Company derives licensing fees from recurring contractual fixed fee contracts with larger financial institutions or enterprise clients. Licensing contracts allow the customer to provide a unique configuration of platform features and investment solutions for their advisors. The licensing fees vary based on the type of services provided and our revenues received under license agreements are recognized over the contractual term. The Company's license agreements do not generally provide its customers the ability to take possession of the software or host the software on the customers' own systems or through a hosting arrangement with an unrelated party.

When the Company enters into arrangements with multiple deliverables, exclusive of arrangements with software deliverables, it applies the FASB's guidance for revenue arrangements with multiple deliverables and evaluates each deliverable to determine whether it represents a separate unit of accounting based on the following criteria: (i) whether the delivered item has value to the customer on a stand-alone basis, and (ii) if the contract includes a general right of return relative to the delivered item, delivery or performance of the undelivered item(s) is considered probable and substantially in the control of the Company. Revenue is allocated to each unit of accounting or element based on relative selling prices. The Company determines relative selling prices by using either (i) vendor-specific objective evidence ("VSOE") if it exists; or (ii) third-party evidence ("TPE") of selling price. When neither VSOE nor TPE of selling price exists for a deliverable, the Company uses its best estimate of the selling price for that deliverable.

After determining which deliverables represent a separate unit of accounting, each unit is then accounted for under the applicable revenue recognition guidance. In cases where elements cannot be treated as separate units of accounting, the elements are combined into a single unit of accounting for revenue recognition purposes.

When the Company enters into arrangements with multiple deliverables involving software, the Company applies the American Institute of Certified Public Accountants' ("AICPA") accounting guidance for software. The entire arrangement fee is allocated to each element in the arrangement based on the respective VSOE of fair value of each element.

# Notes to Consolidated Financial Statements (Continued) (In thousands, except share and per share amounts)

 Professional services—The Company derives professional service fees from providing contractual customized service platform software development, which are recognized under a proportional performance model utilizing an output-based approach. The Company's contracts have fixed prices, and generally specify or quantify interim deliverables.

Revenues are recognized in the periods in which the related services are performed provided that persuasive evidence of an agreement exists, the fee is fixed or determinable, and collectability is reasonably assured. Cash received by the Company in advance of the performance of services is deferred and recognized as revenue when earned. Certain portions of the Company's revenues require management's consideration of the nature of the client relationship in determining whether to recognize as revenue the gross amount billed or net amount retained after payments are made to providers for certain services related to the product or service offering.

The Company uses the following factors to determine whether to record revenue on a gross or net basis is based on whether:

- the Company has a direct contract with the third party service provider;
- the Company has discretion in establishing fees paid by the customer and fees due to the third party service provider; and
- the Company has credit risk

When customer fees include charges for third party service providers where the Company has a direct contract with such third party service providers, gross revenue recognized by the Company equals the fee paid by the customer. The cost of revenues recognized by the Company is the amount due to the third party service provider.

In instances where the Company does not have a direct contract with the third party service provider, the Company cannot exercise discretion in establishing fees paid by the customer and fees due to the third party service provider, and the Company does not have credit risk, the Company records the revenue on a net basis.

Deferred Revenue—Deferred revenue primarily consists of implementation and set up fees, professional services, and license fee payments received in advance from customers.

*Cost of Revenues*—Cost of revenues primarily include expenses related to sub-advisory and clearing, custody and brokerage services. Generally, these expenses are calculated based upon a contractual percentage of the market value of assets held in customer accounts measured as of the end of each quarter and are recognized ratably throughout the quarter based on the number of days in the quarter.

Allowance for Doubtful Accounts—The Company evaluates the need for an allowance for doubtful accounts for potentially uncollectible fee receivables. In establishing the amount of the allowance, if any, customer-specific information is considered related to delinquent accounts, including historical loss experience and current economic conditions. As of December 31, 2012 and 2011, the Company's allowance for doubtful accounts was zero.

Segments—The Company's chief operating decision maker is its chief executive officer, who reviews financial information presented on a consolidated basis. Historically, the Company has determined that it has a single reporting segment and operating unit structure. As a result of the acquisitions as discussed in Note 3, the Company has re-examined its reporting and operating structure and has determined it continues to maintain a single reporting segment and operating unit structure.

# Notes to Consolidated Financial Statements (Continued) (In thousands, except share and per share amounts)

*Fair Value of Financial Instruments*—The carrying amounts of financial instruments, net of any allowances, including cash equivalents, fees receivable, accounts payable and accrued expenses are considered to be reasonable estimates of their fair values due to their short-term nature.

*Cash and Cash Equivalents*—The Company considers all highly liquid investments purchased with an original maturity of three months or less to be cash equivalents. Cash and cash equivalents are recorded at cost, which approximates fair value. The Company's financial instruments that are exposed to concentrations of credit risk consist primarily of cash and cash equivalents. The Company maintains its cash accounts at financial institutions in excess of amounts insured by the Federal Deposit Insurance Corporation ("FDIC"). The Company monitors such credit risk and has not experienced any losses related to such risk.

*Investments*—Investments are recorded at cost and reviewed for impairment. Investments are included in "Other non-current assets" on the consolidated balance sheets and consist of non-marketable investments in privately held companies, as well as other alternative investments. The Company reviews these investments on a regular basis to evaluate the carrying amount and economic viability of these investments. This policy includes, but is not limited to, reviewing each of the investee's cash position, financing needs, earnings/revenue outlook, operational performance, management/ownership changes and competition. The evaluation process is based on information that the Company requests from these investees. This information is not subject to the same disclosure regulations as U.S. publicly traded companies, and as such, the basis for these evaluations is subject to the timing and accuracy of the data received from these investees.

The Company's investments are assessed for impairment when a review of the investee's operations indicates that there is a decline in value of the investment and the decline is other than temporary. Such indicators include, but are not limited to, limited capital resources, limited prospects of receiving additional financing, and prospects for liquidity of the related securities. Impaired investments are written down to estimated fair value. The Company estimates fair value using a variety of valuation methodologies, including comparing the investee with publicly traded companies in similar lines of business, applying valuation multiples to estimated future operating results and estimated discounted future cash flows.

Property and Equipment—Property and equipment are stated at cost less accumulated depreciation and amortization. Depreciation of furniture and equipment is computed using the straight-line method based on estimated useful lives of the depreciable assets. Leasehold improvements are amortized on a straight-line basis over their estimated economic useful lives or the remaining lease term, whichever is shorter. Improvements are capitalized, while repairs and maintenance costs are charged to operations as incurred. Assets are reviewed for recoverability whenever events or circumstances indicate the carrying value may not be recoverable.

Customer Inducements—Payments made to customers as an inducement are capitalized and amortized against revenue on a straight-line basis over the term of the agreement.

Internally Developed Software—Costs incurred in the preliminary stages of development are expensed as incurred. Once an application has reached the development stage, internal and external costs, if direct and incremental, are capitalized until the software is substantially complete and ready for its intended use. Capitalization ceases upon completion of all substantial testing. The Company also capitalizes costs related to specific upgrades and enhancements when it is probable the expenditures will result in additional functionality. Maintenance and training costs are expensed as incurred. Internally developed software is amortized on a straight-line basis over its estimated useful life. Management evaluates the useful lives of these assets on an annual basis and tests for impairment whenever events or changes in circumstances occur that could impact the recoverability of these assets. There were no impairments of internally developed software during the years ended December 31, 2012, 2011 and 2010.

# Notes to Consolidated Financial Statements (Continued) (In thousands, except share and per share amounts)

*Goodwill and Intangible Assets*—Goodwill consists of the excess of the purchase price over the fair value of identifiable net assets of businesses acquired. Goodwill is reviewed for impairment each year using a two-step process that is performed at least annually or whenever events or circumstances indicate that impairment may have occurred. The Company has concluded that it has a single reporting unit. The first step is a comparison of the fair value of an internal reporting unit with its carrying amount, including goodwill. If the fair value of the reporting unit exceeds its carrying value, goodwill of the reporting unit is not considered impaired and the second step is unnecessary. If the carrying value of the reporting unit exceeds its fair value, a second test is performed to measure the amount of impairment by comparing the carrying amount of the goodwill to a determination of the implied fair value of the goodwill. If the carrying amount of the goodwill is greater than the implied value, an impairment loss is recognized for the difference. The implied value of the goodwill is determined as of the test date by performing a purchase price allocation, as if the reporting unit had just been acquired, using currently estimated fair values of the individual assets and liabilities of the reporting unit, together with an estimate of the fair value of the reporting unit taken as a whole. The estimate of the fair value of the reporting unit is based upon information available regarding prices of similar groups of assets, or other valuation techniques including present value techniques based upon estimates of future cash flow. No impairment charges have been recorded for the years ended December 31, 2012, 2011 and 2010.

Intangible assets are recorded at cost less accumulated amortization. Intangible assets are reviewed for impairment whenever events or changes in circumstances may affect the recoverability of the net assets. Such reviews include an analysis of current results and take into consideration the undiscounted value of projected operating cash flows.

*Long-Lived Assets*—Long-lived assets, such as property, equipment, capitalized internal use software and intangible assets subject to amortization are reviewed for impairment whenever events or changes in circumstances indicate that the carrying amount of an asset group may not be recoverable. Recoverability of assets to be held and used is measured by a comparison of the carrying amount of an asset group to estimated undiscounted future cash flows expected to be generated by the asset group. If the carrying amount of an asset group exceeds its estimated future cash flows, an impairment charge is recognized by the amount by which the carrying amount of the asset group exceeds the fair value of the asset group. All long-lived assets of the Company are located in the U.S., except for approximately \$764 and \$822 as of December 31, 2012 and 2011, respectively, which are located in India.

Management evaluates the useful lives of these assets on an annual basis and tests for impairment whenever events or changes in circumstances occur that could impact recoverability of these assets. There were no impairments to long-lived assets during the years ended December 31, 2012, 2011 and 2010.

*Leases*—In certain circumstances, the Company enters into leases with free rent periods, rent escalations or lease incentives over the term of the lease. In such cases, the Company calculates the total payments over the term of the lease and records them ratably as rent expense over that term.

*Income Taxes*—The Company uses the asset and liability method to account for income taxes. Deferred tax assets and liabilities are recognized for the future tax consequences attributable to differences between the financial statement carrying amounts of existing assets and liabilities and their respective tax bases and net operating loss carryforwards. Deferred tax assets and liabilities are measured using enacted tax rates expected to apply to taxable income in the years in which those temporary differences are expected to be recovered or settled. The effect on deferred tax assets and liabilities of a change in tax rates is recognized in income in the period that includes the enactment date. The Company records a valuation allowance to reduce deferred tax assets to an amount that is more likely than not to be realized.

The Company follows authoritative guidance related to how uncertain tax positions should be recognized, measured, disclosed and presented in the consolidated financial statements. This requires the evaluation of tax

# Notes to Consolidated Financial Statements (Continued) (In thousands, except share and per share amounts)

positions taken or expected to be taken in the course of preparing the Company's tax returns to determine whether the tax positions are "more-likely-than-not" of being sustained "when challenged" or "when examined" by the applicable tax authority. Tax positions deemed to meet the more-likely-than-not threshold would be recorded as a tax benefit or expense and liability in the current year. The tax benefits recognized in the consolidated financial statements from tax positions are measured based on the largest benefit that has a greater than 50% likelihood of being realized upon ultimate settlement.

*Advertising Costs*—The Company expenses all advertising costs as incurred and they are classified within general and administration expenses. Advertising costs totaled approximately \$1,504, \$1,388 and \$1,160 for the years ended December 31, 2012, 2011 and 2010, respectively.

*Business Combinations*—The Company accounts for business combinations under the acquisition method. The cost of an acquired company is assigned to the tangible and intangible assets acquired and the liabilities assumed on the basis of their fair values at the date of acquisition. The determination of fair values of assets acquired and liabilities assumed requires management to make estimates and use valuation techniques when market values are not readily available. Any excess of purchase price over the fair value of net tangible and intangible assets acquired is allocated to goodwill. Transaction costs associated with business combinations are expensed as incurred.

Stock-Based Compensation—Compensation cost relating to stock-based awards made to employees and directors is recognized in the consolidated financial statements using the Black-Scholes option-pricing model in the case of non-qualified stock option awards, and intrinsic value in the case of restricted stock awards. The Company measures the cost of such awards based on the estimated fair value of the award measured at the grant date and recognizes the expense on a straight-line basis over the requisite service period, which is the vesting period.

Determining the fair value of stock options requires the Company to make several estimates, including the volatility of its stock price, the expected life of the option, forfeiture rate, dividend yield and interest rates. Prior to July 28, 2010, the Company was not a publicly traded company. Accordingly, the Company had limited historical information on the price of its stock as well as employees' stock option exercise behavior. Because of this limitation, the Company cannot rely on its historical experience alone to develop assumptions for stock-price volatility and the expected life of its options. The Company estimates the expected life of its options using the "Simplified Method." The Company estimates stock-price volatility with reference to a peer group of publicly traded companies. Determining the companies to include in this peer group involves judgment. The Company utilizes a risk-free interest rate, which is based on the yield of U.S. zero coupon securities with a maturity equal to the expected life of the options. The Company has not and does not expect to pay dividends on its common shares.

The Company is required to estimate expected forfeitures of stock-based awards at the grant date and recognize compensation cost only for those awards expected to vest. The forfeiture assumption is ultimately adjusted to the actual forfeiture rate. Therefore, changes in the forfeiture assumptions may impact the total amount of expense ultimately recognized over the vesting period. Estimated forfeitures will be reassessed in subsequent periods and may change based on new facts and circumstances.

Reclassifications—Certain reclassifications were made to the December 31, 2011 consolidated balance sheet to conform to the 2012 presentation.

## **Recent Accounting Pronouncements**

In June 2011, the FASB issued authoritative guidance that amends ASC Topic 220, Comprehensive Income, to require that all non-owner changes in stockholders' equity be presented either in a single continuous statement of comprehensive income or in two separate but consecutive statements, and it eliminates the option to

# Notes to Consolidated Financial Statements (Continued) (In thousands, except share and per share amounts)

present components of other comprehensive income as a part of the statement of changes in stockholders' equity. In addition, this guidance requires an entity to present on the face of the financial statements reclassification adjustments for items that are reclassified from other comprehensive income to net income in the statement(s) where the components of net income and the components of other comprehensive income are presented. These amendments are to be applied retrospectively and are effective for fiscal years, and interim periods within those years, beginning after December 15, 2011; however, early adoption is permitted. The adoption of this guidance on January 1, 2012 did not have a material impact on the Company's consolidated financial statements.

In September 2011, the FASB issued authoritative guidance regarding the testing of goodwill for impairment. This guidance allows companies to perform a "qualitative" assessment to determine whether or not the current two-step quantitative testing method, in which a company compares the fair value of reporting units to their carrying amounts including goodwill, must be followed. If a qualitative assessment indicates that it is more-likely-than-not that the fair value of a reporting unit is greater than its carrying amount, then the quantitative impairment test is not required. A company may choose to use the qualitative assessment on none, some, or all of its reporting units or to bypass the qualitative assessment and proceed directly to the two-step quantitative testing method. This guidance is effective for annual and interim goodwill impairment tests performed for fiscal years beginning after December 15, 2011; however, early adoption is permitted. The adoption of this guidance on January 1, 2012 did not have a material impact on the Company's consolidated financial statements.

# 3. Business Acquisitions

#### FundQuest Incorporated

On December 13, 2011, the Company acquired all of the outstanding shares of FundQuest Incorporated ("FundQuest"), a subsidiary of BNP Paribas Investment Partners USA Holdings, Inc. for total consideration of \$27,796. FundQuest was renamed Envestnet Portfolio Solutions, Inc. ("EPS") subsequent to the acquisition. EPS provides managed account programs, overlay portfolio management, mutual funds, institutional asset management and investment consulting to registered investment advisors, independent advisors, broker-dealers, banks and trust organizations. The goodwill arising from the acquisition represents the expected synergistic benefits of the transaction and the knowledge and experience of the workforce in place. The goodwill recognized is not deductible for income tax purposes.

In February 2010, the Company signed a seven-year platform services agreement (the "Agreement") with FundQuest. Pursuant to the Agreement with FundQuest, the Company provided FundQuest and its clients with the Company's platform technology and support services, replacing FundQuest's technology platform. The Company earned fees based upon a contractual percentage of assets under administration. In connection with the Agreement, the Company was required to make various payments to FundQuest during the contract term as defined in the Agreement. These payments included an up-front payment upon completion of the conversion of FundQuest's clients' assets to the Company's technology platform, five annual payments and a payment after the fifth year of the Agreement calculated based on the average annual revenues the Company was to receive from FundQuest during the first five years of the contract term. In addition, the Company also issued to FundQuest a warrant to purchase 1,388,888 shares of its common stock, with an exercise price of \$10.80 for an estimated fair value of \$2,946 (see Notes 8 and 12). The present value of all payments and the fair value of the warrant was originally accounted for as customer inducement costs and were amortized as a reduction to the Company's revenues from assets under management or administration on a straight-line basis over the contract term of seven years. Customer inducement amortization totaled zero, \$4,568 and \$3,239 for 2012, 2011 and 2010, respectively, and imputed interest totaled zero, \$771 and \$546 for 2012, 2011 and 2010, respectively.
# Notes to Consolidated Financial Statements (Continued) (In thousands, except share and per share amounts)

Upon the acquisition, the Agreement between the Company and FundQuest was effectively settled. The Company analyzed the Agreement to determine the amount by which the contract was favorable or unfavorable when compared to current market pricing. The Company, using the discounted cash flow method, determined the Agreement resulted in a favorable amount of \$4,897. The favorable amount of the Agreement was compared to the net book value of the customer inducement asset and liability at the date of the business combination resulting in a charge of approximately \$1,183, which is included in other expense in the consolidated statements of operations for the year ended December 31, 2011. The net cash portion of the total consideration paid is included in "Cash flows from investing activities" in the consolidated statements of cash flows.

The consideration transferred in the acquisition was as follows:

Cash paid to owners	\$ 24,390
Non-cash consideration:	
Favorable contract	4,897
Other	1,241
Cash acquired	(671)
Working capital adjustment	(2,061)
	\$ 27,796

During 2012, the Company finalized the estimated working capital adjustment, which resulted in a decrease in goodwill of approximately \$889 and an increase in prepaid expense and other current assets, which was retrospectively adjusted in the December 31, 2011 consolidated balance sheet and the related notes to the consolidated financial statements.

The following table summarizes the estimated fair values of the assets acquired and liabilities assumed at the acquisition date of December 13, 2011, adjusted.

	December 31, 2011 (As Adjusted)
Accounts receivable	\$ 2,603
Prepaid expenses and other current assets	46
Property and equipment	442
Intangible assets	11,830
Goodwill	19,303
Accounts payable and accrued liabilities	(1,364)
Deferred income taxes	(4,710)
Deferred revenue	(354)
Total assets acquired	\$ 27,796

A summary of intangible assets acquired, estimated useful lives and amortization method was as follows:

			Weighted-	
			Average	
			Useful	
			Life	Amortization
	А	mount	In Years	Method
Customer list	\$	11,830	7	Accelerated

The results of EPS's operations are included in the consolidated statements of operations beginning December 13, 2011 and were not material to the 2011 results of operations.

## Notes to Consolidated Financial Statements (Continued) (In thousands, except share and per share amounts)

#### Prima Capital Holding, Inc. Acquisition

On April 5, 2012, the Company completed the acquisition of Prima Capital Holding, Inc. ("Prima"). In accordance with the stock purchase agreement, the Company acquired all of the outstanding shares of Prima for total consideration of approximately \$13,925. Prima provides investment management due diligence, research applications, asset allocation modeling and multi-manager portfolios to the wealth management and retirement industries. Prima's clientele includes banks, independent RIAs, regional broker-dealers, family offices and trust companies. The goodwill arising from the acquisition represents the expected synergistic benefits of the transaction and the knowledge and experience of the workforce in place. The goodwill recognized is not deductible for income tax purposes.

The consideration transferred in the acquisition was as follows:

Cash paid to owners	\$ 13,750
Cash acquired	(1,767)
Cash paid for working capital settlement	 1,942
	\$ 13,925

The following table summarizes the estimated fair values of the assets acquired and liabilities assumed at the date of acquisition:

Accounts receivable	\$	72
	ф	
Prepaid expenses and other current assets		36
Notes receivable		860
Property and equipment		103
Deferred income taxes - non current		1,328
Intangible assets		4,940
Goodwill		9,283
Accounts payable and accrued liabilities		(171)
Deferred income tax liabilities		(1,796)
Deferred revenue		(730)
Total net assets acquired	\$	13,925

A summary of intangible assets acquired, estimated useful lives and amortization method is as follows:

	An	iount	Weighted- Average Useful Life in Years	Amortization Method
Customer list	\$	3,740	10	Accelerated
Proprietary technology		700	5	Accelerated
Trade names		500	5	Accelerated
Total	\$	4,940		

The results of Prima's operations are included in the consolidated statement of operations beginning April 5, 2012. Prima's revenues and net loss for the nine months ended December 31, 2012 totaled \$3,626 and (\$791), respectively. The net loss for the nine months ended December 31, 2012 included pretax acquired intangible asset amortization of \$1,005.

# Notes to Consolidated Financial Statements (Continued) (In thousands, except share and per share amounts)

# Tamarac, Inc. Acquisition

On May 1, 2012, the Company completed the acquisition of Tamarac, Inc. ("Tamarac"). In accordance with the merger agreement, a newly formed subsidiary of Envestnet merged with and into Tamarac, and Tamarac became a wholly-owned subsidiary of Envestnet. Under the terms of the merger agreement, total consideration was approximately \$48,427 for all of the outstanding stock of Tamarac. Tamarac provides leading portfolio accounting, rebalancing, trading, performance reporting and client relationship management software, principally to high-end RIAs. The goodwill arising from the acquisition represents the expected synergistic benefits of the transaction and the knowledge and experience of the workforce in place. The goodwill recognized is not deductible for income tax purposes.

The consideration transferred in the acquisition was as follows:

Cash paid to owners	\$ 54,000
Non-cash consideration	101
Cash acquired	(2,533)
Receivable from working capital settlement	 (3,141)
	\$ 48,427

The following table summarizes the estimated fair values of the assets acquired and liabilities assumed at the date of acquisition.

Accounts receivable	\$ 489
Other receivables	681
Prepaid expenses and other current assets	216
Deferred income tax assets	7,235
Property and equipment	444
Deposits	379
Intangible assets	16,150
Goodwill	35,027
Accounts payable and accrued liabilities	(2,356)
Deferred income tax liabilities	(5,907)
Deferred revenue	 (3,931)
Total net assets acquired	\$ 48,427

A summary of intangible assets acquired, estimated useful lives and amortization method is as follows:

	Ап	nount	Weighted- Average Useful Life in Years	Amortization Method
Customer list	\$	8,680	12	Accelerated
Proprietary technology		5,880	8	Accelerated
Trade names		1,590	5	Accelerated
Total	\$	16,150		

### Notes to Consolidated Financial Statements (Continued) (In thousands, except share and per share amounts)

The results of Tamarac's operations are included in the consolidated statement of operations beginning May 1, 2012. Tamarac's revenues and net loss for the eight-month period ended December 31, 2012 totaled \$9,971 and (\$1,236), respectively. The net loss for the eight months ended December 31, 2012 included pre-tax acquired intangible asset amortization of \$1,304.

In accordance with the terms of the merger agreement between Envestnet and Tamarac, Tamarac senior management were required to apply at least 50% (up to 100%) of the aggregate proceeds of the Tamarac change of control payment totaling \$2,759 to purchase registered shares of Envestnet common stock (232,150 shares) in an amount equal to 95% multiplied by the Envestnet closing market price on the day before the merger closed (see Note 12).

In addition, the Company adopted the Envestnet, Inc. Management Incentive Plan for Envestnet | Tamarac Management Employees (the "2012 Plan"). The 2012 Plan provides for the grant of up to 559,551 shares of unvested common stock. The unvested common stock vests based upon Tamarac meeting certain performance conditions and then a subsequent two-year service condition (see Note 13). The Company also granted to certain Tamarac employees 232,150 stock options to acquire Envestnet common stock at an exercise price of \$12.51. These stock options vest on the second anniversary of the grant date (see Note 13).

Acquisition related costs of \$2,317 and \$405 are included in general and administration expenses in the consolidated statements of operations for the years ended December 31, 2012 and 2011, respectively.

### Pro forma results for Envestnet, Inc. giving effect to the FundQuest, Prima and Tamarac acquisitions

The following unaudited pro forma financial information presents the combined results of operations of Envestnet, Prima and Tamarac for the year ended December 31, 2012 and Envestnet, FundQuest, Prima and Tamarac for the year ended December 31, 2011. The unaudited pro forma financial information presents the results as if the acquisitions had occurred as of the beginning of 2011.

The unaudited pro forma results presented include amortization charges for acquired intangible assets and stock-based compensation expense, and the elimination of intercompany transactions, unrealized gain or loss on warrant, imputed interest expense, and transaction-related expenses and the related tax effect on the aforementioned items.

Pro forma financial information is presented for informational purposes and is not indicative of the results of operations that would have been achieved if the acquisitions had taken place as of the beginning of 2011.

	 At December 31,			
	2012		2011	
Revenues	\$ 162,859	\$	148,695	
Net income	528		3,772	
Net income per share:				
Basic	0.02		0.12	
Diluted	0.02		0.11	

# Notes to Consolidated Financial Statements (Continued) (In thousands, except share and per share amounts)

# 4. Property and Equipment

Property and equipment consists of the following:

			ıber 31,
	Estimated Useful Life	2012	2011
Cost:			
Office furniture and fixtures	5-7 years	\$ 3,613	\$ 2,713
Computer equipment and software	3 years	22,098	18,942
Other office equipment	5 years	598	598
Leasehold improvements	Shorter of the lease term or		
	useful life of the asset	7,638	5,833
		33,947	28,086
Less accumulated depreciation and			
amortization		(22,156)	(16,995)
Property and equipment, net		\$ 11,791	\$ 11,091

Depreciation and amortization expense was as follows:

	Year ended December 31,		
	2012	2011	2010
Depreciation and amortization expense	\$ 4,685	\$ 3,862	\$ 2,980

# 5. Internally Developed Software

Internally developed software consists of the following:

			At Dece	mber 3	1,
	Estimated Useful Life	2	012		2011
Internally developed software	5 years	\$	13,232	\$	10,882
Less accumulated amortization			(8,908)	_	(7,358)
Internally developed software, net		\$	4,324	\$	3,524

Amortization expense was as follows:

	Year ended December 31,					
	2	2012		2011	2	2010
Amortization expense	\$	1,550	\$	1,579	\$	1,606

# Notes to Consolidated Financial Statements (Continued) (In thousands, except share and per share amounts)

# 6. Goodwill and Intangible Assets

Changes in the carrying amount of the Company's goodwill were as follows:

Balance at December 31, 2010	\$ 2,031
FundQuest acquisition	 19,303
Balance at December 31, 2011	21,334
Prima acquisition	9,283
Tamarac acquisition	 35,027
Balance at December 31, 2012	\$ 65,644

Intangible assets consist of the following:

		December 31, 2012				December 31, 2011	
	Useful Life	Gross Carrying Amount	Accumulated Amortization	Net Carrying Amount	Gross Carrying Amount	Accumulated Amortization	Net Carrying Amount
Customer lists	4 - 12 years	\$ 28,103	\$ (8,720)	\$ 19,383	\$ 15,683	\$ (3,458)	\$ 12,225
Proprietary technologies	5 - 8 years	6,580	(657)	5,923	-	-	-
Trade names	5 years	2,090	(246)	1,844	-	-	-
Total intangible assets		\$ 36,773	\$ (9,623)	\$ 27,150	\$ 15,683	\$ (3,458)	\$ 12,225

Amortization expense was as follows:

	Year ended December 31,				
2	.012	20	011	2	2010
\$	6,165	\$	935	\$	1,117
	2 \$	2012	<u>2012</u> 20	2012 2011 \$ 6165 \$ 935	

Future amortization expense of the intangible assets as of December 31, 2012, is expected to be as follows:

Years endir	ng December 31:	
	2013	\$ 6,222
	2014	6,222 5,399
	2015	4,582
	2016	3,787
	2017	2,687
	Thereafter	4,473
		\$ 27,150

# Notes to Consolidated Financial Statements (Continued) (In thousands, except share and per share amounts)

### 7. Other Non-Current Assets

Other non-current assets consist of the following:

	At Decem	ber 31,
	2012	2011
Investment in private company	\$ 1,250	\$ 1,250
Deposits:		
Lease	1,655	1,313
Other	264	259
Other	366	340
	\$ 3,535	\$ 3,162

In April 2008, the Company entered into an agreement to purchase 1,250,000 Preferred A Units of a private company for a total purchase price of \$1,250. The Preferred A Units are entitled to a preferred distribution at a cumulative rate of 8% per annum of unreturned capital contributions, as defined in the agreement.

### 8. Fair Value Measurements

Financial assets and liabilities recorded at fair value in the consolidated balance sheets are categorized based upon a fair value hierarchy established by U.S. GAAP, which prioritizes the inputs used to measure fair value into the following levels:

Level 1: Inp	uts based on quote	d market prices in a	ctive markets for identica	al assets or liabilities at t	he measurement date.
--------------	--------------------	----------------------	----------------------------	-------------------------------	----------------------

Level 2:	Quoted prices for similar assets or liabilities in active markets; quoted prices for identical or similar assets and liabilities in
	markets that are not active; or inputs that are observable and can be corroborated by observable market data.

Level 3: Inputs reflect management's best estimates and assumptions of what market participants would use in pricing the asset or liability at the measurement date. The inputs are unobservable in the market and significant to the valuation of the instruments.

#### Fair Value on a Recurring Basis:

The Company periodically invests excess cash in money-market funds not insured by the FDIC. The Company believes that the investments in money market funds are on deposit with creditworthy financial institutions and that the funds are highly liquid. The fair values of the Company's investments in money-market funds are based on the daily quoted market prices for the net asset value of the various money market funds. These money-market funds are considered Level 1 assets and totaled approximately \$20,682 and \$52,383 as of December 31, 2012 and 2011, respectively, and are included in cash and cash equivalents in the consolidated balance sheets.

On February 8, 2010, the Company issued a warrant to FundQuest to acquire a certain amount of the Company's common stock (see Note 3) which was initially recorded as a current liability. As a result of the Company's initial public offering effective July 28, 2010, certain terms of the warrant were determined in accordance with the warrant agreement and the Company reclassified the estimated fair value of the warrant to additional paid-in capital during 2010.

# Notes to Consolidated Financial Statements (Continued) (In thousands, except share and per share amounts)

The table below presents a reconciliation of all assets and liabilities of the Company measured at fair value on a recurring basis using significant unobservable inputs (Level 3) for the period from December 31, 2009 to December 31, 2010:

	Warrant
Balance at December 31, 2009	\$ -
Issuance	2,360
Change in fair value	586
Transfer to additional paid-in capital	 (2,946)
Balance at December 31, 2010	\$ -

The Company assesses the levels of the investments at each measurement date, and transfers between levels are recognized on the actual date of the event or change in circumstances that caused the transfer, in accordance with the Company's accounting policy regarding the recognition of transfers between levels of the fair value hierarchy. There were no transfers between Levels 1, 2 and 3 during the year.

# 9. Accrued Expenses

Accrued expenses consist of the following:

	At Dec	ember 31,
	2012	2011
Accrued investment manager fees	\$ 12,937	\$ 8,451
Accrued compensation and related taxes	5,726	4,230
Accrued professional services	408	481
Accrued restructuring charges	-	290
Other accrued expenses	1,130	1,467
	\$ 20,201	\$ 14,919

Effective March 31, 2010, the Company closed its Los Angeles office in order to more appropriately align and manage the Company's resources. In the year ended December 31, 2010, the Company recognized pretax restructuring charges of \$961 consisting primarily of \$451 for accrued lease payments, \$414 for relocation and other expenses and \$96 for severance charges. In the year ended December 31, 2011, the Company recognized pretax restructuring charges of \$53 consisting primarily of relocation and other expenses.

As a result of the FundQuest acquisition in 2011, the Company incurred restructuring charges of \$381 in the year ended December 31, 2011, primarily severance charges related to the termination of certain FundQuest and Envestnet employees.

As a result of the FundQuest, Prima and Tamarac acquisitions, the Company incurred restructuring charges of \$115 in the year ended December 31, 2012, primarily severance charges for certain Tamarac employees and lease abandonment charges related to Prima.

# Notes to Consolidated Financial Statements (Continued) (In thousands, except share and per share amounts)

The summary of activity in accrued restructuring charges was as follows:

Balance at December 31, 2009	-
	\$
Restructuring provision incurred	961
Payments	(733)
Balance at December 31, 2010	228
Restructuring provision incurred	434
Payments	(372)
Balance at December 31, 2011	290
Restructuring provision incurred	115
Payments	(405)
Balance at December 31, 2012	\$ -

### 10. Note Payable

In connection with the acquisition of B-Ready Outsourcing Solutions, Inc. on April 1, 2010, the Company assumed a note payable in the amount of \$300 that bears simple interest of 8% per annum. A principal payment of \$150 plus interest was paid on April 30, 2011, and an additional principal payment of \$150 plus interest was paid on April 30, 2012.

### 11. Income Taxes

Income before income tax provision was generated in the following jurisdictions:

		Year ended December 31,			
	2012	2011	2010		
Domestic	\$ 2	702 \$ 10,291	\$ 619		
Foreign		366 289	288		
Total	\$ 3,	068 \$ 10,580	\$ 907		

The components of the income tax provision charged to operations are summarized as follows:

010
-
(16)
76
60
1,207
266
-
1,473
1,533

### Notes to Consolidated Financial Statements (Continued) (In thousands, except share and per share amounts)

Net deferred tax assets (liabilities) consist of the following:

		At December 31,		
	2	012	2	2011
Current:				
Deferred revenue	\$	346	\$	30
Prepaid expenses and accruals		(108)		162
Net operating loss and tax credit carryforwards		2,563		3,716
Total current deferred tax assets		2,801		3,908
Less valuation allowance		(712)		(1,303)
Net current deferred tax assets		2,089		2,605
Non-current:				
Deferred rent and lease incentives		2,212		1,666
Net operating loss and tax credit carryforwards		13,980		7,559
Loss on investments		2,157		2,157
Property and equipment and intangible assets		(13,284)		(5,617)
Stock compensation expense		3,058		537
Other		180		118
Total long-term deferred tax assets		8,303		6,420
Less valuation allowance		(2,109)	_	(2,141)
Net long-term deferred tax assets	\$	6,194	\$	4,279

During 2010, the write-off of notes receivable from Fetter Logic (see Note 16) was considered a capital loss for tax purposes. In assessing the realizability of this deferred tax asset, management determined that it was more-likely-than-not that the asset would not be realized and accordingly recorded an increase to the valuation allowance in the amount of \$926. The valuation allowance for net deferred tax assets as of December 31, 2012 and 2011 was \$2,821 and \$3,444, respectively. The valuation allowance as of December 31, 2012 and 2011 was related to capital losses of \$2,157 and federal and state net operating losses of \$644 for 2012 and \$1,287 for 2011, primarily due to Internal Revenue Code Section 382 limitations. In assessing the realizability of deferred tax assets, management considers whether it is more-likely-than-not that some or all of the deferred tax assets will be realized.

The ultimate realization of deferred tax assets depends on the generation of future taxable income during the periods in which net operating losses and temporary differences are deductible. Management considers the scheduled reversal of deferred tax assets and liabilities (including the impact of available carryback and carryforward periods), projected taxable income, and tax-planning strategies in making this assessment. In order to fully realize the deferred tax assets, the Company will need to generate future taxable income before the expiration of the deferred tax assets governed by the tax code. Based on the level of taxable income and projections for future taxable income over the periods for which the net operating losses are available and deferred tax assets are deductible, management believes that it is more-likely-than-not that in consideration of its recorded valuation allowance, it will realize the benefits of the net operating losses and any other deferred tax assets. The amount of the deferred tax assets considered realizable however, could be reduced in the near term if estimates of future taxable income during the carryforward period are reduced.

Upon exercise of stock options, the Company recognizes any difference between GAAP compensation expense and income tax compensation expense as a tax windfall or shortfall. The difference is charged to equity

. ..

# Envestnet, Inc.

# Notes to Consolidated Financial Statements (Continued) (In thousands, except share and per share amounts)

in the case of a windfall. When the exercise results in a windfall and the windfall results in net operating loss ("NOL"), or the windfall increases an NOL carryforward, no windfall is recognized until the deduction reduces the income tax payable. For GAAP purposes, the Company has recognized all previously suspended windfall tax benefits, as they will be utilized on the Company's 2012 tax return to reduce taxes payable. The benefit was recorded in stockholders' equity, and as such, does not impact the Company's effective tax rate.

The expected tax provision calculated at the statutory federal rate differs from the actual provision as follows:

	Year ended December 31,					
	2	012	1	2011	2	010
Tax provision, at U.S. federal statutory tax rate	\$	1,043	\$	3,597	\$	308
State income tax, net of federal tax benefit		64		449		42
Effect of permanent items		414		487		66
Change in assertion over permanent reinvestment of foreign earnings		-		(234)		-
Effect of return to provision adjustment		(81)		(113)		
Change in valuation allowance		(620)		-		927
Effect of contract settlement		-		(1,186)		-
Effect of change in state income tax rate		691		-		-
Unrecognized tax benefits		1,105		(25)		106
Effect of foreign tax credits		(87)		-		-
Effect of adjustments to state deferred taxes		(576)		-		-
Adjustments to state net operating losses		638		-		-
Foreign income taxes		(6)		-		76
Other		18		-		8
Income tax provision	\$	2,603	\$	2,975	\$	1,533

At December 31, 2012, the Company had NOL carryforwards for federal income tax purposes of \$42,912, which are available to offset future federal taxable income, if any, and expire as follows:

Years ending December 31:	
2019	\$ 1,484
2020	-
2021	-
2022	988
2023	10,990
2024	9,630
2025	696
2026	1,386
2027	1,687
2028	3,363
2029	3,969
2030	1,491
2031	1,808
2032	4,322
2033	1,098
	\$ 42,912

Of the \$42,912 in NOLs, due to Internal Revenue Code Section 382 limitations, approximately \$1,938 in NOLs will not be utilized. In addition, as of December 31, 2012, we had net operating loss carryforwards for state income tax purposes of \$30,271, available to reduce future income subject to income taxes. The federal and state net operating loss carryforwards expire through 2033. In addition, the Company has alternative minimum

### Notes to Consolidated Financial Statements (Continued) (In thousands, except share and per share amounts)

tax credit carryforwards of approximately \$445, which are available to reduce future federal regular income taxes, if any, over an indefinite period.

A reconciliation of the beginning and ending amount of unrecognized tax benefit was as follows:

		Year ended December 31,				
	201	2	20	11	20	)10
Unrecognized tax benefits balance at beginning of year	\$	364	\$	415	\$	326
Additions based on tax positions related to the current year		517		128		107
Additions (deletions) based on tax positions related to prior years		474		(55)		(9)
Reductions for lapses of statute of limitations		(258)		(235)		(8)
Unrecognized tax benefits balance at end of year	\$	1,097	\$	364	\$	415

At December 31, 2012, the amount of unrecognized tax benefits that would benefit the Company's effective tax rate, if recognized, was \$844. At this time, the Company estimates it is reasonably possible that the liability for unrecognized tax benefits will decrease by as much as \$400 in the next twelve months due to the completion of reviews by tax authorities, the voluntary filing of certain state income taxes and the expiration of certain statutes of limitations.

The Company recognizes potential interest and penalties related to unrecognized tax benefits in income tax expense. For the years ended December 31, 2012, 2011 and 2010, income tax expense includes \$448, \$14 and \$53, respectively, of potential interest and penalties related to unrecognized tax benefits. The Company had accrued interest and penalties of \$642 and \$194 as of December 31, 2012 and 2011, respectively.

The Company files a consolidated federal income tax return and separate tax returns with various states. Additionally, a subsidiary of the Company files a tax return in a foreign jurisdiction. The Company's tax returns for the calendar years ended December 31, 2011, 2010 and 2009 remain open to examination by the Internal Revenue Service in their entirety. With respect to state taxing jurisdictions, the Company's tax returns for fiscal year ended March 31, 2009, as well as calendar years ended December 31, 2011, 2010 and 2009 remain open to examination by various state revenue services.

The Company's India subsidiary is currently under examination by the India Taxing Authority for the fiscal year ended March 31, 2009 and March 31, 2010. Based on the outcome of examinations of the Company's subsidiary or the result of the expiration of statutes of limitations it is reasonably possible that the related unrecognized tax benefits could change from those recorded in the statement of financial position. It is possible that one or more of these audits may be finalized within the next twelve months. The Company's subsidiary's tax returns for the fiscal years ended March 31, 2007 through March 31, 2012 remain open to examination by the India Taxing Authority in their entirety.

# Notes to Consolidated Financial Statements (Continued) (In thousands, except share and per share amounts)

# 12. Stockholders' Equity

### **Preferred Stock**

Prior to the closing of the Company's initial public offering in July of 2010, the Company had the following \$0.001 par value convertible preferred stock authorized, issued and outstanding:

	Shares Authorized	Shares Issued and Outstanding	А	mount	Liq	gregate uidation eference
Series A Convertible Preferred Stock	66,000	65,649	\$	31,475	\$	65,649
Series B Convertible Preferred Stock	10,000	7,130		5,330		7,130
Series C Convertible Preferred Stock	5,000	3,864		8,787		9,000
Undesignated	119,000	-		-		-
	200,000	76,643	\$	45,592	\$	81,779

Each share of preferred stock was convertible at any time after the date of issuance and was convertible at various prices and into various amounts of common stock. The preferred stock had liquidation and voting rights as defined in each preferred stock agreement. The holders of Series C Convertible Preferred Stock ("Series C") were entitled to receive preferred dividends annually at a rate of 8% of the Series C original issue price, accruing and cumulative from the date of issue, whether or not earned or declared.

As noted below, upon closing of the Company's initial public offering, all of the Company's outstanding Series A, Series B and Series C Convertible Preferred Stock converted into an aggregate 13,002,034 shares of common stock. In addition, the holders of Series C were paid cumulative preferred dividends totaling approximately \$1,346.

On August 3, 2010, the Company amended its certificate of incorporation which increased the amount of authorized preferred stock to 50,000,000 shares and increased the par value to \$0.005 per share. There were no shares of preferred stock issued or outstanding as of December 31, 2012 and 2011.

#### Warrants

On March 24, 2005, in connection with the sale of Series B Convertible Preferred Stock ("Series B"), the Company issued detachable warrants to holders of Series B to purchase 1,497 shares of Series B at a price of \$1,000 per share. In 2010, all of the warrants were exercised and the Company issued 1,497 shares of Series B. In addition, during 2010, the Company acquired 122 shares of Series B for approximately \$378.

On September 18, 2008, in connection with the sale of Series C Convertible Preferred Stock ("Series C"), the Company issued detachable warrants to holders of Series C to purchase 154,548 shares of common stock at a price of \$0.05 per share. During 2010, all of the warrants were exercised and the Company issued 154,548 shares of common stock.

In February 2010, in connection with the Agreement (see Note 3), the Company issued to FundQuest a warrant to purchase shares of the Company's common stock, with an exercise price to be calculated as 120% of the Company's initial public offering price per share of the Company's common stock. As a result of the closing of the Company's initial public offering, the number of shares of common stock issuable to FundQuest under the warrant was determined to be 1,388,888 shares at an exercise price of \$10.80 per share and the estimated fair value of the warrant of \$2,946 is classified in equity as additional paid-in capital. During 2011, the warrant was sold by FundQuest to a third party. As of December 31, 2012, the warrant with issuable shares totaling 1,388,888 was outstanding.

# Notes to Consolidated Financial Statements (Continued) (In thousands, except share and per share amounts)

### **Common Stock**

On June 29, 2010, in connection with the cashless exercise of 179,624 expiring stock options, the Company repurchased 135,827 shares of its common stock from certain of its former employees for an aggregate purchase price of \$1,616 to fund the purchase price and tax obligations of such exercises, resulting in a net issuance of 43,797 shares. Included in the above amounts, were 90,000 expiring stock options from a former officer and director of the Company in which the Company repurchased 63,279 shares for an aggregate purchase price of \$753, resulting in a net issuance of 26,721 shares to this former officer and director.

On July 28, 2010, the Company completed its initial public offering whereby the Company sold 4,705,500 shares of common stock for a price of \$9.00 per share, which resulted in proceeds, before deducting underwriting discounts and commissions and other offering expenses, of approximately \$42,350.

Upon closing of the Company's initial public offering:

- as approved by the Board of Directors on June 22, 2010, one newly issued share of the Company's stock was exchanged for every five
  outstanding shares of stock, effective immediately prior to the effectiveness of the Company's registration statement on July 28, 2010. All
  share amounts and per share information referenced throughout the consolidated financial statements have been retroactively adjusted to reflect
  this reverse stock split;
- all of the Company's outstanding Series A, Series B and Series C convertible preferred stock converted into an aggregate 13,002,034 shares of common stock. In addition, the holders of Series C were paid cumulative preferred dividends totaling approximately \$1,346;
- The EnvestNet Group, Inc., the Company's 41% shareholder prior to the initial public offering (the "Envestnet Shareholder") merged with and
  into the Company, with the Company being the surviving entity. Pursuant to the merger, all of the shareholders of the Envestnet Shareholder
  exchanged their Envestnet Shareholder common shares and preferred shares for 10,680,000 shares of the Company's common stock; and
- each share of the Company's common and Series A convertible preferred stock held by the Envestnet Shareholder was automatically converted into treasury stock of the Company.

On August 3, 2010, the Company amended its certificate of incorporation, which increased the amount of authorized common stock to 500,000,000 shares and increased the par value to \$0.005 per share.

On August 31, 2010, the underwriters exercised their overallotment option to purchase an additional 705,825 shares of common stock for a price of \$9.00 per share, which resulted in proceeds, before deducting underwriting discounts and commissions, of approximately \$6,352.

In accordance with the terms of the merger agreement between Envestnet and Tamarac (see Note 3), Tamarac senior management were required to apply at least 50% (up to 100%) of the aggregate proceeds of the Tamarac change of control payment totaling \$2,759 to purchase registered shares of Envestnet common stock (232,150 shares) in an amount equal to 95% multiplied by the Envestnet closing market price on the day before the merger closed. These shares cannot be sold or otherwise transferred for a period of two years following the date of merger. If a participant terminates their employment with the Company or is terminated for cause, the participant shall be required to pay the Company an amount equal to 5% multiplied by the closing market price on the day before the merger closed for each of the shares purchased by the participant.

During the fourth quarter of 2012, the Company reversed a \$690 net operating loss tax benefit that was recognized incorrectly in 2010 as a result of the EnvestNet Group, Inc. merger.

### Notes to Consolidated Financial Statements (Continued) (In thousands, except share and per share amounts)

### 13. Stock-Based Compensation

On December 31, 2004, the Company adopted a stock incentive plan (the "2004 Plan"). The 2004 Plan provided for the grant of options to employees, consultants, and non-employee directors to purchase common stock, which vest over time and have a ten-year contractual term. To satisfy options granted under the 2004 Plan, the Company made common stock available from authorized but unissued shares or shares held in treasury, if any, by the Company. Stock options granted under the 2004 Plan were non-qualified stock options, as defined in the 2004 Plan agreement. Stock options were granted with an exercise price no less than the fair-market-value price of the common stock at the date of the grant.

The 2004 Plan has a change in control provision whereby if a change in control occurs and the participant's awards are not equitably adjusted, such awards shall become fully vested and exercisable and all forfeiture restrictions on such awards shall lapse. Based on the terms of the 2004 Plan, the Company's initial public offering did not trigger the change in control provision and did not result in any modifications to the outstanding equity awards under the 2004 Plan.

On February 3, 2010, the Board of Directors approved an increase to the number of shares of the Company's common stock available for issuance under the 2004 Plan by 1,875,230 shares.

On June 22, 2010, the Board of Directors approved the 2010 Long-Term Incentive Plan ("2010 Plan"), effective upon the closing of the Company's initial public offering. The 2010 Plan provides for the grant of options, stock appreciation rights, Full Value Awards (as defined in the 2010 Plan) and cash incentive awards to employees, consultants, and non-employee directors to purchase common stock, which vest over time and have a ten-year contractual term. The maximum number of shares of common stock that may be delivered under the 2010 Plan is equal to the sum of 2,700,000 plus the number of shares of common stock that are subject to outstanding awards under the 2004 Plan which are forfeited, expire or are cancelled after the effective date of the Company's initial public offering. Stock options and stock appreciation rights are granted with an exercise price no less than the fair-market-value price of the common stock at the date of the grant.

As a result of the merger between Envestnet and Tamarac (see Note 3), the Company adopted the Envestnet, Inc. Management Incentive Plan for Envestnet | Tamarac Management Employees (the "2012 Plan"). The 2012 Plan provides for the grant of restricted common stock, stock options and the purchase of common stock for certain Tamarac employees. The maximum number of shares of stock which may be issued with respect to awards under the 2012 Plan is 1,023,851.

The 2012 Plan provides for the grant of up to 559,551 shares of unvested common stock ("Target Incentive Awards"). The Target Incentive Awards vest based upon Tamarac meeting certain performance conditions and then a subsequent two-year service condition. The Company measured the cost of these awards based on the estimated fair value of the award as of the market closing price on the day before the acquisition closed. The Company is recognizing the estimated expense on a graded-vesting method over a requisite service period of three to five years, which is the estimated vesting period. The Company has estimated expected forfeitures at the grant date and will recognize compensation expense only for those awards expected to vest. The initial forfeiture assumption will be reassessed in subsequent periods and may change based upon new facts and circumstances. Changes in the forfeiture assumptions may impact the total amount of expense ultimately recognized over the vesting period.

The Company also granted to certain Tamarac employees 232,150 stock options to acquire Envestnet common stock at an exercise price of \$12.51. These stock options vest on the second anniversary of the grant date.

As of December 31, 2012, the maximum number of options and restricted stock available for future issuance under the Company's plans is 1,733,230.

# Notes to Consolidated Financial Statements (Continued) (In thousands, except share and per share amounts)

Employee stock-based compensation expense was as follows:

	Ye	ear ended December 31,	
	2012	2011	2010
Employee stock-based compensation expense	\$ 4,342	\$ 3,062	\$ 1,731
Tax effect on employee stock-based compensation expense	(1,643)	(1,159)	(655)
Net effect on income	\$ 2,699	\$ 1,903	\$ 1,076

Stock Options

The following weighted average assumptions were used to value options granted during the periods indicated:

	Yea	Year ended December 31,		
	2012	2011	2010	
Grant date fair value of options	\$4.96	\$5.14	\$3.71	
Volatility	39.7%	39.4%	37.5%	
Risk-free interest rate	1.2%	2.37%	2.2%	
Dividend yield	0.0%	0.0%	0.0%	
Expected term (in years)	6.0	6.0	6.2	

The following table summarizes option activity under the Company's plans:

	Shares	Weighted- Average Exercise Price	Weighted-Average Remaining Contractual Life (Years)	-	gregate nsic Value
Outstanding as of December 31, 2009	3,285,579	\$ 6.70			
Granted	2,049,461	9.18			
Exercised	(296,188)	6.50			
Forfeited	(40,515)	7.81			
Outstanding as of December 31, 2010	4,998,337	7.64	7.5	\$	47,083
Granted	486,833	12.37			
Exercised	(447,528)	6.14			
Forfeited	(173,924)	9.36			
Outstanding as of December 31, 2011	4,863,718	8.19	6.8		18,704
Granted	738,915	12.53			
Exercised	(298,947)	6.92			
Forfeited	(26,274)	11.03			
Outstanding as of December 31, 2012	5,277,412	8.86	6.3		26,885
Options exercisable	3,378,591	7.70	5.2		21,124

The aggregate intrinsic values in the table above represent the total pre-tax intrinsic value (the aggregate difference between the fair value of the Company's common stock on December 31, 2012, 2011 and 2010 of \$13.95, \$11.96 and \$17.06, respectively, and the exercise price of in-the-money options) that would have been received by the option holders had all option holders exercised their options as of that date.

### Notes to Consolidated Financial Statements (Continued) (In thousands, except share and per share amounts)

Exercise prices of stock options outstanding as of December 31, 2012 range from \$1.10 to \$13.98.

Other information was as follows:

	Yea	Year ended December 31,			
	2012	2011	2010		
Total intrinsic value of options exercised	\$ 1,611	\$ 3,082	\$ 1,511		
Cash received from exercises of stock options	2,069	2,747	1,925		
Cash received from issuance of restricted stock	2,759	-	-		

The following table summarizes the prices whereby the Company granted employee stock options from the period January 1, 2010 through July 27, 2010 (the date prior to the Company's initial public offering):

		Fair Ma	rket Value
		of Comr	non Stock
Grant Date	Options Granted	(per	share)
February 22, 2010	71,000	\$	13.45

Prior to our initial public offering on July 28, 2010, the Board of Directors determined the exercise price was the fair market value on the respective grant dates. Historically, determining the fair value of our common stock required making subjective judgments. The valuation of the Company's common stock considered a market approach and an income approach, incorporating the Company's historical and expected financial performance, relevant market, industry and economic trends, recent capital transactions, involving either the Company or comparable companies, and comparable public company valuations. The resulting calculation assigned a value for 100% of the Company's equity on a marketable equivalent, non-controlling interest basis.

After the value of the Company had been determined, the Company allocated the value to each class of its shares, including common stock. The value allocation methodology applies the principles set forth in the AICPA Practice Aid—Valuation of Privately-Held-Company Equity Securities Issued as Compensation ("Practice Aid"). The Practice Aid defines appropriate methods to allocate enterprise value to common shares when multiple share classes exist. Based on various factors, including the stage of a company's life and the timing and likelihood of various liquidity events, one method of allocation may be more appropriate than the others. The Company used the option pricing method, as defined in the Practice Aid, which treats each class of equity as having a "call option" on the enterprise value. The option pricing method considers the economic preferences and other rights attributable to each share class, resulting in a price for each of the share classes, including common stock. The valuations of common stock also reflected a discount for lack of marketability, adjusted over time to reflect the expected likelihood and timing of a liquidity event subsequent to each valuation date. No other discounts were applied in determining the value of the Company's common stock. There was inherent uncertainty in the estimates used in the valuations. If different discount rates, assumptions or weightings had been used, the valuations would have been different. From January 1, 2010 through July 27, 2010, the Company performed contemporaneous valuations to determine the fair value of the Company's common stock.

Since our initial public offering on July 28, 2010, the Company has not performed internal valuations or obtained independent valuations in order to determine the Company's stock price to reference when determining the fair value of our common stock in connection with the granting of stock options.

# Notes to Consolidated Financial Statements (Continued) (In thousands, except share and per share amounts)

### Restricted Stock Awards

Periodically, the Company grants restricted stock awards under the 2010 Plan to employees that vest one-third on each of the first three anniversaries of the grant date. The Company also granted restricted stock awards under the 2012 Plan that vest upon Tamarac meeting certain performance conditions and then a subsequent two-year service condition.

The following is a summary of the activity for unvested restricted stock awards granted under the Company's plans:

	Number of Shares	Averag Date Fa	ghted- ge Grant hir Value Share
Balance at December 31, 2010	-	\$	-
Granted	77,224		12.38
Vested	-		-
Forfeited	(3,404)		12.55
Balance at December 31, 2011	73,820		12.37
Granted	714,934		12.50
Vested	(24,568)		-
Expired/cancelled	(1,064)		12.45
Forfeited	(4,132)		12.49
Balance at December 31, 2012	758,990		12.49

At December 31, 2012, there was \$5,736 of unrecognized compensation cost related to unvested stock options which the Company expects to recognize over a weighted-average period of 1.7 years. At December 31, 2012, there was \$1,605 of unrecognized compensation cost related to restricted stock awards which the Company expects to recognize over a weighted-average period of 3.6 years. This excludes \$6,865 of potential unrecognized stock compensation cost related to the Target Incentive Awards, which the Company expects to recognize over the remaining estimated vesting period of 2.3 to 4.3 years. The amount of stock compensation cost related to the Target Incentive Awards is dependent upon Tamarac meeting certain performance conditions and then a subsequent two-year service condition.

#### 14. Earnings per Share

Net income per common share reflects the application of the two-class method for the year ended December 31, 2010. Under the two-class method, net income is allocated between common stock and other participating securities based on their respective participating rights. All classes of convertible preferred stock would participate pro rata in dividends and therefore are considered participating securities.

Basic net income per common share excludes dilution for potential common stock issuances and is computed by dividing net income attributable to common stockholders by the weighted average number of common shares outstanding for the period. Diluted net income per common share reflects the potential dilution that could occur if securities or other contracts to issue common stock were exercised or converted into common stock, unless they are anti-dilutive. For the year ended December 31, 2010, convertible preferred securities were excluded from the computation of diluted net income per share as their inclusion on an as if converted basis would have been anti-dilutive. For the year ended December 31, 2010, the convertible preferred securities are

# Notes to Consolidated Financial Statements (Continued) (In thousands, except share and per share amounts)

considered anti-dilutive as a result of such securities not having the contractual obligation to participate in losses of the Company. For the calculation of diluted net income per common share, the basic weighted average number of shares is increased by the dilutive effective of stock options, common warrants and restricted stock using the treasury stock method.

The following table provides a reconciliation of the numerators and denominators used in computing basic and diluted net income (loss) per share attributable to common stockholders:

	Year ended December 31,		
	2012	2011	2010
Basic income per share calculation:			
Net income (loss)	\$ 465	\$ 7,605	\$ (626)
Less: Preferred stock dividends			(422)
Net income (loss) attributable to common stockholders	\$ 465	\$ 7,605	\$ (1,048)
Basic number of weighted-average shares outstanding	32,162,672	31,643,390	20,805,911
Basic net income (loss) per share attributable to common stockholders	\$ 0.01	\$ 0.24	\$ (0.05)
Diluted income (loss) per share calculation:			
Net income (loss) attributable to common stockholders	\$ 465	\$ 7,605	\$ (1,048)
Basic number of weighted-average shares outstanding	32,162,672	31,643,390	20,805,911
Effect of dilutive shares:			
Options to purchase common stock	954,056	974,192	-
Common warrants	177,257	211,495	-
Restricted stock	47,630	34,757	-
Diluted number of weighted-average shares outstanding	33,341,615	32,863,834	20,805,911
Diluted net income (loss) per share attributable to common			
stockholders	\$ 0.01	\$ 0.23	\$ (0.05)

Common share equivalents for securities that were anti-dilutive and therefore excluded from the computation of diluted earnings per share are as follows:

	Year ended December 31,		
	2012	2011	2010
Options to purchase common stock	1,209,748	121,000	4,998,337
Common warrants	-	-	1,388,888
Unvested restricted stock	559,551		
Total	1,769,299	121,000	6,387,225

### Notes to Consolidated Financial Statements (Continued) (In thousands, except share and per share amounts)

### 15. Insurance Recovery

On April 26, 2011, the Company and its directors' and officers' liability insurance carrier entered into an agreement under which the insurance carrier agreed to pay the Company \$1,100 to reimburse the Company for defense fees and expenses incurred by the Company in 2010 related to certain litigation (see Note 16). This amount was received in 2011 and is included in other income in the consolidated statements of operations.

#### 16. Commitments and Contingencies

#### Leases

The Company rents office space under leases that expire at various dates through 2023. Future annual minimum lease commitments under these operating leases were as follows:

Years endi	ng December 31:	
	2013	\$ 4,488
	2014	5,083
	2015	5,281
	2016	5,537
	2017	5,109
	Thereafter	21,073
Total		\$46,571

Rent expense for all operating leases totaled:

	Year ended December 31,					
2012	2	2	011		2	010
\$ 4	4,008	\$	2,930	_	\$	2,529

#### **Purchase Obligations and Indemnifications**

The Company enters into unconditional purchase obligations arrangements for certain of its services that it receives in the normal course of business. Future minimum unconditional purchase obligations are as follows:

Years ending December 31:	
2013	\$ 637
2014	73
2015	-
2016	-
2017	-
Thereafter	-
	\$ 710

The Company includes various types of indemnification and guarantee clauses in certain arrangements. These indemnifications and guarantees may include, but are not limited to, infringement claims related to intellectual property, direct or consequential damages and guarantees to certain service providers. The type and amount of any potential indemnification or guarantee varies substantially based on the nature of each

### Notes to Consolidated Financial Statements (Continued) (In thousands, except share and per share amounts)

arrangement. The Company has experienced no previous claims and cannot determine the maximum amount of potential future payments, if any, related to such indemnification and guarantee provisions. The Company believes that it is unlikely it will have to make material payments under these arrangements and therefore has not recorded a contingent liability in the consolidated balance sheets.

#### Litigation

On November 23, 2009, the Company sued Fetter Logic, Inc. ("Fetter Logic"), a private company, and its chief executive officer seeking, among other things, unspecified damages for breaches of the investment agreement and operating agreement that the Company had entered into with Fetter Logic in December 2008 and a declaratory judgment that the Company owns certain rights in certain intellectual property. Fetter Logic asserted claims against the Company in a separate suit and in a counterclaim filed on November 30, 2009, for breaches of the investment agreement, breach of the implied covenant of good faith and fair dealing, breach of fiduciary duty, copyright infringement, misappropriation of trade secrets, an accounting, unjust enrichment and a declaratory judgment that Fetter Logic owns all rights in the contested intellectual property. Fetter Logic sought declaratory and injunctive relief, as well as unspecified compensatory and punitive damages. Both cases arose out of (1) an investment agreement, pursuant to which the Company purchased shares in Fetter Logic for approximately \$5,700, and (2) an operating agreement, under which the parties agreed to integrate their respective software applications and develop and sell joint product offerings. Fetter Logic alleged that the Company did not comply with the terms of the operating agreement to develop joint product offerings, but instead misappropriated Fetter Logic's intellectual property to develop products for its own benefit.

On July 8, 2010, the Company and Fetter Logic agreed to dismiss their respective lawsuits. As a result, the Company wrote off net fees receivable and notes receivable totaling \$2,283 in the year ended December 31, 2010. This amount is included in general and administration expenses in the consolidated statements of operations.

In connection with this settlement, none of the parties was required to make any payments to any other party, the Company relinquished its ownership interest in Fetter Logic, as well as operating receivables for services the Company provided during the term of the operating agreement and its rights under a promissory note issued to the Company in December 2009 in connection with its redemption of a portion of its ownership interest in Fetter Logic. In addition, the Company has the right to use any intellectual property developed or obtained by the Company in connection with the operating agreement. For the year ended December 31, 2010, the Company incurred legal fees of \$1,933 related to this matter.

The Company is also involved in other litigation arising in the ordinary course of its business. The Company does not believe that the outcome of any of these proceedings, individually or in the aggregate, would, if determined adversely to it, have a material adverse effect on its results of operations, financial condition, cash flows or business.

#### 17. Major Customers

One customer accounted for the following percentage of the Company's fees receivable:

	Determo	ci 51,
	2012	2011
Fidelity	11%	34%

### Notes to Consolidated Financial Statements (Continued) (In thousands, except share and per share amounts)

One customer accounted for the following percentage of the Company's revenues:

		December 31,	
	2012	2011	2010
Fidelity	22%	31%	31%

#### 18. Benefit Plan

The Company sponsors a profit sharing and savings plan under Section 401(k) or the Internal Revenue Code, covering substantially all domestic employees. The Company made voluntary employer matching contributions as follows:

	Year ei	Year ended December 31,		
	2012	2011	2010	
Voluntary employer matching contributions	\$ 660	\$ 474	\$ 427	

#### 19. Net Capital Requirements

PBS is a broker-dealer subject to the SEC Uniform Net Capital Rule (SEC Rule 15c3-1), which requires the maintenance of minimum net capital and requires that the ratio of aggregate indebtedness to net capital ("net capital ratio"), both as defined, shall not exceed 15 to 1. SEC Rule 15c3-1 also provides that equity capital may not be withdrawn or cash dividends paid if the resulting net capital ratio would exceed 10 to 1. At December 31, 2012, the Company had net capital of \$968, which was \$868 in excess of its required net capital of \$100. At December 31, 2012, the Company's net capital ratio was .12 to 1.

Additionally, PBS is subject to net capital requirements of certain self-regulatory organizations and at December 31, 2012, PBS was in compliance with such requirements.

#### 20. Subsequent Events

On April 11, 2013, the Company entered into a definitive agreement to acquire substantially all of the assets of the Wealth Management Solutions ("WMS") division of Prudential Investments for \$10,000 in cash upon closing, subject to certain post-closing adjustments, plus contingent consideration of up to a total of \$23,000 in cash, based upon meeting certain performance targets, to be paid over three years. WMS is a provider of technology solutions that enables financial services to firms develop and enhance their wealth management offerings. The Company anticipates the acquisition will be completed in the third quarter of 2013.

On April 11, 2013, the Company amended the 2012 Plan (see Note 13). The purpose of the amendment was to amend the methodology for determining the vesting requirements of performance awards granted under the 2012 Plan as well as the inclusion of additional Envestnet | Tamarac employees eligible to participate in the 2012 Plan. The amendment may result in changes to the timing or amount of expense to be recognized in future periods in connection with any performance awards granted under the 2012 Plan, compared to that which may have occurred without the amendment.

# 21. Quarterly Financial Data (Unaudited)

Quarterly results for the years ended December 31, 2012 and 2011 were as follows:

		2012		
	First	Second	Third	Fourth
Total revenues	\$ 32,642	\$ 37,962	\$ 42,283	\$ 44,379
Income (loss) from operations (1) (2)	1,232	(1,132)	920	2,022
Net income (loss) (3)	740	(668)	551	(158)
Net income (loss) per share				
Basic	0.02	(0.02)	0.02	(0.01)
Diluted	0.02	(0.02)	0.02	(0.01)
		201	1	
	First	201 Second	1 Third	Fourth
Total revenues	<b>First</b> \$ 29,262			<b>Fourth</b> \$ 30,542
Total revenues Income from operations (4)		Second	Third	
	\$ 29,262	<b>Second</b> \$ 31,334	Third \$ 32,040	\$ 30,542
Income from operations (4)	\$ 29,262 2,554	Second \$ 31,334 3,151	Third           \$ 32,040           3,226	\$ 30,542 2,445
Income from operations (4) Net income	\$ 29,262 2,554	Second \$ 31,334 3,151	Third           \$ 32,040           3,226	\$ 30,542 2,445

(1) During the fourth quarter, the Company recorded an adjustment to non-cash compensation expense that resulted in an increase of \$305 to income from operations.

(2) Included in income (loss) from operations for the first quarter, second quarter, third quarter and fourth quarter of 2012 is \$644, \$1,353, \$215 and \$506, respectively, of restructuring and transaction related costs.

(3) During the fourth quarter, the Company recorded certain adjustments to income taxes including \$848 related to additional India income taxes, and \$392 in additional income taxes primarily to adjust deferred tax assets related to net operating loss carryforwards.

(4) Included in income from operations for the first quarter, second quarter, third quarter and fourth quarter of 2011 is \$10, \$53, \$302 and \$689, respectively, of restructuring and transaction related costs.

#### Item 9. Changes in and Disagreements with Accountants on Accounting and Financial Disclosure

On February 13, 2013, our Audit Committee terminated McGladrey LLP ("McGladrey") as our independent registered public accountants and on February 14, 2013 engaged KPMG LLP ("KPMG") as our independent registered public accounting firm for our fiscal year ended December 31, 2012 and to re-audit our financial statements for the year ended December 31, 2011. In addition, on March 29, 2013, our Audit Committee determined that it needed to retain new independent registered public accountants to re-audit our financial statements for the year ended December 31, 2010. As a result, on April 2, 2013, the Audit Committee engaged KPMG to re-audit our financial statements for the year ended December 31, 2010.

Information regarding the change in independent accountants was reported in our Current Reports on Form 8-K filed with the SEC on February 14, 2013 and on April 3, 2013. McGladrey's reports on our consolidated financial statements for each of the years ended December 31, 2011 and 2010 did not contain an adverse opinion or disclaimer of opinion, and such reports were not qualified or modified as to uncertainty, audit scope, or accounting principle. During the years ended December 31, 2011 and 2010, and the period from January 1, 2012 through March 29, 2013, we did not have any disagreements with McGladrey on any matter of accounting principles or practices, financial statement disclosure or auditing scope or procedure, which disagreements, if not resolved to McGladrey's satisfaction, would have caused them to make reference thereto in their reports on Envestnet's consolidated financial statements for such periods. During the years ended December 31, 2011 and 2010, there were no reportable events as the term is described in Item 304(a)(1)(v) of Regulation S-K.

#### Item 9A. Controls and Procedures

#### Evaluation of Disclosure Controls and Procedures

Our management, with the participation of our chief executive officer and chief financial officer, evaluated the effectiveness of our disclosure controls and procedures as of December 31, 2012. The term "disclosure controls and procedures," as defined in Rules 13a-15(e) and 15d-15(e) under the Exchange Act, means controls and other procedures of a company that are designed to ensure that information required to be disclosed by a company in the reports that it files or submits under the Exchange Act is recorded, processed, summarized and reported, within the time periods specified in the SEC's rules and forms. Disclosure controls and procedures include, without limitation, controls and procedures designed to ensure that information required to be disclosed by a company in the reports that it files or submits under the Exchange Act is accumulated and communicated to the Company's management, including its principal executive and principal financial officers, as appropriate to allow timely decisions regarding required disclosure. Management recognizes that any controls and procedures, no matter how well designed and operated, can provide only reasonable assurance of achieving their objectives, and management necessarily applies its judgment in evaluating the cost-benefit relationship of possible controls and procedures. Based on the evaluation of our disclosure controls and procedures as of December 31, 2012, and due to the material weakness in our internal control over financial reporting described in our accompanying *Management's Report on Internal Control Over Financial Reporting*, our chief executive officer and chief financial officer concluded that, as of such date, our disclosure controls and procedures were not effective.

Management has excluded Tamarac, Inc. and Prima Capital Holding, Inc., from its assessment of internal control over financial reporting as of December 31, 2012, because those entities were acquired by the Company in purchase business combinations in the second quarter of 2012. Tamarac, Inc. and Prima Capital Holding, Inc. are wholly-owned subsidiaries whose total assets represent approximately 39% and 9%, respectively, and whose revenues represent approximately 6% and 2%, respectively, of the Company's consolidated financial statement amounts as of and for the year ended December 31, 2012.

### Management's Report on Internal Control Over Financial Reporting

Our management is responsible for establishing and maintaining adequate internal control over financial reporting for our company. Internal control over financial reporting is defined in Rule 13a-15(f) and 15d-15(f) promulgated under the Exchange Act, as a process designed by, or under the supervision of, a company's

principal executive and principal financial officers and effected by the Company's board of directors, management and other personnel, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles, and includes those policies and procedures that:

- pertain to the maintenance of records that, in reasonable detail, accurately and fairly reflect the transactions and dispositions of the assets of the company;
- provide reasonable assurance that transactions are recorded as necessary to permit preparation of financial statements in accordance with generally accepted accounting principles, and that receipts and expenditures of the company are being made in accordance with authorizations of management and directors of the company; and
- provide reasonable assurance regarding prevention or timely detection of unauthorized acquisition, use or disposition of the company's assets that could have a material effect on the financial statements.

Because of its inherent limitations, internal control over financial reporting may not prevent or detect misstatements. Projections of any evaluation of effectiveness to future periods are subject to the risk that controls may become inadequate because of changes in conditions, or that the degree of compliance with the policies or procedures may deteriorate.

Our management, including our chief executive officer and chief financial officer, assessed the effectiveness of our internal control over financial reporting as of December 31, 2012. In connection with this assessment, we identified a material weakness, as described below, in internal control over financial reporting as of December 31, 2012. A material weakness is a deficiency, or a combination of deficiencies, in internal control over financial reporting such that there is a reasonable possibility that a material misstatement of the annual or interim financial statements will not be prevented or detected on a timely basis. In making this assessment, our management used the criteria set forth by the Committee of Sponsoring Organizations of the Treadway Commission in *Internal Control—An Integrated Framework* (September 1992). Because of the material weakness described below, management concluded that, as of December 31, 2012, our internal control over financial reporting was not effective.

# Inadequate and ineffective controls over accounting for income taxes

We did not have adequate design or operation of controls that provide reasonable assurance that the accounting for income taxes, including the related financial statement disclosures, were in accordance with U.S. GAAP. Specifically, we relied on third-party subject matter experts and did not have sufficient technical expertise in the income tax function to provide adequate review and control with respect to the (a) identification and ongoing evaluation of uncertain tax positions in foreign tax jurisdictions; (b) complete and accurate recording of deferred tax assets and liabilities due to differences in accounting treatment for book and tax purposes; and (c) complete and accurate recording of inputs to the consolidated income tax provision and related accruals.

This material weakness resulted in post-closing adjustments to deferred income tax assets and liabilities, income taxes receivable, uncertain tax liabilities as included in other non-current liabilities, and income tax expense, which were corrected by management prior to the issuance of the Company's consolidated financial statements included herein.

KPMG, our independent registered public accounting firm, has audited our consolidated financial statements and the effectiveness of our internal control over financial reporting as of December 31, 2012. Their report appears below.



#### Remediation Plans

Management is committed to remediating the material weakness in a timely fashion. We have begun the process of executing remediation plans that address the material weakness in internal control over financial reporting relating to accounting for income taxes. Specifically, we replaced our third-party subject matter experts and also hired internal personnel dedicated to managing the income tax function to enhance our expertise in determining the appropriate accounting for material and complex tax transactions. In addition, management's planned actions to further address the material weakness include:

- Review of tax accounting process to identify and implement enhanced tax accounting processes and related internal control procedures;
- Enhancement of our process and internal controls related to the preparation of tax accounting position papers documenting our analysis and conclusions for all technical tax accounting matters, and
- Establish training and education programs for financial personnel responsible for income tax accounting.

The Audit Committee has directed management to develop a detailed plan and timetable for the implementation of the foregoing remedial measures (to the extent not already completed) and will monitor their implementation. In addition, under the direction of the Audit Committee, management will continue to review and make necessary changes to the overall design of the Company's internal control environment, as well as policies and procedures to improve the overall effectiveness of internal control over financial reporting.

Management believes the measures described above and others that will be implemented will remediate the control deficiencies identified and will strengthen our internal control over financial reporting. Management is committed to continuous improvement of the Company's internal control processes and will continue to diligently review our financial reporting controls and procedures. As management continues to evaluate and work to improve internal control over financial reporting, we may take additional measures to address control deficiencies or determine to modify, or in appropriate circumstances not to complete, certain of the remediation measures described above. We expect these remedial actions and or other actions related to this maternal weakness to be effectively implemented in 2013 in order to successfully remediate the material weakness reported within this Form 10-K by December 31, 2013.

If the remedial measures described above are insufficient to address the identified material weakness or are not implemented effectively, or additional deficiencies arise in the future, material misstatements in our interim or annual financial statements may occur in the future. Among other things, any unremediated material weakness could result in material post-closing adjustments in future financial statements. Furthermore, any such unremediated material weakness could have the effects described in "Item 1A, Risk Factors – In preparing our financial statements for 2012, we identified a material weakness in our internal control over financial reporting, and our failure to remedy this or other material weaknesses that we may identify in the future could result in material misstatements in our financial statements" in Part I of this Form 10-K.

### Changes in Internal Control Over Financial Reporting

There were no changes in internal control over financial reporting during the quarter ended December 31, 2012 that have materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.



### **REPORT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM**

To The Board of Directors and Stockholders Envestnet, Inc.:

We have audited Envestnet, Inc.'s (the Company's) internal control over financial reporting as of December 31, 2012, based on criteria established in *Internal Control — Integrated Framework* issued by the Committee of Sponsoring Organizations of the Treadway Commission (COSO). The Company's management is responsible for maintaining effective internal control over financial reporting and for its assessment of the effectiveness of internal control over financial reporting included in the accompanying Management's Annual Report on Internal Control Over Financial Reporting. Our responsibility is to express an opinion on the Company's internal control over financial reporting based on our audit.

We conducted our audit in accordance with the standards of the Public Company Accounting Oversight Board (United States). Those standards require that we plan and perform the audit to obtain reasonable assurance about whether effective internal control over financial reporting was maintained in all material respects. Our audit included obtaining an understanding of internal control over financial reporting, assessing the risk that a material weakness exists, and testing and evaluating the design and operating effectiveness of internal control based on the assessed risk. Our audit also included performing such other procedures as we considered necessary in the circumstances. We believe that our audit provides a reasonable basis for our opinion.

A company's internal control over financial reporting is a process designed to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles. A company's internal control over financial reporting includes those policies and procedures that (1) pertain to the maintenance of records that, in reasonable detail, accurately and fairly reflect the transactions and dispositions of the assets of the company; (2) provide reasonable assurance that transactions are recorded as necessary to permit preparation of financial statements in accordance with generally accepted accounting principles, and that receipts and expenditures of the company are being made only in accordance with authorizations of management and directors of the company; and (3) provide reasonable assurance regarding prevention or timely detection of unauthorized acquisition, use, or disposition of the company's assets that could have a material effect on the financial statements.

Because of its inherent limitations, internal control over financial reporting may not prevent or detect misstatements. Also, projections of any evaluation of effectiveness to future periods are subject to the risk that controls may become inadequate because of changes in conditions, or that the degree of compliance with the policies or procedures may deteriorate.

A material weakness is a deficiency, or a combination of deficiencies, in internal control over financial reporting, such that there is a reasonable possibility that a material misstatement of the company's annual or interim financial statements will not be prevented or detected on a timely basis. A material weakness related to the ineffective controls over income tax accounting and disclosure has been identified and included in management's assessment.

Envestnet, Inc. management excluded Tamarac, Inc. and Prima Capital Holding, Inc., from its assessment of the effectiveness of the Company's internal control over financial reporting as of December 31, 2012, because those entities were acquired by the Company in purchase business combinations in the second quarter of 2012. Tamarac, Inc. and Prima Capital Holding, Inc. are wholly-owned subsidiaries whose total assets represent approximately 39% and 9%, respectively, and whose revenues represent approximately 6% and 2%, respectively, of the related consolidated financial statement amounts as of and for the year ended December 31, 2012. Our audit of internal control over financial reporting of the Company also excluded an evaluation of the internal control over financial reporting of Tamarac, Inc. and Prima Capital Holding, Inc.

We also have audited, in accordance with the standards of the Public Company Accounting Oversight Board (United States), the consolidated balance sheets of the Company as of December 31, 2012 and 2011, and the related consolidated statements of operations, stockholders' equity, and cash flows for each of the years in the

three-year period ended December 31, 2012 and our report dated June 14, 2013 expressed an unqualified opinion on those consolidated financial statements. This material weakness was considered in determining the nature, timing and extent of audit tests applied in our audit of 2012 consolidated financial statements, and this report does not affect our report dated June 14, 2013, which expressed an unqualified opinion on those financial statements.

We do not express an opinion or any other form of assurance on management's statements referring to corrective actions taken after December 31, 2012, relative to the aforementioned material weakness in internal control over financial reporting.

/s/ KPMG LLP

Chicago, Illinois June 14, 2013

Directors

# Item 9B. Other Information

Item 10. Directors, Executive Officers and Corporate Governance

None.

# Part III

The following summa	rizes information concerning each of our directors.
Judson Bergman	Mr. Bergman, age 56, is the founder of our company and has served as our Chairman, Chief Executive Officer and a director since 1999. Prior to founding our company, Mr. Bergman was Managing Director at Nuveen. Mr. Bergman serves as a trustee of RS Investment Trust and RS Variable Products Trust, registered investment companies. Mr. Bergman received an MBA in finance and accounting from Columbia University and a BA from Wheaton College.
	Mr. Bergman has extensive familiarity with the financial services industry acquired through his years with the company and his experience at Nuveen as well as his education in finance and accounting.
Ross Chapin	Mr. Chapin, age 60, has served as a director of our company since 2001. Mr. Chapin is a Managing Director of Parametric Portfolio Associates LLC, a provider of structured portfolio management, which he joined as a senior executive in October 2005. Prior to Parametric, Mr. Chapin co-founded Orca Bay Partners, a private equity firm in 1998. Mr. Chapin received an MBA from Columbia University in finance and accounting, and an undergraduate degree from Denison University.
	Mr. Chapin has broad knowledge of the financial services industry and financial products acquired through his experience at Parametric. In addition, the Board benefits from Mr. Chapin's experience with a broad range of companies and industries acquired as a result of the review and analysis of investments by Orca Bay Partners and his education in finance and accounting.
Gates Hawn	Mr. Hawn, age 64, has served as a director of our company since 2004. Mr. Hawn is currently retired. He had previously worked as a Senior Advisor for Credit Suisse, an investment banking firm or its predecessors from 2000 to 2004. In 2000 Donaldson, Lufkin & Jenrette, or DLJ, merged with Credit Suisse, and Mr. Hawn was with DLJ from 1981 to 2000. Mr. Hawn received an undergraduate degree from Williams College.
	Mr. Hawn's has extensive knowledge of the financial services industry and financial products acquired through his experience at DLJ and Credit Suisse.
James Johnson	Mr. Johnson, age 75, has served as a director of our company since 2000. Mr. Johnson is a General Partner and Founder of Apex Venture Partners, or Apex, a private equity firm, which he founded in 1988. Prior to founding Apex, Mr. Johnson was one of three founding partners of Knightsbridge Partners, a private investment firm. Prior to Knightsbridge, Mr. Johnson served in senior management roles with Beatrice Foods, including corporate Chief Financial Offer and Senior Vice President of the \$6 billion U.S. Foods subsidiary. Mr. Johnson received an MBA from Northwestern University and a BS from Loyola University.

	Mr. Johnson has experience with a broad range of companies and industries acquired as a result of the review and analysis of investments by Apex and Knightsbridge Partners. The Board also benefits from Mr. Johnson's experience in senior financial and management roles at Beatrice Foods and his education in business administration.
Paul Koontz	Mr. Koontz, age 53, has served as a director of our company since 2004. Mr. Koontz has been a general partner at Foundation Capital Management, or Foundation Capital, a venture capital firm since 1996. Mr. Koontz serves as the Chairman of the Board of Financial Engines, Inc. and is on the boards of Yaolan Media (in Beijing), Numerate, eBates, and the Stanford University DAPER Fund. Mr. Koontz received a master's in engineering management from Stanford University and a BS from Princeton University.
	Mr. Koontz has experience with a broad range of early stage and developing companies and industries acquired as a result of the review and analysis of investments by Foundation Capital. The Board also benefits from Mr. Koontz's experience serving on the boards of directors of other companies in the financial services and other industries and his education in engineering management.
Charles Roame	Mr. Roame, age 47, has served as a director of our company since August 2011. Mr. Roame has served as Managing Partner of Tiburon Strategic Advisors, a provider of research, strategy consulting, and other related services primarily to financial services firms, since 1998. Mr. Roame has served on the boards of a variety of start-up ventures. Mr. Roame received an MBA from the University of Michigan and a BA from Michigan State University.
	Mr. Roame's qualifications to serve on our Board of Directors are primarily based on his industry experience.
Yves Sisteron	Mr. Sisteron, age 58, has served as a director of our company since 2004. Mr. Sisteron has been a Managing Partner and Co- Founder of GRP Partners, a private investment firm, since 2000. Mr. Sisteron serves on the boards of GumGum, Kyriba Corp. and Qualys, Inc. Mr. Sisteron holds a JD and an LLM from the University of Law (Lyon) and an LLM degree from the New York University School of Law.
	Mr. Sisteron has experience with a broad range of companies and industries acquired as a result of the review and analysis of investments by GRP Partners. The Board also benefits from Mr. Sisteron's experience serving on the boards of directors of other companies and his education in law.

### The Committees of the Board of Directors

Our Board of Directors, or our Board, has established an Audit Committee, a Compensation Committee and a Nominating and Governance Committee.

## The Audit Committee

The Audit Committee provides oversight of the integrity of our financial statements and financial reporting process, our compliance with legal and regulatory requirements, the system of internal controls, the audit process, the performance of our internal audit program and the performance, qualification and independence of the independent auditors.

The Audit Committee is composed entirely of directors who are independent of us and our management, as defined by the New York Stock Exchange, or NYSE, listing standards.

	The Board has determined that each member of the Audit Committee satisfies the financial literacy requirements of the NYSE, and that Messrs. Chapin, Johnson and Hawn are each audit committee financial experts, as that term is defined under the rules of the SEC.
	The members of the Audit Committee are Mr. Johnson (Chairman), Mr. Chapin and Mr. Hawn.
The Compensation Committee	The Compensation Committee has responsibility for evaluating the performance of our chief executive officer, or our CEO, and senior management and determining executive compensation in conjunction with the independent directors. The Compensation Committee also works with the Nominating and Governance Committee and the CEO on succession planning.
	The Compensation Committee is composed entirely of directors who are independent of us and our management, as defined by the NYSE listing standards.
	The members of the Compensation Committee are Mr. Sisteron (Chairman), Mr. Chapin, Mr. Koontz and Mr. Roame. Mr. Roame joined the Compensation Committee in February 2012.
The Nominating and Governance Committee	The responsibilities of the Nominating and Governance Committee include identifying individuals qualified to become Board members, recommending director nominees to the Board and developing and recommending corporate governance guidelines. The Nominating and Governance Committee also has responsibility to review and make recommendations to the full Board regarding director compensation. In addition to general corporate governance matters, the Nominating and Governance Committee assists the Board and the Board committees in their self-evaluations.
	The Nominating and Governance Committee is composed entirely of directors who are independent of us and our management, as defined by the NYSE listing standards.
	The members of the Nominating and Governance Committee are Mr. Hawn (Chairman), Mr. Johnson, Mr. Koontz and Mr. Roame. Mr. Roame joined the Committee in February 2012.

# Compliance with Section 16(a) of the Exchange Act

Our executive officers and directors are subject to the reporting requirements of Section 16 of the Exchange Act. Except as disclosed in the next sentence, we believe that all of our executive officers and directors complied with all filing requirements imposed by Section 16(a) of the Exchange Act on a timely basis during fiscal year 2012. Mr. Sisteron filed a Form 4 late regarding the purchase of 15,000 shares of our common stock in February and March of 2012 and Mr. Roame filed a Form 4 late regarding the purchase of 2,075 shares of our common stock in August 2012.

### **Executive Officers**

The information required by this item relating to our executive officers and other corporate officers is included under the caption "Executive Officers of the Registrant" in Item 1 of this report.

### **Code of Ethics**

We have adopted a code of ethics that applies to all of our employees, including our principal executive officer, our principal financial officer and our principal accounting officer. This code of ethics is posted on our Website. The Internet address for our Website is www.envestnet.com, and the code of ethics may be found from our main Web page by clicking first on "Investor Information" and then "Corporate Governance," and then on "Code of Business Conduct and Ethics."

We intend to disclose any amendment to, or waiver from, a provision of this code of ethics that are required to be publicly disclosed pursuant to the rules of the SEC or the New York Stock Exchange by posting such information to our Website, at the address and location specified above.

### Item 11. Executive Compensation

#### **Compensation Discussion and Analysis**

We operate in a highly competitive environment and our executive compensation program is designed to attract and retain talented executives who can execute our strategy. The discussion below describes the material elements of the 2012 compensation program for our named executive officers and the manner in which compensation decisions were made.

#### **Philosophy and Objectives**

Our executive compensation philosophy, as established by our Compensation Committee, is designed to:

- Attract and retain skilled executive officers;
- Support our business strategy and objectives; and
- Align the interests of our executive officers with those of our stockholders through a pay-for-performance philosophy.

We do not utilize formulas to determine compensation amounts and have established a set of guiding principles that have provided the foundation for all compensation programs for executive officers and all other employees. These guiding principles are as follows:

- Pay for performance in such a way as to drive our business strategy and objectives and create shareholder value, consistent with an acceptable
  risk profile and through legal and ethical means.
- The amount of overall total compensation should be attractive to executive officers, affordable for the company, proportional to the executive officer's contribution, and fair to shareholders and employees, while providing payouts that are clearly aligned with actual performance.
- Avoid controversial pay practices.
- Compensation should be transparent, understandable and effectively communicated to shareholders and employees.

We are committed to providing a comprehensive total rewards program to attract, retain, and reward highly qualified, diverse and productive employees. The total rewards program emphasizes alignment of employee efforts to support our corporate strategies. The components of the program include compensation, benefits, learning and development opportunities and recognition of employee performance. We strive to remain externally competitive in relevant labor markets while maintaining internal equity. The program also promotes fiscally responsible pay decisions, encourages efficient use of our resources and ensures compliance with applicable legal and contractual requirements.

To our employees, our compensation philosophy means fair pay based on their role in the company, a subjective determination of the market value of their job and their performance in that position. In addition, there is opportunity for additional rewards when we meet or exceed business objectives. Performance rewards provide employees with the opportunity to earn additional compensation beyond their base salary.

Compensation for our executive officers consists of three primary elements. They receive a base salary which is paid in semi-monthly cash installments, they receive an annual incentive-based cash payment, which is typically paid in February of the subsequent year or in quarterly installments and they receive an annual grant of restricted stock and stock options. For details regarding why we pay each element and how the amounts are determined, see "—Our 2012 Executive Compensation Program—Base Salary", "—Our 2012 Executive Compensation Program—Base Salary", "—Our 2012 Executive Compensation Program—Equity Awards". Although these sections discuss our practices employed in 2012, generally we plan to continue these practices in future years.

We do not have a specific policy that governs the allocation of compensation between cash and non-cash compensation or between long-term or current compensation. The allocations are driven primarily through a desire to pay what we view as competitive compensation, as determined solely by us based on our review of broad-based third party surveys and other generally available information, which we have historically used to obtain a general understanding of market compensation practices.

#### **Role of Compensation Committee and Management**

The Compensation Committee consists of four independent non-employee members of our Board of Directors. The Compensation Committee reviews and, as it deems appropriate, recommends to the Board of Directors policies, practices and procedures relating to the compensation of officers and other managerial employees and the establishment and administration of employee benefit plans.

The Compensation Committee determines, and recommends to the Board of Directors for approval, the Chief Executive Officer's compensation without the participation of the Chief Executive Officer. The Compensation Committee is also responsible for reviewing the performance of the Chief Executive Officer. Our Chief Executive Officer is the only executive officer that has a role in determining the compensation of our named executive officers other than himself. The Chief Executive Officer will provide the Compensation Committee with recommendations, which the Compensation Committee has the discretion to approve or disapprove, for (a) changes to base salary, (b) distribution of incentive-based annual bonus and (c) restricted stock and stock option grants.

#### **Competitive Market Review**

In December of 2012, the Compensation Committee retained an independent third-party compensation specialist, McLagan, a subsidiary of Aon Corporation, to assist in identifying and facilitating any future changes to our compensation program, including the amounts of compensation paid to our senior management, including the named executive officers.

McLagan was engaged to analyze the pay levels of senior management, including those of our named executive officers. In particular, McLagan's engagement involved (a) defining the roles and responsibilities of senior management, (b) identifying comparable firms within the relevant competitive marketplace and (c) assessing the comparability of senior management roles and compensation from those firms to our senior management. The comparable firms included companies with similar size, number of employees or business model.

In 2010, the Compensation Committee had engaged McLagan to perform a similar review.

### **Our 2012 Executive Compensation Program**

Our 2012 executive compensation program had three primary components: base salary, cash incentive compensation and equity awards.

**Base Salary.** Base salaries are intended to provide our executives with a degree of financial certainty and stability that does not depend on company performance. In determining the base salaries for our Chief Executive

Officer and the other named executive officers, the Compensation Committee, at the beginning of each year, reviews the overall scope of each executive officer's responsibilities while taking into account the base salaries paid by companies with which we compete for talent. For 2012, base salary adjustments were based on a subjective determination of competitive market rates, individual performance, changes in job duties and responsibilities and our overall budget for base salary increases. We did not follow a specific formula or set of criteria in determining base salary adjustments in 2012.

Market data, such as the base salary of comparable jobs at comparable companies, were reviewed to provide guidance as to what constitutes competitive base salaries. In addition, individual performance of the executive's duties and responsibilities is also considered. If the executive has performed his or her duties above expectations, then an increase in the base salary may be justified. Similarly, if the executive is given different duties or responsibilities or if they have changed jobs within the company, then their base salary may be increased or decreased accordingly.

In all cases where base salaries may be changed, the overall compensation budget must be sufficient for such changes. In certain extreme cases, our financial results and performance may lead to reductions in base salaries as a cost cutting measure.

Annual Incentive-Based Cash Compensation. We maintain an annual incentive-based cash compensation program comprising two elements: the Annual Incentive Program, which is intended to reward executives and employees based on our revenue growth and profitability; and Incentive Compensation, which is intended to reward executives for gross sales.

Annual Incentive Program. Under the Annual Incentive Program, a predetermined percentage of the revenues and profits from the preceding year are distributed to employees, including our executive officers. At the beginning of each year, the Compensation Committee approves the calculation methodology, or formula, which will be used at the end of the year to determine the amount of the annual bonus distribution. The formula includes the distribution amount as a percentage of revenues and Adjusted EBITDA, as well as a stretch incentive target and a minimum threshold. The distribution percentage, as determined by the Compensation Committee, is generally derived by (a) investigating the annual compensation practices of comparable financial services firms and (b) considering the resulting aggregate incentive for management and employees to meet or exceed the firm's financial expectations.

In calculating the amount to be paid under the Annual Incentive Program, the Compensation Committee utilizes revenues and Adjusted EBITDA, which is prior to the payment of any amount paid pursuant to the Annual Incentive Program and certain extraordinary non-cash or non-recurring general and administrative expenses. The amounts paid for 2012 under the Annual Incentive Program were based on our financial performance during the 12-month period ending December 31, 2012. The Compensation Committee established a minimum threshold amount of revenue and Adjusted EBITDA, defined as described above, of \$121.8 million and \$19.4 million, respectively. Performance above either or both of the minimum threshold amounts would result in an aggregate annual bonus pool amount of 0.101% of total revenue of revenue equal to or in excess of the threshold amount plus 0.101% of total revenue for each incremental \$1.5 million of revenue between \$121.8 million and \$129.5 million plus 1.0% of the excess revenue above \$129.5 million plus 2.2% of the incremental revenue above \$144.7 million plus 3.85% of incremental revenue above \$152.3 million plus 1.0% of revenue above \$167.5 million and 3.1% of threshold Adjusted EBITDA if the threshold amount has been achieved plus 3.9% of the excess Adjusted EBITDA above the threshold amount. The Compensation Committee has final authority to exercise its discretion in setting compensation amounts or awards for the company as a whole and for individuals and is not bound by the formula or by recommendations of Mr. Bergman nor of any consultant. For the amount to be earned in 2012, the Compensation Committee agreed with management's recommendation and approved the amount to be paid under the Annual Incentive Program of \$1.8 million.

At the end of each year, an allocation of the Annual Incentive Program to each eligible employee, including executive officers, is made. The Chief Executive Officer ("CEO") recommends to the Compensation

Committee the distribution amounts for each executive officer, including himself, based on a subjective analysis of his or her performance. For the named executive officers, the CEO subjectively assesses their performance broadly with consideration given to four general categories: integrity, intelligence/business knowledge, qualitative considerations and effectiveness. No quantitative criteria are used. The CEO's analysis is based on his sole and absolute discretion when assessing performance. Similarly, the amounts he recommends for individual Annual Incentive Program recommendations are based on his sole and absolute discretion. The CEO presents his recommendations to the Compensation Committee for their consideration along with his individual Annual Incentive Program distribution recommendations. The Compensation Committee reviews and makes the final approval for annual bonus distributions for the named executive officers, including the CEO. For the year ending December 31, 2012, the Compensation Committee chose to accept the individual annual bonus distribution recommendations which it received from the CEO. Distributions of the Annual Incentive Program earned for a fiscal year are paid in the subsequent fiscal year, generally within the first two months.

Incentive Compensation. We maintain a compensation program with amounts paid to eligible employees as incentive compensation based on a target percentage of gross sales. The target percentage is revised annually. Incentive Compensation amounts are calculated quarterly and paid out in 20% installments over the subsequent five quarters. Incentive Compensation payments are not guaranteed. The recipient must still be employed at the time of payment. Incentive Compensation payments are allocated among sales and service personnel and sales management, and the allocation of the incentive compensation payments is approved at the discretion of Mr. Bergman and Mr. Crager. The amount of the Incentive Compensation payments may also be adjusted based on our overall performance. In 2012, approximately \$4.3 million in incentive compensation was earned by eligible employees, of which Mr. Crager received \$142,511. Mr. Crager is the only executive officer who receives Incentive Compensation. The amount of Mr. Crager's Incentive Compensation was based on our total gross sales and the CEO's subjective assessment of his leadership of the sales organization.

Equity Awards. We grant stock options and restricted stock awards to our current and newly hired executive officers to enable them to share in our success and to reinforce a corporate culture that aligns employee interests with the interests of our stockholders. All equity grants to date have been awarded under the Envestnet Asset Management Group, Inc. 2004 Stock Incentive Plan, or the Envestnet, Inc. 2010 Long-Term Incentive Plan (the "Equity Plans"). It has been our practice to annually grant equity awards to employees, including executives, in recognition of performance and as an incentive for retention, as well as to align their interests with the interests of our stockholders. The size of these grants is based on a number of factors, including our subjective analysis of competitive practices, individual performance as determined in the discretion of the Compensation Committee, changes in the scope of the individual's position, internal equity and retention potential. Ultimately, all amounts were determined in the sole and absolute discretion of the Compensation Committee based on its conclusions as to what was appropriate and necessary after consideration of these factors. The Compensation Committee does not use a formula nor does it have formal procedures regarding grants of stock options. Historically, vesting for equity award grants occurs annually on the anniversary of the grant date with one-third vesting on each of the first three anniversaries. In keeping with the annual practice of issuing equity grants in the first quarter of the company's fiscal year, the Compensation Committee authorized equity grants equal to 1.2% of our outstanding number of fully-diluted shares as of December 31, 2011. In total, equity grants with respect to 72,792 shares of common stock were made on February 29, 2012 to our executive officers. Consistent with our executive compensation philosophy and to better align the interests of our employees with those of our stockholders through pay-for-performance, each award was made half in shares of restricted common stock and half in stock options. In addition, the recipient had the option to alter the mix of the equity award by exchanging one share of restricted common stock for options to purchase three shares of common stock, or options to purchase three shares of common stock for one share of restricted common stock, subject to the limitation that in the aggregate no more than 67% of the equity award could consist of shares of restricted common stock or options to purchase common stock, as applicable.

Our CEO, with the help of his management team, recommended to the Compensation Committee individuals who should receive stock option awards and the size of each individual award. The individual distribution amounts that were recommended were based on the CEO's subjective review of each individual's

performance within his or her role in our company since the previous year's grant, as well as a subjective determination of the competitive practices necessary to retain key employees. This distribution was submitted to the Compensation Committee for its consideration and approval. As noted above, the Compensation Committee has final authority to exercise its discretion in setting compensation amounts or awards and is not bound by the use of any formula or recommendations of the CEO nor of any consultant. In its discretion, the Compensation Committee approved the recommendations as submitted. The equity grants to our named executive officers on February 29, 2012 were as follows:

	Number of Shares
	Underlying
Name	2012 Equity Award
Judson Bergman	12,951
William Crager	16,713
Peter D'Arrigo	12,534
Scott Grinis	12,534
Brandon Thomas	10,836

### Supplemental Benefits

We provide the following benefits to our executives on the same basis as provided to all of our employees:

- Health, dental and vision insurance;
- Life insurance;
- Medical and dependent care flexible spending account;
- Short- and long-term disability, accidental death and dismemberment;
- A 401(k) plan, with company match; and
- A college scholarship plan for employees' children.

We believe these benefits are consistent with companies with which we compete for talent. Other than certain parking privileges to certain of our executive officers, we provide no perquisites to any of our employees, including our named executive officers.

#### **Recoupment of earned awards**

We do not currently have a formal policy requiring a fixed course of action with respect to compensation adjustments following later restatements of financial results. Under those circumstances, the Board of Directors or the Compensation Committee would evaluate whether compensation adjustments were appropriate, or required under applicable law, based on the facts and circumstances relating to the restatement.

#### **Regulatory limitations**

Section 162(m) of the Internal Revenue Code generally limits the deductibility for federal income tax purposes of compensation in excess of \$1 million to the chief executive officer or any of the next three most highly paid executive officers of a publicly held corporation (other than the chief financial officer). We expect that on an ongoing basis we will generally consider whether a form of compensation will be deductible under section 162(m) in determining executive compensation, though other factors will also be considered. However, we may authorize compensation payments that do not comply with the exemptions under section 162(m) when we believe that such payments are appropriate to attract and retain executive talent.

In light of the strong shareholder support on our say-on-pay proposal in 2011, we made no significant changes to the executive compensation program based on concerns or issues raised by our shareholders. We continue to desire feedback from our shareholders on our executive compensation program and will consider the views of our shareholders as we evaluate our compensation program in 2013.
# 2012 Summary Compensation Table

The following table contains compensation information for our Chief Executive Officer, our Chief Financial Officer, and the three other most highly compensated executive officers. We refer to these individuals as our "named executive officers" in other parts of this Form 10-K. The information included in this table reflects compensation paid to our named executive officers for services rendered to us.

Name and Principal Position	Year	Salary (\$)	Bonus (\$)	Stock Awards (\$)	Option Awards (\$)	All Other Compensation (\$)	Total
Judson Bergman Chief Executive Officer	2012 2011 2010	\$ 400,000 400,000 391,667	\$ 110,000 115,000 160,000	\$ 62,250	\$ 39,062 52,336 1,366,572	\$ 5,000 4,900 4,900	\$ 616,312 572,236 1,923,139
<i>William Crager</i> President	2012 2011 2010	334,583 330,000 327,500	30,000	38,832	66,786 130,840 596,058	147,511 144,190 211,531	587,712 635,030 1,135,089
Peter D'Arrigo Chief Financial Officer	2012 2011 2010	275,000 275,000 275,000	58,000 50,000 65,000	29,121	50,087 52,336 312,567	5,000 4,900 4,900	417,208 382,236 657,467
Scott Grinis Chief Technology Officer	2012 2011 2010	245,000 245,000 243,333	60,000 55,000 65,000	29,121	50,087 52,336 203,532	5,000 4,900 4,900	389,208 357,236 516,765
Brandon Thomas Chief Investment Officer	2012 2011 2010	240,000 240,000 238,333	54,000 50,000 65,000	39,703	37,547 52,336 225,339	5,000 4,900 4,900	376,250 347,236 533,572

(1) Bonuses earned for a fiscal year are paid in the subsequent fiscal year, generally within the first two months (e.g., the bonuses earned for 2012 were paid in February 2013). The amounts disclosed in the Bonus column relate to amounts paid under our Annual Incentive Program.

(2) Amounts disclosed in the Equity Awards column relate to grants of restricted stock and stock options in the identified year. With respect to each equity grant, the amounts disclosed reflect the full grant-date fair value in accordance with Financial Accounting Standards Board Accounting Standards Codification topic 718. Our assumptions with regard to determining the 2012 stock option values are set forth in note 1 to the 2012 Grants of Plan-Based Awards table.

(3) For each person other than Mr. Crager, the amounts disclosed in the All Other Compensation column reflect matching contributions to the executive's 401(k) account. For Mr. Crager, the amounts disclosed reflect \$142,511 earned as incentive compensation in 2012, \$139,290 earned as incentive compensation in 2011, \$191,638 earned as incentive compensation in 2010 and matching contributions to his 401(k) account of \$5,000 in 2012 and \$4,900 in 2011 and 2010.

## 2012 Grants of Plan-Based Awards

The following table contains information concerning grants of plan-based awards made in 2012 to our named executive officers.

			All Other		
	<b>C</b> and	All Other Stock Awards: Number of	Option Awards: Number of Securities	Exercise or Base Price of Option	Grant Date Fair Value of Stock
	Grant	Shares of	Underlying	Awards	and Option
Name	<b>Date (1)</b>	Stock or Units (#)	Options (#)	(\$/Sh)	Awards
Judson Bergman	2/29/2012	5,000	7,951	\$ 12.45	\$ 101,312
William Crager	2/29/2012	3,119	13,594	12.45	105,618
Peter D'Arrigo	2/29/2012	2,339	10,195	12.45	79,208
a			40.40.		
Scott Grinis	2/29/2012	2,339	10,195	12.45	79,208

 All restricted stock awards and stock option grants were approved by the Compensation Committee and the Board of Directors on February 8, 2012. The grants were made on February 29, 2012.

(2) The fair value of stock options granted was determined using the Black-Scholes model as of the grant date. The model assumes: (i) the stock option would be exercised 6 years after grant date, (ii) expected stock price volatility of 39.76%, (iii) a risk-free yield equal to 1.13% US Treasury STRIPS, and (iv) our dividend yield (0%) would remain constant from grant date to exercise date.

## Narrative to 2012 Summary Compensation Table and 2012 Grants of Plan-Based Awards Table

See "—Compensation Discussion and Analysis" above for a complete description of compensation plans pursuant to which the amounts listed under the 2012 Summary Compensation Table and 2012 Grants of Plan-Based Awards Table were paid or awarded, and the criteria on which such payments were based. The Compensation Discussion and Analysis also describes certain grants of stock options to our named executive officers.

# 2012 Outstanding Equity Awards at Fiscal Year-End

The following table lists all outstanding equity awards held by our named executive officers ("NEO") as of December 31, 2012:

		Option Av	vards (1)				Stock A	Awards (2)	
Name Judson Bergman	Number of Securities Underlying Unexercised Options (#) Exercisable 170,000 200,000 36,000 15,000 185,000	Number of Securities Underlying Unexercised Options (#) Unexercisable	Equity Incentive Plan Awards: Number of Securities Underlying Unexercised Unearned Options (#)	Option Exercise Price (S) 5 5.00 7.50 7.50 7.50 7.50 7.50 7.50 7.50	Option Expiration Date 11/14/2015 11/14/2015 4/26/2017 4/30/2018 5/15/2019 7/28/2020	Number of Shares or Units of Stock that have not Vested (#)	Market Value of Shares or Units of Stock That Have Not Yet Vested (\$)	Equity Incentive Plan Awards: Number of Unearned Shares, Units or Other Rights That Have Not Vested (#)	Equity Incentive Plan Awards: Market or Payout Value of Unearned Shares, Units or Other Rights That Have Not Vested (\$)
	3,333	6,667 7,951		12.55 12.45	2/28/2020 2/28/2021 2/28/2022	5,000	69,750		
William Crager		82,000 16,667 13,594		5.00 7.50 1.08 7.50 0.11 7.50 7.15 9.00 12.55 12.45	11/14/2015 11/14/2015 4/26/2017 4/26/2017 4/26/2017 4/30/2018 5/15/2019 7/28/2020 2/28/2021 2/28/2022	3,119	43,510		
Peter D'Arrigo	220,000 6,000 43,000 3,333	43,000 6,667 10,195		7.50 7.15 9.00 12.55 12.45	6/16/2018 5/15/2019 7/28/2020 2/28/2021 2/28/2022	2,339	32,629		
Scott Grinis	12,000 9,000 6,000 28,000 3,333	28,000 6,667 10,195		7.50 7.50 7.15 9.00 12.55 12.45	4/26/2017 4/30/2018 5/15/2019 7/28/2020 2/28/2021 2/28/2022	2,339	32,629		
Brandon Thomas	32,500 32,500 12,000 9,000 6,000 31,000 3,333	31,000 6,667 7,647		5.00 7.50 7.50 7.50 7.15 9.00 12.55 12.45	11/14/2015 11/14/2015 4/26/2017 4/30/2018 5/15/2019 7/28/2020 2/28/2021 2/28/2022	3,189	44,487		

(1) Except as otherwise noted, vesting for stock option grants that expire on November 14, 2015, April 26, 2017 and June 16, 2018 occurs annually on the anniversary of the grant date where one fourth vests on the grant date, and one fourth vests on the anniversary of the grant date for the next three years. Vesting for stock option grants that expire on April 30, 2018, May 15, 2019, February 28, 2021 and February 28, 2022 occurs annually on the anniversary of the grant date where one third vests on the first anniversary, one third vests on the second anniversary and the remainder vests on the third anniversary. The 20,000 share grant to William Crager, which expires on April 26, 2017, vested immediately. Vesting for stock option grants that expire on July 28, 2020 occurs annually on the anniversary of the grant date where one fourth vests each of the first four anniversaries.

## 2012 Option Exercises and Stock Vested

There were no options exercised or restricted stock that vested with respect to any of our NEOs for the calendar year ending December 31, 2012.

### **Nonqualified Deferred Compensation**

We do not currently have a nonqualified deferred compensation plan. However, we may consider implementing such a plan in the future.

#### **Potential Payments Upon Termination of Change of Control**

None of our named executive officers has an employment agreement or change of control agreement.

*Equity Incentive Plans.* We currently maintain equity-based incentive plans –the 2004 Stock Incentive Plan, the 2010 Long-Term Incentive Plan and the Envestnet, Inc. Management Incentive Plan for Envestnet | Tamarac Management Employees. No new awards are being made under the 2004 Stock Incentive Plan. We established the 2010 Long-Term Incentive Plan to (i) attract and retain key employees and other persons providing services to us and our related companies; (ii) motivate plan participants by means of appropriate incentive to achieve long-range goals; (iii) provide incentive compensation opportunities that are competitive with those of other similar corporations; and (iv) further align plan participants' interests with those of our stockholders. Under the 2010 Long-Term Incentive Plan, we may issue stock options, stock appreciation rights, restricted stock, restricted stock units and other full value awards, as well as cash incentive awards.

*Change in Control.* In the event that (a) a participant's employment or service, as applicable, is terminated by us, our successor or one of our related companies that is the participant's employer for reasons other than cause (as defined in the 2010 Long-Term Incentive Plan) within 24 months following a change in control (as defined in the 2010 Long-Term Incentive Plan) or (b) the 2010 Long-Term Incentive Plan is terminated by us or our successor following a change in control without provision for the continuation of outstanding awards under the 2010 Long-Term Incentive Plan, all stock options and related awards which have not otherwise expired will become immediately exercisable and all other awards will become fully vested.

Under our equity compensation plans, outstanding and unvested stock options will become fully vested and exercisable, and outstanding and unvested restricted stock will become fully vested and be distributed upon a participant's death, disability, retirement, or involuntary termination. In addition, the awards will vest upon a change-in-control irrespective of a termination of employment. In the event of a change-in-control where the company ceases to have publicly traded equity securities, after the consummation of the change-in-control, if no replacement awards are issued in lieu of outstanding awards under the equity plans, then the plans and all outstanding awards granted under the plans will terminate, and the company (or successor) will pay the participants an amount for their outstanding awards determined using the change in control price. These provisions apply to all employees who participate in the company's equity plans. The outstanding equity awards held by the NEO as of December 31, 2012 are described above under "2012 Outstanding Equity Awards at Fiscal Year-End."

We have estimated the amount of incremental compensation for each of Messrs. Bergman, Crager, D'Arrigo, Grinis and Thomas due to accelerated vesting of outstanding restricted stock awards upon termination of the officer's employment in the event of the officer's death, disability, retirement or involuntary termination, or upon a change in control irrespective of a termination of employment, as follows: Mr. Bergman, \$951,860; Mr. Crager, \$449,625; Mr. D'Arrigo \$237,476; Mr. Grinis \$163,226; and Mr. Thomas, \$174,254. These amounts assume that the termination of employment or change in control was effective as of December 31, 2012 and that the price of Common Stock on which the calculations are made was the closing price of \$20.23 on that date. We have estimated the value for the acceleration of stock options as of December 31, 2012 for each NEO as follows: Mr. Bergman, \$69,750; Mr. Crager, \$43,510; Mr. D'Arrigo \$32,629; Mr. Grinis \$32,629; and Mr. Thomas, \$44,487. The amounts shown above are estimates of the incremental compensation these officer's upon such terminations or a change in control. The actual amounts to be received can only be determined at the time of the officer's termination of employment or at the time of a change in control.

*Transferability.* Awards under the 2010 Long-Term Incentive Plan are not transferable except as designated by the participant by will or by laws of descent and distribution or, to the extent provided by the Compensation Committee, pursuant to a qualified domestic relations order or to or for the benefit of the participant's family (including, without limitation, to a trust or partnership for the benefit of a participant's family).

*Withholding.* All awards and other payments under the 2010 Long-Term Incentive Plan are subject to withholding of all applicable taxes. With the consent of the Compensation Committee, withholding obligations may be satisfied with previously-owned shares of common stock or shares of common stock to which the participant is otherwise entitled under the 2010 Long-Term Incentive Plan. Such shares may only be used to satisfy minimum withholding requirements.

*Amendment and Termination.* The Board may, at any time, amend or terminate the 2010 Long-Term Incentive Plan, and the Board or the Compensation Committee may amend any award, provided that no amendment or termination may adversely affect the rights of any participant without the participant's written consent. Adjustments to the 2010 Long-Term Incentive Plan and awards on account of business transactions (as described above) are not subject to the foregoing prohibition. The provisions of the 2010 Long-Term Incentive Plan that prohibit repricing of stock options and stock appreciation rights cannot be amended unless the amendment is approved by our stockholders. The 2010 Long-Term Incentive Plan also permits the Board to amend the 2010 Long-Term Incentive Plan and any awards that are subject to Section 409A of the Internal Revenue Code (relating to nonqualified deferred compensation) as it deems necessary to conform to Section 409A.

## **Director Compensation**

Until July 30, 2012 each non-employee director received an annual retainer of \$30,000 with an additional meeting attendance stipend of \$2,000 for each board meeting attended in person, including all coinciding committee meetings. The chairperson of our Audit Committee received an additional annual retainer of \$15,000. The chairpersons of our other committees received an additional annual retainer of \$10,000. The lead independent director received an additional annual retainer of \$15,000. All non-chairperson committee members received an additional annual retainer of \$5,000 for each committee on which they serve. Directors received one-half of such amounts in cash and could receive one-half in either restricted stock or in options to acquire shares of our common stock at the discretion of the company.

Effective July 31, 2012, the non-employee director compensation policy was amended so that each non-employee director receives an annual retainer of \$60,000 with an additional meeting attendance stipend of \$5,000 for each board meeting attended in person, including all coinciding committee meetings. The chairperson of our Audit Committee receives an additional annual retainer of \$15,000. The chairpersons of our other committees receive an additional annual retainer of \$10,000. The lead independent director receives an additional annual retainer of \$15,000. All non-chairperson committee members receive an additional annual retainer of \$15,000 for each committee on which they serve. Directors receive 33% of such amounts in cash and may receive 67% in either restricted stock or in options to acquire shares of our common stock at the discretion of the company.

We also reimburse all of our directors for their reasonable expenses incurred in attending meetings of our Board of Directors or committees. Equity grants to our non-employee directors vest monthly over a four-year period, except that the shares that would otherwise vest over the first 12 months do not vest until the first anniversary of the grant. All equity grants to our non-employee directors are made pursuant to our 2010 Long-Term Incentive Plan. See " —Compensation Discussion and Analysis—2010 Long-Term Incentive Plan".

The following table sets forth our 2012 independent director compensation:

Director Compensation

	Fees		
	Earned or		
	Paid in	Option	
	Cash	Awards	Total
Name	(\$)	(\$)(*)	(\$)
Ross Chapin	30,150	26,035	56,185
Gates Hawn	33,225	29,538	62,763
James Johnson	35,300	32,044	67,344
Paul Koontz	31,350	26,533	57,883
Charles Roame	31,150	11,515	42,665
Yves Sisteron	34,725	32,044	66,769

\* Option awards were granted on February 28, 2012, in connection with 2011 service.

As of December 31, 2012, the following option awards were outstanding for each director in connection with option awards granted on February 29, 2012, with an exercise price of \$12.45 for 2011 service:

Ross Chapin	5,277 options
Gates Hawn	5,987 options
James Johnson	6,495 options
Paul Koontz	5,378 options
Charles Roame	2,344 options
Yves Sisteron	6,495 options

### **Compensation Committee Report**

The Compensation Committee has reviewed and discussed the above Compensation Discussion and Analysis with management and, based on such review and discussion, the Compensation Committee recommended to the Board of Directors that the Compensation Discussion and Analysis be included in the company's Annual Report on Form 10-K for the year ended December 31, 2012.

The foregoing report has been approved by the following members of the Compensation Committee.

Yves Sisteron, Chairman Ross Chapin Paul Koontz Charles Roame

The section captioned "Compensation Committee Report on Compensation Discussion and Analysis" is hereby "furnished" and not "filed" with this annual report on Form 10-K.

## Item 12. Security Ownership of Certain Beneficial Owners and Management and Related Stockholder Matters

The following table sets forth information, as of May 31, 2013, regarding the beneficial ownership of our common stock by our directors and named executive officers and by our directors and executive officers as a group. Unless otherwise indicated, the named individual has sole voting and investment power over the common stock under the column "Common Stock Beneficially Owned."

		Options Exercisable	Unvested Restricted	Total	Beneficial
	Shares	within 60	Common	Beneficial	Ownership
	Held	Days (1)	Stock	Ownership	Percentages
Jud Bergman (2)	644,746	822,317	6,969	1,534,032	4.6%
Ross Chapin	78,816	3,475		78,351	*
Gates Hawn	128,000	3,870		131,870	*
Jim Johnson (3)	1,117,381	4,193		1,121,574	3.5%
Paul Koontz (4)	1,450,457	3,511		1,453,968	4.5%
Charles Roame	2,075	827		2,902	*
Yves Sisteron (5)	2,269,313	4,108		2,273,421	7.0%
Bill Crager (6)	149,935	433,004	7,720	549,659	1.8%
Pete D'Arrigo	491	300,565	4,679	305,735	*
Scott Grinis	245,138	79,065	4,679	328,882	1.0%
Brandon Thomas (7)	241,184	147,716	5,246	394,146	1.2%
All Directors and Executive Officers as a Group	6,324,038	1,991,920	33,510	8,345,251	24.2%

\* Denotes beneficial ownership of less than one percent.

(1) Includes options vested and exercisable within 60 days of May 31, 2013.

(2) Includes 200,000 shares held as security in a margin account

(3) Includes 1,774 shares held by Mr. Johnson and 1,115,607 shares held by various funds managed by Apex Venture Partners. As a result of positions held with these entities, Mr. Johnson may be deemed to share voting and dispositive power over the securities held by these funds. Mr. Johnson disclaims beneficial ownership of the reported securities except to the extent of his pecuniary interests therein.

(4) Includes 9,610 shares held by Paul Koontz 2010 Annuity Trust, 9,610 shares held by Andrea Koontz 2010 Annuity Trust and 1,431,237 shares held by entities affiliated with Foundation Capital.

(5) Includes 15,000 shares held by Mr. Sisteron individually and 2,254,313 held by entities affiliated with GRP Partners. See footnote (1) under the following table.

(6) Includes 100 shares indirectly held by Mr. Crager's wife.

(7) Includes 13,938 shares held by a trust in which Mr. Thomas is the trustee.

The following table shows all persons we know to be direct or indirect owners of more than 5% of our common stock as of the close of business on May 31, 2013, unless otherwise indicated. Our information is based on reports filed with the SEC by each of the firms listed in the table below. You may obtain these reports from the SEC.

Name and Address of Beneficial Owner	Number of Shares Beneficially Owned	Percent of Class
Entities associated with GRP Partners (1) 2121 Avenue of the Stars Suite 1630 Los Angeles, CA 90067	6,801,017	21.0%
Janus Capital Management LLC (2) 151 Detroit Street Denver, CO 80206	1,997,187	6.2%
William Blair & Company LLC (3) 222 W. Adams Street Chicago, IL 60606	1,777,244	5.5%

- (1) Based on a Schedule 13G filed by GRP II Investors, L.P. ("GRP II Investors"), GRP II Partners, L.P. ("GRP II Partners"), GRP Management Services Corp. ("GRPMS"), GRPVC, L.P. ("GRPVC"), and AOS Partners, L.P. ("AOS") on February 15, 2012, reporting the amount of securities beneficially owned as of December 31, 2011. Hique, Inc. is the general partner of AOS. GRPVC is the general partner of GRP II Partners. GRPMS is the general partner of each of GRPVC and GRP II Investors. Steven Dietz, Brian McLoughlin and Mark Suster are members of the investment committee of AOS. Yves Sisteron, one of our directors, together with Steven Dietz and Brian McLoughlin, is an officer of GRPMS. Mr. Sisteron, together with Steven Dietz, Brian McLoughlin and Mark Suster, is a member of the investment committee of GRP II Partners. Pursuant to contractual arrangements, GRP II Investors has granted GRPMS the authority to vote and dispose of the shares held by it in the same manner as the investment committee votes or disposes of the shares held by GRP II Partners. While Mr. Sisteron may be deemed to possess indirect beneficial ownership of the shares owned by GRPVC, GRP II Partners and GRP II Investors, he does not have sole voting or investment power with respect to such shares and, as a result, disclaims beneficial ownership of any and all such shares.
- (2) Based on a Schedule 13F filed by Janus Capital Management LLC on May 15, 2013, reporting the amount of securities beneficially owned as of March 31, 2013. Janus has sole voting power over 1,770,210 shares and shared dispositive power over 226,977 shares.
- (3) Based on a Schedule 13F filed by William Blair & Company, LLC on May 8, 2013, reporting the amount of securities beneficially owned as of March 31, 2013. William Blair has sole voting power over 1,567,144 shares and shared dispositive power over 210,100 shares.

For a description of securities authorized under our equity compensation plans, see note 13 to the notes to consolidated financial statements in Part II, Item 8.

### Item 13. Certain Relationships and Related Transactions, and Director Independence

#### Registration Rights

On March 22, 2004, we entered into a registration rights agreement with certain holders of our common stock, or the registration rights agreement, pursuant to which these holders of our common stock are entitled to demand registration rights, Form S-3 registration rights and piggyback registration rights with respect to the registration of their shares of our common stock under the Securities Act of 1933, as amended, or the Securities Act. We refer to shares of our common stock that are subject to the registration rights agreement as "registrable securities."

In connection with our initial public offering, The EnvestNet Group, Inc., Envestnet's 41% shareholder prior to the IPO (the "Envestnet Shareholder"), merged with and into Envestnet, with Envestnet being the surviving entity. Upon consummation of the merger of the Envestnet Shareholder with and into Envestnet, certain stockholders of the Envestnet Shareholder are entitled to become party to the registration rights agreement and to receive each of the registration rights described below.

**Demand Registration Rights.** The holders of registrable securities have rights, at their request, to have their shares registered for resale under the Securities Act. Holders of at least 50% of registrable securities may demand the registration of their shares on up to two occasions within any 12-month period if the gross proceeds from the registration of their shares would exceed \$15,000,000.

**Registration on Form S-3.** In addition to the demand registration rights discussed above, holders of at least 20% of the registrable securities may require that we register their shares of our common stock for public resale on Form S-3 or similar short-form registration statement if the gross proceeds from the registration of their shares of our common stock would exceed \$5,000,000 and our company is eligible to use Form S-3.

*Piggyback Registration Rights.* The holders of approximately 13.3 million shares of registrable securities have rights to have their shares of our common stock registered for resale under the Securities Act if we register any of our securities, either for our own account or for the account of other stockholders, subject to the right of the underwriters involved in any such transaction to limit the number of shares of our common stock included in an underwritten offering.

The following Related Parties are currently party to the registration rights agreement: AOS Partners, L.P., GRP II Investors, L.P., GRP II Partners, L.P., GRPVC, L.P., Apex Investment Fund IV, L.P., Apex Investment Fund V, L.P., Apex Strategic Partners IV, LLC, Foundation Capital III Principals, LLC, Foundation Capital III, L.P., Foundation Capital Leadership Principals Fund, LLC, Foundation Capital Leadership Fund, L.P., Judson Bergman (our Chairman and Chief Executive Officer and one of our directors), William Crager (our President), Scott Grinis (our Chief Technology Officer), Brandon Thomas (our Chief Investment Officer), and James Johnson, one of our directors. Holders of our registrable securities are entitled to the registration rights described above. Collectively, these Related Parties hold approximately 10.8 million shares covered by the registration rights agreement as of May 31, 2013.

### Indemnification of Directors and Executive Officers

On July 28, 2010, and upon the election of Mr. Roame as a director in August 2011, we entered into agreements to indemnify our directors and certain of our officers in addition to the right to indemnification provided to such persons in our certificate of incorporation and by-laws. These agreements will, among other things, require us to indemnify these individuals to the fullest extent permitted under Delaware law, including for certain expenses (including attorneys' fees), judgments, fines and settlement amounts incurred by such person in any action or proceeding, including any action by or in our right, on account of services by any such person as a director or officer of our company or as a director or officer of any of our subsidiaries, or as a director or officer of any other company or enterprise if any such person serves in such capacity at our request. We also intend to enter into indemnification agreements with our future directors and executive officers.

### Director Independence

In March 2013, our Board determined that the following directors are independent under the listing standards of the NYSE: Ross Chapin, Gates Hawn, James Johnson, Paul Koontz, Charles Roame and Yves Sisteron. These independent directors constitute substantially more than a majority of the Board of Directors. In making its determination of independence, the Board applied the categorical standards for director independence set forth in the NYSE's rules and determined that no other material relationships existed between us and these directors. The Board also considered the other directorships held by the independent directors and determined that none of these directorships constituted a material relationship with us.

## Item 14. Principal Accountant Fees and Services

On February 14, 2013, the Audit Committee engaged KPMG as our independent accountant for our fiscal year ended December 31, 2012 and to reaudit our financial statements for the year ended December 31, 2011. On February 14, 2013, we disclosed that the Audit Committee dismissed our previous auditor, McGladrey, from its engagement as our independent auditors effective February 13, 2013. McGladrey's dismissal followed a determination that certain services (the "Services") that McGladrey had provided in prior years to us may be inconsistent with the SEC's rules on auditor independence. On April 2, 2013, the Audit Committee engaged KPMG to re-audit our financial statements for the year ended December 31, 2010.

During the years ended December 31, 2011 and 2010 and through February 14, 2013, we had not consulted with KPMG regarding either:

(i) the application of accounting principles to any specified transaction, either completed or proposed, or the type of audit opinion that might be rendered on our financial statements, and neither a written report was provided to us nor oral advice was provided that KPMG concluded was an important factor considered by us in reaching a decision as to the accounting, auditing or financial reporting issue; or

(ii) any matter that was either the subject of a disagreement (as defined in paragraph (a)(1)(iv) of Item 304 of the SEC's Regulation S-K and the related instructions thereto) or a reportable event (as described in paragraph (a)(1)(v) of Item 304 of the SEC's Regulation S-K).

Typically, the appointment of independent auditors is approved annually by the Audit Committee and ratified by our stockholders. The Audit Committee reviews both the audit scope and estimated fees for professional services for the coming year. The Audit Committee has appointed, subject to ratification by our stockholders, KPMG as our independent auditor for the year ending December 31, 2013.

### **Independent Auditor Fee Information**

The following table sets forth the approximate aggregate fees for professional services rendered by KPMG for the audit of our consolidated financial statements for 2012 and for the re-audit of 2011, as well as other fees paid to KPMG in 2011 and 2012 before they were appointed as our auditors.

	2011	2012
Audit fees (1)	\$485,000	\$ 590,000
Audit-related fees	-	-
Tax fees (2)	-	4,000
All other fees (3)	36,000	73,000
Total	\$521,000	\$667,000

(1) Audit fees, related to the years ended December 31, 2011 and December 31, 2012 in connection with:

- the audits of our consolidated financial statements for 2011 and 2012;
- the audit of our internal control over financial reporting as of December 31, 2012;
- reviews of quarterly consolidated financial statements; and
- the statutory audit of one of our subsidiaries for 2012.
- (2) Tax fees for assistance with transfer pricing issues related to foreign income taxes.
- (3) All other fees include a review of the compliance program of the Company, including a review of our compliance policies, procedures and related controls to comply with Rule 206(4)-7 under the Investment Advisers Act of 1940.

### Pre-Approval Policy of Audit and Non-Audit Services

The Audit Committee pre-approved the audit services associated with the fees described above. The tax fees and all other fees described above were incurred by the Company prior to the engagement of KPMG as the Company's independent auditor.

The Audit Committee has adopted policies and procedures for the pre-approval of all audit and permissible non-audit services provided by our independent auditor. The Audit Committee provides a general pre-approval of certain audit and non-audit services on an annual basis. The types of services that may be covered by a general pre-approval include other audit services, audit-related services and permissible non-audit services. If a type of service is not covered by the Audit Committee's general pre-approval, the Audit Committee must review the service on a specific case by case basis and pre-approve it if such service is to be provided by the independent auditor. Annual audit services engagement terms and fees require specific pre-approval of the Audit Committee. Any proposed services exceeding the pre-approved fees also require specific pre-approval by the Audit Committee. For both types of pre-approval, the Audit Committee will consider whether such services are consistent with the SEC's rules on auditor independence. The Audit Committee may delegate either type of pre-approval authority to one or more of its members.

1	1	ç
	-	-

Part IV

# Item 15. Exhibits and Financial Statement Schedules

	Page Number in
	Form 10-K
(a)(1) Consolidated Financial Statements	
Report of Independent Registered Public Accounting Firm	61
Consolidated Balance Sheets as of December 31, 2012 and 2011	62
Consolidated Statements of Operations for each of the years ended December 31, 2012, 2011, and 2010	63
Consolidated Statements of Stockholders' Equity for each of the years ended December 31, 2012, 2011 and 2010	64
Consolidated Statements of Cash Flows for each of the years ended December 31, 2012, 2011 and 2010	65
Notes to Consolidated Financial Statements	66
(a)(2) Evaluation and Qualifying Accounts	

Financial statements and schedules are omitted for the reason that they are not applicable, are not required, or the information is included in the financial statements or the related footnotes.

## INDEX TO EXHIBITS

 Description

 3.1
 Amended and Restated Certificate of Incorporation of Envestnet, Inc. (filed as Exhibit 3.1 to the Company's Registration Statement on Form S-1, as amended (File No. 333-165717), filed with the SEC on July 1, 2010 and incorporated by reference herein).

- 3.2 Amended and Restated Bylaws of Envestnet, Inc. (filed as Exhibit 3.2 to the Company's Registration Statement on Form S-1, as amended (File No. 333-165717), filed with the SEC on July 1, 2010 and incorporated by reference herein).
- 4.1 Registration Rights Agreement dated as of March 22, 2004 (filed as Exhibit 4.2 to the Company's Registration Statement on Form S-1, as amended (File No. 333-165717), filed with the SEC on March 26, 2010 and incorporated by reference herein).
- 4.2 First Amendment to Registration Rights Agreement dated as of August 30, 2004 (filed as Exhibit 4.3 to the Company's Registration Statement on Form S-1, as amended (File No. 333-165717), filed with the SEC on March 26, 2010 and incorporated by reference herein).
- 4.3 Second Amendment to Registration Rights Agreement effective as of March 24, 2005 (filed as Exhibit 4.4 to the Company's Registration Statement on Form S-1, as amended (File No. 333-165717), filed with the SEC on March 26, 2010 and incorporated by reference herein).
- 4.4 Joinder Agreements to Registration Rights Agreement (filed as Exhibit 4.5 to the Company's Registration Statement on Form S-1, as amended (File No. 333-165717), filed with the SEC on March 26, 2010 and incorporated by reference herein).
- 10.1 Technology and Services Agreement dated as of March 31, 2008, between Registrant and FMR LLC (filed as Exhibit 10.1 to the Company's Registration Statement on Form S-1, as amended (File No. 333-165717), filed with the SEC on May 6, 2010 and incorporated by reference herein).
- 10.2 First Amendment to Technology and Services Agreement dated June 26, 2008 (filed as Exhibit 10.2 to the Company's Registration Statement on Form S-1, as amended (File No. 333-165717),
- filed with the SEC on May 6, 2010 and incorporated by reference herein). 10.3 Second Amendment to Technology and Services Agreement dated May 5, 2009 (filed as Exhibit 10.3 to the Company's Registration Statement on Form S-1, as amended (File No. 333-165717), filed with the SEC on May 6, 2010 and incorporated by reference herein).
- 10.4 Third Amendment to Technology and Services Agreement dated November 16, 2009 (filed as Exhibit 10.4 to the Company's Registration Statement on Form S-1, as amended (File No. 333-165717), filed with the SEC on May 6, 2010 and incorporated by reference herein).
- 10.5 Services Agreement dated December 28, 2005 between Registrant and Fidelity Brokerage Services LLC (filed as Exhibit 10.5 to the Company's Registration Statement on Form S-1, as amended (File No. 333-165717), filed with the SEC on May 6, 2010 and incorporated by reference herein).
- 10.6 Services Agreement effective March 24, 2005 between Registrant and National Financial Services LLC (filed as Exhibit 10.6 to the Company's Registration Statement on Form S-1, as amended (File No. 333-165717), filed with the SEC on May 6, 2010 and incorporated by reference herein).
- 10.7 Services Agreement Amendment dated effective March 2008 (filed as Exhibit 10.7 to the Company's Registration Statement on Form S-1, as amended (File No. 333-165717), filed with the SEC on May 6, 2010 and incorporated by reference herein).
- 10.10 2010 Long-Term Incentive Plan (filed as Exhibit 10.10 to the Company's Registration Statement on Form S-1, as amended (File No. 333-165717), filed with the SEC on July 1, 2010 and incorporated by reference herein).\*
- 10.11 2004 Stock Incentive Plan (filed as Exhibit 10.11 to the Company's Registration Statement on Form S-1, as amended (File No. 333-165717), filed with the SEC on July 1, 2010 and incorporated by reference herein).\*
- 10.12 Form of Equity Award, filed as Exhibit 10.12 to the Company's 2010 Form 10-K, (filed with the SEC on March 18, 2011 and incorporated by reference herein).\*
- 10.13 Fourth Amendment to Technology Services Agreement, dated as of December 31, 2011, between Envestnet, Inc. and FMR LLC (filed as Exhibit 10.1 to the Company's Form 8-K filed with the SEC on January 6, 2012 and incorporated by reference herein.).
- 10.14 Amendment to Services Agreement effective December 31, 2011, between Envestnet Asset Management, Inc. and Fidelity (filed as Exhibit 10.2 to the Company's Form 8-K filed with the SEC on January 6, 2012 and incorporated by reference herein.).
- 10.15 Third Amendment to Services Agreement effective December 31, 2011, between Envestnet Asset Management, Inc. and National Financial Services LLC. (filed as Exhibit 10.3 to the Company's Form 8-K filed with the SEC on January 6, 2012 and incorporated by reference herein.).
- 10.16 Stock Purchase Agreement by and among The Sellers of Prima Capital Holding, Inc. Named Herein and Envestnet, Inc. dated as of February 9, 2012, (filed as Exhibit 10.16 to the Company's 10-K filed with the SEC on March 9, 2012 and incorporated by reference herein.).
- 10.17 Merger Agreement by and among Tamarac Inc., Envestnet, Inc. and Titan Merger Corp and KLJ Consulting, LLC (as the Shareholder's Representative) dated as of February 16, 2012 (filed as Exhibit 10.17 to the Company's 10-K filed with the SEC on March 9, 2012 and incorporated by reference herein.)
- 10.18 Envestnet, Inc. Management Incentive Plan for Envestnet | Tamarac Management Employees (filed as Exhibit 4.3 to the Company's Registration Statement on Form S-8 (filed with the SEC on May 1, 2012 and incorporated by reference herein.)\*
- 10.19 First Amendment to Envestnet, Inc. Management Incentive Plan for Envestnet | Tamarac Management Employees (filed as Exhibit 10.1 to the Company's Form 8-K filed with the SEC on April 17, 2013 and incorporated by reference herein.)\*



Exhibit <u>No.</u>

F

Exhibit	
No.	Description
10.20	Second Amendment to Envestnet, Inc. Management Incentive Plan for Envestnet   Tamarac Management Employees (filed as Exhibit 10.1
	to the Company's Form 8-K filed with the SEC on May 13, 2013 and incorporated by reference herein.)*
10.21	Purchase and Sale Agreement between Prudential Investments LLC and Envestnet, Inc. dated as of April 11, 2013, filed herewith. **
16.1	Letter re Change in Certifying Accountant dated February 27, 2013 (filed as Exhibit 16.1 to the Company's 8-K filed with the SEC on
	March 1, 2013 and incorporated by reference herein.)
16.2	Letter re Change in Certifying Accountant dated March 4, 2013 (filed as Exhibit 16.1 to the Company's 8-K filed with the SEC on March
	5, 2013 and incorporated by reference herein.)
16.2	Letter re Change in Certifying Accountant dated April 4, 2013 (filed as Exhibit 16.1 to the Company's 8-K filed with the SEC on April 5,
	2013 and incorporated by reference herein.)
21.1	Subsidiaries of the Company, filed herewith.
23.1	Consent of Independent Registered Public Accounting Firm, filed herewith.
31.1	Certification of Chief Executive Officer pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
31.2	Certification of Chief Financial Officer pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
32.1(1)	Certification of Chief Executive Officer Pursuant to 18 U.S.C. 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of
	2002.
32.2(1)	Certification of Chief Financial Officer Pursuant to 18 U.S.C. 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of
	2002.
101.INS	XBRL Instance Document ***
101.SCH	XBRL Taxonomy Extension Schema Document ***
101.CAL	XBRL Taxonomy Extension Calculation Linkbase Document ***
101.LAB	XBRL Taxonomy Extension Label Linkbase Document ***
101.PRE	XBRL Taxonomy Extension Presentation Linkbase Document ***
101.DEF	XBRL Taxonomy Extension Definition Linkbase Document ***

The material contained in Exhibit 32.1 and 32.2 is not deemed "filed" with the SEC and is not to be incorporated by reference into any filing of the Company under the Securities Act of 1933 or the Securities (1) Exchange Act of 1934, whether made before or after the date hereof and irrespective of any general incorporation language contained in such filing, except to the extent that the registrant specifically incorporates it by reference

Management contract or compensation plan.

\*\* Application has been made with the Securities and Exchange Commission to seek confidential treatment of certain provisions omitted material for which confidential treatment has been requested has been filed separately with the Securities and Exchange Commission.

Attached as Exhibit 101 to this Annual Report on Form 10-K are the following materials, formatted in XBRL (Extensible Business Reporting Language): (i) the Consolidated Balance Sheets as of December 31, 2012 and 2011; (ii) the Consolidated Statements of Operations for the year ended December 31, 2012, 2011 and 2010; (iii) the Consolidated Statements of Stockholders' Equity for the years ended December 31, \*\*\* 2012, 2011 and 2010; (iv) the Consolidated Statements of Cash Flows for the years ended December 31, 2012, 2011 and 2010; (v) Notes to Consolidated Financial Statements.

The XBRL information in this Annual Report on Form 10-K, Exhibit 101, is not deemed "filed" for purposed of Section 11 or 12 of the Securities Act of 1933, as amended (the Securities Act), or Section 18 of the Securities Act of 1934, as amended (the Exchange Act), or otherwise subject to the liabilities of those sections, and is not part of any registration statement to which it may relate, and is not incorporated by reference into any registration statement or other document filed under the Securities Act of the Exchange Act, except as is expressly set forth by specific reference in such filing or document.

# SIGNATURES

Pursuant to the requirements of Section 13 or 15(d) the Securities Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

# ENVESTNET, INC.

Date:	June	14,	2013
-------	------	-----	------

Date: June 14, 2013

/s/ JUDSON BERGMAN

Judson Bergman Chairman and Chief Executive Officer (Principal Executive Officer)

/s/ PETER H. D'ARRIGO

Peter H. D'Arrigo Chief Financial Officer (Principal Financial Officer)

Pursuant to the requirements of the Securities Act of 1934, this report has been signed below by the following persons on behalf of the registrant and in the capacities on June 14, 2013.

Name	Position
/S/ JUDSON BERGMAN	Chairman and Chief Executive Officer; Director (Principal Executive Officer)
Judson Bergman	_
/S/ PETER H. D'ARRIGO	Chief Financial Officer (Principal Financial Officer)
Peter H. D'Arrigo	
/S/ DALE SEIER	Senior Vice President, Finance (Principal Accounting Officer)
Dale Seier	_
/S/ ROSS CHAPIN	Director
Ross Chapin	_
/S/ GATES HAWN	Director
Gates Hawn	_
/S/ JAMES JOHNSON	Director
James Johnson	_
/S/ PAUL KOONTZ	Director
Paul Koontz	
/S/ CHARLES ROAME	Director
Charles Roame	
/S/ YVES SISTERON	Director
Yves Sisteron	
	123

# Exhibit 10.21

EXECUTION VERSION

PURCHASE AND SALE AGREEMENT BETWEEN PRUDENTIAL INVESTMENTS LLC (the "<u>Seller</u>") AND ENVESTNET, INC. (the "<u>Buyer</u>")

April 11, 2013

# TABLE OF CONTENTS

ARTICLE I ASSET PURCHASE	1
<ul> <li>1.1 Purchase and Sale of Assets; Assumption of Liabilities</li> <li>1.2 Purchase Price and Related Matters</li> <li>1.3 Allocation of Purchase Price</li> <li>1.4 The Closing</li> <li>1.5 Post-Closing Purchase Price Adjustment</li> <li>1.6 Consents to Assignment</li> <li>1.7 Further Assurances</li> <li>1.8 Withholding</li> </ul>	1 2 5 5 6 8 8 9
ARTICLE II REPRESENTATIONS AND WARRANTIES OF THE SELLER	9
<ul> <li>2.1 Organization, Qualification and Corporate Power</li> <li>2.2 Authority</li> <li>2.3 Noncontravention</li> <li>2.4 Financial Information</li> <li>2.5 Tax Matters</li> <li>2.6 Acquired Assets</li> <li>2.7 Leased Real Property</li> <li>2.8 Intellectual Property</li> <li>2.9 Contracts</li> <li>2.10 Sufficiency of Assets</li> <li>2.11 Litigation</li> <li>2.12 Employment Matters</li> <li>2.13 Employee Benefits</li> <li>2.14 Legal Compliance</li> <li>2.15 Permits</li> <li>2.16 Business Relationships with Affiliates</li> <li>2.17 Registration</li> <li>2.18 Participating Sponsor Revenue</li> <li>2.19 Absence of Certain Changes</li> <li>2.20 Financial Capability</li> </ul>	9 9 10 10 11 11 12 12 12 14 15 15 16 16 16 16 17 17 17 17 17 18 18 18
ARTICLE III REPRESENTATIONS AND WARRANTIES OF THE BUYER	19
<ul> <li>3.1 Organization</li> <li>3.2 Authority</li> <li>3.3 Noncontravention</li> <li>3.4 Litigation</li> <li>3.5 Financial Capability</li> <li>3.6 Registration</li> <li>3.7 Due Diligence by the Buyer</li> <li>3.8 Disclaimer of Warranties</li> </ul>	19 19 19 20 20 20 20 20 20 20 20 21

ARTICLE IV PRE-CLOSING COVENANTS	21
<ul> <li>4.1 Closing Efforts</li> <li>4.2 Operation of the Business</li> <li>4.3 Access</li> <li>4.4 Sponsor and Manager Contract Consents</li> <li>4.5 Financial Statements</li> </ul>	21 22 24 25 26
ARTICLE V CONDITIONS PRECEDENT TO CLOSING	28
<ul><li>5.1 Conditions to Each Party's Obligations</li><li>5.2 Conditions to Obligations of the Buyer</li><li>5.3 Conditions to Obligations of the Seller</li></ul>	28 28 29
ARTICLE VI INDEMNIFICATION	30
<ul> <li>6.1 Indemnification by the Seller</li> <li>6.2 Indemnification by the Buyer</li> <li>6.3 Claims for Indemnification</li> <li>6.4 Survival</li> <li>6.5 Limitations</li> <li>6.6 Treatment of Indemnification Payments</li> <li>6.7 Setoff Rights</li> </ul>	30 30 30 32 32 32 34 34
ARTICLE VII TAX MATTERS	34
<ul> <li>7.1 Transfer Taxes</li> <li>7.2 Allocation of Certain Taxes</li> <li>7.3 Cooperation on Tax Matters; Tax Audits</li> <li>7.4 Post-Closing Actions</li> <li>7.5 Termination of Tax Sharing Agreements</li> </ul>	34 34 35 35 35
ARTICLE VIII TERMINATION	35
<ul><li>8.1 Termination of Agreement</li><li>8.2 Effect of Termination</li></ul>	35 36
ARTICLE IX EMPLOYEE MATTERS	36
<ul> <li>9.1 Offer of Employment; Continuation of Employment</li> <li>9.2 Cessation of Employee Benefit Plan Participation</li> <li>9.3 Compensation; Employee Benefits; Severance Plans</li> <li>9.4 Employment Related Liabilities</li> <li>9.5 U.S. WARN Act</li> <li>9.6 U.S. COBRA</li> <li>9.7 Non-Solicitation</li> </ul>	36 37 38 39 40 40 40

41
41 43 43 44 45 45
45
45 61
63
63 64 64 64 64 65 65 65 65 66 66 66 66 66 66 66 68 86 88
68 68

Ext		

Exhibit A	-	Closing Adjustment Amount and Closing Net Asset Schedule Example
Exhibit B-1	_	Form of Seller Bill of Sale
Exhibit B-2	-	Form of Affiliate Bill of Sale
Exhibit C	_	Form of Assignment and Assumption Agreement
Exhibit D	_	Form of Transition Services Agreement
Exhibit E	_	Form of Gateway Sublease Agreement
Exhibit F	_	Form of MAPSA Subcontracting Agreement
Exhibit G-1	-	Form of Pruco Contract
Exhibit G-2	_	Form of Pruco/PI Termination Agreement
Exhibit H-1	_	Form of Notice and Consent to Participating Sponsors
Exhibit H-2	_	Form of Notice and Consent to Participating Managers
Exhibit H-3	_	Form of Negative Consent Notice
Exhibit I	_	Form of Buyer Press Release

Exhibit J – Calculation of Acquired Assets Revenue and Pruco Revenue

## PURCHASE AND SALE AGREEMENT

This PURCHASE AND SALE AGREEMENT (this "<u>Agreement</u>") is entered into as of April 11, 2013 between PRUDENTIAL INVESTMENTS LLC, a New York limited liability company (the "<u>Seller</u>") and ENVESTNET, INC., a Delaware corporation (the "<u>Buyer</u>"). The Seller and the Buyer are each referred to herein individually as a "<u>Party</u>" and collectively as the "<u>Parties</u>."

## INTRODUCTION

WHEREAS, the Seller is engaged, among other matters, in the Business;

WHEREAS, the Buyer desires to purchase from the Seller, and the Seller desires to sell to the Buyer, certain assets of the Seller relating exclusively to the Business, subject to the assumption of certain related liabilities and upon the terms and subject to the conditions set forth herein; and

WHEREAS, capitalized terms used in this Agreement shall have the meanings ascribed to them in ARTICLE XI.

NOW, THEREFORE, in consideration of the representations, warranties, covenants and agreements contained in this Agreement and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties agree as follows:

## ARTICLE I ASSET PURCHASE

1.1 Purchase and Sale of Assets; Assumption of Liabilities .

(a) <u>Transfer of Assets</u>. On the terms and subject to the conditions set forth in this Agreement, at the Closing, the Seller shall sell, convey, assign, transfer and deliver to the Buyer, and the Buyer shall purchase and acquire from the Seller, all of the Seller's right, title and interest in and to the Acquired Assets.

(b) Excluded Assets. Notwithstanding anything to the contrary in this Agreement, the Acquired Assets shall not include any of the Excluded Assets.

(c) <u>Assumed Liabilities</u>. On the terms and subject to the conditions set forth in this Agreement, at the Closing, the Buyer shall assume and agree to pay, perform and discharge when due the Assumed Liabilities.

(d) <u>Excluded Liabilities</u>. Notwithstanding anything to the contrary in this Agreement, the Assumed Liabilities shall not include the Excluded Liabilities.

## 1.2 Purchase Price and Related Matters.

(a) <u>Closing Date Adjusted Purchase Price</u>. In consideration for the sale and transfer of the Acquired Assets and the Affiliate Assets, at the Closing the Buyer shall assume the Assumed Liabilities and shall pay to the Seller the Closing Date Adjusted Purchase Price in cash in immediately available funds into an account designated by the Seller. Following the Closing, the Closing Date Adjusted Purchase Price shall be subject to verification and adjustment as set forth in <u>Section 1.5</u>.

(b) Additional Purchase Price Payments and Excess Revenue Payments. As additional consideration for the sale and transfer of the Acquired Assets, following the Closing, (i) for each Post-Closing Period, the Buyer shall make the applicable Additional Purchase Price Payment, if any, in respect of such Post-Closing Period and (ii) for each of the Second Post-Closing Period and the Third Post-Closing Period, the Buyer shall make the Applicable Excess Revenue Payment, if any, in respect of such Post-Closing Period. The payments contemplated by this <u>Section 1.2(b)</u> shall be made shall be made in accordance with and subject to the procedures set forth in Section 1.2(c).

## (c) Calculation of Additional Purchase Price Payments and Excess Revenue Payments; Seller's Audit Rights .

(i) Not later than forty-five (45) days after the end of each Post-Closing Period, the Buyer shall prepare in good faith and deliver to the Seller a statement setting forth the Buyer's calculation in accordance with Exhibit J of the Acquired Assets Revenue, the Annualized Revenue and the Applicable Additional Purchase Price Payment, in each case, as of the end of the applicable Post-Closing Period, and, if applicable, the Applicable Excess Revenue Payment and Lost Pruco Revenue (each such statement, a "Revenue Statement"). Each Revenue Statement (x) shall include for the relevant calendar quarter (A) the amounts called for by <u>clauses (a)(i)</u>, (a)(ii), (b)(i) and (b)(ii) of the definition of "Acquired Assets Revenue" (as set forth in <u>Schedule 11.1</u>) for each Acquired Sponsor Contract and each Acquired Prospect Contract and (B) if applicable, the amounts called for by <u>clauses (a)(i)</u> of the definition of "Annualized Pruco Revenue" (as set forth in <u>Schedule 11.1</u>) and (y) shall be accompanied by the supporting materials set forth on <u>Schedule 1.2(c)</u>.

(ii) As soon as practicable following delivery of the Revenue Statement for the most recently ended Post-Closing Period (but no later than fifty (50) days after the end of such Post-Closing Period), the Buyer shall, subject to the Buyer's setoff right under <u>Section 6.7</u>, pay to the Seller (in cash by wire transfer of immediately available funds into an account designated by the Seller) the Applicable Additional Purchase Payment and Applicable Excess Revenue Payment, if any, for such Post-Closing Period.

(iii) Upon the written request of the Seller provided within sixty (60) days after the date a Revenue Statement has been delivered pursuant to <u>Section 1.2(c)</u>, the Buyer shall permit a mutually acceptable independent accounting firm (the "<u>Section 1.2 Accounting Firm</u>"), at the Seller's expense, to have access to the work papers and back-up materials used in preparing such Revenue Statement (but not any preceding Revenue Statement), as well as the personnel of the Buyer and its Affiliates with knowledge regarding the matters underlying such

Revenue Statement and the calculations set forth therein, at reasonable times and upon reasonable notice, for the purpose of determining the accuracy of the Revenue Statement and such calculations. The Seller shall use commercially reasonable efforts to cause the Section 1.2 Accounting Firm to work in good faith to complete any such inspection and audit within sixty (60) days of the date the Seller has made its request under the first sentence of this <u>Section 1.2(c)(iii)</u>; provided that such sixty (60)-day period shall be extended to the extent the Buyer has delayed in agreeing to the identity of the Section 1.2 Accounting Firm or access to the relevant work papers, back-up materials and personnel of the Buyer and its Affiliates is not provided to the Section 1.2 Accounting Firm promptly. If the Section 1.2 Accounting Firm concludes that Annualized Revenue was underreported, or that Annualized Pruco Revenue was overreported, in each case, in such Revenue Statement by more than five percent (5%), the Buyer shall promptly reimburse the Seller for the reasonable out-of-pocket costs of the Section 1.2 Accounting Firm's inspection and audit. If the Seller does not deliver the written request required under this <u>Section 1.2(c)(iii)</u> within sixty (60) days after the date a Revenue Statement has been delivered to the Seller pursuant to <u>Section 1.2(c)(i)</u>, such Revenue Statement shall be conclusive and binding, absent fraud or manifest error.

(iv) Promptly following any determination by the Section 1.2 Accounting Firm that Annualized Revenue was underreported, or that, if reported, Annualized Pruco Revenue was overreported, in each case in such Revenue Statement in a manner negatively impacting the amount of an Applicable Additional Purchase Price Payment or Applicable Excess Revenue Payment for the applicable Post-Closing Period (but no later than two (2) Business Days following such determination), the Buyer shall pay to the Seller (in cash by wire transfer of immediately available funds into an account designated by the Seller) the total amount by which the Applicable Additional Purchase Price Payment or Applicable Excess Revenue Payment previously made relating to such Post-Closing Period was negatively impacted as a result of such underreporting of Annualized Revenue or, if reported, overreporting of Annualized Pruco Revenue (plus interest from the date that is fifty (50) days following the end of the applicable Post Closing Period to, but not including, the date of payment at a rate equal to one month LIBOR as of such date, calculated on a 360 day basis). In the event that the Section 1.2 Accounting Firm determines that the Buyer paid to the Seller an amount greater than that to which the Seller was entitled pursuant to this Section 1.2, then, no later than two (2) Business Days following such determination, the Seller shall pay to the Buyer in cash by wire transfer of immediately available funds into an account designated by the Buyer the total amount by which the Applicable Purchase Price Payment or Applicable Excess Revenue Payment previously made by the Buyer relating to such Post-Closing Period was overpaid.

(v) The Parties understand and agree that the Post-Closing Payments are an integral part of the consideration payable to the Seller in connection with the transactions contemplated hereby and, other than as specifically contemplated by <u>Section 6.7</u>, shall be payable without setoff or counterclaim. Once paid, any and all amounts paid by the Buyer with respect to any Post-Closing Payments hereunder shall not be refundable or otherwise subject to retroactive reduction.

(vi) For all purposes of this Agreement, the Acquired Assets Revenue, the Annualized Revenue and the Annualized Pruco Revenue shall be calculated in accordance with the principles and procedures set forth on <u>Exhibit J</u> (it being understood and agreed that Annualized Pruco Revenue will be calculated in the same manner the Acquired Assets Revenue and the Annualized Revenue are calculated with respect to the other Sponsor Contracts in existence on the date hereof).

(d) <u>Post-Closing Conduct of the Business</u>. From and after the Closing until the end of the Third Post-Closing Period, the Buyer shall and, if applicable, shall cause the Buyer Subsidiaries to operate the Business in a commercially reasonable manner. From and after the Closing until the Final Payment Date, the Buyer shall not take any action or inaction in bad faith or with the intention of decreasing the amount of any Post-Closing Payment, including transferring any assets among the Buyer and the Buyer Subsidiaries with the intention of impairing the Buyer's ability to make the Post-Closing Payments. Notwithstanding the foregoing, the Parties acknowledge and agree that (i) the provisions of this <u>Section 1.2(d)</u> shall not (x) require the Buyer to continue any line of business conducted by the Business as of any date if the Buyer, in the exercise of its reasonable business judgment, believes that it is not reasonable for the Buyer to continue such line of business or (y) limit the Buyer from improving or modifying any aspect of the Business in the exercise of the Buyer's reasonable business judgment and (ii) the Seller's right to the Post-Closing Payments under this <u>Section 1.2</u> is a contract right and shall not give rise to any rights or duties (including fiduciary duties), express or implied, other than those expressly set forth herein.

(e) <u>Sale of Acquired Assets</u>. In the event that, prior to the Final Payment Date, the Buyer proposes to consummate a transaction pursuant to which all or substantially all of the Acquired Sponsor Contracts and Acquired Prospect Contracts are sold or transferred to an entity that is not a Buyer Subsidiary (or the Buyer, or a Buyer Subsidiary that is party to any such contracts, is sold or transferred to an entity that is not the Buyer or another Buyer Subsidiary) (any such transaction, a "<u>Divestiture</u>") in a transaction in which the proposed purchaser or transferee would assume the obligations of the Buyer under this <u>Section 1.2</u>, the Buyer shall provide notice of such proposed Divestiture to the Seller, together with such information regarding the relevant purchaser or transferee as the Seller shall reasonably request. The Seller shall have ten (10) Business Days from the date the Seller receives any such notice (or any information the Seller may have reasonably requested within the initial ten (10) Business Day period) to provide an objection to any such proposed Divestiture on the basis that there is a reasonable likelihood that the proposed purchaser or transferee is not financially capable of making the remaining Post-Closing Payments. The Buyer shall provide the Seller with prompt notice of the consummation of any Divestiture and thereafter shall have no further obligations pursuant to this <u>Section 1.2</u> except to the extent (x) the Seller provides a timely notice of objection pursuant to the immediately preceding sentence or (y) the definitive documentation for the proposed Divestiture does not contain a written assumption of the Buyer's obligations under this <u>Section 1.2</u>. For the avoidance of doubt, the Seller's timely objection notice shall not prohibit the consummation of the Divestiture.

(f) <u>Books and Records</u>. From the Closing Date until the Final Payment Date, the Buyer agrees to maintain true and accurate books and records relating to the subject matter of this <u>Section 1.2</u>, which it shall make available to the Seller pursuant to <u>Section 10.1</u>.

1.3 <u>Allocation of Purchase Price</u>. The Buyer shall prepare and deliver to the Seller, within sixty (60) days following the determination of the Final Adjusted Purchase Price pursuant to <u>Section 1.5(c)(i)</u>, a schedule setting forth the allocation of the Final Adjusted Purchase Price (including any Assumed Liabilities and any other capitalizable costs to the extent properly taken into account under the Code) among the Acquired Assets in accordance with Section 1060 of the Code and the Treasury Regulations thereunder (the "<u>Allocation</u>"). If within thirty (30) Business Days after receipt of the Allocation, the Seller notifies the Buyer in writing that the Seller objects to one or more items reflected in such Allocation, the Buyer and the Seller shall negotiate in good faith to resolve such dispute. If the Buyer and the Seller resolve any such dispute and agree to a final Allocation, the Buyer and the Seller shall file all Tax Returns (including Internal Revenue Service Form 8594) consistent with the final Allocation, and neither the Buyer nor the Seller shall take any Tax position inconsistent with the final Allocation, except to the extent otherwise required by law. If the Buyer and the Seller shall not be bound by the Allocation as prepared by the Buyer and shall be permitted to file any Tax Returns and take any Tax position consistent with the Seller's determination of the Final Adjusted Purchase Price (including any Assume Liabilities and any other capitalizable costs to the extent properly taken into account under the Code) shall be allocated as provided by Treasury Regulations Section 1.1060-1(c).

#### 1.4 The Closing.

(a) <u>Time and Location</u>. The Closing shall take place at the offices of Wilmer Cutler Pickering Hale and Dorr LLP located at 7 World Trade Center, 250 Greenwich Street, New York, NY, 10007, or remotely via the exchange of documents and signatures, commencing at 10:00 a.m., local time, on the Closing Date.

(b) Actions Prior to the Closing. Not less than three (3) Business Days prior to the Closing Date, the Seller shall prepare in good faith and deliver to the Buyer the Seller's estimate of (x) the Closing Net Asset Schedule (the "Seller's Closing Net Asset Schedule"), which shall include updated information with respect to Schedule 1.1(a)(ii), (y) the Closing Adjustment Amount (the "Seller-Determined Closing Adjustment Amount") and (z) the Final Adjusted Purchase Price (the "Closing Date Adjusted Purchase Price"). The Seller shall make the work papers and back-up materials used in preparing the Seller's Closing Net Asset Schedule and the Seller-Determined Closing Adjustment Amount, as well as personnel of the Seller with knowledge regarding such calculations, available to the Buyer and its Representatives at reasonable times and upon reasonable notice.

### (c) Actions at the Closing. At the Closing:

(i) the Seller shall deliver the Seller Certificate;

(ii) the Buyer shall deliver the Buyer Certificate;

(iii) the Seller shall execute and deliver a Bill of Sale in substantially the form attached hereto as Exhibit B-1 (the "Seller Bill of Sale");

(iv) the Seller Affiliate shall execute and deliver a Bill of Sale in the form attached hereto as Exhibit B-2 (the "Affiliate Bill of Sale");

(v) the Buyer shall execute and deliver to the Seller an Assignment and Assumption Agreement in the form attached hereto as Exhibit C (the "Assignment and Assumption Agreement");

(vi) the Buyer and the Seller shall execute and deliver a Transition Services Agreement in substantially the form attached hereto as <u>Exhibit</u> <u>D</u> (the "<u>Transition Services Agreement</u>");

(vii) the Buyer and the Seller shall execute and deliver (A) a Gateway Sublease Agreement in the form attached hereto as <u>Exhibit E</u> (the "<u>Gateway Sublease Agreement</u>") or (B) such other agreement for the purposes of documenting the alternative to the Gateway Sublease Agreement contemplated by <u>Section 5.2(i)(B)</u>;

(viii) to the extent there are any Non-Consenting Sponsors or Terminating Sponsors as of the Closing, the Buyer and the Seller shall execute and deliver a MAPSA Subcontracting Agreement in substantially the form attached hereto as <u>Exhibit F</u> (the "<u>MAPSA Subcontracting Agreement</u>");

(ix) the Seller shall execute and deliver to the Buyer a certificate as to the non-foreign status of the Seller, dated as of the Closing Date, complying with the requirements of Treasury Regulation Section 1.1445-2(b)(2);

(x) (A) the Buyer and Pruco shall execute and deliver the Pruco Contract in substantially the form attached hereto as Exhibit G-1 and (B) the Seller and Pruco shall execute and deliver a termination agreement in substantially the form attached hereto as Exhibit G-2 (the "Pruco/PI Termination Agreement);

(xi) the Buyer shall pay to the Seller the Closing Date Adjusted Purchase Price in cash by wire transfer of immediately available funds into an account designated by the Seller; and

(xii) the Seller shall cause to be filed such documents as are necessary to change the name of the Canadian Subsidiary to a name chosen by the Seller that does not include the phrase "Wealth Management Solutions" or "WMS".

## 1.5 Post-Closing Purchase Price Adjustment.

(a) As soon as practicable following the Closing Date (but no later than thirty (30) Business Days after the Closing Date), the Buyer shall prepare and deliver to the Seller the Buyer's version of the Closing Net Asset Schedule (the "<u>Buyer's Closing Net Asset Schedule</u>") together with a written statement that sets forth the Buyer's calculation of the Closing Adjustment Amount calculated as of the end of the day immediately preceding the Closing Date (the "<u>Buyer-Determined Closing Adjustment Amount</u>"), together with a summary of differences (if any) between the Buyer's Closing Net Asset Schedule and the Buyer-Determined Closing Adjustment Amount, on the one hand, and the Seller's Closing Net Asset Schedule and the

Seller-Determined Closing Adjustment Amount, on the other hand. The Buyer shall make the work papers and back-up materials used in preparing the Buyer's Closing Net Asset Schedule and the Buyer-Determined Closing Adjustment Amount, as well as the personnel of the Buyer with knowledge regarding such calculations, available to the Seller and its Representatives at reasonable times and upon reasonable notice.

(b) If the Seller fails to deliver to the Buyer a statement describing any objections to the Buyer's Closing Net Asset Schedule or the Buyer-Determined Closing Adjustment Amount within thirty (30) Business Days after receipt by the Seller, then such Buyer-Determined Closing Adjustment Amount shall be deemed to be the "Stipulated Closing Adjustment Amount". However, if the Seller delivers to the Buyer a statement describing any objections to the Buyer's Closing Net Asset Schedule or the Buyer-Determined Closing Adjustment Amount within such thirty (30)-Business Day period, then the Seller and the Buyer shall exercise commercially reasonable efforts to resolve, in good faith, any such objections. If the Buyer and the Seller reach a resolution of all such objections, then the Buyer-Determined Closing Adjustment Amount as modified by such resolution shall be deemed to be the "Stipulated Closing Adjustment Amount". If such a resolution is not reached within twenty (20) Business Days after the Buyer has received a statement describing the Seller's objections to the Buyer's Closing Net Asset Schedule or the Buyer-Determined Closing Adjustment Amount, then the Seller and the Buyer shall submit promptly and in any event no later than five (5) Business Days after the expiration of such twenty (20)-Business Day period any unresolved objections (the "Disputed Items") to a mutually acceptable independent accounting firm (the "Section 1.5 Accounting Firm") for resolution in accordance with the following sentences of this Section 1.5(b). The Buyer, on the one hand, and the Seller, on the other hand, may provide the Section 1.5 Accounting Firm, within ten (10) Business Days of retaining the Section 1.5 Accounting Firm, with a definitive statement of their respective positions with respect to each Disputed Item (and only such Disputed Items). The Section 1.5 Accounting Firm will be provided with access to the books and records of the Buyer and the Seller relevant to the preparation of the Closing Net Asset Schedule and the calculation of the Closing Adjustment Amount. The Section 1.5 Accounting Firm will be asked (i) to review each of the Disputed Items on an expedited basis, and shall in any event have no more than twenty (20) Business Days to carry out the procedures set forth in this Section 1.5(b), (ii) to prepare a written statement of its decision with respect to each Disputed Item, which decision shall be the position of the Buyer or the Seller for such Disputed Item that is closest to the Accounting Firm's independent determination of such Disputed Item based on the principles set forth on Exhibit A, and (iii) to prepare the Closing Net Asset Schedule and the calculation of the Closing Adjustment Amount, in each case, as modified by the Section 1.5 Accounting Firm's decision (as between the Buyer's and the Seller's position) with respect to each Disputed Item. The Closing Adjustment Amount so calculated shall be deemed to be the "Stipulated Closing Adjustment Amount". The decision of the Section 1.5 Accounting Firm shall be conclusive and binding, absent fraud or manifest error. The Parties shall each pay their own respective costs and expenses incurred under this Section 1.5, and the Party whose proposed Closing Adjustment Amount differed the most and was the farthest away from the Stipulated Closing Adjustment Amount, as determined by the Section 1.5 Accounting Firm pursuant to the second preceding sentence, shall be responsible for the fees and expenses of the Section 1.5 Accounting Firm.

(c) Within two (2) Business Days of the determination of the Stipulated Closing Adjustment Amount in accordance with <u>Section 1.5(b)</u>, the following shall occur:

(i) Using the Stipulated Closing Adjustment Amount as the Closing Adjustment Amount, the Seller and the Buyer shall calculate the Final Adjusted Purchase Price.

(ii) If the Final Adjusted Purchase Price is greater than the Closing Date Adjusted Purchase Price, then the Buyer shall pay to the Seller (in cash by wire transfer of immediately available funds into an account designated by the Seller) the total amount by which the Final Adjusted Purchase Price exceeds the Closing Date Adjusted Purchase Price (plus interest from the Closing Date to, but not including, the date of payment at a rate equal to one-month LIBOR as of such date, calculated on a 360-day basis).

(iii) If the Final Adjusted Purchase Price is less than the Closing Date Adjusted Purchase Price, then the Seller shall pay to the Buyer (in cash by wire transfer of immediately available funds into an account designated by the Buyer) the total amount by which the Final Adjusted Purchase Price is less than the Closing Date Adjusted Purchase Price (plus interest from the Closing Date to, but not including, the date of payment at a rate equal to one-month LIBOR as of such date, calculated on a 360-day basis).

1.6 <u>Consents to Assignment</u>. If (a) any Assigned Contract other than any Sponsor Contract or any Manager Contract (each, a "<u>Non-Assigned Contract</u>") may not be assigned, delegated and transferred by the Seller to the Buyer (as a result of either the provisions thereof or applicable law) without the consent or approval of a third party and (b) such consent or approval has not been obtained prior to the Closing, then, with respect to each such Non-Assigned Contract (other than any Non-Assigned Contract that has been cancelled or terminated by the applicable counterparty or has expired), (i) notwithstanding any other provision of this Agreement, such Non-Assigned Contract shall not be assigned, delegated and transferred by the Seller to the Buyer at the Closing, (ii) until the end of the Post-Closing Consent Period, the Seller and the Buyer shall cooperate, in all commercially reasonable respects, to obtain the necessary consent or approval as soon as practicable after the Closing, provided that neither the Seller nor the Buyer shall be required to make any payments or agree to any material undertakings in connection with obtaining any such consent, and (iii) from and after the Closing until the earlier to occur of (x) the assignment of such Non-Assigned Contract, the Seller and the Buyer shall cooperate, in all commercially reasonable respects (including, if permitted by applicable law and contract, by entering into a subcontracting arrangement), to provide to the Buyer all of the benefits under such Non-Assigned Contract and to cause the Buyer to fulfill, satisfy and perform all of the liabilities and obligations under such Non-Assigned Contract.

1.7 <u>Further Assurances</u>. At any time and from time to time after the Closing Date, as and when requested by either Party and at such Party's expense, the other Party shall promptly execute and deliver, or cause to be executed and delivered, all such documents, instruments and certificates and shall take, or cause to be taken, all such further or other actions as are necessary to evidence and effectuate the transactions contemplated by this Agreement.

1.8 Withholding. The Buyer shall be entitled to deduct and withhold from the amounts otherwise payable by it pursuant to this Agreement such amounts as it is required to deduct and withhold with respect to the making of such payment under the Code or any provision of state, local or foreign Tax Law; provided that the Parties agree to cooperate with one another and take all reasonable actions to minimize such amounts that need to be deducted and withheld. In the event any amount is deducted and withheld, such amount shall be promptly paid to the appropriate Governmental Entity and shall be treated for all purposes of this Agreement as having been paid by the Buyer.

## ARTICLE II REPRESENTATIONS AND WARRANTIES OF THE SELLER

The Seller represents and warrants to the Buyer that the statements contained in this <u>ARTICLE II</u> are true and correct as of the date hereof, except as set forth in the Disclosure Schedule. The Disclosure Schedule shall be arranged in sections and subsections corresponding to the numbered and lettered sections and subsections contained in this <u>ARTICLE II</u>. The information disclosed in any section or subsection of the Disclosure Schedule shall be considered to be disclosed with respect to each other section and/or subsection of the Disclosure Schedule to the extent that it is reasonably apparent on the face of the Disclosure Schedule that such disclosed information is applicable to such other section or subsection. For purposes of this Agreement, the phrase "to the knowledge of the Seller" or any phrase of similar import shall mean and be limited to the actual knowledge following reasonable inquiry and investigation of the following individuals: Kevin Osborn, Laurie Campbell, Peter Laskaris, John Yackel, Tim Maroney, Gaurav Wadhwa and Robert Cohen.

2.1 Organization, Qualification and Corporate Power. The Seller is duly formed, validly existing and in good standing under the laws of the State of New York. The Seller has the limited liability company power and authority to own or lease all of its properties and assets and to carry on its business as it is now being conducted, and is duly qualified to conduct business under the laws of each jurisdiction where the character of the properties owned, leased or operated by it or the nature of its activities, in each case as they relate exclusively to the Business, makes such qualification necessary, except for any such failure to be so qualified that would not reasonably be expected to result in a Business Material Adverse Effect.

2.2 <u>Authority</u>. The Seller has all requisite limited liability company power and authority to execute and deliver this Agreement and the Ancillary Agreements to which it will be a party and to perform its obligations hereunder and thereunder. The execution and delivery by the Seller of this Agreement and such Ancillary Agreements and the consummation by the Seller of the transactions contemplated hereby and thereby have been validly authorized by all necessary limited liability company action on the part of the Seller. This Agreement has been, and such Ancillary Agreements will be, validly executed and delivered by the Seller and, assuming this Agreement and each such Ancillary Agreement constitute the valid and binding obligation of the Buyer, constitutes or will constitute a valid and binding obligation of the Seller, enforceable against the Seller in accordance with its terms, except as enforceability may be limited by bankruptcy, insolvency, fraudulent transfer, reorganization, moratorium or other similar laws relating to or affecting the rights of creditors generally and by equitable principles, including those limiting the availability of specific performance, injunctive relief and other equitable remedies and those providing for equitable defenses.

2.3 <u>Noncontravention</u>. Subject to the making of the filings and receipt of the permits, authorizations, consents and approvals listed in <u>Section 2.3 of the</u> <u>Disclosure Schedule</u>, neither the execution and delivery by the Seller of this Agreement or the Ancillary Agreements to which the Seller will be a party, nor the consummation by the Seller of the transactions contemplated hereby or thereby, will:

(a) conflict with or violate any provision of the certificate of formation or limited liability company agreement of the Seller;

(b) require on the part of the Seller any filing with, or any permit, authorization, consent or approval of, any (i) Governmental Entity or (ii) other Person, except in the case of this <u>clause (ii)</u>, for any such filing, permit, authorization, consent or approval which if not obtained or made would not reasonably be expected to result in a Business Material Adverse Effect;

(c) conflict with, result in a breach of, constitute (with or without due notice or lapse of time or both) a default under, result in the acceleration of obligations under, create in any party the right to terminate or modify, or require any notice, consent or waiver under, any (i) Assigned Contract or (ii) other contract or agreement to which the Seller is a party or by which the Seller is bound, except in the case of <u>clause (ii)</u> for (x) any conflict, breach, default, acceleration or right to terminate or modify that would not reasonably be expected to result in a Business Material Adverse Effect or (y) any notice, consent or waiver the absence of which would not reasonably be expected to result in a Business Material Adverse Effect; or

(d) violate any judgment, order, writ, injunction or decree of any Governmental Entity (each, an "<u>Order</u>") specifically naming, or statute, rule or regulation applicable to, the Seller or any of or its properties or assets.

### 2.4 Financial Information.

(a) Section 2.4 of the Disclosure Schedule sets forth the pro forma divisional profit and loss statements of the Business for (i) each of the fiscal years ended December 31, 2010, December 31, 2011 and December 31, 2012 and (ii) the two (2) month period ended February 28, 2013. Taking into account the methodology set forth on Section 2.4(a) of the Disclosure Schedule, the pro forma divisional profit and loss statements present fairly, in all material respects, the results of operations of the Business for the periods indicated in such profit and loss statements.

(b) As of the Closing, taking into account the applicable methodologies set forth on <u>Section 2.4(b) of the Disclosure Schedule</u>, the Required Financial Statements and the Specified Quarterly Financial Statements will present fairly, in all material respects, the results of operations of the Business for the period indicated in such financial statements.

(c) As of the Closing, the Specified Quarterly Financial Statements will have been prepared in a manner consistent with (i) if the Abbreviated Financial Statements constitute the Required Financial Statements, the financial statements described in <u>clause (ii)</u> of the definition of Abbreviated Financial Statements or (ii) if the Full Financial Statements constitute the Required Financial Statements, the applicable financial statements described in <u>clause (ii)</u> of the definition of Full Financial Statements.

## 2.5 Tax Matters.

(a) The Seller has filed or had filed on its behalf all material Tax Returns that it was required to file with respect to the Business and all such Tax Returns were correct and complete in all material respects to the extent they relate to the Business. The Seller has paid (or had paid on its behalf) all material Taxes due and payable to the extent they relate to the Business or could reasonably be expected to give rise to a lien on the Acquired Assets (unless such Taxes are being contested in good faith).

(b) There are no liens for Taxes against the Acquired Assets, other than Permitted Liens.

(c) No claim has been made in writing by any Governmental Entity in a jurisdiction where Tax Returns are not filed with respect to the Business asserting that the Seller may be subject to taxation in such jurisdiction with respect to the Business, and there is no dispute or claim pending concerning Tax liability with respect to the Business that has been raised in writing by any Governmental Entity.

(d) The Seller has complied in all material respects with all withholding and information reporting requirements with respect to the Business.

### 2.6 Acquired Assets.

(a) Except as set forth in Section 2.6 of the Disclosure Schedule, the Seller has good and valid title to, a valid leasehold interest in or a valid license or right to use, all of the Sponsor Contracts, Manager Contracts and all of the remaining material Acquired Assets, free and clear of all Liens (other than Permitted Liens). As of the date of this Agreement, except as set forth in Section 2.6 of the Disclosure Schedule, the Seller has the full right to sell, convey, transfer, assign and deliver the Sponsor Contracts, the Manager Contracts and the remaining material Acquired Assets to the Buyer, free and clear of all Liens (other than Permitted Liens) (it being understood and agreed that any and all Liens set forth in Section 2.6 of the Disclosure Schedule shall be released as of the Closing). As of the Closing, except as set forth in Section 2.6 of the Disclosure Schedule, the Seller will have the full right to sell, convey, transfer, assign and deliver the Assigned Sponsor Contracts, the Assigned Manager Contracts and the remaining material Acquired Assets to the Buyer, free and clear of all Liens (other than Permitted Liens).

(b) All of the material tangible assets included in the Acquired Assets have been well maintained and are in good operating condition and repair (with the exception of normal wear and tear), and are free from defects other than such minor defects as do not interfere with the intended use thereof in the conduct of normal operations.

# 2.7 Leased Real Property.

(a) Section 2.7 of the Disclosure Schedule includes a correct and complete list of the lease and amendments or other modifications thereto to which Seller's Affiliate is a party covering the Leased Real Property (the "<u>Real Property Lease</u>"). The Seller has delivered to the Buyer a correct and complete copy of the Real Property Lease.

(b) The Real Property Lease is in full force and effect, valid and enforceable in accordance with its terms. Neither the Seller nor any of its Affiliates has received as of the date hereof any written notice of any dispute, claim, event of default or event that constitutes or would constitute (with notice or lapse of time or both) a default under the Real Property Lease which remains outstanding.

(c) To the knowledge of the Seller, neither the Seller nor any of its Affiliates has received written notice from a Governmental Entity regarding any pending, threatened or proposed proceeding or governmental action to modify the zoning classification of, or to condemn or take by the power of eminent domain (or to purchase in lieu thereof) all or any part of the Leased Real Property.

### 2.8 Intellectual Property.

(a) <u>Ownership</u>. The Seller, together with Seller Affiliate, is the sole and exclusive owner of all material Seller Owned Intellectual Property, free and clear of any Liens, other than (i) Permitted Liens and (ii) Liens existing pursuant to any license of Seller Intellectual Property that is described in <u>Section 2.8(d)</u> below. There are no Intellectual Property Registrations in the Seller Owned Intellectual Property.

(b) <u>Infringement by Seller</u>. To the knowledge of the Seller, none of the Seller Owned Intellectual Property, or the Exploitation thereof by the Seller, infringes or violates, or constitutes a misappropriation of, any Intellectual Property rights of any third party. <u>Section 2.8(b) of the Disclosure Schedule</u> sets forth a true and correct list, as of the date of this Agreement, of any written complaint, claim or notice received by the Seller, or any pending proceedings which are known to Seller, alleging any such infringement, violation or misappropriation. To the knowledge of the Seller, there is no outstanding Order regarding the Seller Owned Intellectual Property affecting the Seller's use of any of the Seller Owned Intellectual Property or any of its rights thereto.

(c) Infringement by Third Parties. To the knowledge of the Seller, no Person is misappropriating any of the Software.

(d) <u>Outbound IP Agreements</u>. <u>Section 2.8(d) of the Disclosure Schedule</u> contains, as of the date of this Agreement, a true and complete list of each license or other agreement pursuant to which the Seller has assigned, transferred, licensed, distributed or otherwise granted any right or access to any Person, or covenanted not to assert any right, with respect to any Seller Intellectual Property.

(e) Inbound IP Agreements. Section 2.8(e) of the Disclosure Schedule sets forth a true and complete list, as of the date of this Agreement, of (i) each item of Seller Licensed Intellectual Property (excluding licenses for off the shelf software that is generally or freely available or licenses with license fees of \$5,000 or less per seat, user or workstation per annum) and the license or agreement pursuant to which the Seller Exploits it (excluding currently-available, off the shelf software programs that are licensed by the Seller pursuant to "shrink wrap" licenses) and (ii) each agreement, contract, assignment or other instrument pursuant to which the Seller has obtained any joint or sole ownership interest in or to each item of Seller Owned Intellectual Property, other than employee and consultant agreements in the ordinary course of business.

(f) <u>Protection Measures</u>. The Seller has taken reasonable measures to protect the proprietary nature of each item of Seller Intellectual Property, and to maintain in confidence all trade secrets and confidential information comprising a part thereof. The Seller has never disclosed or delivered to any escrow agent or any other Person any source code form of the Software (other than to the Seller's employees and subcontractors who have a "need to know" in connection with the performance of their duties to the Seller and who have executed reasonable non-disclosure agreements with the Seller), and no other Person has the right, contingent or otherwise, to obtain access to or use any such source code. No event has occurred, and no circumstance or condition exists, that is likely to require the delivery, license or disclosure of such source code of the Software to any Person. Seller has taken and maintains reasonable measures to protect the Software against unauthorized use, modification, or access thereto, or the introduction of any viruses or other unauthorized or damaging or corrupting elements. To the knowledge of the Seller, none of the Software contains any disabling codes or instructions (including "viruses" or "worms").

(g) <u>Public Rights</u>. Except as set forth in <u>Section 2.8(g) of the Disclosure Schedule</u>, none of the Software contains or is derived in any manner (in whole or in part) from any software that is distributed under the GNU General Public License, Lesser/Library GPL (LGPL), Artistic License (e.g., PERL), Mozilla Public License, Netscape Public License, Sun Community Source License (SCSL), Sun Industry Standards License (SISL) or any similar licenses or distribution models ("<u>Open Source Materials</u>"), to the extent that such use of such Open Source Materials would create, or purport to create, an obligation of the Seller that the Software be (x) disclosed or distributed in source code form, (y) licensed to a third party for the purpose of making derivative works, or (z) redistributable at no charge or minimal charge.

(h) <u>Privacy</u>. Solely as related to the Business and the Acquired Assets, and except as set forth in <u>Section 2.8(h) of the Disclosure Schedule</u>, the Seller is in compliance in all material respects with its own privacy and security policies, in each case, as applicable to Personal Information of customers and employees of the Business. As of the date of this Agreement, the Seller is not prohibited by any Privacy Law or its own privacy or security policies from providing the Buyer with, or transferring to the Buyer, all or any portion of Personal Information of customers and employees of the Business collected, stored, acquired and used in the conduct of the Business by the Seller. Except as set forth in <u>Section 2.8(h) of the Disclosure Schedule</u>, neither the Seller nor, to the knowledge of the Seller, any third party service provider working on behalf of the Seller has had, at any time since September 30, 2010, a breach, security incident or unauthorized access, disclosure, use or loss of any information, in each case, solely to the extent related to the Business, for which the Seller is either required to notify individuals and/or to notify any Governmental Entity.

# 2.9 Contracts.

(a) Section 2.9 of the Disclosure Schedule lists, as of the date of this Agreement, all of the following contracts or agreements to which the Seller or any Affiliate of the Seller is a party and has or may have any rights or obligations and that (x) with respect to <u>clauses (i)</u>, (ii), (iii), (iv), (v) and (xii), relate exclusively to the Business and (y) with respect to <u>clauses (vi)</u>, (vii), (xi), (x), and (xi), relate to the Business:

(i) Sponsor Contracts, Manager Contracts and all other similar contracts or agreements;

(ii) each license or agreement pursuant to which the Seller Exploits the Seller Licensed Intellectual Property;

(iii) any agreement (or group of related agreements with the same party) for the purchase of products or services under which the undelivered balance of such products and services is in excess of \$25,000, other than the Assigned Contracts;

(iv) any agreement (or group of related agreements with the same party) for the sale of goods or services that involves a payment to be made to the Seller in excess of \$25,000, other than agreements described in <u>clause (i)</u> above;

(v) any agreement for capital expenditures that requires aggregate future payments in excess of \$25,000;

(vi) any agreement for the acquisition by the Seller of any operating business, whether by merger, equity purchase or asset purchase, except for any such business which did not or will not become part of the Business;

(vii) any agreement establishing a partnership or joint venture;

(viii) any agreement (or group of related agreements with the same party) under which the Seller has consented to the imposition of a Lien (other than a Permitted Lien) on any of the Acquired Assets;

(ix) any agreement containing covenants that prohibit the Business from competing in any line of business or geographic area or that prohibit the Business from soliciting customers, clients or employees, in each case whether in any specified geographic region or business or generally;

(x) any employment agreement with any employee or officer of the Seller or the Canadian Subsidiary;

(xi) any severance, "stay-pay" or termination agreement with any Seller Business Employee other than any such employee who is not a Scheduled Employee or any such agreement that is an individual retention agreement or bonus guarantee agreement under which all liabilities will be retained by Seller or its Affiliates as provided in <u>Section 9.4(a)</u>; and

(xii) any other contract that is material to the Business.

(b) As of the date hereof, the Seller has made available to the Buyer a complete and accurate copy of each Sponsor Contract, Manager Contract and Assigned Contract listed in <u>Schedule 1.1(a)(vi)</u> (as constituted as of the date hereof), together with each amendment or modification thereto, other than Manager Contracts, together with any amendments or modifications, that contain confidentiality provisions prohibiting their disclosure (the "<u>Confidential Manager Contracts</u>"). The terms of each Confidential Manager Contract are substantially similar, in all material respects, to the terms provided for in the relevant template agreements identified on <u>Section 2.9(b) of the Disclosure Schedule</u>, copies of which have been made available to the Buyer as of the date hereof. As of the Closing, the Seller will have made available to the Buyer a complete and accurate copy of each Assigned Contract.

(c) Each Sponsor Contract, Manager Contract and Assigned Contract is a valid and binding obligation of the Seller and, to the knowledge of the Seller, of each other party thereto.

(d) As of the date hereof, neither the Seller nor, to the knowledge of the Seller, any third party is in breach or default, in any material respect, of any Sponsor Contract, Manager Contract or contract listed in Schedule 1.1(a)(vi) (as constituted as of the date hereof) and, to the knowledge of the Seller, no event has occurred which, with notice or lapse of time or both, would constitute a breach or default, in any material respect, or permit acceleration, termination or modification under any Sponsor Contract, Manager Contract or contract listed in Schedule 1.1(a)(vi) (as constituted as of the date hereof). As of the Closing, neither the Seller nor, to the knowledge of the Seller, any third party is in breach or default, in any material respect, of any Assigned Contract and, to the knowledge of the Seller, no event has occurred which, with notice or lapse of time or both, would constitute a breach or default, in any material respect, of any Assigned Contract and, to the knowledge of the Seller, no event has occurred which, with notice or lapse of time or both, would constitute a breach or default, in any material respect, or permit acceleration, termination or modification under any Assigned Contract.

2.10 <u>Sufficiency of Assets</u>. Taking into account (i) the items set forth in <u>Section 2.10 of the Disclosure Schedule</u>, (ii) any services to be provided to the Buyer pursuant to the Transition Services Agreement or the Gateway Sublease Agreement (or the alternative to the Gateway Sublease Agreement contemplated by <u>Section 5.2(i)(B)</u>), (iii) the alternative arrangements contemplated by <u>Section 1.6</u>, (iv) the MAPSA Subcontracting Agreement, (v) the Pruco Contract, (vi) the termination or expiration of any Sponsor Contract or Manager Contract and (vii) the Seller's current access to the Overhead and Shared Services to which the Buyer will not have access following the Closing (unless access is otherwise provided for pursuant to the Transition Services Agreement or the Gateway Sublease Agreement), the Acquired Assets and the Affiliate Assets, when utilized by a labor force substantially similar to that employed by the Seller in connection with the Business on the date hereof, constitute all of the assets, properties and rights that are necessary to conduct the Business immediately following the Closing, in all material respects, as the Business is currently conducted as of the date hereof.

2.11 <u>Litigation</u>. <u>Section 2.11 of the Disclosure Schedule</u> lists, as of the date of this Agreement, each material (a) Order specifically naming the Seller or the Canadian Subsidiary that relates to the Business and (b) action, suit, claim or legal, administrative or arbitratorial proceeding by or before any Governmental Entity (each, an "<u>Action</u>") to which the Seller or the Canadian Subsidiary is a party that relates primarily to the Business.

# 2.12 Employment Matters.

(a) The Seller has provided to the Buyer a list, as of the date of this Agreement, of all Seller Business Employees and Canadian Employees and with respect to each such employee, the following information: (i) the name of the employee's employer, (ii) position, (iii) annual rate of base compensation or salary, (iv) bonus or incentive compensation paid for such individuals during the preceding three calendar years, (v) whether full-time or part-time, (vi) whether exempt or non-exempt (or such analogous classifications as may apply with respect to the Canadian Employees), and (vii) principal workplace location.

(b) Neither the Seller nor the Canadian Subsidiary is a party to or bound by any collective bargaining agreement relating to the Business. With respect to the Business, the Seller and the Canadian Subsidiary are not currently subject to any material strikes, grievances, claims of unfair labor practices or other collective bargaining disputes. No representation election petition has been filed by any employee or is pending with the NLRB and no union organizing campaign involving or affecting any employee has occurred, is in progress or is threatened against the Business. No NLRB unfair labor practice charge (or litigation alleging such claim) has been filed or threatened or is presently pending relating to a Seller Business Employee.

(c) Except as would not have a Business Material Adverse Effect, the Seller and the Canadian Subsidiary are in compliance with all applicable laws respecting employment, employment practices and standards, terms and conditions of employment, wages and hours, overtime pay, occupational health and safety, human rights and equal opportunity, immigration, labor relations, pay equity and workers compensation with respect to the Seller Business Employees and with laws and regulations regarding tax and benefits treatment applicable to independent contractors, consultants, temporary and leased employees.

### 2.13 Employee Benefits.

(a) Section 2.13 of the Disclosure Schedule lists, as of the date of this Agreement, all Employee Benefit Plans that are Business Benefit Plans (other than any Business Benefit Plan that relates solely to Seller Business Employees who are not Scheduled Employees).

(b) The Seller has provided the Buyer with true, complete and correct copies of plan documents, written summaries and other such summary documentation as is reasonably necessary for the Buyer to fulfill its obligations under <u>ARTICLE IX</u>, including, without limitation, to the extent applicable, the most recent summary plan description and all summaries of material modifications of each Business Benefit Plan.

(c) Except as applicable solely to a Seller Business Employee who is not a Scheduled Employee or as set forth in <u>Section 2.13 of the Disclosure</u> <u>Schedule</u>, neither the execution of this Agreement nor consummation of the transactions contemplated by this Agreement will accelerate the time of vesting or the time of payment, or increase the amount, of compensation due to any Seller Business Employee under any Business Benefit Plan.
(d) There are no circumstances under which it would be reasonably likely that the Buyer or any Affiliate thereof would be subject to liability under Section 4069 or 4212 of ERISA or otherwise under Title IV of ERISA (or comparable provisions of Canadian law relating to defined benefit plans) with respect to any Employee Benefit Plan.

(e) Except to the extent any such Action would not reasonably be expected to result in any liability to the Buyer or its Affiliates, with respect to each Business Benefit Plan, no Action is pending or, to the knowledge of the Seller, threatened other than claims for benefits in the ordinary course.

2.14 Legal Compliance. Except for laws relating to Taxes, employment matters, and employee benefits, which shall be governed exclusively by <u>Sections</u> 2.5, 2.12 and 2.13, respectively, with respect to the Business, each of the Seller and the Canadian Subsidiary is in compliance in all material respects with all applicable laws (including rules and regulations thereunder) of any federal, state, provincial or foreign government, currently in effect with respect to the Business. Neither the Seller nor the Canadian Subsidiary has received written notice of any pending Action, hearing, investigation or demand relating to the Business alleging any failure to so materially comply.

2.15 <u>Permits</u>. <u>Section 2.15 of the Disclosure Schedule</u> lists, as of the date of this Agreement, all Permits. As of the date of this Agreement, (a) each Permit listed in <u>Section 2.15 of the Disclosure Schedule</u> is in full force and effect and the Seller is not in material violation of or default under any Permit and (b) no suspension or cancellation of any such Permit is currently threatened in writing.

2.16 <u>Business Relationships with Affiliates</u>. <u>Section 2.16 of the Disclosure Schedule</u> lists those agreements with respect to the Business to which the Seller or any of its Affiliates, on the one hand, and the Buyer or any of its Affiliates, on the other hand, will be party after the Closing.

2.17 <u>Registration</u>. The Seller is duly registered as an investment adviser under the Investment Advisers Act of 1940. The Seller is duly registered, licensed or qualified as in investment adviser in each state or any other jurisdiction where the conduct of the Business requires such registration, licensing or qualification, and the Seller is in good standing under the rules and regulations thereof, except where the failure to be so registered, licensed or qualified or to be in good standing would not reasonably be likely to result in a Business Material Adverse Effect. With respect to the Business, (i) the Seller is not (taking into account any applicable exemption) ineligible pursuant to applicable laws to act as an investment adviser, and (ii) to the knowledge of Seller, no employee of the Seller or other person associated with the Seller who has performed any functions of an investment adviser representative or a person associated with an investment adviser. With respect to the Business, there is no proceeding or investigation pending and served on the Seller or, to the knowledge of the Seller, pending and not so served or

threatened by any Governmental Entity, which would result in (A) the ineligibility under such laws of the Seller to act as an investment adviser or (B) the ineligibility under such applicable laws of such investment adviser representative or a person associated with the Seller to serve as an investment adviser representative or a person associated with an investment adviser.

2.18 Participating Sponsor Revenue. Section 2.18 of the Disclosure Schedule sets forth, with respect to each Participating Sponsor, as of the Base Date:

(a) the name of such Participating Sponsor;

(b) the Assets Under Administration of such Participating Sponsor for each program type, including Separately Managed Accounts, Mutual Fund Wrap and Unified Managed Account;

(c) the annualized Fee rates payable to the Seller by such Participating Sponsor under the applicable Sponsor Contract for each program type, including Separately Managed Accounts, Mutual Fund Wrap and Unified Managed Account;

(d) the Revenue Run-Rate in respect of each program type, including Separately Managed Accounts, Mutual Fund Wrap and Unified Managed Account, for each Participating Sponsor; and

(e) a detailed schedule of the actual invoiced amounts, together with related invoices and proof of payment of such invoices, for the quarterly billings for the quarters ended December 31, 2012 and March 31, 2013.

Section 2.18 of the Disclosure Schedule also sets forth the aggregate amount representing the Revenue Run-Rate as of the Base Date for all program types for all Participating Sponsors (such aggregate amount, the "<u>Base Revenue Run-Rate</u>"). As of the date of this Agreement, since June 1, 2012, other than as disclosed in <u>Section 2.18 of the Disclosure Schedule</u>, (A) with respect to the Business, no Participating Sponsor has canceled any contract or agreement with the Seller or indicated in writing that it intended to cease business with the Seller or to materially reduce such business and (B) the Business has not experienced a material net outflow of assets under administration by Participating Clients. There are no (and since June 1, 2012 there have not been any) material disputes between the Seller or the Canadian Subsidiary, on the one hand, and any Participating Sponsor, on the other hand.

2.19 <u>Absence of Certain Changes</u>. Except as set forth in <u>Section 2.19 of the Disclosure Schedule</u>, since December 31, 2012 through the date hereof, the Seller has, in all material respects, carried on the Business in the ordinary course consistent with its past practices.

2.20 <u>Financial Capability</u>. The Seller has, and at the Closing will have, sufficient funds in order to consummate the transactions contemplated by this Agreement and to perform its obligations hereunder and under the Ancillary Agreements, including any amounts that may be required to be paid to Buyer hereunder and thereunder. Immediately after giving effect to the transactions contemplated by this Agreement, the Seller and its subsidiaries, on a consolidated basis shall be able to pay its debts as they become due.

# ARTICLE III REPRESENTATIONS AND WARRANTIES OF THE BUYER

The Buyer represents and warrants to the Seller that the statements contained in this ARTICLE III are true and correct as of the date hereof.

3.1 Organization. The Buyer is a corporation duly organized, validly existing and in good standing under the laws of the State of Delaware.

3.2 <u>Authority</u>. The Buyer has all requisite corporate power and authority to execute and deliver this Agreement and the Ancillary Agreements to which it will be a party and to perform its obligations hereunder and thereunder. The execution and delivery by the Buyer of this Agreement and such Ancillary Agreements and the consummation by the Buyer of the transactions contemplated hereby and thereby have been validly authorized by all necessary corporate action on the part of the Buyer. This Agreement has been, and such Ancillary Agreements will be, validly executed and delivered by the Buyer and, assuming this Agreement and each such Ancillary Agreement constitute the valid and binding obligation of the Seller, constitutes or will constitute a valid and binding obligation of the Buyer, enforceable against the Buyer in accordance with its terms, except as enforceability may be limited by bankruptcy, insolvency, fraudulent transfer, reorganization, moratorium or other similar laws relating to or affecting the rights of creditors generally and by equitable principles, including those limiting the availability of specific performance, injunctive relief and other equitable remedies and those providing for equitable defenses.

3.3 <u>Noncontravention</u>. Neither the execution and delivery by the Buyer of this Agreement or the Ancillary Agreements to which the Buyer will be a party, nor the consummation by the Buyer of the transactions contemplated hereby or thereby, will:

(a) conflict with or violate any provision of the charter or bylaws of the Buyer;

(b) require on the part of the Buyer any filing with, or any permit, authorization, consent or approval of, any (i) Governmental Entity or (ii) other Person, except in the case of this <u>clause (ii)</u>, for any such filing, permit, authorization, consent or approval which if not obtained or made would not reasonably be expected to result in a Buyer Material Adverse Effect;

(c) conflict with, result in a breach of, constitute (with or without due notice or lapse of time or both) a default under, result in the acceleration of obligations under, create in any party the right to terminate or modify, or require any notice, consent or waiver under, any contract or agreement to which the Buyer is a party or by which the Buyer is bound, except for (x) any conflict, breach, default, acceleration or right to terminate or modify that would not reasonably be expected to result in a Buyer Material Adverse Effect or (y) any notice, consent or waiver the absence of which would not reasonably be expected to result in a Buyer Material Adverse Effect; or

(d) violate any Order specifically naming, or statute, rule or regulation applicable to, the Buyer or any of its properties or assets, except for any such violation that would not reasonably be expected to result in a Buyer Material Adverse Effect.

3.4 <u>Litigation</u>. There are no Actions pending against, or, to the Buyer's knowledge, threatened against, the Buyer or any of its Affiliates which would be reasonably likely to materially impair or delay the Buyer's ability to perform its obligations under this Agreement and the Ancillary Agreements to which it will be a party or the consummation of the transactions contemplated by this Agreement.

3.5 <u>Financial Capability</u>. The Buyer has, and at the Closing will have, sufficient funds in order to consummate the transactions contemplated by this Agreement and to perform its obligations hereunder and under the Ancillary Agreements, including payment to the Seller of the Closing Date Adjusted Purchase Price at the Closing and any additional amounts that may be required to be paid pursuant to <u>Sections 1.2</u> and <u>1.4</u>. Immediately after giving effect to the transactions contemplated by this Agreement, the Buyer and the Buyer Subsidiaries, on a consolidated basis shall be able to pay their debts as they become due.

3.6 <u>Registration</u>. Envestnet Asset Management Inc., a subsidiary of the Buyer, is duly registered as an investment adviser under the Investment Advisers Act of 1940. Such subsidiary is duly registered, licensed or qualified as in investment adviser in each state or any other jurisdiction where the conduct of its business requires such registration, licensing or qualification, except where the failure to be so registered, licensed or qualified would not reasonably be likely to result in a Buyer Material Adverse Effect.

3.7 Due Diligence by the Buyer. The Buyer acknowledges that none of the Seller, its Affiliates or Representatives, or any other Person, has made, and the Buyer has not relied upon, any representation or warranty whatsoever, express or implied (including any implied warranty or representation as to the value, condition, merchantability or suitability as to any of the Acquired Assets), regarding the Business or other matters beyond those expressly given in <u>ARTICLE II</u> of this Agreement. Without limiting the generality of the foregoing, the Buyer hereby acknowledges that (i) none of the Seller, its Affiliates or Representatives, or any other Person, has made a representation or warranty to the Buyer with respect to (A) any estimates, forecasts, projections, or other predictions for the Business or business plan information of the Business or (B) any materials, documents or information relating to the Business or Acquired Assets, whether written or oral, made available to the Buyer or any of its Affiliates or Representatives in the confidential information memorandum, provided to the Buyer on January 30, 2013, or in any data room, presentation by management of the Business, due diligence discussion with Representatives of the Seller or Seller Business Employees or otherwise, except as expressly covered by a representation or warranty set forth in <u>ARTICLE II</u>, and (ii) the financial statements referred to in <u>Section 2.4(a)</u> and the Required Financial Statements have been and will have been prepared using different methodologies, as more fully described in <u>Section 2.4 of the Disclosure Schedule</u> and that the Seller disclaims any and all liability for any difference in the information contained in such financial statements to the extent that such difference in information relates to the difference in methodologies (in and of themselves, and not as any such difference relates to any error in the application of such methodologies or any error in the financial information therein contained) used to prepare such finan

The Buyer will not assert, except as provided in <u>ARTICLE VI</u>, any claim against the Seller or any of its Representatives or seek to hold the Seller (except as provided in <u>ARTICLE VI</u>) or any of its Representatives liable, for any inaccuracies, misstatements or omissions with respect to information furnished by the Seller or its Representatives concerning the Business, the Acquired Assets, this Agreement or the transactions contemplated hereby.

3.8 <u>Disclaimer of Warranties</u>. Notwithstanding anything herein to the contrary, it is the explicit intent of the Parties hereto that the Buyer is not making, nor has it made, any representation or warranty whatsoever, express or implied, beyond those expressly given in this <u>ARTICLE III</u>.

## ARTICLE IV PRE-CLOSING COVENANTS

4.1 Closing Efforts. Subject to the terms hereof, including Section 4.5, each of the Parties shall use commercially reasonable efforts to take all actions and to do all things reasonably necessary or advisable to consummate the transactions contemplated by this Agreement, including using commercially reasonable efforts to: (i) satisfy the conditions precedent to the obligations of either Party, (ii) pursuant to and consistent with the terms of Section 4.4, satisfy the applicable Assignment Requirements, if any, for each Assigned Contract, (iii) obtain all consents, licenses, permits, waivers, approvals, authorizations from any Governmental Entity required to be obtained by a Party in connection with the consummation of the transactions contemplated by this Agreement, (iv) otherwise comply in all material respects with all applicable laws and regulations in connection with the consummation of the transactions contemplated by this Agreement, (v) cause to be lifted or rescinded any Order adversely affecting the ability of the Parties to consummate the Closing and (vi) defend any Actions challenging this Agreement or any of the Ancillary Agreements or the consummation of the transactions contemplated hereby or thereby or the performance of such Party's obligations hereunder or thereunder. In furtherance of and not in limitation of the foregoing, in the event that a material development with respect to the Business occurs between the date of this Agreement and the Closing as a result of the execution, announcement or pendency of this Agreement, which development (A) relates to the identity of, or the effect of any facts or circumstances relating to, the Buyer and (B) has had an adverse impact on the Business, the Parties shall collaborate in good faith to identify any such development and mitigate the further impact of such development on the Business. In addition, neither Party shall take any action after the date hereof that could reasonably be expected to delay the satisfaction of, or result in not satisfying, (1) the conditions set forth in ARTICLE V or (2) the applicable Assignment Requirements with respect to any Assigned Contract. Each of the Parties shall promptly notify the other Party of any fact, condition or event known to it that would reasonably be expected to prohibit, make unlawful or delay the consummation of the transactions contemplated by this Agreement. The failure of any Assignment Requirement to be satisfied with respect to any Assigned Contract shall not constitute a failure to satisfy any condition set forth in Section 5.2 or relieve the Buyer from its obligation to consummate the transactions contemplated by this Agreement and the Ancillary Agreements.

#### 4.2 Operation of the Business.

(a) Except (i) as contemplated by this Agreement including the consent process set forth in <u>Section 4.4</u>, (ii) as required by applicable law, (iii) as provided in Employee Benefit Plans in effect as of the date hereof, (iv) as set forth in <u>Section 4.2 of the Disclosure Schedule</u>, or (v) as consented to in writing by the Buyer, during the period from the date of this Agreement until the earlier of the Closing Date or proper termination of this Agreement, the Seller shall conduct the operations of the Business in all material respects in the ordinary course, and without limiting the generality of the foregoing, the Seller shall not, and shall cause the Canadian Subsidiary not to, with respect to the Business and without written consent provided by the Buyer (which consent may be withheld in Buyer's sole discretion):

(i) sell, assign or transfer any Sponsor Contract, Manager Contract or Acquired Asset (notwithstanding the foregoing, it is understood and agreed that the Seller shall be permitted to replace obsolete or worn-out Equipment in the ordinary course of business and to make updates to <u>Schedule</u> <u>1.1(a)(iii)</u> to reflect such replacement);

(ii) knowingly consent to the imposition of any Lien (other than a Permitted Lien) upon any of the Sponsor Contracts, Manager Contracts or Acquired Assets;

(iii) enter into any contract similar to a Manager Contract unless such contract (A) allows the Seller to terminate at will upon thirty (30) days' or less notice, (B) does not require any obligations to be performed by Seller after the date that is six (6) months after any termination notice is delivered thereunder and (C) includes a consent from the counterparty thereto that allows the Seller to assign such contract to the Buyer in connection with the consummation of the Closing;

(iv) enter into any contract similar to a Sponsor Contract unless such contract (A) allows the Seller to terminate at will upon thirty (30) days' or less notice, (B) does not require any obligations to be performed by Seller after the date that is six (6) months after any termination notice is delivered thereunder and (C) includes a consent from the counterparty thereto that allows the Seller to assign such contract to the Buyer in connection with the consummation of the Closing;

(v) make any change in its accounting systems, policies, principles, practices or methods, in each case, that would impact the method for calculating the Revenue Run-Rate for the Participating Sponsors;

(vi) with respect to the Acquired Assets or the Business, (A) make, revoke or amend any Tax election (outside the ordinary course of business), (B) compromise any claim, investigation, audit or controversy related to any material amount of Taxes, (C) consent to or execute any waiver of restrictions on a Tax claim or assessment, (D) enter into or amend any Tax agreement with a Tax authority or (E) file any amended Tax Return or claim for a refund of material Taxes, in the case of each of the foregoing <u>subclauses (A)</u> through (E), in a manner that would reasonably be expect to increase the Buyer's Tax liability with respect to the Acquired Assets or the Business following the Closing;

(vii) grant any rights to severance benefits, "stay-pay" or termination pay to any Scheduled Employee or increase the compensation or other benefits payable or potentially payable to any Scheduled Employee, in each case, other than grants or increases that are required by applicable law or the terms of any existing contract or Employee Benefit Plan as in effect as of the date hereof (or as amended in a manner also applicable generally to similarly situated employees of Seller and/or its ERISA Affiliates who are not Scheduled Employees);

(viii) enter into any employment, compensation or deferred compensation agreement (or any amendment to any such existing agreement) with any Scheduled Employee whose annual rate of base compensation exceeds \$125,000;

(ix) (A) accelerate, terminate, modify, amend, waive or otherwise alter or change any of the terms or provisions of any Assigned Sponsor Contract or pay any material amount in connection therewith not required by law or by any such contract or (B) accelerate, terminate, modify, amend, waive or otherwise alter or change any of the terms or provisions of any Assigned Manager Contract outside of the ordinary course of business or pay any material amount in connection therewith not required by law or by any such contract, in the case of <u>clauses (A)</u> and <u>(B)</u>, other than as contemplated hereby in connection with obtaining the required consents for the assignment thereof; or

## (x) agree in writing to take any of the foregoing actions.

(b) During the period from the date of this Agreement until the earlier of the Closing Date or proper termination of this Agreement, (i) in the event that the Seller, with respect to the Business, (A) waives, releases or cancels any material claims against a Participating Sponsor or Participating Manager or debts owing to it by a Participating Sponsor or Participating Manager, (B) pays any material amount or agrees to pay any material amount or perform any obligation in settlement or compromise of any Action or claims of liability against the Seller or any of its directors, officers, employees or agents or (C) agrees in writing to take any of the actions described in the foregoing <u>clauses (A)</u> and (B) or (ii) (A) a Participating Sponsor cancels a Sponsor Contract or indicates in writing that it intends to cease business with the Seller or to materially reduce such business or (B) the Business experiences a material net outflow of assets under administration by Participating Clients, in the case of the foregoing <u>clauses (i)</u> and <u>(ii)</u>, the Seller shall provide prompt (and in any event, within five (5) Business Days) notice of the occurrence of such event to the Buyer.

(c) In the event that, prior to the Closing, the Seller enters into a contract similar to a Manager Contract or a contract similar to a Sponsor Contract, as contemplated by <u>clauses (iii)</u> or <u>(iv)</u> of <u>Section 4.2(a)</u>, the Seller shall provide prompt (and in any event, within five (5) Business Days) notice of its entry into such contract to the Buyer, together with a copy of any such contract. The Buyer shall have an opportunity to review such contract and object to such contract becoming an Assigned Manager Contract or Assigned Sponsor Contract, as applicable, for purposes hereof. If the Buyer does not notify the Seller within ten (10) Business Days of receiving a copy thereof that Buyer agrees that such contract shall be an Assigned Manager Contract or an Assigned Sponsor Contract, such contract shall be deemed to be rejected by the Buyer and shall, in the case of a contract similar to a Manager Contract, be terminated at the Closing as contemplated by <u>Section 5.2(h)</u> or, in the case of a contract similar to a Sponsor Contract, be terminated at the Closing as contemplated by <u>Section 5.2(h)</u> or become subject to the MAPSA Subcontracting Agreement, as applicable, to the extent the Seller has any remaining obligations under such contract following the Closing.

#### 4.3 Access.

(a) The Seller shall permit the Representatives of the Buyer to have access (at reasonable times during normal business hours, on reasonable prior written notice and in a manner so as not to interfere with the conduct of the Seller's and its Affiliates' business operations) to the premises, properties, financial and accounting records, contracts, and other records and documents, of or pertaining to the Business; <u>provided</u>, <u>however</u>, that any such access shall be conducted at the Buyer's sole expense. Notwithstanding the foregoing, the Seller shall not be obligated (i) to provide any information, documents or access to any Person unless the Buyer is responsible, pursuant to the terms of the Confidentiality Agreement, for the use and disclosure of any information obtained by such Person from the Seller, (ii) to provide any information, documents or access (A) that would violate the provisions of any applicable laws or regulations or any agreement to which it is a party or, in the case of information or documents, the sharing of which would legally require the consent of any third party (including any employee of the Seller or any of its Affiliates), (B) that would cause the loss of a claim of privilege (including attorney-client and work product privileges) with respect thereto or (C) that would result in the disclosure of any information of the Seller or any of its Affiliates unrelated to the Business (and the Seller shall be permitted to redact any such information from any materials provided to the Buyer or its Representatives) or (iii) to provide any consolidated, combined, affiliated or unitary Tax Return that includes the Seller or any of its Affiliates or any Tax-related work papers. Prior to the Closing, the Buyer and its Representatives shall not contact or communicate with the employees, customers or suppliers of or other third parties having business relationships with the Seller in connection with the transactions contemplated by this Agreement, except with the prior written consent of the Seller.

(b) The Buyer and the Seller acknowledge and agree that the Confidentiality Agreement remains in full force and effect and that information provided by the Seller or any of its Representatives to or on behalf of the Buyer pursuant to this Agreement prior to the Closing shall be treated in accordance with the Confidentiality Agreement. If this Agreement is terminated prior to the Closing, the Confidentiality Agreement shall remain in full force and effect in accordance with its terms. If the Closing occurs, the Confidentiality Agreement shall terminate effective as of the Closing.

(c) Notwithstanding any provision of this Agreement to the contrary, the Buyer and its Representatives shall not have any access at any time prior to the Closing to any information regarding pending or proposed bids for new contracts or any related information where the Buyer or any of its Affiliates also has submitted or intends to submit a bid to provide the same or substantially similar services.

## 4.4 Sponsor and Manager Contract Consents.

(a) Subject to this <u>Section 4.4</u>, the Seller shall use commercially reasonable efforts to obtain the consents necessary to satisfy prior to the Closing Date the Assignment Requirements with respect to each Sponsor Contract and Manager Contract. The Buyer shall use commercially reasonable efforts to cooperate with the Seller in its efforts to satisfy the Assignment Requirements with respect to each such contract.

(b) Each of the Parties shall provide promptly in writing all information concerning itself and its Affiliates required to be included in the materials contemplated by this <u>Section 4.4</u>. Each of the Parties shall promptly correct such information if and to the extent that such information becomes false or misleading in any material respect.

(c) Notwithstanding anything herein to the contrary, none of the Seller, the Canadian Subsidiary, the Buyer or any of their respective Affiliates shall have any obligation under this Agreement to make any payments or agree to any material undertakings in connection with the assignment of any Sponsor Contract or Manager Contract or to initiate any Action against any counterparty thereto in order to obtain any consent necessary to satisfy any Assignment Requirement.

(d) With respect to each counterparty to a Sponsor Contract or Manager Contract (other than Pruco), as promptly as practicable following the date of this Agreement and to the extent required by applicable law or the terms of the applicable contract, the Seller shall (i) provide notice of the transactions contemplated by this Agreement to such counterparty (it being understood that any such notice to a Participating Sponsor or Participating Manager shall be in substantially the form set forth as <u>Exhibit H-1</u> and <u>Exhibit H-2</u>, respectively) and (ii) use commercially reasonable efforts to obtain any consents or approvals necessary to satisfy the applicable Assignment Requirements (it being understood that the countersignature by a Participating Sponsor or, subject to <u>subsection</u> (c) below, a Participating Manager of the notice referred to in <u>clause (i)</u> above shall be sufficient for purposes of this <u>clause (ii)</u>).

(e) Notwithstanding <u>clause (ii)</u> of <u>subsection (d)</u> above, with respect to each Participating Manager that has not responded in writing within fifteen (15) days of the date the initial notice referred to in <u>clause (i)</u> of <u>subsection (d)</u> above was sent, the Seller shall provide a second notice to such Participating Manager with respect to such Participating Manager's Manager Contract (other than any Canadian Manager Contract, the assignment of which shall be subject to <u>clause (ii)</u> of <u>subsection (d)</u> above) in substantially the form set forth as <u>Exhibit H-3</u> (such notice a "<u>Negative Consent Notice</u>") requesting written consent as aforesaid and informing such Participating Manager, inter alia, that the consent of such Participating Manager will be deemed to have been granted if such Participating Manager continues to provide services through the Business for a period of at least thirty (30) days after the date the Negative Consent Notice was sent without otherwise terminating the applicable Manager Contract; <u>provided</u> that no consent shall be deemed to have been given for any purpose under this Agreement if at any time prior to the Closing a Participating Manager terminates or notifies the Seller in writing that such Participating Manager has not consented to the transfer of, or is terminating, its Manager Contract (any such Participating Manager, a "<u>Terminating Manager</u>").

(f) On or before the Closing Date, the Seller shall (1) provide each Participating Sponsor with a list of the Terminating Managers and Non-Consenting Canadian Managers pertaining to such Participating Sponsor; (2) with respect to each Terminating Manager and Non-Consenting Canadian Managers, to the extent any such Terminating Manager

or Non-Consenting Canadian Manager has not previously terminated each of its Manager Contracts (other than, in the case of a Non-Consenting Canadian Manager, any Manager Contract to which such Non-Consenting Manager has consented to (or is deemed to have consented to) the assignment of), give a notice of termination to such Terminating Manager or Non-Consenting Canadian Manager with respect to each such Manager Contract (with the effective date of such termination to be Closing Date); and (3) provide the Buyer will a list of Non-Consenting Sponsors and Terminating Sponsors, as applicable, so that the Buyer and the Seller can execute and deliver the MAPSA Subcontracting Agreement at the Closing pursuant to <u>Section 1.4(c)(viii)</u>. Following the terminations contemplated by <u>clause (2)</u> of the first sentence of this <u>Section 4.4(f)</u> and a reasonable period of good-faith consultation with the Seller, the Buyer shall provide to each affected Participating Sponsor replacement recommendations for the applicable Terminating Managers and Non-Consenting Canadian Managers.

#### 4.5 Financial Statements.

(a) <u>Requested Relief</u>. Promptly, and in any event, within ten (10) days of the date hereof, the Buyer shall prepare and submit to the Chief Accountant of the Division of Corporation Finance of the U.S. Securities and Exchange Commission (the "<u>SEC</u>") a request ("<u>Buyer's Request Letter</u>") for a letter from the SEC staff expressing no objection to the Buyer's use of the Abbreviated Financial Statements for purposes of Rule 3-05 of Regulation S-X (such letter expressing no objection, the "<u>Requested Relief</u>"). The Buyer shall provide the Seller and its designated Representatives with a reasonable opportunity to comment on the Buyer's Request Letter and shall consider in good faith all comments provided to the Buyer in connection with such review. Each of the Parties shall (i) use commercially reasonable efforts to obtain (and in the case of the Seller, to assist the Buyer in obtaining) the Requested Relief and (ii) make any further information submissions pursuant to the SEC's consideration of the Buyer's Request Letter that may be reasonably necessary or advisable. Each of the Parties shall use commercially reasonable efforts to resolve any objections that may be asserted by the SEC with respect to the Requested Relief; <u>provided</u> that neither Party shall be required to initiate any Action in pursuit of the Requested Relief; and <u>provided further</u> that Buyer shall not be required to agree to any condition to the receipt of the Requested Relief that would reasonably be expected to significantly increase Buyer's costs of preparing its public SEC reports. Each of the Parties shall promptly inform each other of any material communication received by such Party from the SEC regarding the Requested Relief (unless the provision of such information would (i) violate the provisions of any applicable laws or regulations or any confidentiality agreement or (ii) cause the loss of the attorney-client privilege with respect thereto).

(b) <u>Required Financial Statements</u>. With respect to the Seller's obligation to deliver the Required Financial Statements and Specified Quarterly Financial Statements as contemplated by <u>Section 5.2(g)</u>, the Seller shall use commercially reasonable efforts to prepare, in accordance with the applicable methodologies set forth in <u>Section 2.4(b) of the Disclosure Schedule</u>, and deliver the Abbreviated Financial Statements and the Specified Quarterly Financial Statements as soon as is reasonably practicable; <u>provided</u> that, if, despite the Parties commercially reasonable efforts pursuant to <u>Section 4.5(a)</u> above, the SEC has not provided the Requested Relief by the date that is fifteen (15) Business Days from the date the Buyer submitted the Buyer's Request Letter to the SEC, the Seller shall use commercially reasonable efforts to prepare, in accordance with the applicable methodologies set forth in <u>Section 2.4(b) of the</u>

<u>Disclosure Schedule</u>, and deliver the Full Financial Statements and the Specified Quarterly Financial Statements as soon as is reasonably practicable, it being understood, however, that in the event that the Requested Relief is provided by the SEC, the covenant set forth in this <u>Section 4.5(b)</u> shall apply only with respect to the Abbreviated Financial Statements and the Specified Quarterly Financial Statements; and <u>provided further</u> that if, despite Seller's commercially reasonable efforts, the Seller is unable to prepare and deliver the Required Financial Statements and the Specified Quarterly Financial Statements by the Outside Date, the Seller shall have no further obligation or liability under this <u>Section 4.5</u> (it being understood that the foregoing proviso shall not affect the closing condition set forth in <u>Section 5.2(g)</u>).

(c) In addition, prior to the Closing and, if not completed by the Closing, promptly thereafter, the Seller shall use commercially reasonable efforts to obtain reviews prepared by the Seller's independent auditor in respect of such independent auditor's performance of the procedures specified by the PCAOB for a review of interim financial information as described in AU 722 with respect to the Specified Quarterly Financial Statements.

(d) (i) In the event that the Closing occurs on or prior to June 30, 2013, the Seller shall prepare and deliver to the Buyer, on or prior to August 15, 2013, unaudited financial statements of the same type as those constituting the Specified Quarterly Financial Statements with respect to the Business, as of, and for the period beginning on April 1, 2013 and ending on, the Closing Date, together with associated notes to such unaudited financial statements.

(ii) In the event that the Closing occurs on or after July 1, 2013 and prior to August 15, 2013, the Seller shall prepare and deliver to the Buyer, on or prior to September 30, 2013, unaudited financial statements of the same type as those constituting the Specified Quarterly Financial Statements with respect to the Business, as of, and for the quarter ended June 30, 2013, together with associated notes to such unaudited financial statements.

(iii) In the event that the Closing occurs on or after July 1, 2013, the Seller shall prepare and deliver to the Buyer, on or prior to the later of (A) September 30, 2013 and (B) the date that is forty-five (45) days following the Closing, unaudited financial statements of the same type as those constituting the Specified Quarterly Financial Statements with respect to the Business, as of, and for the period beginning on July 1, 2013 and ending on, the Closing Date, together with associated notes to such unaudited financial statements.

The Seller shall use commercially reasonable efforts to obtain reviews prepared by the Seller's independent auditor in respect of such independent auditor's performance of the procedures specified by the PCAOB for a review of interim financial information as described in AU 722 with respect to the applicable financial statements delivered pursuant to the immediately preceding <u>clauses (i), (ii)</u> or <u>(iii)</u>.

# ARTICLE V CONDITIONS PRECEDENT TO CLOSING

5.1 <u>Conditions to Each Party's Obligations</u>. The respective obligations of each Party to consummate the transactions to be consummated at the Closing are subject to the satisfaction (or waiver by both Parties) of the following conditions:

(a) no law, rule or regulation shall have been enacted by a Governmental Entity that prohibits the consummation of the transactions contemplated by this Agreement, and no Order enjoining or preventing the consummation of the transactions contemplated by this Agreement shall be in effect;

(b) the Closing Run-Rate Percentage shall be equal to or greater than the percentage set forth in <u>Schedule 11.1</u>; and

(c) to the extent any Negative Consent Notices are sent to Participating Managers, as contemplated by <u>Section 4.4(e)</u>, at least seven (7) Business Days shall have elapsed following the passage of the thirty- (30-) day period referred to in <u>Section 4.4(e)</u>.

5.2 <u>Conditions to Obligations of the Buyer</u>. The obligation of the Buyer to consummate the transactions to be consummated at the Closing is subject to the satisfaction (or waiver by the Buyer) of the following conditions:

(a) (i) the representations and warranties of the Seller set forth in <u>ARTICLE II</u> (other than the representations and warranties set forth in <u>Sections</u> 2.1, 2.2 and 2.10) shall be true and correct as of the date of this Agreement and as of the Closing Date as if made as of the Closing Date, except (i) for changes permitted by <u>Section 4.2</u>, (ii) for those representations and warranties that address matters only as of a particular date (which shall be true and correct as of such date, subject to <u>clause (iii)</u> below) and (iii) where the failure of such representations and warranties to be true and correct would not reasonably be expected to result, in the aggregate, in a Business Material Adverse Effect (it being agreed that any materiality or Business Material Adverse Effect qualification in a representation and warranty shall be disregarded in determining whether any such failure would reasonably be expected to result in a Business Material Adverse Effect for purposes of this <u>clause (iii)</u>);

(b) the representations and warranties of the Seller set forth in <u>Sections 2.1</u>, <u>2.2</u> and <u>2.10</u> shall be true and correct in all respects as of the date of this Agreement and as of the Closing Date as if made as of the Closing Date, except for changes contemplated or permitted by this Agreement;

(c) the Seller shall have performed or complied in all material respects with the agreements and covenants required to be performed or complied with by it under this Agreement as of or prior to the Closing;

(d) the Buyer shall have received the Seller Certificate;

(e) the Buyer shall have received from Pruco an executed copy of the Pruco Contract;

(f) since the date of the Agreement, there shall not have occurred any Business Material Adverse Effect;

(g) the Buyer shall have received the Required Financial Statements and the Specified Quarterly Financial Statements;

(h) on or prior to the Closing Date, the Seller shall have sent a termination notice under (A) each contract which has been deemed rejected by Buyer pursuant to <u>Section 4.2(c)</u> and thereafter not agreed in writing by Buyer to be an Assigned Sponsor Contract or Assigned Manager Contract and (B) each Sponsor Contract and each Manager Contract for which the applicable contract counterparty has not provided (or has not been deemed to have provided) its consent to the assignment of its contract pursuant to <u>Section 4.4</u> prior to the date that is three (3) Business Days prior to the Closing (and any such termination notice given in respect of a Sponsor Contract or a contract similar to a Sponsor Contract shall include a notification by the Seller that certain non-advisory services following the Closing shall be performed on behalf of the Seller by the Buyer); and

(i) (A) the consent set forth in <u>Schedule 5.2(i)</u> shall have been received, and evidence thereof shall have been delivered to the Buyer prior to the Closing or (B) the Seller shall have provided the Buyer with a reasonably suitable alternative to the Gateway Sublease Agreement upon substantially the same or better economic terms, with the location of such alternative space being no more than ten (10) miles from the location of the Leased Real Property.

5.3 <u>Conditions to Obligations of the Seller</u>. The obligation of the Seller to consummate the transactions to be consummated at the Closing is subject to the satisfaction (or waiver by the Seller) of the following conditions:

(a) the representations and warranties of the Buyer set forth in <u>ARTICLE III</u> (other than the Buyer Fundamental Representations) shall be true and correct as of the date of this Agreement and as of the Closing Date as if made as of the Closing Date, except (i) for those representations and warranties that address matters only as of a particular date (which shall be true and correct as of such date, subject to <u>clause (ii)</u> below) and (ii) where the failure of such representations and warranties to be true and correct would not reasonably be expected to result, in the aggregate, in a Buyer Material Adverse Effect (it being agreed that any materiality or Buyer Material Adverse Effect qualification in a representation and warranty shall be disregarded in determining whether any such failure would reasonably be expected to result in a Buyer Material Adverse Effect for purposes of this <u>clause (ii)</u>);

(b) the Buyer Fundamental Representations shall be true and correct in all respects as of the date of this Agreement and as of the Closing Date as if made as of the Closing Date, except for changes contemplated or permitted by this Agreement;

(c) the Buyer shall have performed or complied in all material respects with its agreements and covenants required to be performed or complied with by it under this Agreement as of or prior to the Closing; and

(d) the Seller shall have received the Buyer Certificate; and

(e) Pruco shall have received from the Buyer an executed copy of the Pruco Contract.

# ARTICLE VI INDEMNIFICATION

6.1 <u>Indemnification by the Seller</u>. Subject to the terms and conditions of this <u>ARTICLE VI</u>, from and after the Closing, the Seller shall indemnify the Buyer in respect of, and hold the Buyer harmless against, all Damages incurred or suffered by the Buyer or any Affiliate thereof resulting from:

(a) any breach of a representation or warranty of the Seller contained in <u>ARTICLE II</u> of this Agreement or of a representation or warranty of the Seller Affiliate contained in the Affiliate Bill of Sale;

(b) any failure by the Seller to perform any covenant or agreement contained in this Agreement;

(c) any Excluded Liabilities; or

(d) any claim brought by a Seller Business Employee, a successor to the rights of a Seller Business Employee or a Governmental Authority (i) relating to such Seller Business Employee not being a Scheduled Employee or due to Buyer's failure to offer employment to a Seller Business Employee or (ii) arising from or related to the Buyer's determination of which Seller Business Employees are to be Scheduled Employees, Transitional New Buyer Employees and Non-Transitional New Buyer Employees, in each case to the extent related to a charge of discrimination or disparate treatment by Buyer or by Seller (to the extent Seller was acting, or deemed to be acting, at the direction of Buyer).

6.2 Indemnification by the Buyer. Subject to the terms and conditions of this <u>ARTICLE VI</u> from and after the Closing, the Buyer shall indemnify the Seller in respect of, and hold the Seller harmless against, any and all Damages incurred or suffered by the Seller or any Affiliate thereof resulting from:

(a) any breach of a representation or warranty of the Buyer contained in ARTICLE III of this Agreement;

- (b) any failure by the Buyer to perform any covenant or agreement contained in this Agreement; or
- (c) any Assumed Liabilities.
- 6.3 Claims for Indemnification.

(a) <u>Third-Party Claims</u>. All claims for indemnification made under this Agreement resulting from, related to or arising out of a third-party claim against an Indemnified Party shall be made in accordance with the following procedures. An Indemnified Party shall give prompt (but in any event within thirty (30) days) written notification to the Indemnifying Party of the commencement of any Action relating to a third-party claim for which indemnification may be sought or, if earlier, upon the assertion of any such claim by a third party. Such notification shall include a description in reasonable detail (to the extent known by

the Indemnified Party) of the facts constituting the basis for such third-party claim and a good faith estimate of the amount of the Damages claimed, if then determinable. Notwithstanding the foregoing, the failure or delay in notifying the Indemnifying Party will not relieve the Indemnifying Party of any liability that it may have to any Indemnified Party, except to the extent that the defense of such Action is prejudiced by the Indemnifying Party's failure or delay in giving such notice. Within thirty (30) days after delivery of such notification, the Indemnifying Party may, upon written notice thereof to the Indemnified Party, assume control of the defense of such Action with counsel reasonably satisfactory to the Indemnified Party. If the Indemnifying Party does not assume control of such defense, the Indemnified Party shall control such defense. The Party not controlling such defense may participate therein at its own expense. The Party controlling such defense shall keep the other Party advised of the status of such Action and the defense thereof and shall consider recommendations made by the other Party with respect thereto, and the Parties shall use commercially reasonable efforts to avoid production of confidential information (consistent with applicable local law), and to cause all communications among employees, counsel and others representing either Party with respect to a thirdparty claim to be made so as to preserve any applicable attorney-client or work-product privileges. If the Indemnifying Party assumes the defense of an Action, the Indemnified Party (1) shall not agree to any settlement of such Action without the prior written consent of the Indemnifying Party if such settlement would result in the Indemnifying Party becoming liable for an indemnification claim hereunder and (2) shall agree to any settlement of such Action that the Indemnifying Party may recommend that by its terms (or pursuant to a binding commitment of the Indemnifying Party) obligates the Indemnifying Party to pay the full amount of any Damages payable in connection with such settlement; provided, however, that the sole relief provided is such Damages paid in full by the Indemnifying Party and such settlement would not result in (i) the imposition of a consent order, injunction or decree that would restrict the future activity or conduct of the Indemnified Party or any of its Affiliates or (ii) a finding or admission of a violation of law by the Indemnified Party.

(b) <u>Direct Claims</u>. An Indemnified Party wishing to assert a claim for indemnification under this <u>ARTICLE VI</u> (other than a claim for indemnification arising out of a third-party claim) shall deliver to the Indemnifying Party a Claim Notice. Notwithstanding the foregoing, the failure or delay in delivering a Claim Notice to the Indemnifying Party will not relieve the Indemnifying Party of any liability that it may have to any Indemnified Party, except to the extent that the defense of such claim is prejudiced by the Indemnifying Party's failure or delay in delivering such Claim Notice. Within thirty (30) days after delivery of a Claim Notice, the Indemnifying Party shall deliver to the Indemnified Party a written response in which the Indemnifying Party shall: (i) agree that the Indemnified Party is entitled to receive all of the Claimed Amount (in which case such response shall, subject to the limitations set forth in <u>Sections 6.5(a)(ii)</u> and <u>6.5(a)(ii)</u>, be accompanied by a payment by the Indemnifying Party to the Indemnified Party of the Claimed Amount, by check or by wire transfer), (ii) agree that the Indemnified Party is entitled to receive an Agreed Amount with respect to such claim for indemnifying Party to the Indemnified Party of the Agreed Amount, by check or by wire transfer), (ii) agree that the Indemnified Party is entitled to receive an Agreed Amount with respect to such claim for indemnifying Party to the Indemnified Party of the Agreed Amount, by check or by wire transfer) or (iii) contest that the Indemnified Party is entitled to receive any of the Claimed Amount, and the Indemnifying Party and the Indemnified Party disputes such response, the Indemnifying Party and the Indemnified Party.

shall use good faith efforts to resolve such dispute. If such dispute is not resolved within sixty (60) days following the delivery by the Indemnifying Party of such response, the Indemnifying Party and the Indemnified Party shall each have the right to submit such dispute to a court of competent jurisdiction in accordance with the provisions of <u>Section 12.12</u>.

#### 6.4 Survival.

(a) The representations and warranties of the Seller and the Buyer set forth in this Agreement shall survive the Closing and the consummation of the transactions contemplated hereby and continue until the second anniversary of the Closing Date, at which time they shall expire. Notwithstanding the foregoing, (i) the Seller Fundamental Representations and the Buyer Fundamental Representations shall survive the Closing and the consummation of the transactions contemplated hereby until the fifth anniversary of the Closing Date, at which time they shall expire and (ii) the representations and warranties of the Seller contained in <u>Section 2.5</u> shall survive until sixty (60) days after the expiration of the applicable statute of limitations (including any applicable extensions).

(b) The covenants and agreements of the Parties contained herein (i) that by their terms are to be performed in whole or in part, or that prohibit actions, prior to the Closing shall expire upon the occurrence of the Closing, except that claims for indemnification in respect of any breach thereof shall survive until the date that is twelve (12) months after the Closing and (ii) that by their terms are to be performed in whole or in part, or that prohibit actions, following the Closing shall survive until the date that is twelve (12) months after the expiration in accordance with their terms.

(c) If an indemnification claim is properly and in good faith asserted in writing pursuant to <u>Section 6.3</u> prior to the expiration as provided in <u>Section 6.4(a)</u> or <u>Section 6.4(b)</u> of the representation or warranty or covenant or agreement that is the basis for such claim, then such representation or warranty or covenant or agreement of such indemnification claim.

#### 6.5 Limitations.

(a) Notwithstanding anything to the contrary contained in this Agreement, the following limitations shall apply to indemnification claims under this Agreement:

(i) no individual claim (or series of related claims) for indemnification under <u>Sections 6.1(a)</u> or <u>6.2(a)</u> shall be valid and assertable unless it is (or they are) for an amount in excess of \$10,000;

(ii) (A) except with respect to breaches of the Seller Fundamental Representations, the Seller shall be liable with respect to claims under Section 6.1(a) for only that portion of the aggregate Damages related to such claims (excluding any claims disallowed under Section 6.5(a)(i)), considered together, which exceeds \$250,000 and (B) except with respect to breaches of the Buyer Fundamental Representations, the Buyer shall be liable with respect to claims under Section 6.2(a) for only that portion of the aggregate Damages related to such claims (excluding any claims disallowed under Section 6.5(a)(i)), considered together, which exceeds \$250,000;

(iii) (A) the aggregate liability of the Seller for indemnification claims under <u>Section 6.1(a)</u> (other than indemnification claims with respect to breaches of the Seller Fundamental Representations) shall not exceed an amount equal to twenty-five percent (25%) of the Purchase Price and (B) the aggregate liability of the Buyer for indemnification claims under <u>Section 6.2(a)</u> (other than indemnification claims with respect to breaches of the Buyer Fundamental Representations) shall not exceed an amount equal to twenty-five percent (25%) of the Purchase Price; and

(iv) with respect to breaches of the Seller Fundamental Representations, the aggregate liability of the Seller for indemnification claims under <u>Section 6.1(a)</u> shall not exceed the Purchase Price.

(b) Each Indemnified Party shall (and shall cause its Affiliates to) use commercially reasonable efforts (including by pursuing all legal rights and remedies available) in order to mitigate and minimize the Damages for which indemnification is contemplated by this <u>ARTICLE VI</u>, including pursuing commercially reasonable insurance claims with respect to any Damages or pursuing in a commercially reasonable manner any indemnification obligations of third parties with respect to any Damages that are or may be the subject of an indemnification claim hereunder.

(c) The amount of Damages recoverable by an Indemnified Party under this <u>ARTICLE VI</u> with respect to an indemnity claim shall be reduced by the amount of any payment received by such Indemnified Party (or an Affiliate thereof), with respect to the Damages to which such indemnity claim relates, from an insurance carrier or any other Person. If an Indemnified Party (or an Affiliate) receives any insurance payment in connection with any claim for Damages for which it has already received an indemnification payment from the Indemnifying Party, it shall pay to the Indemnifying Party, within thirty (30) days of receiving such insurance payment, an amount equal to the excess of (A) the amount previously received by the Indemnified Party under this <u>ARTICLE VI</u> with respect to such claim plus the amount of the insurance payments received, over (B) the amount of Damages with respect to such claim which the Indemnified Party has become entitled to receive under this <u>ARTICLE VI</u>.

(d) The representations and warranties of the Parties, and the right of any Indemnified Party with respect thereto, shall not be affected or deemed waived by reason of any investigation made by or on behalf of such Indemnified Party (including by any of its Representatives) or by reason of the fact that such Indemnified Party or any of its Representatives knew or should have known that any such representation or warranty is, was or might be inaccurate or by reason of such Indemnified Party's waiver of any condition set forth in <u>ARTICLE V</u>, as the case may be.

(e) The Parties agree that where one and the same set of facts qualifies under more than one provision entitling the Buyer or the Seller to a claim or remedy under this Agreement, there shall be only one claim or remedy.

(f) In no event shall any Indemnified Party be entitled to indemnification pursuant to this <u>ARTICLE VI</u> to the extent indemnifiable Damages were attributable to such Indemnified Party's gross negligence, willful misconduct or fraud.

(g) Except with respect to claims for fraud or equitable relief, including specific performance, made with respect to breaches of any covenant or agreement contained in this Agreement, the rights of the Indemnified Parties under this <u>ARTICLE VI</u> and under <u>ARTICLE VII</u> shall be the sole and exclusive remedies of the Indemnified Parties and their respective Affiliates with respect to claims covered by <u>Section 6.1</u>, <u>Section 6.2</u>, or <u>ARTICLE VII</u> or otherwise relating to the transactions contemplated by this Agreement, the Seller Bill of Sale, the Affiliate Bill of Sale and the Assignment and Assumption Agreement. Without limiting the generality of the foregoing, in no event shall any Party, its successors or permitted assigns be entitled to claim or seek rescission of any transactions consummated pursuant to this Agreement.

6.6 <u>Treatment of Indemnification Payments</u>. Unless otherwise required by applicable law, all indemnification payments made under this Agreement shall be treated by the Parties as an adjustment to the Final Adjusted Purchase Price.

6.7 <u>Setoff Rights</u>. The Buyer shall have the right to setoff, against each Post-Closing Payment, the amount of any Damages with respect to any Resolved But Unsatisfied Claims that the Buyer is entitled to recover under this <u>ARTICLE VI</u> (after giving effect to any applicable limitations provided for in <u>Section 6.5</u>).

# ARTICLE VII TAX MATTERS

7.1 <u>Transfer Taxes</u>. The Buyer and the Seller shall each be responsible for one-half of any transfer, sales, use, stamp, conveyance, value added, recording, registration, documentary, filing and other non-Income Taxes and administrative fees (including notary fees) ("<u>Transfer Taxes</u>") arising in connection with the consummation of the transactions contemplated by this Agreement. The Party customarily responsible under applicable law shall file all necessary Tax Returns and other documentation with respect to such Transfer Taxes. The Party responsible for filing any such Tax Returns shall provide to the other Party evidence of timely filing and payment of all such Transfer Taxes. Within fifteen (15) days after receipt of such evidence, the receiving Party shall pay to the filing Party its one-half share of such Transfer Taxes.

#### 7.2 Allocation of Certain Taxes.

(a) Any real property Taxes, personal property Taxes or other Taxes imposed on a periodic basis, in each case, applicable to the Acquired Assets for a taxable period that includes but does not end on the Closing Date (a "<u>Straddle Period</u>") shall be paid by the Buyer or the Seller, as required under applicable law, and such Taxes shall be apportioned between the Buyer and the Seller based on the number of days in the portion of the taxable period that ends on the Closing Date (the "<u>Pre-Closing Tax Period</u>") and the number of days in the entire taxable period. All Taxes imposed with respect to the Business and the Assumed Assets for any taxable period that ends before the Closing Date and for any Pre-Closing Tax Period shall be borne by the Seller. All Taxes imposed for any taxable period beginning after the Closing date and any Straddle Period Taxes that are not allocable to the Pre-Closing Tax Period shall be borne by the Buyer.

(b) For purposes of this <u>Section 7.2</u>, transactions that occur on the Closing Date but after the Closing and that are not incurred in the ordinary course of business of the Business shall be considered attributable to the period that commences on the day following the Closing Date.

7.3 <u>Cooperation on Tax Matters</u>; <u>Tax Audits</u>. The Buyer and the Seller and their respective Affiliates shall cooperate in the preparation of all Tax Returns for any taxable periods for which one Party could reasonably require the assistance of the other Party in obtaining any necessary information (including the provision of applicable Tax books and records relating exclusively to the Business or reasonably redacted so as to relate exclusively to the Business). The Seller shall have the right to control any Tax audit, initiate any claim for refund, and contest, resolve and defend against any assessment, notice of deficiency, or other adjustment or proposed adjustment relating to any and all Taxes for any taxable period ending on or before the Closing Date with respect to the Business.

7.4 <u>Post-Closing Actions</u>. The Buyer shall not file, amend, re-file or otherwise modify any Tax Return or Tax election relating in whole or in part to the Business or agree to the waiver or any extension of the statute of limitations with respect to any taxable period (or portion thereof) ending on or before the Closing Date without the prior written consent of the Seller (which consent shall not be unreasonably withheld, delayed or conditioned).

7.5 <u>Termination of Tax Sharing Agreements</u>. All Tax sharing agreements or similar arrangements with respect to or involving the Business (other than any such agreements or arrangements that are ordinary course commercial agreements, credit agreements or other agreements not primarily related to Taxes) shall be terminated prior to the Closing Date and, after the Closing Date, the Buyer and its Affiliates shall not be bound thereby or have any liability thereunder for amounts due in respect of periods ending on or before the Closing Date.

## ARTICLE VIII TERMINATION

8.1 Termination of Agreement. The Parties may terminate this Agreement prior to the Closing as provided below:

(a) the Parties may terminate this Agreement by mutual written consent;

(b) the Buyer may terminate this Agreement by giving written notice to the Seller in the event the Seller is in material breach of any representation, warranty, covenant or agreement contained in this Agreement, and such breach, individually or in combination with any other such breach, (i) would cause the conditions set forth in Sections 5.2(a), 5.2(b) or 5.2(c) not to be satisfied by the Outside Date and (ii) is not cured within thirty (30) days following delivery by the Buyer to the Seller of written notice of such breach; provided that the Buyer is not then in breach of any representations, warranties, covenants or agreements contained in this Agreement that would cause any of the conditions in Sections 5.3(a), 5.3(b) or 5.3(c) not to be satisfied by the Outside Date;

(c) the Seller may terminate this Agreement by giving written notice to the Buyer in the event the Buyer is in material breach of any representation, warranty, covenant or agreement contained in this Agreement, and such breach, individually or in combination with any other such breach, (i) would cause the conditions set forth in Sections 5.3(a), 5.3(b) or 5.3(c) not to be satisfied by the Outside Date and (ii) is not cured within thirty (30) days following delivery by the Seller to the Buyer of written notice of such breach; provided that the Buyer is not then in breach of any representations, warranties, covenants or agreements contained in this Agreement that would cause any of the conditions in Sections 5.2(a), 5.2(b) or 5.2(c) not to be satisfied by the Outside Date;

(d) the Buyer may terminate this Agreement by giving written notice to the Seller if the Closing shall not have occurred on or before the Outside Date by reason of the failure of any condition precedent under Section 5.1 or 5.2 (unless the failure results primarily from a breach by the Buyer of any representation, warranty, covenant or agreement contained in this Agreement); and

(e) the Seller may terminate this Agreement by giving written notice to the Buyer if the Closing shall not have occurred on or before Outside Date by reason of the failure of any condition precedent under Section 5.1 or 5.3 (unless the failure results primarily from a breach by the Seller of any representation, warranty, covenant or agreement contained in this Agreement).

8.2 Effect of Termination. If any Party terminates this Agreement pursuant to Section 8.1, all obligations of the Parties hereunder shall terminate without any liability of any Party to the other Parties, except that the provisions of Section 3.7, Section 3.8, this Section 8.2 and ARTICLE XII and the Confidentiality Agreement shall survive the termination of this Agreement. Notwithstanding the foregoing, termination of this Agreement shall not relieve any Party for any breach by such Party, prior to the termination of this Agreement, of any covenant or agreement (but not any representation or warranty) contained in this Agreement or impair the right of any Party to obtain such remedies as may be available to it at law or in equity with respect to such a breach by any other Party.

# ARTICLE IX EMPLOYEE MATTERS

### 9.1 Offer of Employment; Continuation of Employment.

(a) As soon as practicable (but not more than ten (10) Business Days) following the date of this Agreement, the Buyer shall, or shall cause Affiliate to, make an offer of employment effective as of the Closing Date to each individual designated as a Scheduled Employee on <u>Schedule 9.1(b)</u> (i) in the same job or position as in effect immediately prior to the Closing Date, (ii) at a worksite no more than twenty-five (25) miles from the current job's position in effect immediately prior to the Closing Date, (iii) unless otherwise agreed in writing by the Parties on the date hereof, at a salary or wage level and with a bonus opportunity, benefits

(other than severance benefits, any benefits provided under any agreement excluded from <u>Section 2.9(a)(xi)</u> or any defined benefit pension or retiree medical or other retiree welfare benefits), perquisites and other terms and conditions of employment that are substantially comparable in the aggregate to the salary or wage level, bonus opportunity, benefits (other than severance benefits, any benefits provided under any agreement excluded from <u>Section 2.9(a)(xi)</u> or any defined benefit pension or retiree medical or other retiree welfare benefits), perquisites, and other terms and conditions that such Scheduled Employee was entitled to receive immediately prior to the date of this Agreement and (iv) requiring such employee to accept or decline such offer within ten (10) Business Days after receiving such offer. The Seller has provided the Buyer the information set forth in <u>clauses (i)</u>, (ii) and (iii). Notwithstanding the foregoing, following the Closing Date, the Buyer or one of its Affiliates shall provide each New Buyer Employee, as of the Closing Date, with compensation and benefits (including incentive and equity-based compensation, but excluding any defined benefit pension or retiree medical or other retiree welfare benefits) at least as favorable as the compensation and benefits provided to similarly-situated employees of the Buyer and its Affiliates as of the Closing Date. The Buyer or its Affiliates will condition the offers to each Canadian Employee on such employee's resigning in a manner that does not impose severance obligations on the Canadian Subsidiary nor trigger any rights to statutory or other benefits. The obligation of the Buyer to provide, or cause to provide, the conditions of employment relating to pay and employee benefits in accordance with this <u>Section 9.1</u> shall continue for a period ending not earlier than one (1) year following the Closing Date. The Buyer shall provide a list to the Seller twenty-one (21) Business Days following the date of this Agreement of all Scheduled Employe

(b) Notwithstanding any provision herein, any Scheduled Employee who is on short-term disability leave as of the Closing Date (an "<u>On-Leave</u> <u>Business Employee</u>") will not become a New Buyer Employee until such time, if any, as such On-Leave Business Employee has been approved to return to active employment, at which time such employee shall be offered continued employment (with any reasonable accommodation, as necessary) with the Buyer or one of its Affiliates, but only if such employee is able to return to active employment within nine months after the Closing Date. If and when the On-Leave Business Employee returns to active status (and has accepted the offer of employment from the Buyer and its Affiliates), such On-Leave Business Employee shall be considered a New Buyer Employee and the On-Leave Business Employee shall become eligible for coverage and benefits under all employee benefit plans or programs maintained by the Buyer or its Affiliates under the same terms and conditions that apply to other New Buyer Employees, effective as of such date of return and with service credit under <u>Section 9.3(b)</u> as though it referred to his or her date of return rather than the Closing Date.

9.2 <u>Cessation of Employee Benefit Plan Participation</u>. Except as otherwise required by applicable law, the New Buyer Employees shall cease to participate in or accrue further benefits as active employees of the Seller or the Canadian Subsidiary under the Employee Benefit Plans effective as of the Closing Date (or, in the case of any On-Leave Business Employee who becomes a New Buyer Employee, such later date as of which such employee becomes a New Buyer Employee).

### 9.3 Compensation; Employee Benefits; Severance Plans.

(a) The Seller and its Affiliates will retain, bear and discharge all liabilities for any accrued compensation, unused accrued vacation and other paid time off, "stay-pay," and other accrued benefits (other than commissions and the 2013 annual incentive award payments) payable with respect to Seller Business Employees, applicable as of periods of service with the Seller or the Canadian Subsidiary, as the case may be, as of the Closing Date. The Buyer will assume, bear, and discharge all liabilities with respect to commissions and 2013 annual incentive award payments that are accrued but unpaid as of the Closing Date or that, with respect to the period in 2013 following the Closing Date, will continue to accrue, other than any commission payments due to be paid by the Seller or its Affiliates on or before the Closing Date.

(b) The Buyer will take into account periods of service of the New Buyer Employees with the Seller and its Affiliates through the Closing Date, as credited by the Seller or its Affiliates through the Closing Date under the Business Benefit Plans, for purposes of determining, as applicable, the eligibility of New Buyer Employees for participation in and vesting under (but not for benefit accrual) the employee benefit plans offered by the Buyer or any of its Affiliates to the New Buyer Employees, including vacation plans or arrangements, 401(k) or other retirement savings plans and any severance or welfare plans, and the New Buyer Employee shall receive service credit for purposes of benefits levels under the vacation/paid time off and severance plans of Buyer and its Affiliates based on such prior service.

(c) The Buyer shall cause the insurance carriers and administrators under the group health plans of the Buyer and its Affiliates to (i) waive any limitation on health coverage of New Buyer Employees due to pre-existing conditions and/or waiting periods, active employment requirements, and requirements to show evidence of good health under the applicable health plan of the Buyer or any of its Affiliates to the extent such New Buyer Employees are currently covered under a health plan of the Seller or its Affiliates (including the Canadian Subsidiary) and (ii) to credit any expenses incurred by the New Buyer Employees and their beneficiaries or dependents under similar plans of the Seller or any of its Affiliates (including the Canadian Subsidiary) during the portion of the calendar year in which the Closing Date occurs for purposes of satisfying applicable deductible, co-insurance and maximum out-of-pocket expenses or any like or similar expenses or costs.

(d) If the Buyer or its Affiliates terminates the employment of a Transitional New Buyer Employee, a Non-Transitional New Buyer Employee or a Specified Employee at any time after the Closing Date and on or prior to the Severance Outside Date without cause, the Buyer shall provide such Transitional New Buyer Employee, Non-Transitional New Buyer Employee or Specified Employee, as applicable, that executes and delivers a fully enforceable release of any and all claims against the Buyer and its Affiliates, as well as Seller and its Affiliates as an additional released party, with a cash severance payment in an amount equal to the greatest of (i) the cash payment provided under the Buyer's severance plan, policy, practice, or arrangement as may be in effect, (ii) the cash payment provided under any applicable Business Benefit Plan that is a severance plan, policy, practice, or arrangement or any other agreement entitling such employee to severance as in effect as of the date of this Agreement and as set forth in <u>Schedule 9.3(d)</u>, and (iii) the cash payment otherwise required to be provided pursuant to local law, in the case of each of <u>clauses (i)</u> and (ii), taking into account such employee's aggregate

periods of services with the Seller and its Affiliates and the Buyer and its Affiliates and, in the case of clause (iii), the periods of service as may be required by local law. The Seller or one of its Affiliates shall reimburse the Buyer and its Affiliates for such cash severance payments made by the Buyer or one of its Affiliates with respect to each such terminated New Buyer Employee (other than any Specified Employee) if the sum of (x) total number of New Buyer Employees who remain employed by the Buyer or one of its Affiliates immediately after the Severance Outside Date (the "Remaining Employees") plus (y) the total number of New Buyer Employees who are terminated by Buyer or one of its Affiliates for cause on or prior to the Severance Outside Date or who voluntarily terminate employment with the Buyer or one of its Affiliates on or prior to the Severance Outside Date (and in each case are not entitled to severance on or prior to the Severance Outside Date under this Section 9.3(d)) (the "Non-Severance Terminated Employees") is equal to, or exceeds, [\*\*\*]. If the sum of (I) the Remaining Employees plus (II) the Non-Severance Terminated Employees is less than [\*\*\*], the Buyer shall retain all liability for severance costs and expenses for the number of such terminated New Buyer Employees that is equal to (AA) [\*\*\*] minus (BB) the sum of the Remaining Employees and the Non-Severance Terminated Employees, and the Seller shall reimburse the Buyer and its Affiliates for any and all cash severance payments made by the Buyer and its Affiliates to all such other terminated New Buyer Employees; provided that the Buyer, in such case, shall be permitted in its sole discretion to designate the terminated New Buyer Employees (other than any Specified Employee) for whom the Seller or its Affiliates shall provide reimbursement to Buyer or its Affiliates of cash severance payments. The Buyer shall provide the Seller with written notice of the making of any such severance payment to any such employee and the Seller or its Affiliates shall provide such reimbursement to the Buyer or its applicable Affiliates, as the case may be, within five (5) Business Days after receipt from the Buyer of such notice. The severance benefits, if any, to be provided by the Buyer and its Affiliates to any New Buyer Employee during the period following the Severance Outside Date shall be determined by the Buyer in its sole discretion.

#### 9.4 Employment Related Liabilities.

(a) Except as otherwise provided in Section 9.3(a) or (d) or this Section 9.4(a) of the Agreement, the Seller and its Affiliates shall solely be responsible for and shall pay any and all liabilities or expenses arising out of or related to the employment of each New Buyer Employee with respect to the period on or before the Closing Date, including any individual retention or bonus guarantee agreements with Seller Business Employees, which will remain the responsibility of Seller, and Buyer and its Affiliates shall solely be responsible for and shall pay any and all liabilities and expenses arising out of or related to the employment of each New Buyer Employee after the Closing Date. For the avoidance of doubt, and except as provided in this Section 9.4(a), the Seller and its Affiliates (including the Canadian Subsidiary) shall retain, bear and discharge all liabilities, including, without limitation, liabilities for severance benefits, with respect to each Seller Business Employee who does not become a New Buyer Employee. Buyer will use its reasonable best efforts to make offers of employment in compliance with applicable laws.

[\*\*\*] Certain information has been omitted and filed separately with the Commission. Confidential treatment has been requested with respect to the omitted portions.

(b) Nothing in this Agreement shall be construed to grant any employee any right to continued employment after the Closing Date or be construed to modify, amend or establish any benefit plan, program or arrangement. No provision of this Agreement (including, for the avoidance of doubt, <u>Section 6.1(d)</u>) shall create any third party beneficiary rights in any New Buyer Employee, Canadian Employee, or former employee of the Seller with respect to the Business or former employee of the Canadian Subsidiary in respect of continued employment (or resumed employment) or any other matter.

9.5 <u>U.S. WARN Act</u>. The Buyer agrees to provide any required notice under WARN and any other similar applicable law and to otherwise comply with any such law with respect to any "plant closing" or "mass layoff" (as defined in WARN) or similar event affecting employees and occurring on or after the Closing Date. The Buyer shall assume sole responsibility for any liabilities or obligations arising under WARN or other applicable law resulting from the actions (or inactions) of the Buyer or its Affiliates on or after the Closing Date or from the transactions contemplated hereby.

9.6 U.S. COBRA. The Seller and its Affiliates shall retain, bear and discharge all liabilities with respect to the provision of COBRA rights and benefits with respect to all qualifying events occurring on or prior to the Closing Date with respect to the Seller Business Employees and former employees of the Seller with respect to the Business (and, with respect to the On-Leave Business Employees, all qualifying events with respect to such employees occurring on or prior to the date of their termination from employment with the Seller and its Affiliates). The Buyer or its Affiliates shall have responsibility for any liabilities or obligations arising under COBRA or other similar applicable law with respect to the New Buyer Employees resulting from the actions (or inactions) of the Buyer or its Affiliates after the Closing Date.

#### 9.7 Non-Solicitation.

(a) Except where such restrictions are prohibited by applicable law, from the Closing Date until the first (1 s) anniversary of the Closing Date, none of the Buyer or any of its Affiliates shall (i) solicit any individual who was a Seller Business Employee on the date hereof or on the Closing Date and who was not a Scheduled Employee, to terminate his or her employment with the Seller (or any Affiliate of the Seller) or to become an employee of or consultant to the Buyer (or any Affiliate of the Buyer) or any other Person or (ii) hire or engage any individual who was such an employee on the date hereof or on the Closing Date (other than as provided in Section 9.1); provided, however, that the foregoing shall not restrict general soliciting activity not specifically targeted at the Seller or its Affiliates (including the placement of general advertisements in trade media and the engagement of search firms that are not instructed to target the Seller or its Affiliates) (provided that, for the avoidance of doubt, this proviso shall not be construed to permit the hiring of any Seller Business Employee that was not a Scheduled Employee).

(b) From the Closing Date until the first (1 st) anniversary of the Closing Date, Seller shall not and shall cause each of its Affiliates not to (i) solicit any Specified Employee, to terminate his or her employment with the Buyer (or any Affiliate of the Buyer) or to become an employee of or consultant of the Seller (or any Affiliate of the Seller) or any other Person or (ii) hire or engage any Specified Employee.

# ARTICLE X OTHER POST-CLOSING COVENANTS AND LICENSES

#### 10.1 Access to Information; Record Retention; Cooperation.

(a) Access to Information. Following the Closing, each Party shall afford to the other Party and to such other Party's Representatives (at reasonable times, during normal business hours, on reasonable prior written notice and in a manner so as not to interfere with the conduct of the applicable Party's and its Affiliates' business operations) (i) reasonable access and duplicating rights to all Information within the possession or control of such Party and (ii) reasonable access to the personnel of such Party who have knowledge of the Information or the Business; provided, however, that neither Party (nor its Representatives) shall be required (x) to provide any Information or access (A) that would violate the provisions of any applicable laws or regulations or any agreement to which it is a party or, in the case of Information, the sharing of which would legally require the consent of any third party (including any employee of such Party of any of its Affiliates), (B) that would cause the loss of a claim of privilege (including attorney-client and work product privileges) with respect thereto or (C) that would result in the disclosure of any information of such Party or any of its Affiliates unrelated to the Business (and such Party shall be permitted to redact any such information from any materials provided to the other Party or its Representatives) or (y) to provide any consolidated, combined, affiliated or unitary Tax Return that includes such Party or any of its Affiliates or any Tax-related work papers. Requests may be made under this Section 10.1(a) for financial reporting and accounting matters, preparing financial statements, preparing and filing of any Tax Returns, prosecuting any claims for refund, defending any Tax claims or assessment, preparing securities law or securities exchange filings, prosecuting, defending or settling any litigation, or insurance claim, performing obligations under this Agreement and the Ancillary Agreements, and all other proper business purposes. With respect to any Information, the Seller (i) shall not use, and shall cause its Representatives not to use, such Information for any purpose other than to confirm the accuracy of each Revenue Statement and (ii) shall, consistent with Section 10.1(e), keep all such Information confidential and not disclose such Information to any third party (other than its Representatives, each of whom shall not use such Information for any purpose other than to confirm the accuracy of each Revenue Statement).

(b) <u>Preparation of the Financial Statements</u>. Without limitation of the provisions of, and subject to the limitations and standards of reasonableness set forth in, <u>Section 10.1(a)</u>, following the Closing, (i) the Buyer shall provide to the Seller all information relating to the Business prior to the Closing that is reasonably requested by the Seller for the preparation of the financial statements of the Seller and its Affiliates and (ii) the Seller shall provide to the Buyer all financial and accounting information relating the Business prior to the Closing that is reasonably requested by the Buyer required to be included in any periodic reports on Forms 10-K and 10-Q, any Current Report on Form 8-K (including the financial statements required by Rule 3-05 of Regulation S-X in respect of the Business) and any registration statement to be filed with the SEC. In connection with the preparation of such financial statements, upon the reasonable request of the either Party, the requested Party shall provide the requesting Party (and its independent auditors) with reasonable access to the books, accounts and records relating to the Business prior to the Closing.

(c) <u>Reimbursement</u>. A Party making Information or personnel available to the other Party under <u>Section 10.1</u> shall be entitled to receive from such other Party, upon the presentation of invoices therefor, payments for such amounts relating to supplies, disbursements and other out-of-pocket expenses, as may reasonably be incurred in making such Information or personnel available; <u>provided</u>, <u>however</u>, that no such reimbursements shall be required for the salary or cost of fringe benefits or similar expenses pertaining to employees of the providing Party.

(d) <u>Retention of Records</u>. Except as may otherwise be required by law or agreed to in writing by the Parties, each Party shall use commercially reasonable efforts to preserve, until seven (7) years after the Closing Date, all Information in its possession or control that relates to the period prior to the Closing. Notwithstanding the foregoing, in lieu of retaining any specific Information, any Party may offer in writing to the other Party to deliver such Information to the other Party, and if such offer is not accepted within ninety (90) days, the offered Information may be disposed of at any time. From the Closing Date until the termination of the MAPSA Subcontracting Agreement the Buyer agrees to maintain true and accurate books and records relating to the subject matter of the performance of its obligations under such agreement, which it shall make available to the Seller pursuant to this <u>Section 10.1</u>.

(e) Confidentiality. From and after the Closing, each Party shall keep confidential, and shall cause its Representatives to keep confidential, all of the other Party's Proprietary Information (except to the extent that such Proprietary Information (i) is or becomes generally available to the public other than as a result of any disclosure directly or indirectly by the receiving Party or its Representatives, (ii) was available to the receiving Party on a non-confidential basis prior to its disclosure to the receiving Party by the other Party or the other Party's Representatives or (iii) is or becomes available on a non-confidential basis to the receiving Party from a Person other than the disclosing Party, provided that, the receiving Party reasonably believes that the source of such Proprietary Information was not prohibited from disclosing such Proprietary Information to it by a legal, contractual, fiduciary or other obligation to the disclosing Party or any of its Representatives or to any other Person), and the receiving Party shall not (and shall cause its Representatives not to) use such Proprietary Information for purposes of trading securities, for competitive purposes or to obtain commercial advantage with respect to the disclosing Party. If a receiving Party or one of its Representatives is required (by oral questions, interrogatories, requests for information or documents in legal proceedings, subpoena, civil investigative demand or other similar legal process) to disclose such Proprietary Information, the receiving Party or its Representative, as applicable, shall provide the disclosing Party with prompt notice of such requirement in order to enable the disclosing Party to seek an appropriate protective order or other remedy, to consult with the receiving Party or its Representative, as applicable, with respect to taking steps to resist or narrow the scope of such requirement, or to waive compliance, in whole or in part, with the terms of this Section 10.1(e). In the event that such protective order or other remedy is not obtained or the disclosing Party waives compliance with the terms of this Section 10.1(e), the receiving Party or its Representatives, as the case may be, shall disclose only that portion of the Proprietary Information which, in the opinion of its counsel, is legally required to be disclosed. In any such event, the receiving Party will use its reasonable best efforts to ensure that all Proprietary Information that is so disclosed will be accorded confidential treatment. For purposes of this Section 10.1, the Seller shall be deemed to be a "receiving Party" of all Business information.

#### 10.2 Trademarks and Tradenames.

(a) Subject to the occurrence of the Closing, during the ninety- (90-) day period commencing on the Closing Date (the "Transition Trademark Period"), the Buyer may use the Trademarks Prudential, the Prudential logo and the Rock design (the "Licensed Marks"), solely to the extent utilized by the Business in electronic or written materials immediately prior to the Closing, and solely (i) to use and distribute such materials, and/or (ii) to disclose that the Business was formerly associated with the Seller. The Buyer shall comply with such other reasonable requests as are made by the Seller or any of its Affiliates to enable the Seller and its Affiliates to maintain and preserve the validity and enforceability of such trademarks and the goodwill associated therewith. The Buyer agrees to display, within five (5) days of the Closing until the end of the Transition Trademark Period, on each page of its public-facing website and on any materials bearing the Seller's or any of its Affiliates' trademarks distributed after the Closing, the following statement (the "Non-Affiliation Statement"): "All Prudential trademarks and logos are used under license. Envestnet is not affiliated with The Prudential Insurance Company of America or with Prudential Financial, Inc." The Buyer acknowledges that the Licensed Marks and all rights therein and thereto and the goodwill pertaining thereto belong exclusively to the Seller and its Affiliates and agrees that it has no right, title or interest, express or implied, in and to the Licensed Marks, except as specifically provided in this Agreement. Any and all goodwill generated by the Buyer's use of the Licensed Marks or associated therewith shall inure solely to the benefit of The Prudential Insurance Company of America. Except as otherwise provided in this Section 10.2, the Buyer shall have no rights to use any Retained Marks and will not hold itself out as having any affiliations with the Seller or any of its Affiliates. During and after the Transition Trademark Period, the Buyer shall use commercially reasonable efforts to ensure that there is no confusion regarding the Buyer's lack of affiliation with the Seller or the Seller's Affiliates and/or, if any such confusion occurs, to promptly remediate such confusion and the causes thereof; provided that, the Buyer's display of the Non-Affiliation Statement as noted above, during or after the Transition Trademark Period, shall constitute commercially reasonable efforts to ensure there is no such confusion.

(b) Subject to the occurrence of the Closing, during the two- (2-) year period commencing on the Closing Date, the Seller and its Affiliates will not use the phrase "Wealth Management Solutions" to denote the name or brand of any product or service competitive with those being offered by the Business on the Closing Date; <u>provided</u> that (x) nothing in this <u>Section 10.2(b)</u> shall limit the Seller's and its Affiliates' rights to utilize any other phrase or any portion of such phrase, including "Wealth Management", "Wealth Solutions" and "Management Solutions", and (y) Seller and its Affiliates may utilize the phrase "Wealth Management Solutions" to generically describe the attributes of any of their products or services that provide wealth management solutions.

10.3 License Back. The Buyer hereby grants to the Seller and each of its Affiliates and their respective successors and assigns a perpetual, irrevocable, non-terminable, worldwide, royalty-free, fully paid-up, non-exclusive, sublicenseable to Affiliates, customers, end users and contractors (provided that such customers, end users and contractors are not licensed to use the Licensed Software for any purpose other than for its intended use to provide services or to process business of the Seller or the applicable Affiliate), and non-transferable (except that the Seller and each of its Affiliates may transfer, as limited by the proviso to this parenthetical, the

foregoing rights and licenses to a Person in connection with sales or divestitures of a business using the Licensed Software, or mergers, reorganizations, consolidations or other change-in-control transactions; provided that the transferee's license shall be no broader than that of the transferor) right and license, effective as of the Closing Date, in, to and under the Licensed Software for the continued use in the respective businesses of the Seller and/or any of its Affiliates (other than in the field of the Business ) as of or prior to the Closing Date, including the right to use, reproduce, create derivative works of, perform and display the Licensed Software. Any transfer made in violation of the foregoing shall be void and of no legal force and effect. The Buyer and its Affiliates are not obligated to provide support to the Seller or any of its Affiliates for the Licensed Software or provide the Seller or any of its Affiliates with any modifications, updates or upgrades to or any maintenance or other services for the Licensed Software. THE LICENSED SOFTWARE LICENSED PURSUANT TO THIS SECTION 10.3 IS FURNISHED "AS IS" AND WITH ALL FAULTS AND BUYER EXPRESSLY DISCLAIMS ALL REPRESENTATIONS AND WARRANTIES WITH RESPECT TO THE LICENSED SOFTWARE, WHETHER EXPRESS OR IMPLIED, EITHER IN FACT OR BY OPERATION OF LAW, BY STATUTE OR OTHERWISE, INCLUDING ANY WARRANTY OF NON-INFRINGEMENT, MERCHANTABILITY, FITNESS FOR ANY PARTICULAR PURPOSE. ADEQUACY, UNINTERRUPTED OR ERROR-FREE PERFORMANCE, OR COMPLIANCE WITH ANY LAW. The Seller shall at all times treat Licensed Software as confidential information of the Buyer; provided, however, that the foregoing shall not limit the rights and licenses granted pursuant to this Section 10.3. For the purposes of this Section 10.3, "Licensed Software" means Software that was used in any of the Seller's or its Affiliates' systems, software or applications prior to the Closing Date (other than systems, software or applications used exclusively by the Business) as well as any immaterial portions of Software that may also have been included in any part of other software of the Seller or any of its Affiliates in an incidental manner, and the Intellectual Property rights of the Buyer and its Affiliates associated therewith.

# 10.4 Accounts Receivable and Accounts Payable.

(a) Each Party agrees that it shall forward promptly to the other Party any monies, checks or instruments received by such Party after the Closing with respect to any accounts receivable to which the other Party is entitled to the benefit of hereunder. The Seller hereby authorizes the Buyer, following the Closing, to endorse and cash any checks or instruments payable or endorsed to the Seller or its order which are received by the Buyer and which relate to accounts receivable constituting Acquired Assets that were purchased by the Buyer pursuant hereto.

(b) In the event that, following the Closing (i) the Seller or its Affiliate inadvertently pays or discharges any trade payable that represented an Assumed Liability or (ii) the Buyer or its Affiliate inadvertently pays or discharges any trade payable that represented an Excluded Liability, in the case of <u>clause (i)</u> above, the Buyer shall, and in the case of <u>clause (ii)</u> above, the Seller shall, reimburse the other Party (or its Affiliate) for the amount so paid or discharged within ten (10) days of being presented with written evidence of such payment or discharge.

10.5 Deletion of Non-Transferred Software. The Buyer shall not use and shall cause each of its Affiliates not to use any third-party software loaded on the Equipment included in the Acquired Assets as of the Closing Date if such software is not included in the Acquired Assets or otherwise licensed to the Buyer. The Buyer shall, as soon as is reasonably practical, and in any event no later than forty-five (45) days following the Closing, delete all such software from any of the Equipment on which it is installed.

10.6 <u>Brokers</u>. Regardless of whether the Closing shall occur, (a) the Seller shall indemnify the Buyer and its Affiliates against, and hold the Buyer and its Affiliates harmless from, any liability for any brokers' or finders' fees or other commissions arising with respect to brokers, finders, financial advisors or other Persons retained or engaged by the Seller or its Affiliates in respect of the transactions contemplated by this Agreement and (b) the Buyer shall indemnify the Seller and its Affiliates against, and hold the Seller and its Affiliates harmless from, any liability for any brokers' or finders' fees or other commissions arising with respect to brokers, finders, financial advisors or other Persons retained or engaged by the Seller and its Affiliates harmless from, any liability for any brokers' or finders' fees or other commissions arising with respect to brokers, finders, financial advisors or other Persons retained or engaged by the Buyer or its Affiliates in respect of the transactions contemplated by this Agreement.

### ARTICLE XI DEFINITIONS

11.1 Defined Terms. For purposes of this Agreement, each of the following terms shall have the meaning set forth below.

"Abbreviated Financial Statements" shall mean (a) (i) an audited statement of assets acquired and liabilities assumed and statement of revenues and direct operating expenses, in each case, with respect to the Business as of, and for each of the years ended, December 31, 2011 and December 31, 2012, together with associated notes to those financial statements, (ii) unaudited statements of assets acquired and liabilities assumed and statements of revenues and direct operating expenses, in each case, with respect to the Business as of, and for each of the quarters ended, March 31, 2012 and March 31, 2013, together with associated notes to those unaudited financial statements (and, if the Closing Date occurs on or after August 15, 2013, as of, and for each of the quarters ended, June 30, 2012 and June 30, 2013), in the case of clauses (i) and (ii), with such financial statements to be prepared in a manner consistent with the applicable methodologies set forth in Section 2.4(b) of the Disclosure Schedule, (iii) quarterly reviews prepared by the Seller's independent auditor in respect of such independent auditor's performance of the procedures specified by the PCAOB for a review of interim financial information as described in AU 722 with respect to the statements of assets acquired and liabilities assumed and statements of revenues and direct operating expenses described in the foregoing clause (ii) and (iv) to the extent the parties reasonably determine the same are required to be filed by the Buyer under Rule 3-05 of Regulation S-X, an audited statement of assets acquired and liabilities assumed and statement of revenues and direct operating expenses, in each case, with respect to the Business, as of, and for the year ended, December 31, 2010, together with associated notes to those financial statements, with such financial statements to be prepared consistent with the applicable methodologies set forth in Section 2.4(b) of the Disclosure Schedule or (b) (i) the Accepted Financial Statements, as of and for the years ended December 31, 2011 and December 31, 2012 and as of and for the quarters ended March 31, 2012 and March 31, 2013 and (ii) to the extent the parties reasonably determine the same are required to be filed by the Buyer under Rule 3-05 of Regulation S-X, the Accepted Financial Statement most comparable to a statement of income, for the year ended December 31, 2010.

"Accepted Financial Statements" such financial statements as the staff of the SEC may advise the Buyer it will not raise an objection to for purposes of Rule 3-05 of Regulation S-X; provided that such financial statements are no more expansive than the Full Financial Statements.

"Acquired Assets" shall mean the following assets:

(a) all accounts receivable and other receivables related exclusively to the Business, whether or not billed, as set forth on <u>Schedule 1.1(a)(ii)</u>, as such Schedule shall be updated pursuant to <u>Section 1.4(b)</u>;

- (b) all Equipment set forth on Schedule 1.1(a)(iii);
- (c) the Assigned Contracts, except as provided in Section 1.6;
- (d) all Seller Intellectual Property; and

(e) all books, records, accounts, ledgers, files, documents, correspondence, studies, reports and other printed or written materials related exclusively to the Business, in each case, to the extent related to any items set forth in this definition and subject to any restrictions imposed by applicable law.

"Acquired Assets Revenue" shall mean, for any Post-Closing Period, the amount calculated pursuant to the method set forth in <u>Schedule 11.1</u>. If the Buyer and the Buyer Subsidiaries do not continue the same invoicing practices used by the Business prior to the Closing, the amount of fees "invoiced" under the Acquired Sponsor Contracts and the Acquired Prospect Contracts for any period shall be calculated based on the amount of fees that would have been invoiced under the Acquired Sponsor Contracts or Acquired Prospect Contracts during such period had the Buyer and the Buyer Subsidiaries continued the same invoicing practices used by the Business prior to the Closing for invoicing under the Sponsor Contracts in existence on the date hereof.

"Acquired Prospect Contract" shall mean any contract (as the same may be amended, restated or otherwise modified, replaced or succeeded from time to time) for the provision of wealth management solutions services by the Business or any Successor to the Business that is entered into or otherwise executed prior to December 31, 2013 with any of the prospective customers of the Business set forth on <u>Schedule 11.1(a)</u>.

"Acquired Sponsor Contract" shall mean, as applicable, (a) an Assigned Sponsor Contract (as the same may be amended, restated or otherwise modified from time to time), (b) the Pruco Contract (as the same may be amended, restated or otherwise modified from time to time) or (c) any successor or replacement contract to an Assigned Sponsor Contract with a Participating Sponsor that relates to the provision of wealth management solutions services substantially similar to the services provided under such Assigned Sponsor Contract by the Business or any Successor to the Business.

"Affiliate" shall have the meaning assigned to it in Rule 12b-2 of the Securities and Exchange Act of 1934.

"<u>Affiliate Assets</u>" shall mean (a) the personal computers and personal communications devices used immediately prior to the Closing by the New Buyer Employees and (b) the Seller Affiliate's right title and interest in the Seller Owned Intellectual Property.

"Agreed Amount" shall mean, with respect to a claim for indemnification made hereunder, a portion, but not all, of the Claimed Amount.

"Ancillary Agreements" shall mean the agreements and instruments referred to in clauses (iii) through (x) in Section 1.4(c).

"Annualized Pruco Revenue" shall mean, for any Post-Closing Period, the amount calculated pursuant to the method set forth in <u>Schedule 11.1</u>. The amount of fees "invoiced" under the Pruco Contract for any period shall be calculated based on the amount of fees that would have been invoiced under the Pruco Contract during such period had the Buyer and the Buyer Subsidiaries continued the same invoicing practices used by the Business prior to the Closing for invoicing under the Sponsor Contracts in existence on the date hereof.

"Annualized Revenue" shall mean, for each Post-Closing Period, (a) the related Acquired Assets Revenue, multiplied by (b) four (4).

"Applicable Additional Purchase Price Payment" shall mean, for any Post-Closing Period, the amount set forth on line 4 of <u>Schedule 11.1</u>; provided, that if the Annualized Revenue relating to such Post-Closing Period is less than the amount set forth on line 5 of <u>Schedule 11.1</u>, such amount set forth on line 4 of <u>Schedule 11.1</u> shall be reduced by \$1.00 for every \$1.00 in shortfall until the Applicable Additional Purchase Price Payment for such Post-Closing Period is reduced to zero.

"<u>Applicable Excess Revenue Payment</u>" shall mean, for the applicable Post-Closing Period the, the amount calculated pursuant to the method set forth in <u>Schedule 11.1</u>.

"Assets Under Administration" shall mean, as of any date of determination, for any Participating Sponsor, the sum amount of assets of such Participating Sponsor's Participating Clients that are managed through the Business.

"Assigned Contracts" shall mean the contracts or agreements to which the Seller is party that are (i) Assigned Sponsor Contracts, (ii) Assigned Manager Contracts or (iii) listed in <u>Schedule 1.1(a)(vi)</u>; provided that, for any such contracts listed on <u>Schedule 1.1(a)(vi)</u> that are specifically identified therein as not relating exclusively to the Business, only the portions of such contracts that relate to the Business and may be assigned shall be "Assigned Contracts" hereunder.

"Assigned Domain Names" shall mean the domain names listed under the heading "Domain Names" on Schedule 1.1(a)(i).

"Assigned Manager Contracts" shall mean (i) each Manager Contract listed in <u>Schedule 1.1(a)(iv)</u> for which consent has been obtained in accordance with <u>Section 4.4</u> and (ii) each other contract similar to a Manager Contract for which the Buyer has agreed in writing to such contract becoming an Assigned Manager Contract pursuant to <u>Section 4.2(c)</u>.

"Assigned Sponsor Contracts" shall mean (i) each Sponsor Contract listed in Schedule 1.1(a)(v) for which consent has been obtained in accordance with Section 4.4 and (ii) each other contract similar to a Sponsor Contract for which the Buyer has agreed in writing to such contract becoming an Assigned Sponsor Contract pursuant to Section 4.2(c).

"Assignment Requirements" shall mean, with respect to and Sponsor Contract, Manager Contract or other Assigned Contract, the necessary consents under applicable law or under such contract to effect the assignment and delegation or continuation of such contract in connection with the transactions contemplated by this Agreement.

"Assumed Liabilities" shall mean the following liabilities and obligations (whether known or unknown, whether absolute or contingent, whether liquidated or unliquidated and whether due or to become due):

(a) all liabilities reflected on the Closing Net Asset Schedule;

(b) all liabilities and obligations arising under the Assigned Contracts to be satisfied or performed on or after the Closing, except (x) as the same may be limited pursuant to the subcontracting arrangements contemplated by Section 1.6 or (y) to the extent relating to any pre-Closing breach thereof;

(c) all liabilities and obligations arising from the operation of the Business or the ownership of the Acquired Assets on or after the Closing except to the extent (x) relating to any indemnification obligation of the Seller pursuant to Section 6.1 or (y) relating to any indemnification obligation of the Seller pursuant to any Ancillary Agreement;

(d) all liabilities and obligations in respect of employees or employee benefits that are assumed by the Buyer pursuant to <u>ARTICLE IX</u>; and

(e) all liabilities and obligations for any Taxes that have been apportioned to the Buyer pursuant to ARTICLE VII.

"Base Date" shall mean December 31, 2013.

"Base Purchase Price" shall mean \$10 million.

"Business" shall mean the business of the Seller and its Affiliates known as Wealth Management Solutions, as conducted on the date hereof and as of the Closing Date, which consists of providing outsourced wealth management services to financial institutions, insurance companies and independent broker/dealer networks through a program under which asset managers provide asset management services to customers of such financial institutions, insurance companies and independent broker/dealer networks.

"Business Benefit Plan" shall mean any Employee Benefit Plan maintained by the Seller or any other Person (other than a Governmental Entity) for the benefit of Seller Business Employees (and their respective beneficiaries) other than any agreement excluded from Section 2.9(a) by clause (xi) thereunder.

"Business Day" shall mean any day other than (i) a Saturday or Sunday or (ii) a day on which banking institutions located in New York, New York or Chicago, Illinois are permitted or required by law, executive order or governmental decree to remain closed.

"Business Material Adverse Effect" shall mean any change, event, occurrence, state of facts or effect that (a) is materially adverse to the business, financial condition or results of operations of the Business, taken as a whole, or (b) materially impairs the ability of the Seller to consummate the transactions contemplated by this Agreement; provided, however, that none of the following (or the results thereof) shall be considered in determining whether a "Business Material Adverse Effect" shall have occurred: (i) any changes in law or accounting standards or interpretations thereof applicable to the Business, (ii) any changes in general political, economic, regulatory or business conditions (including acts of war, declared or undeclared, armed hostilities, sabotage and terrorism), or in financial, securities or other market conditions generally, including changes in interest rates or exchange rates, whether globally, in the United States or in Canada, (iii) any changes generally affecting any industry in which the Business operates, (iv) any changes resulting from or arising out of natural disasters, (v) any changes arising out of the negotiation, execution, announcement, pendency or consummation of this Agreement or the transactions contemplated hereby, to the extent the same relate to the identity of, or the effects of any facts or circumstances relating to, the Buyer, (vi) so long as the condition set forth in <u>Section 5.1(b)</u> has been satisfied, any changes arising out of any failure to retain Participating Sponsors or Participating Managers, (vii) any changes arising solely as the result of any failure of the Business to meet any estimates of revenues or earnings for any period ending on or after the date hereof.

"Buyer Certificate" shall mean a certificate to the effect that each of the conditions specified in clauses (a), (b) and (c) of Section 5.3 is satisfied.

"Buyer Fundamental Representations" shall mean the representations and warranties of the Buyer contained in Sections 3.1 and 3.2.

"Buyer Material Adverse Effect" shall mean a material adverse effect on the ability of the Buyer to consummate the transactions contemplated by this Agreement.

"Buyer Subsidiary" shall mean any direct or indirect subsidiary of the Buyer.

"Canadian Assigned Manager Contract" shall mean each Canadian Manager Contract for which consent has been obtained in accordance with Section 4.4.

"Canadian Employees" shall mean the employees of the Canadian Subsidiary, each of whom, as of the date of this Agreement, is identified on <u>Schedule</u> <u>9.1(a)</u> as a Canadian Employee.

"Canadian Manager Contract" shall mean each Manager Contract listed in Schedule 1.1(a)(vi) identified therein as a "Canadian Manager Contract".

"Canadian Subsidiary" shall mean Prudential Investments Wealth Management Solutions of Canada Ltd.

"<u>Claim Notice</u>" shall mean a written notice that contains (i) a statement that the Indemnified Party is entitled to indemnification under <u>ARTICLE VI</u> and a reasonable explanation of the basis therefor, (ii) a description and a good faith estimate of the Claimed Amount of any Damages incurred by the Indemnified Party, if then determinable and (iii) a demand for payment of Damages related thereto.

"Claimed Amount" shall mean the amount of any Damages claimed by an Indemnified Party.

"Closing" shall mean the closing of the transactions contemplated by this Agreement and shall be deemed effective as of 11:59 p.m. on the Closing Date.

"<u>Closing Adjustment Amount</u>" shall mean an amount equal to (i) the aggregate sum of the liabilities reflected on the Closing Net Asset Schedule <u>minus</u> (ii) the aggregate sum of the assets reflected on the Closing Net Asset Schedule (it being understood that if the aggregate sum of the assets reflected on the Closing Net Asset Schedule exceeds the aggregate sum of the liabilities reflected on the Closing Net Asset Schedule, the Closing Adjustment Amount shall be a negative number). An example of the calculation of the Closing Adjustment Amount as if the Closing shall have occurred as of June 30, 2013 is set forth in <u>Exhibit A</u>.

"<u>Closing Date</u>" shall mean, unless otherwise agreed by the Parties, the last calendar day of the month in which the conditions to the obligations of the Parties to consummate the transactions contemplated hereby (excluding the delivery of any documents to be delivered at the Closing by any of the Parties, it being understood that the occurrence of the Closing shall remain subject to the delivery of such documents) have been satisfied or waived.

"<u>Closing Net Asset Schedule</u>" shall mean a statement setting forth the balances of each of the specific categories of assets and liabilities of the Business set forth on <u>Exhibit A</u> as of the end of the Closing Date, which statement shall be prepared consistent with the principles described on <u>Exhibit A</u>. For the avoidance of doubt, the Parties acknowledge that not all of the Acquired Assets or Assumed Liabilities will be reflected on the Closing Net Asset Schedule.

"Closing Revenue Run-Rate" shall mean the aggregate Continuing Revenue Run-Rate for all Continuing Sponsors as of immediately prior to the Closing Date.

"Closing Run-Rate Percentage" shall mean the fraction (expressed as a percentage), the numerator of which is the Closing Revenue Run-Rate and the denominator of which is the Base Revenue Run-Rate.

"COBRA" shall mean the Consolidated Omnibus Budget Reconciliation Act of 1985.

"Code" shall mean the Internal Revenue Code of 1986.

"Confidentiality Agreement" shall mean the confidentiality letter agreement dated January 30, 2013 between the Buyer and Prudential Financial, Inc.

"Continuing Assets Under Administration" shall mean, as of any date of determination, for any Continuing Sponsor, the sum amount of assets of such Continuing Sponsor's Participating Clients that are managed through the Business.

"<u>Continuing Revenue Run-Rate</u>" shall mean, with respect to a Continuing Sponsor as of any date of determination, the product of the Continuing Assets Under Administration with respect to such Continuing Sponsor, multiplied by the applicable Fee rates for such Continuing Sponsor as of such date of determination (calculated in a manner consistent with the Revenue Run-Rate reflected in <u>Section 2.18 of the Disclosure Schedule</u>).

"Continuing Sponsor" shall mean (i) any Participating Sponsor that has consented to the assignment of its Sponsor Contract to the Buyer and (ii) Pruco.

"Damages" shall mean any and all liabilities, monetary damages, fines, fees, penalties, costs and expenses (including reasonable attorneys' fees and expenses), in each case, other than damages or other amounts that are consequential; in the nature of lost profits, lost revenues or lost opportunities or opportunity costs; based upon a multiple of earnings or similar financial measure, the loss of anticipated savings, claims or customer or business interruptions; special; incidental; exemplary; indirect or punitive, in each case, whether or not advised of the possibility of such damages, except (x) to the extent the same are required to be paid to a third-party pursuant to a third-party claim or (y) to the extent the Damages being measured relate to an Assigned Sponsor Contract, lost revenues directly attributable to such Assigned Sponsor Contract may be included in the determination of Damages relating thereto.

"Disclosure Schedule" shall mean the disclosure schedule provided by the Seller to the Buyer on the date hereof; provided that the mere inclusion of an item in the Disclosure Schedule as an exception to a representation or warranty will not be deemed an admission by the Seller that such item (or any non-disclosed item or information of comparable or greater significance) represents a material exception or fact, event or circumstance, that such item has had, or is expected to result in, a Business Material Adverse Effect, or that such item actually constitutes noncompliance with, or a violation of, any law, permit or contract or other topic to which such disclosure is applicable, nor shall it establish a standard of materiality for any purpose whatsoever.

"Employee Benefit Plan" shall mean (a) any "employee pension benefit plan" (as defined in Section 3(2) of ERISA) (whether or not subject to ERISA), (b) any "employee welfare benefit plan" (as defined in Section 3(1) of ERISA) (whether or not subject to ERISA), (c) any other plan, agreement, program, policy or arrangement providing vacation or other paid time off benefits, severance benefits, sales or other commissions, change in control benefits, stay incentive benefits, disability or other welfare-type benefits, retirement benefits, deferred compensation, bonuses, stock options, stock purchase, phantom stock, stock appreciation or other forms of incentive compensation or post-retirement or other post-employment compensation, or flexible spending or other fringe benefits, and (d) any employment, consulting, services, termination, and severance plan, agreement or arrangement, in the case of (a)-(d),

maintained or contributed to by the Seller, the Canadian Subsidiary or any of its or their ERISA Affiliates, to which the Seller, the Canadian Subsidiary or any of its or their ERISA Affiliates has an obligation to contribute, or with respect to which the Seller, the Canadian Subsidiary or any of its or their ERISA Affiliates otherwise has any liability or contingent liability, whether or not funded and whether or not terminated.

"Equipment" shall mean computers (including personal computers), furniture, fixtures, telecommunications, and other equipment and other interests in tangible personal property, excluding in all cases any Intellectual Property covering, embodied in or connected to any of the foregoing.

"ERISA" shall mean the Employee Retirement Income Security Act of 1974, as amended.

"ERISA Affiliate" shall mean any entity that is, on the date hereof, a member of (a) a controlled group of corporations (as defined in Section 414(b) of the Code), (b) a group of trades or businesses under common control (as defined in Section 414(c) of the Code) or (c) an affiliated service group (as defined under Section 414(m) of the Code or the regulations under Section 414(o) of the Code), any of which includes the Seller or the Canadian Subsidiary.

"Excluded Assets" shall mean all assets, properties and rights of the Seller and its Affiliates that are not Acquired Assets, including:

(a) all cash and cash equivalents or similar investments, bank accounts, commercial paper, certificates of deposit, Treasury bills and other marketable securities;

(b) all rights to insurance claims, related refunds and proceeds arising from or related to the Excluded Assets and Excluded Liabilities;

(c) all rights that accrue or will accrue to the benefit of the Seller under this Agreement or the Ancillary Agreements;

(d) all Tax refunds or credits and Tax deposits of the Seller and its Affiliates;

(e) all Tax books and records and Tax Returns;

(f) all Intellectual Property rights other than the Seller Intellectual Property;

(g) all actions, claims, causes of action, rights of recovery, choses in action and rights of setoff of any kind arising before, on or after the Closing relating to the items set forth in this definition or to any Excluded Liabilities;

(h) the minute books, stock ledgers, charter documents, member interest record books, financial records and other books and records as pertain to the organization, existence or capitalization of the Seller and its Affiliates (other than the Canadian Subsidiary);
(i) all books, records, accounts, ledgers, files, documents, correspondence, studies, reports and other printed or written materials related to any Excluded Assets or Excluded Liabilities and all personnel records;

(j) any assets used primarily for the purpose of providing Overhead and Shared Services and, other than as provided in the Transition Services Agreement or the Gateway Sublease Agreement, any rights of the Business to receive from the Seller or any of its Affiliates any Overhead and Shared Services;

(k) all Intellectual Property rights relating to the marks "Pru", "Prudential", the Prudential logo, the Rock design and all other Trademarks owned by The Prudential Insurance Company of America (or any of its Affiliates) or any variation of the foregoing, together with all goodwill associated with such Trademarks, including the right to use the name "Pru" or "Prudential" in any format and (ii) all Intellectual Property rights relating to websites owned by The Prudential Insurance Company of America (or any of its Affiliates), except for the Assigned Domain Names;

(l) other than to the extent the same specifically constitute Acquired Assets, any computer software systems that are a part of the systems of Prudential Financial, Inc. or any of its Affiliates (including the Seller) or that are under a license agreement to Prudential Financial, Inc. or any of its Affiliates (including the Seller);

(m) all stock or other equity interests in any Person;

(n) all qualifications to do business as a foreign company and arrangements with registered agents relating to foreign qualifications; and

(o) all assets, properties or rights listed on, or arising under any contracts or agreements listed on <u>Schedule 1.1(b)</u>.

"Excluded Liabilities" shall mean all liabilities and obligations of the Seller that are not Assumed Liabilities, including:

(a) other than those liabilities and obligations arising under the Assigned Contracts to be satisfied or performed on or after the Closing that do not relate to any pre-Closing breach thereof, all liabilities and obligations arising from the operation of the Business or ownership of the Acquired Assets prior to the Closing;

(b) all liabilities and obligations arising from the operation or ownership of the Canadian Subsidiary prior to the Closing;

(c) all Taxes, except to the extent apportioned to the Buyer pursuant to <u>ARTICLE VII</u>, (i) imposed with respect to the Business or the Acquired Assets for any taxable period ending on or prior to the Closing Date; (ii) with respect to the Excluded Assets; (iii) of the Seller (including all liabilities and obligations for any Taxes that have been apportioned to the Seller pursuant to <u>ARTICLE VII</u>; and (iv) of another Person for which the Seller is liable by reason of Treasury Regulations Section 1.1502-6 (or any comparable or similar provision of federal, state, local or foreign law) as a transferee or successor, by contract or pursuant to any law, rule or regulation;

(d) all liabilities and obligations of the Seller in respect of employees or employee benefits retained by the Seller pursuant to <u>ARTICLE IX</u>, including all liabilities under or with respect to the Employee Benefit Plans;

(e) all liabilities and obligations of the Seller under this Agreement and the Ancillary Agreements; and

(f) except as otherwise specifically provided herein, all liabilities and obligations of the Seller for costs and expenses incurred in connection with this Agreement and the Ancillary Agreements or the consummation of the transactions contemplated by this Agreement and the Ancillary Agreements (including any fees for financial advisors engaged by or on behalf of the Seller).

"Exploit" shall mean develop, design, test, modify, make, use, sell, have made, used and sold, import, reproduce, market, distribute, commercialize, support, maintain, correct and create derivative works of.

"Fees" shall mean all fees payable by a Participating Sponsor to the Seller pursuant to such Participating Sponsor's Sponsor Contract (after giving effect to, and taking into account, any amounts owing to or to be collected for the account of Participating Managers).

"Final Adjusted Purchase Price" shall mean an amount equal to (i) the Base Purchase Price minus (ii) the Closing Adjustment Amount.

"Final Payment Date" means such time as the Buyer has made each payment, if any, to the Seller of the Applicable Additional Purchase Price Payments and Applicable Excess Revenue Payments under Section 1.2(c)(ii).

"First Post-Closing Period" shall mean the period beginning on the Closing Date and ending on the first anniversary of the Closing Date.

"<u>Full Financial Statements</u>" shall mean (i) an audited unconsolidated balance sheet, statement of income, statement of stockholders' equity and statement of cash flows for the Business as of, and for each of the years ended, December 31, 2011 and December 30, 2012, together with associated notes to those financial statements, (ii) an unaudited unconsolidated balance sheet, statement of income, statement of stockholders' equity and statement of cash flows for the Business as of, and for each of the quarters ended, March 31, 2012 and March 31, 2013, together with associated notes to those unaudited financial statements (and, if the Closing Date occurs on or after August 15, 2013, as of, and for each of the quarters ended, June 30, 2012 and June 30, 2013), in the case of each of <u>clauses (i)</u> and (ii), with such financial statements to be prepared in a manner consistent with the applicable methodologies set forth in <u>Section 2.4(b) of the Disclosure Schedule</u>, (iii) quarterly reviews prepared by the Seller's independent auditor in respect of such independent auditor's performance of the procedures specified by the PCAOB for a review of interim financial information as described in AU 722 with respect to the financial statements described in the foregoing <u>clause (ii)</u> and (iv) to the extent the parties reasonably

determine the same are required to be filed by the Buyer under Rule 3-05 of Regulation S-X, an audited statement of income, statement of stockholders' equity and statement of cash flows for the Business for the year ended December 31, 2010, together with the associated notes to those audited financial statements, with such financial statements to be prepared in a manner consistent with the applicable methodologies set forth in <u>Section 2.4(b) of the Disclosure Schedule</u>.

"Governmental Entity" shall mean any court, arbitrational tribunal, administrative agency or commission or other governmental or regulatory (including any self-regulatory) authority, organization or agency.

"Income Taxes" shall mean any taxes imposed upon or measured by net income.

"Indemnified Party" shall mean, with respect to any claim for indemnification hereunder, the applicable Party seeking indemnification under <u>ARTICLE</u> <u>VI</u>.

"Indemnifying Party" shall mean, with respect to any claim for indemnification hereunder, the applicable Party from whom indemnification is sought by the Indemnified Party.

"Information" shall mean all non-privileged records, books, contracts, instruments, documents, correspondence, computer data and other data and information (a) if requested by the Seller, (i) to the extent relating to the Business prior to the Closing, (ii) to the extent relating to the Business following the Closing and the Buyer's performance of its obligations under the MAPSA Subcontracting Agreement or (iii) as contemplated by Section 1.2(e) or 10.1(b), or (b) if requested by the Buyer, to the extent relating to the Business; provided, however, that personnel records shall not constitute "Information".

"Intellectual Property" shall mean the following subsisting throughout the world:

(a) Patent Rights;

(b) Trademarks and all goodwill in Trademarks;

(c) copyrights, designs, data and database rights and registrations and applications for registration thereof, including moral rights of authors;

(d) Know How; and

(e) other proprietary rights relating to any of the foregoing (including remedies against infringement thereof and rights of protection of interest therein under the laws of all jurisdictions).

"Intellectual Property Registrations" shall mean Patent Rights, registered Trademarks, registered copyrights and designs, and applications for each of the foregoing.

"Know How" shall mean inventions, invention disclosures, statutory invention registrations, trade secrets and confidential business information, know-how, manufacturing and product processes and techniques, research and development information, financial, marketing and business data, pricing and cost information, business and marketing plans and customer and supplier lists and information, whether patentable or nonpatentable, whether copyrightable or noncopyrightable and whether or not reduced to practice.

"Leased Real Property" shall mean the real property leased by the Seller Affiliate located at Three Gateway Center, Newark, New Jersey 07102, including the thirteenth (13th) floor of the building located at such address.

"Liens" shall mean any pledge, lien, charge, mortgage, claim, encumbrance or security interest of any kind or nature.

"Lost Pruco Revenue" shall mean (a) if, prior to the end of the applicable Post Closing Period (including, for the avoidance of doubt, during any prior Post-Closing Period), Pruco has delivered a Pruco Termination Notice, an amount equal to the Annualized Pruco Revenue at the time the Pruco Termination Notice is delivered and, (b) in all other cases, zero.

"<u>Manager Contract</u>" shall mean each contract between the Seller, on the one hand, and a Participating Manager, on the other hand, listed in <u>Schedule</u> <u>1.1(a)(iv)</u>.

"New Buyer Employee" shall mean each Scheduled Employee who accepts an offer of employment from the Buyer.

"Non-Consenting Canadian Manager" shall mean each Participating Manager, if any, that has not consented to the assignment of its Canadian Assigned Manager Contract to the Buyer prior to the Closing.

"<u>Non-Consenting Sponsor</u>" shall mean each Participating Sponsor, if any, that has not consented to the assignment of its Assigned Sponsor Contract to the Buyer prior to the Closing.

"Non-Transitional New Buyer Employees" shall mean the employees identified on Schedule 9.3(d)(i), in each case, that become New Buyer Employees.

"Outside Date" shall mean September 30, 2013.

"Overhead and Shared Services" shall mean ancillary corporate or shared services provided to or in support of the Business that are general corporate or other overhead services or provided to both the Business and other businesses of the Seller, including access to hardware and software related to research and development services, software development services provided by information technology consultants, travel and entertainment services, temporary labor services, office supplies services (including copiers and faxes), personal telecommunications services, computer hardware and software services, fleet services, energy/utilities services, treasury services, public relations, legal and risk management services (including workers' compensation), payroll services, information technology and telecommunications services, property management services, environmental support services and customs and excise services, in each case including services relating to the provision of access to information, operating and reporting systems and databases and all hardware and software used in connection therewith, including those services that are set forth on <u>Schedule 11.1(b)</u>.

"Participating Clients" shall mean, as of any date of determination, with respect to each Participating Sponsor, such Participating Sponsor's clients whose assets are managed through the Business.

"Participating Manager" shall mean, as of any date of determination, a manager or sub-advisor that is participating in the Business as of such date, with such participation being evidenced by the existence of a Master Manager Agreement or a Master MSP Sub-Advisor Agreement with the Seller as of such date.

"<u>Participating Sponsor</u>" shall mean, as of any date of determination, a financial institution, insurance company, independent broker/dealer network or other third party that is participating in the Business as of such date, with such participation being evidenced by the existence of a Managed Account Program Service Agreement with the Seller as of such date.

"<u>Patent Rights</u>" shall mean all patents, patent applications, utility models, design registrations and certificates of invention and other governmental grants for the protection of inventions or industrial designs (including all related continuations, continuations-in-part, divisionals, reissues and reexaminations).

"Permits" shall mean all licenses, permits, franchises or authorizations issued by any Governmental Entity to the Seller relating exclusively to the Business.

"<u>Permitted Lien</u>" shall mean (i) mechanics', carriers', workers', repairers', materialmen's, warehousemen's and other Liens which have arisen in the ordinary course of business; (ii) Liens for Taxes that are not yet due and payable or that are being contested in good faith; (iii) Liens which, in the aggregate, are not reasonably likely to impair, in any material respect, the continued use of the asset or property to which they relate, as used on the date hereof; (iv) Liens under Intellectual Property contracts; and (v) Liens discharged at or prior to the Closing.

"Person" shall mean any natural person, firm, limited liability company, general or limited partnership, association, corporation, unincorporated organization, company, joint venture, trust, Governmental Entity or other entity.

"<u>Personal Information</u>" means: (x) any information that (i) identifies an individual (by name, signature, address, telephone number or other unique identifier), or (ii) that can be used to authenticate that individual (including passwords or PINs, biometric data, unique identification numbers, answers to security questions, or other personal identifiers).

"Post-Closing Consent Period" shall mean the 90-day period beginning on the Closing Date.

"Post-Closing Payments" means, collectively, the Applicable Additional Purchase Price Payments and Applicable Excess Revenue Payments.

"Post-Closing Period" shall mean each of the First Post-Closing Period, the Second Post-Closing Period and the Third Post-Closing Period.

"Privacy Law" shall mean any federal, state, local, municipal, foreign, international, multinational or other statute, law, order, decree, constitution, rule, regulation, ordinance, principle of common law, treaty or other law relating to the collection, processing, storage, use, disclosure, loss, access, transfer or security and safeguarding of Personal Information, or relating to notices, consents, sharing, opt-out rights, disclosure to parties other than the party that collected the Personal Information. Notwithstanding the foregoing, the Health Insurance Portability and Accountability Act, and any other federal, state, local, municipal, foreign, international, multinational or other statute, law, order, decree, constitution, rule, regulation, ordinance, dealing with personal health information or protected health information shall not be considered a Privacy Law.

"Proprietary Information" shall mean, (i) with respect to the Seller, all Information provided by or on behalf of the Seller hereunder and all other information that does not relate to the Business, the Acquired Assets or the Affiliate Assets and was provided by or on behalf of the Seller under the Confidentiality Agreement, including any analyses, compilations, forecasts, studies, summaries, notes, reports, memoranda, interpretations, data and other documents and materials, and (ii) with respect to the Buyer, all Information provided by or on behalf of the Buyer hereunder and all other information in the Seller's or its Affiliates' possession to the extent relating to the Business, the Acquired Assets or the Affiliate Assets, including any analyses, compilations, forecasts, studies, summaries, notes, reports, memoranda, interpretations, data and other documents and materials.

"Pruco" shall mean Pruco Securities, LLC.

"Pruco Contract" shall mean the contract entered into by Pruco and the Buyer at closing in the form attached hereto as Exhibit G-1, as such contract may be amended, restated or otherwise modified, succeeded or replaced from time to time.

"Pruco Termination Notice" shall mean a notice given by Pruco to the applicable counterparty to the Pruco Contract declaring Pruco's intention to terminate the Pruco Contract other than pursuant to Section V.B.(1).(y) or (z) thereof; provided that, any such notice shall be deemed not to be a Pruco Termination Notice if Pruco and the Buyer or the applicable Buyer Subsidiary or Successor to the Business have separately provided for a successor or replacement contract to the Pruco Contract.

"<u>Purchase Price</u>" shall mean, as of any date of determination, an amount equal to the sum of (a) the Final Adjusted Purchase Price <u>plus</u> (b) to the extent the same have been paid hereunder, the aggregate amount of Applicable Additional Purchase Price Payments.

"Regulation S-X" shall mean Regulation S-X under the Securities Exchange Act of 1934.

"<u>Representatives</u>" shall mean, with respect to each Party, its Affiliates and its and their respective directors, officers, managers, employees, controlling Persons, investment bankers, advisors, accountants, consultants, counsel and other designated representatives and agents.

"<u>Required Financial Statements</u>" shall mean, (a) in the event the Requested Relief is provided by the SEC prior to the Closing, the Abbreviated Financial Statements and (b) in the event the Requested Relief is not provided by the SEC prior to the Closing, the Full Financial Statements.

"<u>Resolved But Unsatisfied Claim</u>" shall mean a claim by the Buyer for indemnification pursuant to <u>ARTICLE VI</u>, which (A) has been agreed to in writing by the Buyer and the Seller or (B) has been determined by a court of competent jurisdiction in a final, non-appealable order, but, in each case, only to the extent such claim remains unsatisfied by the Seller and its Affiliates within two (2) Business Days of such claim becoming so agreed or otherwise resolved.

"Retained Marks" shall mean any Trademarks of the Seller or one of its Affiliates or any contraction, abbreviation or simulation thereof.

"<u>Revenue Run-Rate</u>" shall mean, with respect to a Participating Sponsor as of any date of determination, on an annualized basis, (a) the Assets Under Administration with respect to such Participating Sponsor, <u>multiplied by</u> (b) the applicable Fee rates for such Participating Sponsor as of such date of determination.

"Scheduled Employees" shall mean (i) the Seller Business Employees as selected by the Buyer and identified as such by the Buyer on Schedule 9.1(b) on the date of this Agreement and (ii) the Canadian Employees.

"Schedules" shall mean the schedules provided by the Seller to the Buyer on the date hereof and that are attached to this Agreement.

"Second Post-Closing Period" shall mean the period beginning on the day after first anniversary of the Closing Date and ending on the second anniversary of the Closing Date.

"Seller Affiliate" shall mean The Prudential Insurance Company of America.

"Seller Business Employees" shall mean all employees of the Seller and the Canadian Subsidiary exclusively or primarily engaged in the Business as of the date hereof and listed on Schedule 9.1(a).

"Seller Certificate" shall mean a certificate to the effect that each of the conditions specified in clauses (a), (b), (c) and (h) of Section 5.2 is satisfied.

"Seller Fundamental Representations" shall mean the representations and warranties of the Seller contained in Sections 2.1, 2.2 and 2.6(a) (with respect to all Acquired Assets, tangible and intangible) and the representations and warranties of Seller Affiliate contained in Sections 3 and 5 of the Affiliate Bill of Sale.

"Seller Intellectual Property" shall mean the Seller Owned Intellectual Property and the Seller Licensed Intellectual Property.

"Seller Licensed Intellectual Property" shall mean third party software components licensed to Seller from a third party for use in connection with the Software.

"Seller Owned Intellectual Property" shall mean the Software, the Assigned Domain Names, all Know How and documentation exclusively used in the Business, and all rights to sue for past infringement or misappropriation of the foregoing.

"Severance Outside Date" shall mean the date that is the later of (a) the date that is one (1) year following the Closing Date and (b) the earlier of (x) the date that is the last day on which services are provided by the Seller under the Transition Services Agreement and (y) unless otherwise agreed in writing by the parties, the date that is two (2) years following the Closing Date.

"Software" shall mean the computer software product and service offering set forth under the heading "Software" in <u>Schedule 1.1(a)(i)</u>, whether in source code, interpreted code or object code form, excluding, for the avoidance of doubt, the Seller Licensed Intellectual Property.

"Specified Employees" shall mean the employees identified on Schedule 9.3(d)(iii), in each case, that become New Buyer Employees.

"Specified Quarterly Financial Statements" shall mean, (a) in the event the Requested Relief is provided by the SEC prior to the Closing, unaudited statements of assets acquired and liabilities assumed and statements of revenues and direct operating expenses, in each case, with respect to the Business as of, and for each of the quarters ended, June 30, 2012 and September 30, 2012 or (b) in the event the Requested Relief is not provided by the SEC prior to the Closing, (i) the Accepted Financial Statements (other than any statement of cash flows or stockholders' equity, or comparable financial statements) as of, and for each of the quarters ended, June 30, 2012 and September 30, 2012 or (ii) an unaudited unconsolidated balance sheet and statement of income for the Business as of, and for each of the quarters ended, June 30, 2012 and September 30, 2012 or (ii) an unaudited unconsolidated balance sheet and statement of income for the Business as of, and for each of the quarters ended, June 30, 2012 and September 30, 2012, in the case of <u>clauses (a)</u> and (b), prepared in a manner consistent with the applicable methodologies set forth in <u>Section 2.4(b) of the Disclosure Schedule</u>.

"Sponsor Contract" shall mean each contract between the Seller, on the one hand, and a Participating Sponsor, on the other hand, listed in Schedule 1.1(a)(v).

"Subsidiary" shall mean any corporation, partnership, trust, limited liability company or other non-corporate business enterprise in which the Seller (or another Subsidiary) holds stock or other ownership interests representing more than fifty percent (50%) of the voting power of all outstanding stock or ownership interests of such entity.

"Successor to the Business" shall mean any entity that has succeeded (whether by merger, consolidation, operation of law or otherwise) to the Buyer's rights to all or substantially all of the Acquired Assets and/or the Business as it may be conducted by the Buyer or any of the Buyer Subsidiaries following the Closing.

"Taxes" shall mean all taxes, charges, fees, levies, tariffs, duties, liabilities, impositions or other assessments in the nature of a tax, including income, gross receipts, ad valorem, value-added, goods and services, excise, real property, personal property, sales, use, transfer, withholding, employment, social security, escheat, unclaimed property, retirement, unemployment, workers' compensation, occupation, license, payroll and franchise taxes imposed

by the United States of America or any state, local or foreign government, or any agency thereof, or other political subdivision of the United States or any such government, and any interest, penalties, assessments or additions to tax resulting from, attributable to or incurred in connection with any tax or any contest or dispute thereof.

"Taxing Authority" shall mean any applicable governmental authority responsible for the imposition of Taxes.

"Tax Returns" shall mean all reports, returns, declarations, schedules, forms or statements required to be supplied to a Taxing Authority in connection with Taxes.

"Terminating Sponsor" shall mean (a) each Participating Sponsor, if any, that terminates its Sponsor Contract and (b) each Participating Sponsor, if any, that, at the Closing, has its Sponsor Contract terminated pursuant to Section 5.2(h).

"Third Post-Closing Period" shall mean the period beginning on the day after the second anniversary of the Closing Date and ending on third anniversary of the Closing Date.

"Trademarks" shall mean registered trademarks and service marks, corporate names and doing business designations and all registrations and applications for registration of the foregoing, common law trademarks and service marks and trade dress.

"Transitional New Buyer Employees." shall mean the employees identified on Schedule 9.3(d)(ii), in each case, that become New Buyer Employees.

Section

"WARN" shall mean the Worker Adjustment and Retraining Notification Act.

11.2 Other Defined Terms. The following capitalized terms have the meanings assigned to them in the corresponding Sections set forth in the table below:

#### Defined Term

" <u>Action</u> "	Section 2.11
" <u>Affiliate Bill of Sale</u> "	Section 1.4(c)(iv)
" <u>Agreement</u> "	Introduction
" <u>Allocation</u> "	Section 1.3
"Assignment and Assumption Agreement"	Section 1.3(c)(v)
"Base Revenue Run-Rate"	Section 2.18
"Buyer's Closing Net Asset Schedule"	Section 1.5(a)
" <u>Buyer</u> "	Introduction

"Buyer-Determined Closing Adjustment Amount"	Section 1.5(a)
"Buyer's Request Letter"	Section 4.5(a)
"Closing Date Adjusted Purchase Price"	Section 1.4(b)
"Confidential Manager Contracts"	Section 2.9(b)
" <u>Disputed Items</u> "	Section 1.5(b)
" <u>Divestiture</u> "	Section 1.2(e)
"Final Payment Date"	Section 1.2(d)
"Gateway Sublease Agreement"	Section 1.4(c)(vii)
"Licensed Marks"	Section 10.2(a)
"Licensed Software"	Section 10.3
"MAPSA Subcontracting Agreement"	Section 1.4(c)(viii)
"Negative Consent Notice"	Section 4.4(e)
"Non-Affiliation Statement"	Section 10.2(a)
"Non-Assigned Contract"	Section 1.6
"Non-Severance Terminated Employees"	Section 9.3(d)
"On-Leave Business Employee"	Section 9.1(b)
"Open Source Materials"	Section 2.8(g)
" <u>Order</u> "	Section 2.3(d)
"Parties"	Introduction
"Post-Closing Payments"	Section 1.2(d)
"Pre-Closing Tax Period"	Section 7.2(a)
"Pruco/PI Termination Agreement"	Section 1.4(c)(x)
"Real Property Lease"	Section 2.7(a)
"Remaining Employees"	Section 9.3(d)

"Requested Relief"	Section 4.5(a)
"Revenue Statement"	Section 1.2(c)(i)
"Section 1.2 Accounting Firm"	Section 1.2(c)(iii)
"Section 1.5 Accounting Firm"	Section 1.5(b)
" <u>SEC</u> "	Section 4.5(a)
"Seller's Closing Net Asset Schedule"	Section 1.4(b)
" <u>Seller</u> "	Introduction
"Seller Bill of Sale"	Section 1.3(c)(iii)
"Seller-Determined Closing Adjustment Amount"	Section 1.4(b)
"Seller Severance Arrangement"	Section 9.3(d)
"Stipulated Closing Adjustment Amount."	Section 1.5(b)
"Straddle Period"	Section 7.2(a)
"Terminating Manager"	Section 4.4(e)
"Transfer Taxes"	Section 7.1
"Transition Services Agreement"	Section 1.4(c)(vi)
"Transition Trademark Period"	Section 10.2(a)

# ARTICLE XII MISCELLANEOUS

12.1 Press Releases and Announcements. As agreed between the Parties, at the opening of business on the day following the execution and delivery of this Agreement, the Buyer will issue a press release announcing the execution and delivery of this Agreement, substantially in the form set forth in <u>Exhibit I</u> hereto. No Party shall issue or make (and each Party shall cause its Affiliates not to issue or make) any other press release or public disclosure relating to the subject matter of this Agreement without the prior written approval of the other Party; provided, however, that either Party may make any press release or public announcement it believes in good faith is required by law, regulation or stock exchange rule (in which case the disclosing Party shall advise the other Party and the other Party shall, if practicable, have the right to review such press release or public announcement prior to its publication) and the Seller (or its Affiliates) may disclose any information concerning the transactions contemplated by this Agreement or any of the Ancillary Agreements that it or its direct or indirect parent entity deems appropriate in light of its or their status as a public company, including in any filing with a Governmental Entity or to securities analysts or institutional investors or in press interviews.

12.2 No Third Party Beneficiaries. This Agreement shall not confer any rights or remedies upon any Person or entity other than the Parties and their respective successors and permitted assigns and, to the extent specified herein, their respective Affiliates.

12.3 Action to be Taken by Affiliates. The Parties shall cause their respective Affiliates to comply with all of the obligations specified in this Agreement to be performed by such Affiliates.

12.4 Entire Agreement. This Agreement (including the Disclosure Schedule, Schedules and other documents referred to herein), the Ancillary Agreements (to the extent the same are executed and delivered) and, prior to the Closing, the Confidentiality Agreement constitute the entire agreement among the Buyer, on the one hand, and the Seller, on the other hand with respect to the subject matter hereof. This Agreement and the Ancillary Agreements supersede any prior agreements or understandings among the Buyer, on the one hand, and the Seller, on the other hand, with respect to the subject matter hereof.

12.5 <u>Succession and Assignment</u>. No Party may assign either this Agreement or any of its rights, interests, or obligations hereunder without the prior written approval of the other Party; provided, that the Buyer may, without the prior written consent of, but with notice to, the Seller, assign all or any portion of its rights, interests, or obligations under this Agreement to any Buyer Subsidiary. No assignment shall relieve the assigning party of any of its obligations hereunder, except as contemplated by <u>Section 1.2(e)</u>. This Agreement shall be binding upon and inure to the benefit of the Parties and their respective successors and permitted assigns.

12.6 <u>Notices</u>. All notices, requests, demands, claims and other communications hereunder shall be in writing. Any notice, request, demand, claim or other communication hereunder shall be deemed duly delivered four (4) Business Days after it is sent by registered or certified mail, return receipt requested, postage prepaid, or one (1) Business Day after it is sent for next Business Day delivery via a reputable nationwide overnight courier service, in each case to the intended recipient as set forth below:

### If to the Buyer:

Envestnet, Inc. 35 East Wacker Drive, Suite 2400 Chicago, IL 60601 Attn: Shelly O'Brien Fax: (312) 827-2801

#### Copy to:

Mayer Brown LLP 71 South Wacker Drive Chicago, IL 60606 Attn: Edward S. Best, Esq. Fax: (312) 706-2801

If to the Seller: c/o PRUDENTIAL FINANCIAL, INC. 751 Broad Street Newark, NJ 07102-3777 Attn: Anthony Torre Fax: (973) 802-2290

### Copies to:

Wilmer Cutler Pickering Hale and Dorr LLP 1875 Pennsylvania Avenue, NW Washington, DC 20006 Attn: Stephanie C. Evans Fax: (202) 663-6363

Either Party may give any notice, request, demand, claim, or other communication hereunder using any other means (including personal delivery, expedited courier, messenger service, telecopy, telex, ordinary mail, or electronic mail), but no such notice, request, demand, claim or other communication shall be deemed to have been duly given unless and until it actually is received by the Party for whom it is intended. Any Party may change the address to which notices, requests, demands, claims and other communications hereunder are to be delivered by giving the other Party notice in the manner herein set forth.

12.7 <u>Amendments and Waivers</u>. The Parties may mutually amend or waive any provision of this Agreement at any time. Other than the updates to the Schedules contemplated by the definition of "Acquired Assets", "Assigned Manager Contracts" and "Assigned Sponsor Contracts", no amendment or waiver of any provision of this Agreement shall be valid unless the same shall be in writing and signed by both Parties. No waiver by either Party of any default, misrepresentation, or breach of warranty or covenant hereunder, whether intentional or not, shall be deemed to extend to any prior or subsequent default, misrepresentation or breach of warranty or covenant hereunder or affect in any way any rights arising by virtue of any such prior or subsequent occurrence.

12.8 Severability. Any term or provision of this Agreement that is invalid or unenforceable in any situation in any jurisdiction shall not affect the validity or enforceability of the remaining terms and provisions hereof or the validity or enforceability of the offending term or provision in any other situation or in any other jurisdiction. If the final judgment of a court of competent jurisdiction declares that any term or provision hereof is invalid or unenforceable, the Parties agree that the body making the determination of invalidity or unenforceability shall have the power to reduce the scope, duration or area of the term or provision, to delete specific words or phrases, or to replace any invalid or unenforceable term or provision with a term or provision that is valid and enforceable and that comes closest to expressing the intention of the invalid or unenforceable term or provision, and this Agreement shall be enforceable as so modified.

12.9 Expenses. Except as otherwise specifically provided to the contrary in this Agreement, each of the Parties shall bear its own costs and expenses (including legal fees and expenses) incurred in connection with this Agreement, the negotiation hereof and the transactions contemplated hereby. Notwithstanding the foregoing, the Buyer shall be responsible for, and shall promptly reimburse the Seller upon presentation of an invoice for, the reasonable expenses of the Seller's independent auditor incurred by the Seller in connection with the preparation, delivery, audit and review of the Required Financial Statements and the Specified Quarterly Financial Statements and assistance in pursuing the Requested Relief.

12.10 Specific Performance. Each Party acknowledges and agrees that the other Party would be damaged irreparably in the event that any of the provisions of this Agreement are not performed in accordance with their specific terms or otherwise are breached. Accordingly, each Party agrees that the other Party may be entitled to an injunction or injunctions, specific performance or other equitable relief to prevent breaches or threatened breaches of the provisions of this Agreement and to enforce specifically this Agreement and the terms and provisions hereof in any action instituted in the courts described in Section 12.12, without proof of damages or otherwise, this being in addition to any other remedy to which the Parties may be entitled to under this Agreement. The Parties agree not to assert that a remedy of specific enforcement is unenforceable, invalid, contrary to law or inequitable for any reason, and not to assert that a remedy of monetary damages would provide an adequate remedy or that the Parties otherwise have an adequate remedy at law. The Parties acknowledge and agree that either Party seeking an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions of this Agreement in accordance with this Section 12.10 shall not be required to provide any bond or other security in connection with any such order or injunction.

12.11 <u>Governing Law</u>. This Agreement and any disputes hereunder shall be governed by and construed in accordance with the internal laws of the State of New York without giving effect to any choice or conflict of law provision or rule (whether of the State of New York or any other jurisdiction) that would cause the application of laws of any jurisdiction other than those of the State of New York.

12.12 <u>Submission to Jurisdiction</u>. Other than as provided for in <u>Sections 1.2(c)(iii)</u> and <u>1.5(b)</u>, each Party (a) submits to the exclusive jurisdiction of the United States District Court for the Southern District of New York (or only if such court will not accept jurisdiction, the Supreme Court of the State of New York or a court of competent civil jurisdiction, in either case, sitting in New York County, New York) in any action or proceeding arising out of or relating to this Agreement, (b) agrees that all claims in respect of such action or proceeding may be heard and determined only in any such court, (c) waives any claim of inconvenient forum or other challenge to venue in such court and (d) agrees not to bring any action or proceeding arising out of or relating to this Agreement or the transactions contemplated hereby in any other court; provided that a final judgment in any such action or proceeding shall be conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by applicable law. Each Party agrees to accept service of any summons, complaint or other initial pleading made in the manner provided for the giving of notices in <u>Section 12.6</u>. Nothing in this <u>Section 12.12</u>, however, shall affect the right of any Party to serve such summons, complaint or initial pleading in any other manner permitted by law.

12.13 <u>Bulk Transfer Laws</u>. The Buyer and the Seller hereby waive compliance with the provisions of any and all applicable bulk transfer laws that may be applicable to the transactions contemplated by this Agreement; <u>provided</u>, that any liability for any claim against the Buyer based upon any such law shall constitute an Excluded Liability.

# 12.14 Construction.

(a) The language used in this Agreement shall be deemed to be the language chosen by the Parties to express their mutual intent, and no rule of strict construction shall be applied against any Party.

(b) Any reference to any federal, state, local, or foreign statute or law shall be deemed also to refer to all rules and regulations promulgated thereunder, unless the context requires otherwise.

(c) Any agreement, instrument or statute defined or referred to herein or in any agreement or instrument that is referred to herein means such agreement, instrument or statue as from time to time amended, modified or supplemented, including (in the case of agreements or instruments) by waiver or consent and (in the case of statutes) by succession or comparable successor statutes and references to all attachments thereto and instruments incorporated therein.

(d) The section headings contained in this Agreement are inserted for convenience only and shall not affect in any way the meaning or interpretation of this Agreement.

(e) Any reference herein to an ARTICLE, Section, subsection, clause, Exhibit or Schedule shall be deemed to refer to an ARTICLE, Section, subsection or clause of or Exhibit or Schedule to this Agreement, unless the context clearly indicates otherwise.

(f) All references to "\$", "Dollars" or "US\$" refer to currency of the United States of America.

(g) Whenever the words "include", "includes" or "including" are used in this Agreement, they shall be deemed to be followed by the words "without limitation".

(h) The words "date hereof" when used in this Agreement shall refer to the date of this Agreement.

(i) The terms "or", "any" and "either" are not exclusive.

(j) The word "extent" in the phrase "to the extent" shall mean the degree to which a subject or other thing extends, and such phrase shall not mean simply "if".

(k) The word "will" shall be construed to have the same meaning and effect as the word "shall".

(1) All terms defined in this Agreement shall have the defined meanings when used in any document made or delivered pursuant hereto unless otherwise defined therein.

(m) The definitions contained in this Agreement are applicable to the singular as well as the plural forms of such terms and to the masculine as well as to the feminine and neuter genders of such term.

(n) References to a Person are also to its permitted successors and assigns.

12.15 <u>Waiver of Jury Trial</u>. To the extent permitted by applicable law, each Party hereby irrevocably waives all rights to trial by jury in any action, proceeding or counterclaim (whether based on contract, tort or otherwise) arising out of or relating to this Agreement or the transactions contemplated hereby or the actions of any Party in the negotiation, administration, performance and enforcement of this Agreement. Each Party (a) certifies that no Representative, agent or attorney of the other Party has represented, expressly or otherwise, that such Party would not, in the event of any action, proceeding or counterclaim, seek to enforce the foregoing waiver and (b) acknowledges that it and the other Party have been induced to enter into this Agreement, by among other things, the mutual waiver and certifications in this <u>Section 12.15</u>.

12.16 Incorporation of Exhibits and Schedules. The Exhibits and Schedules identified in this Agreement are incorporated herein by reference and made a part hereof.

12.17 <u>Negotiation of Agreement</u>. Each of the Buyer and the Seller acknowledges that it has been represented by independent counsel of its choice throughout all negotiations that have preceded the execution of this Agreement and that it has executed the same with consent and upon the advice of said independent counsel. Each Party and its counsel cooperated in the drafting and preparation of this Agreement and the documents referred to herein, and any and all drafts relating thereto shall be deemed the work product of the Parties and may not be construed against any Party by reason of its preparation. Accordingly, any rule of law or any legal decision that would require interpretation of any ambiguities in this Agreement against the Party that drafted it is of no application and is hereby expressly waived. The provisions of this Agreement shall be interpreted in a reasonable manner to effect the intentions of the Parties and this Agreement.

12.18 <u>Counterparts and Facsimile Signature</u>. This Agreement may be executed in one or more counterparts, each of which shall be deemed an original but all of which together shall constitute one and the same instrument. This Agreement may be executed by facsimile signature.

[Remainder of page intentionally left blank. Signature pages follow.]

IN WITNESS WHEREOF, the Parties have executed this Agreement as of the date first above written.

# PRUDENTIAL INVESTMENTS LLC

By:	/s/ Stuart S. Parker
Name:	Stuart S. Parker
Title:	President, Chief Executive Officer and Chief Operating Officer

ENVESTNET, INC.

By:	/s/ Judson Bergman
Name:	Judson Bergman
Title:	Chairman and Chief Executive Officer

[Signature page to Purchase and Sale Agreement]

Exhibit 21.1

Envestnet, Inc. Subsidiaries of the Registrant

Name

SIGMA Asset Management, LLC Oberon Financial Technology, Inc. NetAssetManagement, Inc. Envestnet Asset Management, Inc. Envestnet Portfolio Solutions, Inc. Envestnet Securities, Inc. ERS, Inc. PMC International, Inc. Premier Advisors Fund Offshore, Ltd. Premier Advisors Fund, L.L.C. Envestnet Asset Management (India) Pvt. Ltd. Portfolio Management Consultants, Inc. Portfolio Brokerage Services, Inc. Prima Capital Holding, Inc. Tamarac, Inc. Jurisdiction of Incorporation Delaware Delaware Delaware Delaware Delaware Delaware Delaware Colorado Cayman Islands Delaware India Colorado Colorado Colorado Washington

## Consent of Independent Registered Public Accounting Firm The Board of Directors Envestnet, Inc.:

We consent to the incorporation by reference in the registration statements (Nos. 333-169050 and 333-181071) on Form S-8 of Envestnet, Inc. (the Company) of our reports dated June 14, 2013, with respect to the consolidated balance sheets of Envestnet, Inc. as of December 31, 2012 and 2011, and the related consolidated statements of operations, stockholders' equity, and cash flows for each of the years in the three-year period ended December 31, 2012, and the effectiveness of internal control over financial reporting as of December 31, 2012, which reports appear in the December 31, 2012 annual report on Form 10-K of Envestnet, Inc.

Our report dated June 14, 2013, on the effectiveness of internal control over financial reporting as of December 31, 2012, expresses our opinion that Envestnet, Inc. did not maintain effective internal control over financial reporting as of December 31, 2012 because of the effect of a material weakness on the achievement of the objectives of the control criteria and contains an explanatory paragraph that states the Company did not have adequate design or operation of controls that provide reasonable assurance that the accounting for income taxes, including the related financial statement disclosures, were in accordance with GAAP.

Our report dated June 14, 2013, on the effectiveness of internal control over financial reporting as of December 31, 2012, also contains an explanatory paragraph that states that management excluded Tamarac, Inc. and Prima Capital Holding, Inc. from its assessment of the effectiveness of internal control over financial reporting as of December 31, 2012, because those entities were acquired by the Company in purchase business combinations in the second quarter of 2012.

/s/ KPMG LLP

Chicago, Illinois June 14, 2013

# CHIEF EXECUTIVE OFFICER CERTIFICATION

I, Judson Bergman, certify that:

- 1. I have reviewed this Annual Report on Form 10-K for the period ended December 31, 2012, of Envestnet, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - a. Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - b. Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - c. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - d. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - a. All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: June 14, 2013

<u>/S/ JUDSON BERGMAN</u> Judson Bergman Chairman and Chief Executive Officer (Principal Executive Officer)

# CHIEF FINANCIAL OFFICER CERTIFICATION

I, Peter H. D'Arrigo, certify that:

- 1. I have reviewed this Annual Report on Form 10-K for the period ended December 31, 2012, of Envestnet, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - b. Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - c. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - d. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - a. All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: June 14, 2013

<u>/S/ PETER H. D'ARRIGO</u> Peter H. D'Arrigo Chief Financial Officer (Principal Financial Officer)

# CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Annual Report of Envestnet, Inc. (the "Company") on Form 10-K for the period ending December 31, 2012 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, Judson Bergman, certify, pursuant to 18 U.S.C. § 1350, as adopted pursuant to §906 of the Sarbanes-Oxley Act of 2002, that, to the best of my knowledge and belief:

(1) The Report fully complies with the requirements of section 13(a) or 15(d) of the Securities Exchange Act of 1934; and

(2) The information contained in the Report fairly presents, in all material respects, the financial condition and result of operations of the Company.

/s/ Judson Bergman By: Judson Bergman Chairman and Chief Executive Officer (Principal Executive Officer)

Dated: June 14, 2013

A signed original of this written statement required by Section 906, or other document authenticating, acknowledging, or otherwise adopting the signature that appears in typed form within the electronic version of this written statement has been provided to the Company and will be retained by the Company and furnished to the Securities and Exchange Commission or its staff upon request.

# CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Annual Report of Envestnet, Inc. (the "Company") on Form 10-K for the period ending December 31, 2012 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, Peter H. D'Arrigo, certify, pursuant to 18 U.S.C. § 1350, as adopted pursuant to §906 of the Sarbanes-Oxley Act of 2002, that, to the best of my knowledge and belief:

(1) The Report fully complies with the requirements of section 13(a) or 15(d) of the Securities Exchange Act of 1934; and

(2) The information contained in the Report fairly presents, in all material respects, the financial condition and result of operations of the Company.

<u>/s/ Peter H. D'Arrigo</u> By: Peter H. D'Arrigo Chief Financial Officer (Principal Financial Officer)

Dated: June 14, 2013

A signed original of this written statement required by Section 906, or other document authenticating, acknowledging, or otherwise adopting the signature that appears in typed form within the electronic version of this written statement has been provided to the Company and will be retained by the Company and furnished to the Securities and Exchange Commission or its staff upon request.